World Journal of *Clinical Oncology*

World J Clin Oncol 2021 December 24; 12(12): 1089-1263





Published by Baishideng Publishing Group Inc

World Journal of Clinical Oncology

Contents

Monthly Volume 12 Number 12 December 24, 2021

OPINION REVIEW

1089 Advances and controversies in the management of early stage non-small cell lung cancer

Cilleruelo-Ramos A, Cladellas-Gutiérrez E, de la Pinta C, Quintana-Cortés L, Sosa-Fajardo P, Couñago F, Mielgo-Rubio X, Trujillo-Reyes JC

REVIEW

- 1101 Liver regeneration biology: Implications for liver tumour therapies Hadjittofi C, Feretis M, Martin J, Harper S, Huguet E
- 1157 Primary vascular tumours of the kidney Omiyale AO

MINIREVIEWS

- 1169 Detection of circulating tumour cells in colorectal cancer: Emerging techniques and clinical implications Yadav A, Kumar A, Siddiqui MH
- 1182 Immunotherapy combinations and chemotherapy sparing schemes in first line non-small cell lung cancer Sereno M, Higuera O, Cruz Castellanos P, Falagan S, Mielgo-Rubio X, Trujillo-Reyes JC, Couñago F
- 1193 Role of liver transplantation in the management of colorectal liver metastases: Challenges and opportunities

Tasoudis PT, Ziogas IA, Alexopoulos SP, Fung JJ, Tsoulfas G

ORIGINAL ARTICLE

Basic Study

Increased tensin 4 expression is related to the histological type of gastric cancer 1202

Nizioł M, Zińczuk J, Zaręba K, Guzińska-Ustymowicz K, Pryczynicz A

1215 Tumor irradiation may facilitate the detection of tumor-specific mutations in plasma

Kuligina E, Moiseyenko F, Belukhin S, Stepanova E, Zakharova M, Chernobrivtseva V, Aliev I, Sharabura T, Moiseyenko V, Aleksakhina S, Laidus T, Martianov A, Kholmatov M, Whitehead A, Yanus G, Imyanitov E

SYSTEMATIC REVIEWS

Mixed odontogenic tumors: A review of the clinicopathological and molecular features and changes in the 1227 WHO classification

Sánchez-Romero C, Paes de Almeida O, Bologna-Molina R

1244 Carcinosarcoma of gallbladder: A world review

Teng TZJ, Chua BQY, Shelat VG



Contents

Monthly Volume 12 Number 12 December 24, 2021

ABOUT COVER

Editorial Board Member of World Journal of Clinical Oncology, Sutatip Pongcharoen, MD, PhD, Associate Professor, Division of Immunology, Department of Internal Medicine, Faculty of Medicine, Naresuan University, Meaung 65000, Phitsanulok, Thailand. sutatipp@nu.ac.th

AIMS AND SCOPE

The primary aim of World Journal of Clinical Oncology (WJCO, World J Clin Oncol) is to provide scholars and readers from various fields of oncology with a platform to publish high-quality basic and clinical research articles and communicate their research findings online.

WJCO mainly publishes articles reporting research results and findings obtained in the field of oncology and covering a wide range of topics including art of oncology, biology of neoplasia, breast cancer, cancer prevention and control, cancer-related complications, diagnosis in oncology, gastrointestinal cancer, genetic testing for cancer, gynecologic cancer, head and neck cancer, hematologic malignancy, lung cancer, melanoma, molecular oncology, neurooncology, palliative and supportive care, pediatric oncology, surgical oncology, translational oncology, and urologic oncology.

INDEXING/ABSTRACTING

The WJCO is now abstracted and indexed in PubMed, PubMed Central, Emerging Sources Citation Index (Web of Science), China National Knowledge Infrastructure (CNKI), China Science and Technology Journal Database (CSTJ), and Superstar Journals Database. The 2021 edition of Journal Citation Reports® cites the 2020 Journal Citation Indicator (JCI) for WJCO as 0.48.

RESPONSIBLE EDITORS FOR THIS ISSUE

Production Editor: Hua-Ge Yu; Production Department Director: Yu-Jie Ma; Editorial Office Director: Ze-Mao Gong.

INSTRUCTIONS TO AUTHORS https://www.wjgnet.com/bpg/gerinfo/204
GUIDELINES FOR ETHICS DOCUMENTS
https://www.wjgnet.com/bpg/GerInfo/287
GUIDELINES FOR NON-NATIVE SPEAKERS OF ENGLISH
https://www.wjgnet.com/bpg/gerinfo/240
PUBLICATION ETHICS
https://www.wjgnet.com/bpg/GerInfo/288
PUBLICATION MISCONDUCT
https://www.wjgnet.com/bpg/gerinfo/208
ARTICLE PROCESSING CHARGE
https://www.wjgnet.com/bpg/gerinfo/242
STEPS FOR SUBMITTING MANUSCRIPTS
https://www.wjgnet.com/bpg/GerInfo/239
ONLINE SUBMISSION
https://www.f6publishing.com

© 2021 Baishideng Publishing Group Inc. All rights reserved. 7041 Koll Center Parkway, Suite 160, Pleasanton, CA 94566, USA E-mail: bpgoffice@wjgnet.com https://www.wjgnet.com



World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1089-1100

DOI: 10.5306/wjco.v12.i12.1089

ISSN 2218-4333 (online)

OPINION REVIEW

Advances and controversies in the management of early stage nonsmall cell lung cancer

Angel Cilleruelo-Ramos, Esther Cladellas-Gutiérrez, Carolina de la Pinta, Laura Quintana-Cortés, Paloma Sosa-Fajardo, Felipe Couñago, Xabier Mielgo-Rubio, Juan Carlos Trujillo-Reyes

Angel Cilleruelo-Ramos, Department of Thoracic Surgery, Clinic Universitary Hospital, Valla-ORCID number: Angel Cilleruelodolid 47005, Spain Ramos 0000-0002-1872-0714; Esther Cladellas-Gutiérrez 0000-0001-7379-Angel Cilleruelo-Ramos, Department of Surgery, Universidad de Valladolid, Valladolid 47001, 0443; Carolina de la Pinta 0000-0002-5425-3471; Laura Quintana-Cortés Spain 0000-0002-0090-3192; Paloma Sosa-Esther Cladellas-Gutiérrez, Juan Carlos Trujillo-Reyes, Department of Thoracic Surgery, Hospital Fajardo 0000-0002-0130-321X; Felipe de la Santa Creu i Sant Pau, Barcelona 08029, Spain Couñago 0000-0001-7233-0234; Xabier Mielgo-Rubio 0000-0002-Carolina de la Pinta, Department of Radiation Oncology, Hospital Universitario Ramón y Cajal, 0985-6150; Juan Carlos Trujillo-Reyes 0000-0002-3370-0869. Madrid 28034, Spain Author contributions: All authors Laura Quintana-Cortés, Department of Medical Oncology, Hospital Don Benito-Villanueva, Badajoz 06400, Spain wrote the paper. Conflict-of-interest statement: No Paloma Sosa-Fajardo, Department of Radiation Therapy, Complejo Hospitalario Universitario, Santiago de Compostela, La Coruña 15706, Spain conflict of interest. Country/Territory of origin: Spain Felipe Couñago, Department of Radiation Oncology, Hospital Universitario Quirónsalud Madrid, Madrid 28223, Spain Specialty type: Oncology Felipe Couñago, Department of Radiation Oncology, Hospital La Luz, Madrid 28223, Spain Provenance and peer review: Felipe Couñago, Department of Medicine, School of Biomedical Sciences, Universidad Euro-Unsolicited article; Externally peer pea, Madrid 28223, Spain reviewed. Xabier Mielgo-Rubio, Department of Medical Oncology Unit, Hospital Universitario Fundación Peer-review model: Single blind Alcorcón, Madrid 28922, Spain Peer-review report's scientific Juan Carlos Trujillo-Reyes, Department of Surgery, Universitat Autónoma, Barcelona 08029, quality classification Spain Grade A (Excellent): A Grade B (Very good): 0 Corresponding author: Angel Cilleruelo-Ramos, MD, PhD, Surgeon, Department of Thoracic Grade C (Good): 0 Surgery, Clinic Universitary Hospital, Ramón y Cajal 3, Valladolid 47005, Spain. Grade D (Fair): 0 ancillera@hotmail.com Grade E (Poor): 0 Open-Access: This article is an Abstract open-access article that was

> Complete resection continues to be the gold standard for the treatment of earlystage lung cancer. The landmark Lung Cancer Study Group trial in 1995 estab-



selected by an in-house editor and

fully peer-reviewed by external

reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt ps://creativecommons.org/Licens es/by-nc/4.0/

Received: April 15, 2021 Peer-review started: April 15, 2021 First decision: July 16, 2021 Revised: July 20, 2021 Accepted: December 10, 2021 Article in press: December 10, 2021 Published online: December 24, 2021

P-Reviewer: Kermenli T S-Editor: Fan JR L-Editor: Webster JR P-Editor: Fan JR



lished lobectomy as the minimum intervention necessary for the management of early-stage non-small cell lung cancer, as it was associated with lower recurrence and metastasis rates than sublobar resection and lower postoperative morbidity and mortality than pneumonectomy. There is a growing tendency to perform sublobar resection in selected cases, as, depending on factors such as tumor size, histologic subtype, lymph node involvement, and resection margins, it can produce similar oncological results to lobectomy. Alternative treatments such as stereotactic body radiotherapy and radiofrequency ablation can also produce good outcomes in inoperable patients or patients who refuse surgery.

Key Words: Video-assisted thoracoscopic surgery; Sublobar resection; Radiofrequency ablation; Stereotactic radiosurgery; Early stage; Lung cancer

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: Complete resection continues to be the gold standard for the treatment of early-stage lung cancer. Lobectomy remains the gold standard for the treatment of early-stage non-small cell lung cancer, but there is a growing tendency to perform sublobar resection in selected cases. Alternative treatments such as stereotactic body radiotherapy and radiofrequency ablation can also produce good outcomes in inoperable patients or patients who refuse surgery.

Citation: Cilleruelo-Ramos A, Cladellas-Gutiérrez E, de la Pinta C, Quintana-Cortés L, Sosa-Fajardo P, Couñago F, Mielgo-Rubio X, Trujillo-Reyes JC. Advances and controversies in the management of early stage non-small cell lung cancer. *World J Clin Oncol* 2021; 12(12): 1089-1100

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1089.htm **DOI:** https://dx.doi.org/10.5306/wjco.v12.i12.1089

INTRODUCTION

Complete resection continues to be the gold standard for the treatment of early-stage lung cancer. The landmark Lung Cancer Study Group trial in 1995 established lobectomy as the minimum intervention necessary for the management of early-stage non-small cell lung cancer (NSCLC), as it was associated with lower recurrence and metastasis rates than sublobar resection and lower postoperative morbidity and mortality than pneumonectomy[1]. The development of lung-sparing techniques (*e.g.*, sleeve resection with vascular and/or bronchial reconstruction) has reduced the number of pneumonectomies performed and with this the risk of adverse outcomes, as the proportion of pneumonectomies is a quality indicator in thoracic surgery[2]

While lobectomy remains the gold standard for the treatment of early-stage NSCLC, there is a growing tendency to perform sublobar resection in selected cases, as, depending on factors such as tumor size, histologic subtype, lymph node involvement, and resection margins, it can produce similar oncological results to lobectomy[3]. Two randomized clinical trials comparing lobectomy and sublobar resection are currently underway: The United States Cancer and Leukemia Group B trial (CALGB 140503)[4] and the Japanese JCOG0802/WJOG4607L trial[5]. The results so far have shown no significant differences in postoperative morbidity or mortality, but as discussed in greater detail below, data on survival and pulmonary function are pending.

The use of minimally invasive techniques for the surgical treatment of early-stage NSCLC has increased in recent years. Video-assisted thoracoscopic surgery (VATS) is the current procedure of choice for most resections in this setting. A recent nationwide cohort study conducted in Spain reported that over 50% of recent anatomic lung resections had been performed by VATS[6]. The main advantages of VATS compared with open surgery are decreased postoperative pain, fewer postoperative complications, and in some cases even, better oncological outcomes. There are, however, substantial geographic variations in the use of VATS.

Advances in VATS techniques and the design of specific surgical material have led to a progressive reduction in the number of incisions required. Most thoracic surgeons now use between one and three incisions and describe similar oncological results [6,7]. Subxiphoid VATS is another minimally invasive technique associated with good outcomes when performed by teams with extensive experience in VATS; it has been linked to a lower incidence of postoperative neuropathic pain[8].

The increasing adoption of VATS has favored its use in more locoregionally advanced lung cancers. Data from large series of angio-bronchoplastic or extended lung resections performed by experienced thoracic surgeons show similar outcomes to thoracotomy[9].

Good outcomes have also been described with robotic-assisted thoracoscopic surgery in the setting of anatomic resections, although the cost-effectiveness of the technique is not so clear[10].

As we discuss below, alternative treatments such as stereotactic body radiotherapy (SBRT) and radiofrequency ablation (RFA) can also produce good outcomes in inoperable patients or patients who refuse surgery.

ROLE OF SUBLOBAR RESECTION IN LUNG CANCER

Anatomic sublobar resections have produced comparable oncological results to lobectomy in the treatment of tumors < 2 cm without nodal involvement or distant metastasis^[11]. These favorable results have led to an increased use of segmentectomy, which, depending on tumor stage and resection margins, can produce similar oncological results to lobectomy in selected patients[12].

Anatomic segmentectomy is oncologically more valuable than atypical (wedge) resection in early-stage cancer as it permits the performance of hilar and mediastinal lymph node dissection[13].

Its main advantage, however, is its parenchyma-sparing effect, which results in better postoperative respiratory function than lobectomy. In view of the above, anatomic sublobar resection can be considered an appropriate treatment for patients with compromised respiratory function unable to tolerate standard lobectomy. Patients considered to be at high operative risk include patients with FEV₁ < 50% or DLCO < 50% and elderly patients with impaired lung function, pulmonary hypertension, and poor left ventricular function[14,15].

Compared with lobectomy, VATS sublobar resection has been linked to shorter hospital stays and drainage times, a lower incidence of supraventricular arrhythmia, and fewer postoperative respiratory complications[11].

In certain cases, anatomic segmentectomy involves a higher risk of air leakage when electrocautery is used for intersegmental plane dissection (as reported by several Japanese groups)[5,13]. Air leakage is not common when absorbable sutures are used, which is the case in most lung resections.

The only randomized prospective trial to compare lobectomy and sublobar resection in T1N0M0 lung cancer (the Lung Cancer Study Group trial) concluded that patients treated with sublobar resection had a higher risk of locoregional recurrence and death[1]. It should be noted, however, that these results were published in 1995 and that lung tumors are now diagnosed earlier.

Several retrospective studies published since 2000 have reported good oncological outcomes in patients with small peripheral tumors (stage I and < 2 cm) treated with segmentectomy[13,16-19].

As mentioned, the ongoing CALGB[4] and Japanese[5] trials have not detected any differences between lobectomy and sublobar resection for postoperative morbidity or mortality, but survival and pulmonary function outcomes are not yet available[4,13, 20].

Thus, it remains to be determined whether segmentectomy is a valid alternative to lobectomy for the treatment of early-stage NSCLC in patients fit for both procedures[4, 5].

POSTRESECTION ADJUVANT THERAPY IN NSCLC

Thirty percent of lung cancer patients have early-stage disease when diagnosed. The standard treatment is surgery, followed or not by chemotherapy with or without radiotherapy.



Data from retrospective series show that less invasive surgical procedures result in fewer complications, allowing earlier initiation of chemotherapy, but do not appear to have an impact on overall survival (OS).

Postoperative radiotherapy in stage I and II NSCLC is indicated for patients with positive margins. According to the recent results of the phase III LUNG ART trial, postoperative radiotherapy did not have any beneficial effects in patients with pathologic mediastinal involvement (N2), in addition, it induced high levels of toxicity. Chemotherapy, however, was associated with a 5.4% increase in OS at 5 years, regardless of age [hazard ratio (HR) = 0.89]. Chemotherapy is indicated for resected stage II and IIIA NSCLC^[21], but its use in stage I disease is more controversial. The standard treatment is four cycles of doublet cisplatin-based chemotherapy. The only clinical trial to investigate the use of carboplatin in this setting reported negative results^[22]. Survival outcomes, however, are poor, mainly because of high rates of distant recurrence. Five-year OS rates range from 73% for stage IB disease to 65% for stage IIA disease, 56% for stage IIB disease, and 41% for stage IIIA disease[23]. It is therefore important to continue to explore new treatments and prognostic and predictive biomarkers.

Attempts to improve treatment outcomes with the addition of antiangiogenics^[24] or vaccine-based therapy^[25] have been unsuccessful. The potential benefits of immunotherapy are being investigated, as good results have been reported for adjuvant immunotherapy in more advanced stages of disease and other types of tumor [26]. Ongoing trials include PEARLS (pembrolizumab), BR31 (durvalumab), ANVIL (nivolumab), Impower 010 (atezolizumab), and Canopy-A (canakinumab). No results, however, are available yet. Immunotherapy, both alone and combined with chemotherapy, has shown promising results in the neoadjuvant setting. Chemoimmunotherapy has significantly improved complete and major pathological responses in NSCLC (by approximately 36% and 65%, respectively) and has also led to downstaging in over 70% of patients[26,27]. It remains to be determined whether immunotherapy is more effective as a neoadjuvant or adjuvant treatment[28].

Agents targeting driver mutations are being investigated in multiple trials, but results are still pending. We do have results from the ADAURA trial, where patients with completely resected EGFR mutation-positive NSCLC, regardless of whether or not they had received prior chemotherapy, were randomized to receive osimertinib [a third-generation tyrosine-kinase inhibitor (TKI)] or placebo for 3 years. The progression-free survival (PFS) outcomes for patients with stage II and IIIA disease in the osimertinib group were unprecedented, with an HR for disease recurrence or death of 0.17. In addition, the benefits were observed in all the subgroup analyses. The adverse events were to be expected based on the experience with this drug. Osimertinib was also associated with a reduction in brain recurrences (HR = 0.18)[29]. These results were sufficient for the United States Food and Drug Administration to approve osimertinib as an adjuvant treatment for NSCLC with EGFR mutations. Recent results from another trial showed that icotinib, a first-generation TKI, improved PFS (HR = 0.36) in patients with resected stages II and IIA disease; results on OS have not been published yet[30]. Nonetheless, in the CTONG trial of adjuvant treatment with gefitinib, the improvement observed for PFS was not carried over to OS, reflecting previous findings for other targeted therapies. It remains to be seen whether osimertinib will achieve a survival benefit in the ADAURA trial.

Little has been reported on the use of biomarkers in this setting, as they were not a requirement in most of the trials conducted to date. Thus, the potential values of BRCA1 and of ERCC1 and thymidylate synthase were not validated in the respective SCAT and ITACA trials. Contradictory results have been reported for the prognostic value of PDL-1 expression and tumor mutational burden[31-33]. Nonetheless, nextgeneration sequencing is a promising strategy for the detection of residual disease after surgery [34,35]. A recent meta-analysis showed that residual molecular disease detected by circulating tumor DNA analysis after complete resection was associated with a higher risk of recurrence and death.

Despite the available evidence, treatment should always be individualized, with careful assessment of risks and benefits, particularly in the current scenario of COVID-19[36].

SBRT IN EARLY-STAGE LUNG CANCER

SBRT is a high-precision technique that delivers high doses of radiation over a short period of time[37]. Conceptually derived from cranial stereotactic radiosurgery, it is



now used in multiple anatomic locations. It is the treatment of choice for early-stage lung cancer in medically inoperable patients or patients who refuse surgery, with a 5year local control rate of 90% [38]. It improves survival in older patients and reduces the number of untreated patients. When SBRT is not feasible, hypofractionated radiotherapy is preferred to conventionally fractionated schedules^[39]. Acute toxicity is rare in SBRT, and includes mild fatigue 1-2 wk after treatment; quality of life is rarely affected [40]. The risk of severe toxicity is low [41], and the most common adverse effect is decreased lung capacity. SBRT can be highly toxic in patients with a history of interstitial lung disease and its use should be assessed by a multidisciplinary committee. Late adverse effects include pain, rib fractures, dyspnea, and ventricular tachycardia^[38]. Other effects are esophagitis, epithelitis, and brachial plexopathy. Complications are largely influenced by tumor location and size, radiation dose and target volume[42]. Pathological confirmation is not always possible, and some authors have suggested that up to 16% of lung nodules may be benign[4].

SBRT has certain technical characteristics that need to be taken into account when planning and administering treatment. Four-dimensional computed tomography (CT) is recommended for preoperative simulation, and multiple beams or arcs should be used for planning purposes as they help limit toxicity[43].

Dose schedules for peripheral tumors vary, but mostly consist of 3-8 fractions of 7.5-20 Gy each; results for a dose of 54 Gy in 3 fractions include a 3-year local control rate of 91%, a 3-year disseminated failure rate of 22%[44], a 5-year local control rate of 80%, and a 5-year local control rate of 31% [41]. A phase II trial comparing 30 Gy in 1 fraction and 60 Gy in 3 fractions showed 2-year survival rates of 71% and 61%, respectively, with no differences in toxicity[45]. On comparing 34 Gy in 1 fraction and 48 Gy in 4 fractions, Nagata et al[46] found OS rates of 61% and 78%, respectively, and no differences in survival, primary tumor control, or toxicity. In their meta-analysis of 34 observational studies involving 2597 patients, Zhang et al[47] determined that the most beneficial dose regimens were those that achieved a biologically equivalent dose of 83.3-146 Gy[47].

Centrally located tumors are tumors located within 2 cm, in any direction, of a critical mediastinal structure, such as the bronchial tree, esophagus, heart, brachial plexus, major vessels, spinal cord, phrenic nerve, and recurrent laryngeal nerve. SBRT is not suitable for ultracentral tumors, but hypofractionated schedules consisting of 6-15 fractions could be considered [48]. Risk-adapted schedules have achieved high local control rates and limited toxicity. Evidence to date shows a 5-year OS rate of 50% and a local control rate of 93% [49,50]. A systematic review of SBRT efficacy and toxicity in centrally located NSCLC showed similar local control and survival rates to those achieved in peripheral tumors.

Three randomized clinical trials have compared SBRT and surgery, although they had problems with accrual. A pooled analysis of the STARS and ROSEL trials showed comparable 3-year recurrence-free survival. Results from the ACOSOG Z4099 trial have not been reported. In the RTOG 0813 trial, 100 medically inoperable patients with central tumors were treated with 50-60 Gy in 5 fractions on alternating days. This resulted in 2-year local control, OS, and PFS rates of 88%, 70%, and 53%, respectively; 15 patients experienced grade 3 or higher toxicity (grade 3, 10 patients; grade 4, one patient; and grade 5, four patients). The standard treatment for patients with operable tumors is surgery, lobectomy, and mediastinal lymph node dissection. The RTOG 0236 [41] and 0915[47] trials showed a 3-year OS rate of 56% over a median follow-up of 4 years and a 5-year OS rate of 40%. The local control and 3-year survival rates were 87.3% and 59.9%, respectively. High recurrence rates, however, were observed in the SBRT group during follow-up[51,52]. Results from the VALOR, SABRTooth, RTOG 3502, and STABLE-MATES trials are pending (Table 1).

When used to treat multiple synchronous tumors vs solitary tumors, SBRT offers similar local control and toxicity rates and worse survival rates⁵³]. The role of SBRT is being investigated in T3-4N0M0 tumors with schedules of 8-10 Gy per fraction in 8-10 fractions. Two-year local control rates of 68%-73.2% have been described [54-56].

A recent study demonstrated that SBRT after contralateral pneumonectomy was safe. Arifin et al [57] analyzed 59 studies with a mean follow-up of 25.4 mo and found a mean 1-year OS rate of 80.6%, a 2-year local control rate of 89.4%, and a grade \geq 3 rate of 13.2%.

RFA IN EARLY-STAGE NSCLC

RFA is a minimally invasive CT-guided procedure originally approved for use in liver



Table 1 Studies analy	vzina s	surgerv	and stereots	etic hod	v radiotherany	in non-small	cell lung cancer
Table T Studies alla	yzing s	surgery	and stereou		y rauloulerapy	III IIUII-SIIIaii	centrung cancer

Ref.	Туре	Surgery-RT, No.	Surgery-RT, Local failure	Surgery-RT, PFS	Surgery-RT, OS	Surgery-RT, Toxicity	LoE
Grills <i>et al</i> [74], 2010	R	69 wedge resection; 58 SBRT; Unfit for lobectomy	20%-4% (<i>P</i> = 0.07)	65% <i>vs</i> 77% (<i>P</i> = 0.37)	87% vs 72% ($P = 0.01$)	Readmission 10%; Pneumonitis 2%; Fracture 11%	3
Varlotto <i>et al</i> [75], 2013	R	48 sublobar resection +132 lobectomy; 137 SBRT	At 5 yr 18.8% lobectomy <i>vs</i> SBRT 12.2% (<i>P</i> = 0.382); Resection 7.1%	No differences (<i>P</i> = 0.378)	At 5 yr lobectomy <i>vs</i> SBRT 33.7%; Resection 86.3% (<i>P</i> = 0.04, <i>P</i> = 0.003)		3
Verstegen <i>et al</i> [76], 2013	R	64 VATS; 64 SBRT; 54% inoperable	At 3 yr 3.1% vs 1.6% (P = 0.04)	79.7% vs 75%	76.9% vs 90.8% ($P = 0.83$)	23.4% vs 6.3% G \geq 3 (P = 0.03)	3
Matsuo <i>et al</i> [77], 2014	R	53 sublobar resection; 53 SBRT	At 5 yr 14.1% vs 28.3% ($P = 0.059$)		55.6% vs 40.4% ($P = 0.124$)		3
Zheng et al [78], 2014	МА	11921; 7071 surgery; 4850 SBRT	At 1 yr 93% lobectomy vs 91.5% sublobar resection vs 96.3% SBRT. At 3 yr 85% vs 78.4% vs 87.8%. At 5 yr 80% vs 63.4% vs 83.9% (P = 0.45)	At 1 yr 93.5% lobectomy vs 90.3% sublobar resection vs 87.1% SBRT. At 3 yr 82.9% vs 82.1% vs 65.8%. At 5 yr 74.8% vs 71.2% vs 65.8% (P = 0.46)	At 1 yr 92.5% lobectomy vs 93.2% sublobar resection vs 83.4% SBRT. At 3 yr 77.9% vs 80.7% vs 56.6%. At 5 yr 66.1% vs 71.7% vs 41.2% HR = 0.52, 95% CI: 0.2-1.36 for lobectomy and HR = 0.49, 95% CI: 0.19-1.3 for sublobar resection		1
Yu et al <mark>[79]</mark> , 2015	R	1078; 711 surgery; 367 SBRT			At 2 yr 77.7% vs 59.9% (P = 0.01)	Acute 54.9% <i>vs</i> 7.9% (<i>P</i> < 0.001). Chronic 73.9% <i>vs</i> 69.7% <i>P</i> = 0.31)	3
Rosen <i>et al</i> [80], 2016	R	1781 lobectomy; 1781 SBRT			At 5 yr 59% vs 29%; 58% vs 40% for patients who refused surgery ($P = 0.010$)		3
Ma <i>et al</i> [81], 2016 (adjusted for operable patients)	MA	6969; 3436 VATS; 4433 SBRT		No differences (<i>P</i> = 0.378)	No differences HR = 2.02, 95%CI: 0.45-3.07 (<i>P</i> = 0.36)		2
Deng <i>et al</i> [82], 2017	MA	13598	No differences ($P = 0.453$)		At 3 yr 68.1% vs 47.7% ($P < 0.001$)		1
Grills <i>et al</i> [74], 2010	P. III	222 Lobectomy; 254 SBRT	At 5 yr 5% <i>vs</i> 8% (<i>P</i> = 0.388)	At 5 yr 72% <i>vs</i> 53% (<i>P</i> = 0.018)	At 5 yr 78% <i>vs</i> 61% (<i>P</i> = 0.006)		1
Ackerson <i>et al</i> [83], 2018	R	151 surgery; 70 SBRT	At 3 yr 10% <i>vs</i> 15% (<i>P</i> = 0.71)	42% vs 29% (P = 0.004)	At 3 yr 63% vs 35% (P < 0.001)	23%-17%	3
Tamura <i>et al</i> [84], 2019	R	141 surgery; 106 SBRT	Higher for SBRT (<i>P</i> = 0.0082)	At 5 yr 69.7%-50.2% (<i>P</i> = 0.036)	At 5 yr 69.7% vs 50.2% (P = 0.036)	8.6% surgery; SBRT G≥27.5%	3

G: Grade; LoE: Level of evidence; MA: Meta-analysis; P: Phase; OS: Overall survival; RT: Radiotherapy; VATS: Video-assisted thoracoscopic surgery.

tumors. It is a percutaneous technique that consists of applying an alternating current (420-500 kHz) to the tumor tissue, resulting in high temperatures (> 70 °C) that cause tissue necrosis and protein denaturation[58].

Because air is a poor conductor of electricity and a good thermal insulator, the lung is theoretically an ideal site for the application of RFA as the surrounding parenchyma is barely affected[59]. The use of RFA to treat lung tumors was first described by Dupuy *et al*[60] in 2000.

The main advantages of RFA over surgery are that it is minimally invasive (percutaneous technique performed with local anesthesia), can be administered on an outpatient basis or under 24-h hospitalization, and does not require thoracotomy[59].

The use of RFA is limited to the treatment of lesions < 3 cm located in the outer twothirds of the lung parenchyma. Tumor size affects the homogeneity of the temperature distribution within the lesion. Tumors > 3 cm require the use of several overlapping electrical fields to achieve a high enough temperature, and this increases the risk of complications. As with surgery, a margin of healthy parenchymal tissue must be included in the radiofrequency field, but this is difficult to achieve because of the



thermal insulation effect mentioned above[61-63].

Central lesions carry a higher risk of complications due to their proximity to the bronchial tree, esophagus, and heart. RFA may be less effective when applied to tumors located close to blood vessels with a diameter > 0.3 cm due to what is known as a "heat sink" effect (a cooling effect caused by the constant renewal of blood within the vessel)[59].

The main adverse effects associated with RFA are pneumothorax [the most common complication (11%-67%) following removal of the electrode from the parenchyma], pleural effusion (related to the increase in pleural temperature), hemoptysis, and more rarely, infections, bronchial fistula, and nerve or cardiac injuries.

In a recent meta-analysis comparing RFA and sublobar resection, Chen et al[59] analyzed four retrospective studies involving 309 patients: 155 treated with RFA and 154 with sublobar resection. The patients who underwent sublobar resection had significantly higher 1- and 3-year OS and PFS rates (97% vs 91% for 1-year OS, 67% vs 52% for 3-year OS, 91% vs 81% for 1-year PFS, and 67% vs 48% for 3-year PFS). Patients in the RFA group had more complications, but they were milder than those seen in the sublobar resection group.

In their prospective phase II trial of 42 patients with inoperable early-stage lung cancer, Palussière et al[64] concluded that RFA was a well-tolerated technique with 1and 3-year local control rates of 84.38% and 81.25%, respectively, and comparable OS rates to those achieved with SBRT. Good tolerability has also been described by other authors[65], including Li et al[61] in their meta-analysis of 1989 patients.

Few studies have compared local treatments (RFA and SBRT), and the little evidence that exists is based on unbalanced, retrospective data. Randomized prospective studies are needed. Authors who have compared RFA and SBRT, however, agree that SBRT should be the technique of choice for inoperable early-stage cancer because of its favorable safety profile and greater survival benefits. RFA, in turn, should be reserved for small tumors not located near vessels or mediastinal structures[66-68].

At the molecular level, hypoxia-inducible factor-1a has been proposed as an independent prognostic marker in the setting of RFA, as high levels have been linked to an increased risk of mortality[69].

In conclusion, RFA may be useful for treating inoperable early-stage lung cancer, in particular tumors < 3 cm located far from the mediastinum and vessels with a diameter > 0.3 cm[70,71]. The poorer outcomes reported for RFA compared with sublobar resection may be due to the lack of randomized, prospective studies comparing the two treatments, as studies to date have included patients who are unfit for surgery, that is older, more frail patients with more comorbidities and as a result a worse prognosis[72,73].

CONCLUSION

Complete resection continues to be the gold standard for the treatment of early-stage lung cancer. Lobectomy remains the gold standard for the treatment of early-stage NSCLC, but there is a growing tendency to perform sublobar resection in selected cases. Alternative treatments such as SBRT and RFA can also produce good outcomes in inoperable patients or patients who refuse surgery.

REFERENCES

- Ginsberg RJ, Rubinstein LV. Randomized trial of lobectomy vs limited resection for T1 N0 nonsmall cell lung cancer. Lung Cancer Study Group. Ann Thorac Surg 1995; 60: 615-622; discussion 622-623 [PMID: 7677489 DOI: 10.1016/0003-4975(95)00537-u]
- 2 Brunelli A. European Society of Thoracic Surgeons Risk Scores. Thorac Surg Clin 2017; 27: 297-302 [PMID: 28647076 DOI: 10.1016/j.thorsurg.2017.03.009]
- 3 Postmus PE, Kerr KM, Oudkerk M, Senan S, Waller DA, Vansteenkiste J, Escriu C, Peters S; ESMO Guidelines Committee. Early and locally advanced non-small-cell lung cancer (NSCLC): ESMO Clinical Practice Guidelines for diagnosis, treatment and follow-up. Ann Oncol 2017; 28: iv1-iv21 [PMID: 28881918 DOI: 10.1093/annonc/mdx222]
- Altorki NK, Wang X, Wigle D, Gu L, Darling G, Ashrafi AS, Landrenau R, Miller D, Liberman M, Jones DR, Keenan R, Conti M, Wright G, Veit LJ, Ramalingam SS, Kamel M, Pass HI, Mitchell JD, Stinchcombe T, Vokes E, Kohman LJ. Perioperative mortality and morbidity after sublobar vs lobar resection for early-stage non-small-cell lung cancer: post-hoc analysis of an international, randomised, phase 3 trial (CALGB/Alliance 140503). Lancet Respir Med 2018; 6: 915-924 [PMID: 30442588



DOI: 10.1016/S2213-2600(18)30411-9]

- Suzuki K, Saji H, Aokage K, Watanabe SI, Okada M, Mizusawa J, Nakajima R, Tsuboi M, 5 Nakamura S, Nakamura K, Mitsudomi T, Asamura H; West Japan Oncology Group; Japan Clinical Oncology Group. Comparison of pulmonary segmentectomy and lobectomy: Safety results of a randomized trial. J Thorac Cardiovasc Surg 2019; 158: 895-907 [PMID: 31078312 DOI: 10.1016/j.jtcvs.2019.03.090]
- Embun R, Royo-Crespo I, Recuero Díaz JL; Spanish Video-Assisted Thoracic Surgery Group. 6 Spanish Video-Assisted Thoracic Surgery Group: Method, Auditing, and Initial Results From a National Prospective Cohort of Patients Receiving Anatomical Lung Resections. Arch Bronconeumol 2020; 56: 718-724 [DOI: 10.1016/j.arbres.2020.01.005]
- 7 Perna V, Carvajal AF, Torrecilla JA, Gigirey O. Uniportal video-assisted thoracoscopic lobectomy vs other video-assisted thoracoscopic lobectomy techniques: a randomized study. Eur J Cardiothorac Surg 2016; 50: 411-415 [PMID: 27174549 DOI: 10.1093/ejcts/ezw161]
- Hernandez-Arenas LA, Lin L, Yang Y, Liu M, Guido W, Gonzalez-Rivas D, Jiang G, Jiang L. 8 Initial experience in uniportal subxiphoid video-assisted thoracoscopic surgery for major lung resections. Eur J Cardiothorac Surg 2016; 50: 1060-1066 [PMID: 27401700 DOI: 10.1093/eicts/ezw189]
- Gao HJ, Jiang ZH, Gong L, Ma K, Ren P, Yu ZT, Wei YC. Video-Assisted Vs Thoracotomy Sleeve Lobectomy for Lung Cancer: A Propensity Matched Analysis. Ann Thorac Surg 2019; 108: 1072-1079 [PMID: 31163131 DOI: 10.1016/j.athoracsur.2019.04.037]
- 10 Song KJ, Flores RM. Commentary: Robot-assisted segmentectomy is safe and expensive-What is the debate? J Thorac Cardiovasc Surg 2020; 160: 1373-1374 [PMID: 32087955 DOI: 10.1016/j.jtcvs.2020.01.014]
- Galvez C, Bolufer S, Corcoles JM, Lirio F, Sesma J, Mafe JJ, Cerezal J. Sublobar minimally invasive 11 surgery vs. stereotactic ablative radiotherapy for early stage non-small cell lung cancer. Mini-invasive Surg 2020; 4:86
- 12 Ohtaki Y, Shimizu K. Anatomical thoracoscopic segmentectomy for lung cancer. Gen Thorac Cardiovasc Surg 2014; 62: 586-593 [PMID: 24791926 DOI: 10.1007/s11748-014-0409-7]
- Mimae T, Okada M. Are segmentectomy and lobectomy comparable in terms of curative intent for 13 early stage non-small cell lung cancer? Gen Thorac Cardiovasc Surg 2020; 68: 703-706 [PMID: 31691886 DOI: 10.1007/s11748-019-01219-y]
- 14 Schneider BJ, Daly ME, Kennedy EB, Stiles BM. Stereotactic Body Radiotherapy for Early-Stage Non-Small-Cell Lung Cancer: American Society of Clinical Oncology Endorsement of the American Society for Radiation Oncology Evidence-Based Guideline Summary. J Oncol Pract 2018; 14: 180-186 [PMID: 29257717 DOI: 10.1200/JOP.2017.028894]
- 15 Tosi D, Nosotti M, Bonitta G, Mendogni P, Bertolaccini L, Spaggiari L, Brunelli A, Ruffini E, Falcoz PE. Anatomical segmentectomy vs pulmonary lobectomy for stage I non-small-cell lung cancer: patients selection and outcomes from the European Society of Thoracic Surgeons database analysis. Interact Cardiovasc Thorac Surg 2021; 32: 546-551 [PMID: 33313840 DOI: 10.1093/icvts/ivaa298]
- Fan J, Wang L, Jiang GN, Gao W. Sublobectomy vs lobectomy for stage I non-small-cell lung 16 cancer, a meta-analysis of published studies. Ann Surg Oncol 2012; 19: 661-668 [PMID: 21769464 DOI: 10.1245/s10434-011-1931-9]
- Bedetti B, Bertolaccini L, Rocco R, Schmidt J, Solli P, Scarci M. Segmentectomy vs lobectomy for 17 stage I non-small cell lung cancer: a systematic review and meta-analysis. J Thorac Dis 2017; 9: 1615-1623 [PMID: 28740676 DOI: 10.21037/jtd.2017.05.79]
- 18 Li H, Liu Y, Ling BC, Hu B. Efficacy of thoracoscopic anatomical segmentectomy for small pulmonary nodules. World J Clin Cases 2020; 8: 2227-2234 [PMID: 32548153 DOI: 10.12998/wjcc.v8.i11.2227]
- 19 Charloux A, Quoix E. Lung segmentectomy: does it offer a real functional benefit over lobectomy? Eur Respir Rev 2017; 26 [PMID: 29070582 DOI: 10.1183/16000617.0079-2017]
- 20 Nakamura K, Saji H, Nakajima R, Okada M, Asamura H, Shibata T, Nakamura S, Tada H, Tsuboi M. A phase III randomized trial of lobectomy vs limited resection for small-sized peripheral nonsmall cell lung cancer (JCOG0802/WJOG4607L). Jpn J Clin Oncol 2010; 40: 271-274 [PMID: 19933688 DOI: 10.1093/jjco/hyp156]
- Burdett S, Pignon JP, Tierney J, Tribodet H, Stewart L, Le Pechoux C, Aupérin A, Le Chevalier T, 21 Stephens RJ, Arriagada R, Higgins JP, Johnson DH, Van Meerbeeck J, Parmar MK, Souhami RL, Bergman B, Douillard JY, Dunant A, Endo C, Girling D, Kato H, Keller SM, Kimura H, Knuuttila A, Kodama K, Komaki R, Kris MG, Lad T, Mineo T, Piantadosi S, Rosell R, Scagliotti G, Seymour LK, Shepherd FA, Sylvester R, Tada H, Tanaka F, Torri V, Waller D, Liang Y; Non-Small Cell Lung Cancer Collaborative Group. Adjuvant chemotherapy for resected early-stage non-small cell lung cancer. Cochrane Database Syst Rev 2015; CD011430 [PMID: 25730344 DOI: 10.1002/14651858.CD011430
- Strauss GM, Herndon JE 2nd, Maddaus MA, Johnstone DW, Johnson EA, Harpole DH, Gillenwater 22 HH, Watson DM, Sugarbaker DJ, Schilsky RL, Vokes EE, Green MR. Adjuvant paclitaxel plus carboplatin compared with observation in stage IB non-small-cell lung cancer: CALGB 9633 with the Cancer and Leukemia Group B, Radiation Therapy Oncology Group, and North Central Cancer Treatment Group Study Groups. J Clin Oncol 2008; 26: 5043-5051 [PMID: 18809614 DOI: 10.1200/JCO.2008.16.4855]
- 23 Goldstraw P, Chansky K, Crowley J, Rami-Porta R, Asamura H, Eberhardt WE, Nicholson AG,



Groome P, Mitchell A, Bolejack V; International Association for the Study of Lung Cancer Staging and Prognostic Factors Committee, Advisory Boards, and Participating Institutions; International Association for the Study of Lung Cancer Staging and Prognostic Factors Committee Advisory Boards and Participating Institutions. The IASLC Lung Cancer Staging Project: Proposals for Revision of the TNM Stage Groupings in the Forthcoming (Eighth) Edition of the TNM Classification for Lung Cancer. J Thorac Oncol 2016; 11: 39-51 [PMID: 26762738 DOI: 10.1016/j.jtho.2015.09.009]

- Wakelee HA, Dahlberg SE, Keller SM, Tester WJ, Gandara DR, Graziano SL, Adjei AA, Leighl NB, 24 Aisner SC, Rothman JM, Patel JD, Sborov MD, McDermott SR, Perez-Soler R, Traynor AM, Butts C, Evans T, Shafqat A, Chapman AE, Kasbari SS, Horn L, Ramalingam SS, Schiller JH; ECOG-ACRIN. Adjuvant chemotherapy with or without bevacizumab in patients with resected non-small-cell lung cancer (E1505): an open-label, multicentre, randomised, phase 3 trial. Lancet Oncol 2017; 18: 1610-1623 [PMID: 29129443 DOI: 10.1016/S1470-2045(17)30691-5]
- Vansteenkiste JF, Cho BC, Vanakesa T, De Pas T, Zielinski M, Kim MS, Jassem J, Yoshimura M, 25 Dahabreh J, Nakayama H, Havel L, Kondo H, Mitsudomi T, Zarogoulidis K, Gladkov OA, Udud K, Tada H, Hoffman H, Bugge A, Taylor P, Gonzalez EE, Liao ML, He J, Pujol JL, Louahed J, Debois M, Brichard V, Debruyne C, Therasse P, Altorki N. Efficacy of the MAGE-A3 cancer immunotherapeutic as adjuvant therapy in patients with resected MAGE-A3-positive non-small-cell lung cancer (MAGRIT): a randomised, double-blind, placebo-controlled, phase 3 trial. Lancet Oncol 2016; 17: 822-835 [PMID: 27132212 DOI: 10.1016/S1470-2045(16)00099-1]
- 26 Remon J, Passiglia F, Ahn MJ, Barlesi F, Forde PM, Garon EB, Gettinger S, Goldberg SB, Herbst RS, Horn L, Kubota K, Lu S, Mezquita L, Paz-Ares L, Popat S, Schalper KA, Skoulidis F, Reck M, Adjei AA, Scagliotti GV. Immune Checkpoint Inhibitors in Thoracic Malignancies: Review of the Existing Evidence by an IASLC Expert Panel and Recommendations. J Thorac Oncol 2020; 15: 914-947 [PMID: 32179179 DOI: 10.1016/j.jtho.2020.03.006]
- Provencio M, Nadal E, Insa A, García-Campelo MR, Casal-Rubio J, Dómine M, Majem M, 27 Rodríguez-Abreu D, Martínez-Martí A, De Castro Carpeño J, Cobo M, López Vivanco G, Del Barco E, Bernabé Caro R, Viñolas N, Barneto Aranda I, Viteri S, Pereira E, Royuela A, Casarrubios M, Salas Antón C, Parra ER, Wistuba I, Calvo V, Laza-Briviesca R, Romero A, Massuti B, Cruz-Bermúdez A. Neoadjuvant chemotherapy and nivolumab in resectable non-small-cell lung cancer (NADIM): an open-label, multicentre, single-arm, phase 2 trial. Lancet Oncol 2020; 21: 1413-1422 [PMID: 32979984 DOI: 10.1016/S1470-2045(20)30453-8]
- Liu J, Blake SJ, Yong MC, Harjunpää H, Ngiow SF, Takeda K, Young A, O'Donnell JS, Allen S, 28 Smyth MJ, Teng MW. Improved Efficacy of Neoadjuvant Compared to Adjuvant Immunotherapy to Eradicate Metastatic Disease. Cancer Discov 2016; 6: 1382-1399 [PMID: 27663893 DOI: 10.1158/2159-8290.CD-16-0577]
- Wu YL, Tsuboi M, He J, John T, Grohe C, Majem M, Goldman JW, Laktionov K, Kim SW, Kato T, 29 Vu HV, Lu S, Lee KY, Akewanlop C, Yu CJ, de Marinis F, Bonanno L, Domine M, Shepherd FA, Zeng L, Hodge R, Atasoy A, Rukazenkov Y, Herbst RS; ADAURA Investigators. Osimertinib in Resected EGFR-Mutated Non-Small-Cell Lung Cancer. N Engl J Med 2020; 383: 1711-1723 [PMID: 32955177 DOI: 10.1056/NEJMoa2027071]
- 30 Liu YT, Hao XZ, Liu DR, Cheng G, Zhang SC, Xiao WH, Hu Y, Liu JF, He M, Ding CM, Zhang L, Wang J, Li H, Dong GL, Zhi XY, Li J, Shi YK. Icotinib as Adjuvant Treatment for Stage II-IIIA Lung Adenocarcinoma Patients with EGFR Mutation (ICWIP Study): Study Protocol for a Randomised Controlled Trial. Cancer Manag Res 2020; 12: 4633-4643 [PMID: 32606956 DOI: 10.2147/CMAR.S240275
- 31 Ilie M, Long-Mira E, Bence C, Butori C, Lassalle S, Bouhlel L, Fazzalari L, Zahaf K, Lalvée S, Washetine K, Mouroux J, Vénissac N, Poudenx M, Otto J, Sabourin JC, Marquette CH, Hofman V, Hofman P. Comparative study of the PD-L1 status between surgically resected specimens and matched biopsies of NSCLC patients reveal major discordances: a potential issue for anti-PD-L1 therapeutic strategies. Ann Oncol 2016; 27: 147-153 [PMID: 26483045 DOI: 10.1093/annonc/mdv489]
- 32 Zaric B, Brcic L, Buder A, Brandstetter A, Buresch JO, Traint S, Kovacevic T, Stojsic V, Perin B, Pirker R, Filipits M. PD-1 and PD-L1 Protein Expression Predict Survival in Completely Resected Lung Adenocarcinoma. Clin Lung Cancer 2018; 19: e957-e963 [PMID: 30197262 DOI: 10.1016/j.cllc.2018.08.014]
- 33 Owada-Ozaki Y, Muto S, Takagi H, Inoue T, Watanabe Y, Fukuhara M, Yamaura T, Okabe N, Matsumura Y, Hasegawa T, Ohsugi J, Hoshino M, Shio Y, Nanamiya H, Imai JI, Isogai T, Watanabe S, Suzuki H. Prognostic Impact of Tumor Mutation Burden in Patients With Completely Resected Non-Small Cell Lung Cancer: Brief Report. J Thorac Oncol 2018; 13: 1217-1221 [PMID: 29654927 DOI: 10.1016/j.jtho.2018.04.003]
- Abbosh C, Birkbak NJ, Swanton C. Early stage NSCLC challenges to implementing ctDNA-based 34 screening and MRD detection. Nat Rev Clin Oncol 2018; 15: 577-586 [PMID: 29968853 DOI: 10.1038/s41571-018-0058-3
- Yin JX, Hu WW, Gu H, Fang JM. Combined assay of Circulating Tumor DNA and Protein 35 Biomarkers for early noninvasive detection and prognosis of Non-Small Cell Lung Cancer. J Cancer 2021; 12: 1258-1269 [PMID: 33442424 DOI: 10.7150/jca.49647]
- Maitland ML, Heyer D, Gomberg-Maitland M. Risk of COVID-19 in Patients With Cancer. JAMA 36 Oncol 2020; 6: 1471 [PMID: 32614413 DOI: 10.1001/jamaoncol.2020.2583]



- 37 Abel S, Hasan S, Horne ZD, Colonias A, Wegner RE. Stereotactic body radiation therapy in earlystage NSCLC: historical review, contemporary evidence and future implications. Lung Cancer Manag 2019; 8: LMT09 [PMID: 31044018 DOI: 10.2217/lmt-2018-0013]
- 38 Lindberg K, Nyman J, Riesenfeld Källskog V, Hoyer M, Lund JÅ, Lax I, Wersäll P, Karlsson K, Friesland S, Lewensohn R. Long-term results of a prospective phase II trial of medically inoperable stage I NSCLC treated with SBRT - the Nordic experience. Acta Oncol 2015; 54: 1096-1104 [PMID: 25813471 DOI: 10.3109/0284186X.2015.1020966]
- Cheung P, Faria S, Ahmed S, Chabot P, Greenland J, Kurien E, Mohamed I, Wright JR, Hollenhorst 39 H, de Metz C, Campbell H, Vu TT, Karvat A, Wai ES, Ung YC, Goss G, Shepherd FA, O'Brien P, Ding K, O'Callaghan C. Phase II study of accelerated hypofractionated three-dimensional conformal radiotherapy for stage T1-3 N0 M0 non-small cell lung cancer: NCIC CTG BR.25. J Natl Cancer Inst 2014; 106 [PMID: 25074417 DOI: 10.1093/jnci/dju430]
- Chen H, Louie AV, Boldt RG, Rodrigues GB, Palma DA, Senan S. Quality of Life After Stereotactic 40 Ablative Radiotherapy for Early-Stage Lung Cancer: A Systematic Review. Clin Lung Cancer 2016; 17: e141-e149 [PMID: 26791542 DOI: 10.1016/j.cllc.2015.12.009]
- 41 Timmerman RD, Hu C, Michalski JM, Bradley JC, Galvin J, Johnstone DW, Choy H. Long-term Results of Stereotactic Body Radiation Therapy in Medically Inoperable Stage I Non-Small Cell Lung Cancer. JAMA Oncol 2018; 4: 1287-1288 [PMID: 29852036 DOI: 10.1001/jamaoncol.2018.1258]
- 42 Bahig H, Filion E, Vu T, Chalaoui J, Lambert L, Roberge D, Gagnon M, Fortin B, Béliveau-Nadeau D, Mathieu D, Campeau MP. Severe radiation pneumonitis after lung stereotactic ablative radiation therapy in patients with interstitial lung disease. Pract Radiat Oncol 2016; 6: 367-374 [PMID: 27068780 DOI: 10.1016/j.prro.2016.01.009]
- Louie AV, Palma DA, Dahele M, Rodrigues GB, Senan S. Management of early-stage non-small cell 43 lung cancer using stereotactic ablative radiotherapy: controversies, insights, and changing horizons. Radiother Oncol 2015; 114: 138-147 [PMID: 25497873 DOI: 10.1016/j.radonc.2014.11.036]
- Timmerman R, Paulus R, Galvin J, Michalski J, Straube W, Bradley J, Fakiris A, Bezjak A, Videtic 44 G, Johnstone D, Fowler J, Gore E, Chov H, Stereotactic body radiation therapy for inoperable early stage lung cancer. JAMA 2010; 303: 1070-1076 [PMID: 20233825 DOI: 10.1001/jama.2010.261]
- 45 Ma SJ, Syed YA, Rivers CI, Gomez Suescun JA, Singh AK. Comparison of single- and five-fraction schedules of stereotactic body radiation therapy for central lung tumours: a single institution experience. J Radiother Pract 2017; 16: 148-154 [PMID: 30713468 DOI: 10.1017/S1460396917000061
- 46 Nagata Y, Hiraoka M, Shibata T, Onishi H, Kokubo M, Karasawa K, Shioyama Y, Onimaru R, Kozuka T, Kunieda E, Saito T, Nakagawa K, Hareyama M, Takai Y, Hayakawa K, Mitsuhashi N, Ishikura S. Prospective Trial of Stereotactic Body Radiation Therapy for Both Operable and Inoperable T1N0M0 Non-Small Cell Lung Cancer: Japan Clinical Oncology Group Study JCOG0403. Int J Radiat Oncol Biol Phys 2015; 93: 989-996 [PMID: 26581137 DOI: 10.1016/j.ijrobp.2015.07.2278]
- 47 Zhang J, Yang F, Li B, Li H, Liu J, Huang W, Wang D, Yi Y, Wang J. Which is the optimal biologically effective dose of stereotactic body radiotherapy for Stage I non-small-cell lung cancer? Int J Radiat Oncol Biol Phys 2011; 81: e305-e316 [PMID: 21658853 DOI: 10.1016/j.ijrobp.2011.04.034]
- Bang A, Bezjak A. Stereotactic body radiotherapy for centrally located stage I non-small cell lung 48 cancer. Transl Lung Cancer Res 2019; 8: 58-69 [PMID: 30788235 DOI: 10.21037/tlcr.2018.10.07]
- 49 Tekatli H, Senan S, Dahele M, Slotman BJ, Verbakel WF. Stereotactic ablative radiotherapy (SABR) for central lung tumors: Plan quality and long-term clinical outcomes. Radiother Oncol 2015; 117: 64-70 [PMID: 26472316 DOI: 10.1016/j.radonc.2015.09.028]
- 50 Senthi S, Haasbeek CJ, Slotman BJ, Senan S. Outcomes of stereotactic ablative radiotherapy for central lung tumours: a systematic review. Radiother Oncol 2013; 106: 276-282 [PMID: 23462705 DOI: 10.1016/j.radonc.2013.01.004]
- Dickhoff C, Dahele M, Paul MA, van de Ven PM, de Langen AJ, Senan S, Smit EF, Hartemink KJ. 51 Salvage surgery for locoregional recurrence or persistent tumor after high dose chemoradiotherapy for locally advanced non-small cell lung cancer. Lung Cancer 2016; 94: 108-113 [PMID: 26973215 DOI: 10.1016/j.lungcan.2016.02.005
- Verstegen NE, Maat AP, Lagerwaard FJ, Paul MA, Versteegh MI, Joosten JJ, Lastdrager W, Smit 52 EF, Slotman BJ, Nuyttens JJ, Senan S. Salvage surgery for local failures after stereotactic ablative radiotherapy for early stage non-small cell lung cancer. Radiat Oncol 2016; 11: 131 [PMID: 27716240 DOI: 10.1186/s13014-016-0706-7]
- 53 Romaszko AM, Doboszyńska A. Multiple primary lung cancer: A literature review. Adv Clin Exp Med 2018; 27: 725-730 [PMID: 29790681 DOI: 10.17219/acem/68631]
- 54 Verma V, Shostrom VK, Kumar SS, Zhen W, Hallemeier CL, Braunstein SE, Holland J, Harkenrider MM, S Iskhanian A, Neboori HJ, Jabbour SK, Attia A, Lee P, Alite F, Walker JM, Stahl JM, Wang K, Bingham BS, Hadzitheodorou C, Decker RH, McGarry RC, Simone CB 2nd. Multi-institutional experience of stereotactic body radiotherapy for large (\geq 5 centimeters) non-small cell lung tumors. Cancer 2017; 123: 688-696 [PMID: 27741355 DOI: 10.1002/cncr.30375]
- Eriguchi T, Takeda A, Sanuki N, Nishimura S, Takagawa Y, Enomoto T, Saeki N, Yashiro K, 55 Mizuno T, Aoki Y, Oku Y, Yokosuka T, Shigematsu N. Stereotactic body radiotherapy for T3 and T4N0M0 non-small cell lung cancer. J Radiat Res 2016; 57: 265-272 [PMID: 26983978 DOI: 10.1093/jrr/rrw023]



- Berriochoa C, Videtic GM, Woody NM, Djemil T, Zhuang T, Stephans KL. Stereotactic Body 56 Radiotherapy for T3N0 Lung Cancer With Chest Wall Invasion. Clin Lung Cancer 2016; 17: 595-601 [PMID: 27301539 DOI: 10.1016/j.cllc.2016.04.007]
- 57 Arifin AJ, Al-Shafa F, Chen H, Boldt RG, Warner A, Rodrigues GB, Palma DA, Louie AV. Is lung stereotactic ablative radiotherapy safe after pneumonectomy? Transl Lung Cancer Res 2020; 9: 348-353 [PMID: 32420074 DOI: 10.21037/tlcr.2020.01.18]
- 58 Abbas G, Pennathur A, Landreneau RJ, Luketich JD. Radiofrequency and microwave ablation of lung tumors. J Surg Oncol 2009; 100: 645-650 [PMID: 20017162 DOI: 10.1002/jso.21334]
- 59 Chen S, Yang S, Xu S, Dong S. Comparison between radiofrequency ablation and sublobar resections for the therapy of stage I non-small cell lung cancer: a meta-analysis. PeerJ 2020; 8: e9228 [PMID: 32509468 DOI: 10.7717/peerj.9228]
- 60 Dupuy DE, Zagoria RJ, Akerley W, Mayo-Smith WW, Kavanagh PV, Safran H. Percutaneous radiofrequency ablation of malignancies in the lung. AJR Am J Roentgenol 2000; 174: 57-59 [PMID: 10628454 DOI: 10.2214/ajr.174.1.1740057]
- 61 Li G, Xue M, Chen W, Yi S. Efficacy and safety of radiofrequency ablation for lung cancers: A systematic review and meta-analysis. Eur J Radiol 2018; 100: 92-98 [PMID: 29496085 DOI: 10.1016/j.ejrad.2018.01.009]
- 62 Yuan Z, Wang Y, Zhang J, Zheng J, Li W. A Meta-Analysis of Clinical Outcomes After Radiofrequency Ablation and Microwave Ablation for Lung Cancer and Pulmonary Metastases. J Am Coll Radiol 2019; 16: 302-314 [PMID: 30642784 DOI: 10.1016/j.jacr.2018.10.012]
- 63 Gao Y, Chen J, Zhang J, Sun L, Zhuang Y. Radiofrequency ablation of primary non-small cell lung cancer: A retrospective study on 108 patients. J BUON 2019; 24: 1610-1618 [PMID: 31646816]
- 64 Palussière J, Chomy F, Savina M, Deschamps F, Gaubert JY, Renault A, Bonnefoy O, Laurent F, Meunier C, Bellera C, Mathoulin-Pelissier S, de Baere T. Radiofrequency ablation of stage IA nonsmall cell lung cancer in patients ineligible for surgery: results of a prospective multicenter phase II trial. J Cardiothorac Surg 2018; 13: 91 [PMID: 30143031 DOI: 10.1186/s13019-018-0773-y]
- 65 Akhan O, Güler E, Akıncı D, Çiftçi T, Köse IÇ. Radiofrequency ablation for lung tumors: outcomes, effects on survival, and prognostic factors. Diagn Interv Radiol 2016; 22: 65-71 [PMID: 2661111] DOI: 10.5152/dir.2015.14378]
- Tetta C, Carpenzano M, Algargoush ATJ, Algargoosh M, Londero F, Maessen JG, Gelsomino S. 66 Non-surgical Treatments for Lung Metastases in Patients with Soft Tissue Sarcoma: Stereotactic Body Radiation Therapy (SBRT) and Radiofrequency Ablation (RFA). Curr Med Imaging 2021; 17: 261-275 [PMID: 32819261 DOI: 10.2174/1573405616999200819165709]
- 67 Lam A, Yoshida EJ, Bui K, Fernando D, Nelson K, Abi-Jaoudeh N. A National Cancer Database Analysis of Radiofrequency Ablation vs Stereotactic Body Radiotherapy in Early-Stage Non-Small Cell Lung Cancer. J Vasc Interv Radiol 2018; 29: 1211-1217.e1 [PMID: 30061058 DOI: 10.1016/j.jvir.2018.04.029
- Ager BJ, Wells SM, Gruhl JD, Stoddard GJ, Tao R, Kokeny KE, Hitchcock YJ. Stereotactic body 68 radiotherapy vs percutaneous local tumor ablation for early-stage non-small cell lung cancer. Lung Cancer 2019; 138: 6-12 [PMID: 31593894 DOI: 10.1016/j.lungcan.2019.09.009]
- 69 Wan J, Ling X, Rao Z, Peng B, Ding G. Independent prognostic value of HIF-1a expression in radiofrequency ablation of lung cancer. Oncol Lett 2020; 19: 849-857 [PMID: 31897199 DOI: 10.3892/ol.2019.11130
- Hiyoshi Y, Miyamoto Y, Kiyozumi Y, Sawayama H, Eto K, Nagai Y, Iwatsuki M, Iwagami S, Baba 70 Y, Yoshida N, Kawanaka K, Yamashita Y, Baba H. CT-guided percutaneous radiofrequency ablation for lung metastases from colorectal cancer. Int J Clin Oncol 2019; 24: 288-295 [PMID: 30328530 DOI: 10.1007/s10147-018-1357-5]
- 71 Ambrogi MC, Fanucchi O, Dini P, Melfi F, Davini F, Lucchi M, Massimetti G, Mussi A. Wedge resection and radiofrequency ablation for stage I nonsmall cell lung cancer. Eur Respir J 2015; 45: 1089-1097 [PMID: 25700387 DOI: 10.1183/09031936.00188014]
- 72 Hiraki T, Gobara H, Iguchi T, Fujiwara H, Matsui Y, Kanazawa S. Radiofrequency ablation for early-stage nonsmall cell lung cancer. Biomed Res Int 2014; 2014: 152087 [PMID: 24995270 DOI: 10.1155/2014/152087
- Iguchi T, Hiraki T, Gobara H, Fujiwara H, Matsui Y, Soh J, Toyooka S, Kiura K, Kanazawa S. 73 Percutaneous radiofrequency ablation of lung cancer presenting as ground-glass opacity. Cardiovasc Intervent Radiol 2015; 38: 409-415 [PMID: 24938905 DOI: 10.1007/s00270-014-0926-x]
- Grills IS, Mangona VS, Welsh R, Chmielewski G, McInerney E, Martin S, Wloch J, Ye H, Kestin 74 LL. Outcomes after stereotactic lung radiotherapy or wedge resection for stage I non-small-cell lung cancer. J Clin Oncol 2010; 28: 928-935 [PMID: 20065181 DOI: 10.1200/JCO.2009.25.0928]
- 75 Varlotto J, Fakiris A, Flickinger J, Medford-Davis L, Liss A, Shelkey J, Belani C, DeLuca J, Recht A, Maheshwari N, Barriger R, Yao N, DeCamp M. Matched-pair and propensity score comparisons of outcomes of patients with clinical stage I non-small cell lung cancer treated with resection or stereotactic radiosurgery. Cancer 2013; 119: 2683-2691 [PMID: 23605504 DOI: 10.1002/cncr.28100]
- 76 Verstegen NE, Oosterhuis JW, Palma DA, Rodrigues G, Lagerwaard FJ, van der Elst A, Mollema R, van Tets WF, Warner A, Joosten JJ, Amir MI, Haasbeek CJ, Smit EF, Slotman BJ, Senan S. Stage I-II non-small-cell lung cancer treated using either stereotactic ablative radiotherapy (SABR) or lobectomy by video-assisted thoracoscopic surgery (VATS): outcomes of a propensity score-matched analysis. Ann Oncol 2013; 24: 1543-1548 [PMID: 23425947 DOI: 10.1093/annonc/mdt026]
- Matsuo Y, Chen F, Hamaji M, Kawaguchi A, Ueki N, Nagata Y, Sonobe M, Morita S, Date H, 77



Hiraoka M. Comparison of long-term survival outcomes between stereotactic body radiotherapy and sublobar resection for stage I non-small-cell lung cancer in patients at high risk for lobectomy: A propensity score matching analysis. Eur J Cancer 2014; 50: 2932-2938 [PMID: 25281527 DOI: 10.1016/j.ejca.2014.09.006]

- 78 Zheng X, Schipper M, Kidwell K, Lin J, Reddy R, Ren Y, Chang A, Lv F, Orringer M, Spring Kong FM. Survival outcome after stereotactic body radiation therapy and surgery for stage I non-small cell lung cancer: a meta-analysis. Int J Radiat Oncol Biol Phys 2014; 90: 603-611 [PMID: 25052562 DOI: 10.1016/j.ijrobp.2014.05.055]
- Yu JB, Soulos PR, Cramer LD, Decker RH, Kim AW, Gross CP. Comparative effectiveness of 79 surgery and radiosurgery for stage I non-small cell lung cancer. Cancer 2015; 121: 2341-2349 [PMID: 25847699 DOI: 10.1002/cncr.29359]
- Rosen JE, Salazar MC, Wang Z, Yu JB, Decker RH, Kim AW, Detterbeck FC, Boffa DJ. Lobectomy 80 vs stereotactic body radiotherapy in healthy patients with stage I lung cancer. J Thorac Cardiovasc Surg 2016; 152: 44-54.e9 [PMID: 27131846 DOI: 10.1016/j.jtcvs.2016.03.060]
- 81 Ma L, Xiang J. Clinical outcomes of video-assisted thoracic surgery and stereotactic body radiation therapy for early-stage non-small cell lung cancer: A meta-analysis. Thorac Cancer 2016; 7: 442-451 [PMID: 27385987 DOI: 10.1111/1759-7714.12352]
- 82 Deng HY, Wang YC, Ni PZ, Li G, Yang XY, Lin YD, Liu LX. Radiotherapy, lobectomy or sublobar resection? Eur J Cardiothorac Surg 2017; 51: 203-210 [PMID: 28186277 DOI: 10.1093/ejcts/ezw272]
- 83 Ackerson BG, Tong BC, Hong JC, Gu L, Chino J, Trotter JW, D'Amico TA, Torok JA, Lafata K, Chang C, Kelsey CR. Stereotactic body radiation therapy vs sublobar resection for stage I NSCLC. Lung Cancer 2018; 125: 185-191 [PMID: 30429018 DOI: 10.1016/j.lungcan.2018.09.020]
- 84 Tamura M, Matsumoto I, Tanaka Y, Saito D, Yoshida S, Kakegawa S, Kumano T, Shimizu Y, Tamamura H, Takanaka T, Takemura H. Comparison Between Stereotactic Radiotherapy and Sublobar Resection for Non-Small Cell Lung Cancer. Ann Thorac Surg 2019; 107: 1544-1550 [PMID: 30458155 DOI: 10.1016/j.athoracsur.2018.10.015]



WJC0

World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1101-1156

DOI: 10.5306/wico.v12.i12.1101

ISSN 2218-4333 (online)

REVIEW

Liver regeneration biology: Implications for liver tumour therapies

Christopher Hadjittofi, Michael Feretis, Jack Martin, Simon Harper, Emmanuel Huguet

ORCID number: Christopher

Hadjittofi 0000-0001-8872-8757; Michael Feretis 0000-0001-9844-9620; Jack Martin 0000-0001-9545-0298; Simon Harper 0000-0001-6016-8049; Emmanuel Huguet 0000-0001-5816-5308.

Author contributions: Hadjittofi C authored text in sections on animal and in vitro models, and alternative regeneration pathways; Feretis M authored text in sections on in vitro and human models; Martin J authored text in sections on zebrafish models; Hadjittofi C, Feretis M and Martin J complex mitogens; Harper S assisted in design of manuscript and manuscript review; Huguet E designed the structure of the overall manuscript and authored text in all sections; all authors have read and approved the final manuscript.

Conflict-of-interest statement: No conflict of interests.

Country/Territory of origin: United Kingdom

Specialty type: Oncology

Provenance and peer review: Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification Grade A (Excellent): 0

Christopher Hadjittofi, Michael Feretis, Jack Martin, Simon Harper, Emmanuel Huguet, University Department of Surgery, Addenbrookes Hospital, NIHR Comprehensive Biomedical Research and Academic Health Sciences Center, Cambridge University Hospitals NHS Foundation Trust, Cambridge CB2 0QQ, United Kingdom

Corresponding author: Emmanuel Huguet, BSc, DPhil, FRCS, MBChB, Surgeon, Surgical Oncologist, University Department of Surgery, Addenbrookes Hospital, NIHR Comprehensive Biomedical Research and Academic Health Sciences Center, Cambridge University Hospitals NHS Foundation Trust, Hills Road, Cambridge CB2 0QQ, United Kingdom. emmanuel.huguet@addenbrookes.nhs.uk

Abstract

The liver has remarkable regenerative potential, with the capacity to regenerate after 75% hepatectomy in humans and up to 90% hepatectomy in some rodent models, enabling it to meet the challenge of diverse injury types, including physical trauma, infection, inflammatory processes, direct toxicity, and immunological insults. Current understanding of liver regeneration is based largely on animal research, historically in large animals, and more recently in rodents and zebrafish, which provide powerful genetic manipulation experimental tools. Whilst immensely valuable, these models have limitations in extrapolation to the human situation. In vitro models have evolved from 2-dimensional culture to complex 3 dimensional organoids, but also have shortcomings in replicating the complex hepatic micro-anatomical and physiological milieu. The process of liver regeneration is only partially understood and characterized by layers of complexity. Liver regeneration is triggered and controlled by a multitude of mitogens acting in autocrine, paracrine, and endocrine ways, with much redundancy and cross-talk between biochemical pathways. The regenerative response is variable, involving both hypertrophy and true proliferative hyperplasia, which is itself variable, including both cellular phenotypic fidelity and cellular trans-differentiation, according to the type of injury. Complex interactions occur between parenchymal and non-parenchymal cells, and regeneration is affected by the status of the liver parenchyma, with differences between healthy and diseased liver. Finally, the process of termination of liver regeneration is even less well understood than its triggers. The complexity of liver regeneration biology combined with limited understanding has restricted specific clinical interventions to enhance liver regeneration. Moreover, manipulating the fundamental biochemical pathways involved would require cautious assessment, for fear of unintended consequences. Nevertheless, current knowledge provides guiding principles for strategies to optimise liver regeneration potential.



Grade B (Very good): B Grade C (Good): 0 Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt p://creativecommons.org/License s/by-nc/4.0/

Received: April 27, 2021 Peer-review started: April 28, 2021 First decision: June 16, 2021 Revised: June 20, 2021 Accepted: November 26, 2021 Article in press: November 26, 2021 Published online: December 24, 2021

P-Reviewer: Zharikov YO S-Editor: Wu YXJ L-Editor: A P-Editor: Wu YXJ



Key Words: Liver; Liver regeneration potential; Regeneration biology; Tumour; Therapies

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: The liver has remarkable regenerative potential, allowing recovery from 90% hepatectomy in some rodent models. Current understanding of liver regeneration comes from *in vitro* and animal models. Liver regeneration is controlled by mitogens acting in autocrine, paracrine, and endocrine ways. Complex cross talk occurs between parenchymal and non-parenchymal cells. Regeneration involves hypertrophy and hyperplasia, with both cellular phenotypic fidelity and transdifferentiation, which come into play according to the nature and magnitude of the injury, and the presence of underlying liver disease. Current knowledge provides guiding principles for strategies to optimise liver regeneration potential in the treatment of liver tumours.

Citation: Hadjittofi C, Feretis M, Martin J, Harper S, Huguet E. Liver regeneration biology: Implications for liver tumour therapies. World J Clin Oncol 2021; 12(12): 1101-1156 URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1101.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1101

INTRODUCTION

The process of liver regeneration is highly complex, and incompletely understood. Moreover, the components of this complexity are multiple. Firstly, liver regeneration may be triggered by a wide range of diverse injury types, occurring in isolation or combination, and including physical trauma, infection, inflammatory processes, direct toxicity, and immunological insults. Commensurate with the range of injuries, the biochemical mechanisms which trigger liver regeneration in the first place are also diverse, but only partly identified. Second, the response to injury is not only dependent on the type of injury, but also its magnitude. For example, liver growth after 30% partial hepatectomy in the rat model is predominantly by hepatocyte hypertrophy (liver growth by hepatocyte volume increase), in contrast to the hyperplasia (liver growth by hepatocyte proliferation) seen after 70% hepatectomy. The mechanism underlying this observation is poorly understood. Third, even in the context of liver regeneration by proliferation, different pathways are activated depending on the magnitude of the injury and the status of the background liver. Thus, when the default pathway of phenotypic fidelity (hepatocytes dividing to produce more hepatocytes, cholangiocytes dividing to produce more cholangiocytes, and so on) fails, alternative pathways are recruited whereby intrahepatic bipotential cells transdifferentiate to hepatocytes or cholangiocytes to meet the deficit. Fourth, the triggers and drivers to liver regeneration are an expanding multitude of cytokines, hormones, and growth factors (collectively referred to as hepatic mitogens), from hepatic and extra-hepatic sources, acting either synchronously or metachronously, each subject to complicated and ill-defined control mechanisms and feedback loops. The mitogen maelstrom is characterized by much redundancy (ablation of particular mitogens is compensated by others) and overlapping 'biochemical promiscuity' (with mitogens impacting on more than one receptor, or intracellular signalling pathway). This degree of overlap and redundancy is understandably a highly valuable evolutionary adaptation to meet the diverse insults the liver is exposed to. Whilst many mitogens have been identified and characterized, the complexity of the interactions make it extremely difficult to assign quantitative relative contributions or importance. Fifth, the complexity of interactions in mitogenic stimuli is further enhanced by the interplay between parenchymal cells (hepatocytes and cholangiocytes) and non-parenchymal cells [Kupffer cells (KC), hepatic stellate cells (HSC), liver sinusoidal endothelial cells (LSEC)], with the latter group, though present in much smaller numbers, playing critical roles. Sixth, the regenerative response is significantly affected not only by the nature of the injury and its magnitude, but also by the health of the underlying liver. Thus, liver regeneration in the face of established steatosis, steatohepatitis, fibrosis, cirrhosis, or biliary outflow obstruction is much altered to that in healthy liver tissue. Seventh, although the processes driving liver



regeneration are only partially understood, those controlling the stop signals, once the liver has grown sufficiently, are even less well defined. Lastly, although many in vitro and animal models are available for the study of liver regeneration, all have their limitations, and their results cannot necessarily be extrapolated to the human situation where information is most limited.

This review provides an overview of liver regeneration biology, and the implications of our current understanding for the treatment of liver tumours. We discuss the subject in separate sections listed below. It is emphasized that the presentation of the subject in this way, though designed to orientate the reader, is somewhat artificial in the context of a biological process characterized by multiple synchronous and overlapping events. There is therefore a degree of overlap between sections, with references made to key events in one section subsequently expanded upon in later ones

Section 1 describes the models of liver regeneration and provides an account of the in vitro, animal, and human models that provide our current knowledge of liver regeneration.

Section 2 describes the very early events post liver injury (provided by the hepatectomy model) and provides an account of the known early triggers to liver regeneration.

Section 3 provides an account of the multiple hepatic mitogens which contribute to initiating and maintaining liver regeneration.

Section 4 describes the contribution of non-parenchymal cells to liver regeneration.

Section 5 describes the 'alternative pathways' of liver regeneration, in which stem cell trans differentiation is recruited as a mechanism to deal with situations when phenotypic fidelity fails.

Section 6 describes the influence of underlying liver disease to liver regeneration.

Section 7 describes current knowledge of the mechanisms underlying ceasing of liver regeneration.

Section 8 considers how our current knowledge of liver regeneration affects therapy for liver tumours currently and in terms of future developments.

SECTION 1: MODELS OF LIVER REGENERATION

Although the clinician's perspective may aim to use understanding of liver regeneration to optimize and develop therapeutic interventions in humans, much of our current knowledge of liver regeneration is based on animal and *in vitro* models. This section describes the historical evolution of liver regeneration research, the current predominant animal models: Rat, mouse, and zebrafish, the modern tissue culture in vitro models, and finally human studies of liver regeneration.

Historical evolution of liver regeneration research

Early research and the flow theory: The history and evolution of animal models used for the study of liver regeneration is described in detail within the excellent review by Mortensen *et al*[1]. The very first liver regeneration research is attributed to Nicolas Eck, a 29-year-old Russian military surgeon, in his investigation of portocaval fistula in dogs[2].

From this early period and into the early 1900s, the prevailing view, referred to as 'the flow theory' hypothesized that liver homeostasis and regeneration could be maintained provided that the liver sinusoids were supplied with mechanical flow of blood, irrespective of its source.

The theory was seemingly supported by experiments showing liver regeneration in dogs after 70% hepatectomy who had undergone total portocaval transposition (thus delivering exclusively systemic venous blood to the sinusoids)[3], and by similar experiments showing liver regeneration in dogs after 42% hepatectomy who had undergone portocaval shunt and arterialization of the hepatic portal stump^[4] (thus delivering exclusively arterial blood to the sinusoids). With hindsight, the interpretation of these results was incorrect, in that in both cases, liver regeneration was in fact supported by growth factors of portal origin spilling into the systemic circulation.

The humoral theory: The concept that constituents of portal blood were essential to liver homeostasis and regeneration only gradually gained acceptance, despite early evidence form Hahn who described liver failure in dogs undergoing portocaval shunts [5]. In the 1920s, Rous and Larimore reported that unilateral portal ligation produced ipsilateral atrophy with contralateral hypertrophy in a rabbit model[6]. From the 1960s



onwards many more investigators pursued the idea of portal flow as critical in liver regeneration, in experiments including those of Marchioro et al[7], who carried out canine split portocaval transposition in which one portal branch is supplied with venous blood from the inferior vena cava and the second portal branch receives portal blood, showing atrophy and hypertrophy of the respective parts of liver parenchyma. Furthermore it was demonstrated that adjusting flow and oxygenation alone did not, in a dog model, compensate for the absence of portal blood[8].

Characterising portal blood constituents: With the recognition of the importance of portal blood came an impetus to define the source and nature of vital portal blood constituents. Thus splanchnic portal flow separation experiments were carried out separating portal flow of distal stomach, duodenum, pancreas and spleen from that of small intestine, with the overall finding that the grafts supplied with small intestinal portal flow atrophied, in contrast to those supplied with portal blood from the upper intestinal tract[9,10].

Thereafter, searches for candidate hepato-trophic factors were carried out by infusing individual growth factors and hormones in portal deprived parenchyma to see if rescue could be achieved. In this way, it was demonstrated that insulin infusion into one portal branch of liver after portocaval shunt could partially rescued atrophy of the liver[11], though insulin was unable to prevent liver atrophy following complete splanchnic evisceration[12].

This portocaval shunt rescue model of experimentation allowed the identification of other factors which promoted liver regeneration including thyroxine (T 3), insulin-like growth factor II, transforming growth factor alpha (TGFa) and hepatocyte growth factor (HGF)[13]. Although portal in origin, the systemic blood dissemination of the factors involved in liver regeneration were shown in canine experiments with autotransplantation of small liver grafts to the jejunal mesentery, then randomising animals to sham surgery or 70% hepatectomy. In contrast to sham surgery, autografts in hepatectomised animals did not atrophy, indicating a growth stimulus via the systemic circulation[14].

Similar results were obtained in parabiosis experiments. Thus, using rats with surgically conjoined systemic circulations, partial hepatectomy in one rat, resulted in liver hypertrophy in the non-hepatectomised rat[15,16].

Thus, the early experiments establishing the underlying principles of liver regeneration were performed using predominantly large animal models. In the more recent era, small animal models have preferentially been used because, as well as providing similar physiology and anatomy to the large animal models, they presented advantages in terms of cost, animal husbandry, rapidity of experimentation, and, in the mouse in particular, greater opportunity for genetic modification as an investigative tool. The sections below follow on to describe the rat and mouse models, with the subsequent evolution to the zebrafish model.

It should be emphasised that drawing conclusions from these different models presents additional complexity per se, in that the observations of one species model may not necessarily be extrapolated to the others. Moreover, even within one species, different liver injury types may present differing characteristics. For example, in the mouse model, epidermal growth factor receptor (EGFR) blockade markedly inhibits liver regeneration after paracetamol injury[17], but only delays it after partial hepatectomy (PH)[18].

Current predominant animal models: rat, mouse, and zebrafish

Rat model: The rat model has gained favour over larger animal models (i.e., dogs, rabbits, baboons and pigs) due to advantages in terms of ethics, costs, and practicalities such as husbandry, handling, and shorter experimental times[19] although their size renders surgery more intricate.

As early as 1931, Higgins & Anderson described a standardized technique for partial hepatectomy in rats, which resulted in liver regeneration[20]. Two decades later, Bucher et al^[15] reported on parabiotic experiments, whereby rats that underwent partial hepatectomy were joined to partner rats with intact livers by way of an abdominal wall anastomosis. The authors found that mitosis increased both within the operated and the intact livers, thus concluding that liver regeneration is influenced by factors in the systemic circulation. In a contemporaneous report of parabiotic rats, Wenneker & Sussman[16] found that liver weight and number of hepatic cells increased both in hepatectomized and "normal" rats, thus reaching the same conclusion. Moolten & Bucher^[21] investigated this further by establishing carotid-tojugular cross-circulation from partial hepatectomy to normal rats, and demonstrating



that DNA synthesis increased in the normal livers, dependent on the extent of hepatectomy in the parabiotic partner. Since these early experiments, a variety of surgical and hepatotoxic rat models have been developed for the study of regeneration in acute and chronic liver disease.

The rat liver consists of four main lobes: middle (38% liver mass), left lateral (30%), right (22%), and caudate (8%)[19]. In descriptions where the paracaval portion is considered separate from the caudate, this amounts to 2% of liver mass. These lobes, and their subdivisions, are analogous to the human liver segments described by Couinaud^[22]. Specifically, the caudate lobe (which consists of the Spiegel lobe and paracaval portion) corresponds to the human segments (Sg) I and IX, the left lobe to Sg II, the left component of the middle lobe to Sg III, the right component to Sg IV, V, and VIII, and the right lobe to Sg VI and VII[23].

The classical surgical model involves a 70% (2/3) hepatectomy, as described by Higgins & Anderson^[20], and remains the most common surgical model for liver regeneration. Impressively, rat liver can completely regenerate within 8 d of 70% hepatectomy[24]. Variations to this model can result in 5%-97% partial hepatectomies, depending on the combination of liver lobes resected^[19]. Impressively, 90% hepatectomy in rats is survivable[25]. Furthermore, survival can (perhaps counterintuitively) be enhanced by suppressing the abrupt early regenerative response of the remnant liver via the mitogenactivated protein kinase pathway, thus rendering regeneration linear in the acute phase[26] or by selective bowel decontamination with gentamicin^[27]. These phenomena point towards a substantial regenerative reserve in rats, which unfortunately is not found in humans and which limits extrapolation from rodent models to humans. Bile duct ligation (BDL) is another commonly used surgical model, which involves dividing the common bile duct between ligatures, thus providing a model for the study of cholestatic disease[28]. Yet another surgical model is portal branch ligation, after which ipsilateral atrophy and contralateral hyperplasia is observed in rats^[29] analogous to human clinical scenarios such as portal vein embolisation (PVE) or associating liver partition and portal vein ligation for staged hepatectomy (ALPPS).

Hepatotoxic models have been extensively studied in both rats and mice, shown in Table 1, with the aim of replicating acute or chronic liver disease. Their mechanisms are also described below in the context of mouse models. The hepatotoxic approach has been used to demonstrate the protective effects of flavonoids[30], thiamine[31] protocatechuic acid[32], Lactococcus lactis in probiotic preparation[33] and 5-methoxytryptophan[34], to mention a few examples. An alternative approach to hepatotoxicity is the manipulation of the cell cycle. Specifically, 2-acetaminofluorene (AAF) has been shown to inhibit hepatocyte proliferation, whilst inducing the proliferation and transdifferentiation of oval cells (hepatic progenitor cells) to hepatocytes after partial hepatectomy[35,36], thus shedding light on alternative liver regenerative pathways.

Mouse model: Although much knowledge on liver regeneration has been generated from partial hepatectomy rat models, the mouse model provides an attractive alternative due to lower costs (mice generally require fewer expensive reagents and less expensive housing)[39] relative ease of handling, and immense experimental potential afforded by genetically altered (transgenic and knockout) mice[19].

Mouse models of liver regeneration have been described in various contexts, including: partial hepatectomy[40], portal branch occlusion[41], bile duct ligation[42], chemical, pharmacological or immune-mediated injury[43-47], and chronic conditions such as non-alcoholic fatty liver disease^[48] and liver cancer^[49].

The lobar anatomy of the mouse and rat liver is broadly similar, and the inferior vena cava is intrahepatic in both species[39]. A significant distinction is the absence of a gallbladder in the rat^[19]. In the mouse, the normal liver consists of seven lobes with the following mass proportions: left posterior (37%), left anterior (12%), right anterior (22%), right posterior (14%), right middle (8%), and two omental lobes (7%)[50]. The classical surgical model in rodents is the partial hepatectomy, which most commonly results in removal of 70% of the liver mass (also referred to as a "2/3 hepatectomy") [19]. Portal branch occlusion can be performed radiologically in humans and in large animals but requires an open surgical approach in rodents[19]. BDL has also been developed as a model of cholestasis^[42], with relevance to the study of malignant biliary obstruction. Although the total BDL rat model has existed for decades, mice have been used more recently in the partial BDL model, whereby (rather than transecting the bile duct between ligatures) a 7-0 needle is ligated onto the duct. When the needle is removed, a reproducibly narrow bile duct lumen is left, which results in less liver necrosis^[42] and may more closely resemble chronic cholestasis. These surgical models are of particular interest with regards to single or staged hepatec-



Table 1 Hepatotoxins used in rodent models						
Toxin	Mechanism	Necrosis pattern				
Acetaminophen (paracetamol)[19,36,37]	Free radical enhancement and Kupffer cell activation	Pericentral				
Carbon tetrachloride[19,30,37]	Free radical enhancement and Kupffer cell activation	Pericentral				
Concanavalin A[37]	T-cell activation; cytokine release; ICAM-1 & VCAM-1 upregulation.	Centrilobular				
D-Galactosamine[19,37]	Uridine metabolite deficiency	Random				
Ethanol[19,31]	Increases production of reactive oxygen species and infiltration of inflammatory cells	None				
Lipopolysaccharide[37]	Kupffer cell activation	Centrilobular				
Thioacetamide[19,37,38]	Increases production of toxic metabolites and reactive oxygen species	Pericentral				

ICAM-1: Intercellular adhesion molecule 1; VCAM-1: Vascular cell adhesion molecule 1.

tomies with or without portal vein occlusion in human patients with liver tumours, where physiological reserve, oncological and technical resectability, as well as liver tumour burden and status of background parenchyma will determine the most optimal approach. However, recapitulating human liver procedures in mouse models is limited by the fact that mice are relatively very small, and (as in humans) there is significant anatomical variability in their hepatic vascular and biliary systems[39]. Furthermore, rodents can typically survive with much smaller liver remnants than humans, and the kinetics of liver regeneration vary between species. Nevertheless, surgical techniques in mice are well established and are characterized by reproducibility and minimal operative mortality[39].

The most frequently used hepatotoxins used to induce liver injury in mouse models are carbon tetrachloride (CCl₄), d-galactosamine, paracetamol (acetaminophen), ethanol^[51] and thioacetamide^[19]. CCl₄ can induce acute and chronic liver injury through its action on cytochrome P450, leading to the production of free peroxide radicals which cause lipid peroxidation of hepatocyte[19]. The disadvantage of the CCl₄ model is the inflammatory and immune response caused during hepatocyte injury, which may confound models of liver regeneration^[19]. D-galactosamine is thought to induce liver injury via intracellular deficiency in uridine metabolites and can additionally induce hepatocyte apoptosis when combined with lipopolysaccharide [19]. Paracetamol is metabolized by cytochrome P450 and in overdose leads to toxic levels of N-acetyl-benzoquinone imine, free radical formation and centrilobular apoptosis/necrosis[19]. The kinetics of liver regeneration after CCl_4 D-Galactosamine and paracetamol-induced injury are similar[19]. Ethanol induces liver injury via mitochondrial dysfunction, oxidative stress, inflammatory cell infiltration and translocation of intestinal bacteria which can then enter the portal and systemic circulation [19]. Finally, thioacetamide leads to oxidative stress *via* its conversion to thioacetamide disulfoxide which increases the production of reactive oxygen species^[19].

In addition to the hepatotoxic models mentioned above, several dietary models are used in mice to model liver disease. These include the 1,4dihydro2,4,6trimethylpyridine-3,5-dicarboxylate (DDC) diet, which leads to biliary injury and regeneration [52], the modified choline-deficient ethionine diet, which leads to hepatocellular injury, steatosis and spread of ductular cells from the portal tract^[53]. More recently, a mouse model with rapid progression from normal liver to non-alcoholic steatohepatitis (NASH), fibrosis, cirrhosis, and hepatocellular carcinoma (HCC) within 24 wk has been described by Tsuchida et al[49]. This was achieved by feeding C57BL/6J mice a western diet (high-fat, high-fructose and high- cholesterol) and administering weekly intraperitoneal doses of CCl₄.

The development of transgenic and knockout mouse models has enabled closer scrutiny of pathophysiological mechanisms with regards to liver regeneration after surgery or chemical/diet-induced injury, also highlighting the importance of the innate and adaptive immune system in liver regeneration[54].

The opportunities offered by these models and their relevance to the treatment of liver tumours in humans will be elaborated in the sections to follow. Table 2, whilst non-exhaustive, gives an impression of the breadth and potential of transgenic and KO mouse models in the study of liver regeneration.

Zebrafish model: Following their discovery in the Ganges River in the late 19th century, zebrafish (Danio rerio) were initially used by embryologists to investigate



Tabl	e 2 Studies of	liver regenera	tion involving transgenic or knockout mice	
Yr	First author	Gene product	Study title	Ref.
1994	Webber	TGF-α	"Overexpression of transforming growth factor-alpha causes liver enlargement and increased hepatocyte proliferation in transgenic mice"	[55]
1996	Cressman	IL-6	"Liver failure and defective hepatocyte regeneration in interleukin-6-deficient mice"	[<mark>56</mark>]
1997	Yamada	TNF	"Initiation of liver growth by tumor necrosis factor: deficient liver regeneration in mice lacking type I tumor necrosis factor receptor"	[<mark>57</mark>]
1998	Greenbaum	C/EBP-β	"CCAAT enhancer-binding protein beta is required for normal hepatocyte proliferation in mice after partial hepatectomy	[58]
1998	Rai	iNOS	"Impaired liver regeneration in inducible nitric oxide synthase-deficient mice"	[<mark>59</mark>]
1998	Roselli	uPA	"Liver regeneration is transiently impaired in urokinase-deficient mice"	[<mark>60</mark>]
1998	Yamada	TNFR- 1TNFR-2	"Analysis of liver regeneration in mice lacking type 1 or type 2 tumor necrosis factor receptor: requirement for type 1 but not type 2 receptor"	[<mark>61</mark>]
2002	Anderson	PPAR-α	"Delayed liver regeneration in peroxisome proliferator-activated receptor-alpha-null mice"	[<mark>62</mark>]
2003	Leu	IGFBP-1	"Impaired hepatocyte DNA synthetic response posthepatectomy in insulin-like growth factor binding protein 1-deficient mice with defects in C/EBP beta and mitogen-activated protein kinase/extracellular signal- regulated kinase regulation"	[<mark>63</mark>]
2003	Strey	C3a/C5a	"The proinflammatory mediators C3a and C5a are essential for liver regeneration"	[<mark>64</mark>]
2004	Borowiak	Met	"Met provides essential signals for liver regeneration"	[<mark>65</mark>]
2004	Mohammed	TIMP3	"Abnormal TNF activity in Timp3(-/-) mice leads to chronic hepatic inflammation and failure of liver regeneration	[<mark>66</mark>]
2004	Nakamura	OSM	"Hepatocyte proliferation and tissue remodeling is impaired after liver injury in oncostatin M receptor knockout mice"	[<mark>67</mark>]
2004	Oe	TGF-β	"Intact signaling by transforming growth factor beta is not required for termination of liver regeneration in mice"	[<mark>68</mark>]
2005	Duffield	DTR	"Selective depletion of macrophages reveals distinct, opposing roles during liver injury and repair"	[<mark>69</mark>]
2005	Mitchell	HB-EGF	"Heparin-binding epidermal growth factor-like growth factor links hepatocyte priming with cell cycle progression during liver regeneration"	[70]
2005	Oliver	MT	"Impaired hepatic regeneration in metallothionein-I/II knockout mice"	[71]
2005	Seki	MyD88	"Contribution of Toll-like receptor/myeloid differentiation factor 88 signaling to murine liver regeneration"	[72]
2006	Fernández	Caveolin-1	"Caveolin-1 is essential for liver regeneration"	[73]
2006	Olle	MMP9	"Matrix metalloproteinase-9 is an important factor in hepatic regeneration after partial hepatectomy in mice"	[74]
2007	Mayoral	Caveolin-1	"Dispensability and dynamics of caveolin-1 during liver regeneration and in isolated hepatic Cells"	[75]
2009	Tumanov	Rag1LT	"T cell-derived lymphotoxin regulates liver regeneration"	[54]
2010	Erhardt	CCR5, CXCR3	"Tolerance induction in response to liver inflammation"	[47]
2010	Liu	GPC3	"Suppression of liver regeneration and hepatocyte proliferation in hepatocyte-targeted glypican 3 transgenic mice"	[76]
2012	Borude	FXR	"Hepatocyte-Specific Deletion of Farnesoid X Receptor Delays But Does Not Inhibit Liver Regeneration After Partial Hepatectomy in Mice"	[77]
2013	Bhave	GPC3	"Regulation of Liver Growth by Glypican 3, CD81, Hedgehog, and Hhex"	[78]
2014	Kong	FGF15	"Fibroblast growth factor 15 deficiency impairs liver regeneration in mice"	[79]
2014	Yang	Lrp5/6	" β -catenin signaling in murine liver zonation and regeneration: a Wnt-Wnt situation!"	[<mark>80</mark>]
2015	Lu	Mdm2	"Hepatic progenitor cells of biliary origin with liver repopulation capacity"	[<mark>81</mark>]
2016	Swiderska- Syn	Cre recombinase	"Hedgehog regulates yes-associated protein 1 in regenerating mouse liver"	[<mark>82</mark>]
2018	Tsagianni	MET	"Combined Systemic Disruption of MET and Epidermal Growth Factor Receptor Signaling Causes Liver Failure in Normal Mice"	[<mark>83</mark>]
2019	Asrud	Epac	"Mice depleted for Exchange Proteins Directly Activated by cAMP (Epac) exhibit irregular liver regeneration	[84]



		in response to partial hepatectomy"	
2019 Fortier	р38а МАРК	"Hepatospecific ablation of p38α MAPK governs liver regeneration through modulation of inflammatory response to CCl 4-induced acute injury"	[<mark>85</mark>]
2019 Modares	s IL-6R	"IL-6 Trans-signaling Controls Liver Regeneration After Partial Hepatectomy"	[<mark>86</mark>]
2019 Zhou	Rictor	"Mammalian Target of Rapamycin Complex 2 Signaling Is Required for Liver Regeneration in a Cholestatic Liver Injury Murine Model"	[87]
2020 Lasching	ger CGRP- RAMP1	"The CGRP receptor component RAMP1 links sensory innervation with YAP activity in the regenerating liver"	[88]
2020 Seguin	Mfrn1, Mfrn2	"The mitochondrial metal transporters mitoferrin1 and mitoferrin2 are required for liver regeneration and cell proliferation in mice"	[<mark>89</mark>]
2020 Xue	GPC3	"Phosphorylated Ezrin (Thr567) Regulates Hippo Pathway and Yes-Associated Protein (Yap) in Liver"	[<mark>90</mark>]

developmental biology[91]. Their relative low cost, rapid development from one-cell embryo to free-swimming larva 5 d post-fertilisation, optical transparency enabling direct observation using light and fluorescent microscopy, and relative genetic conservation compared to the human genome with approximately 70% of human genes having a zebrafish orthologue[92] has led to their role within medical research expanding considerably. In the realm of liver biology, applications include the study of high throughput drug discovery and hepatotoxicity screening, forward genetic screening, heritable and developmental liver diseases, the molecular and cellular factors that contribute to human liver disease, liver cancer biology and liver regeneration[91,93-95]. The research opportunities and disadvantages presented by zebrafish are summarized in Table 3.

Cell types are highly conserved between zebrafish and mammalian livers, with the exception of hepatic immune cells (Kupffer cells), which have not been identified in zebrafish. Whilst zebrafish provide immensely useful models, this difference highlights the caution needed in the extrapolation of results between species. As discussed below, Kupffer cells play an important part in cytokine priming of hepatocytes, implying that a different priming mechanism operates in zebrafish, or that this role is played by different cell type. Cellular morphology and physiology are also largely conserved with zebrafish livers demonstrating similar functions to mammalian livers including secretion of bile, glycogen and lipid storage, insulin responsiveness, ammonia metabolism and the production and secretion of proteins including complement, clotting factors and a protein resembling albumin. The morphological composition of the zebrafish, however, is distinct to the mammalian liver with the liver arranged into 3 Lobes that lack a pedicle that separates the lobes in the mammalian liver. Moreover, the portal architecture of mammalian livers is not observed. In fish, the hepatocytes are arranged into tubules with bile ductules running between two rows of hepatocytes[91].

Liver regeneration in mammalian livers involves a compensatory regeneration with hepatocyte proliferation and hypertrophy. In contrast, zebrafish demonstrate true epimorphic regeneration in response to partial hepatectomy with regrowth of the resected lobe, again highlighting significant inter-species differences. Genome-wide gene expression studies have demonstrated that liver regeneration is the result of a coordinated expression of thousands of genes, and whilst several pathways have been identified as important in liver regeneration in both mammals and zebrafish including WNT, fibroblast growth factor receptor (FGFR) and bone morphogenic protein receptor, in isolation they are insufficient to drive the complex process of liver regeneration. The mechanisms underlying the difference between mammalian liver regeneration and zebrafish, and epimorphic regeneration are still to be elucidated[96].

In vitro models

The observation of interspecies variability, the ethical guiding principles of the 3R principles (replacement, refinement, and reduction of animal testing), and the opportunity of better-defined experimental conditions have motivated the development in vitro models to study liver biology. Thus, two- and three-dimensional (2D and 3D) in vitro models have increased our understanding of the mechanisms of liver injury, hepatotoxicity, and mechanisms of liver regeneration[97].

2D culture models: 2D in vitro liver models, have traditionally used immortalised cell lines such as the HepG2 and the HepaRG cell lines derived from human progenitor cells[98], or mechanically and enzymatically dissociated primary cells[99] expanded on



Table 3 Advantages and disadvantages of zebrafish as a model for human liver pathophysiology

Advantages	Disadvantages				
Vertebrate body plan	Partial genome duplication in teleosts				
Ease of husbandry	Differences in microanatomy and liver architecture				
Inexpensive to maintain	Less conserved physiology than mammalian models				
Large numbers of embryos produced rapidly	Less conserved morphogenesis than mammals				
External development	Less developed cell culture technology				
Optical clarity during development	Poorly developed embryonic stem cell technology				
Zebrafish liver not required for foetal haematopoiesis					
Amenable to forward and reverse genetics					
Molecular conservation of development					
Amenable to high-throughput screening: (1) Phenotype assessment; and (2) Drug/chemical screening					

plastic surfaces, or supported by extracellular matrix (ECM) scaffolding[100]. Though presenting advantages in terms of ease of tissue culture, such systems have limitations; for example, cells lines have fundamentally different gene expression profiles to primary hepatocytes, owing to their immortalised nature^[101]. Primary hepatocytes have some benefits in this regard, but are difficult to source (in the human case), exhibit donor variability [102], and rapidly lose their differentiation and function (such as morphology and toxicant related genes expression) in plastic culture[100,103,104].

The presence of ECM partially addresses these shortcomings. Culturing primary hepatocytes between two layers of collagen, termed sandwich-cultured hepatocytes (SCH)[105], results in extended viability, retained cellular polarity with correct localization of basolateral and canalicular transporters[106] as well as formation of functional bile networks[107]. However, despite their promising properties compared to monolayer cultures, sandwich cultures have their own disadvantages including the barrier to introduced materials created by the collagen layers, and low levels of expression of cell-to-cell adhesion proteins that are critical for cell function and differentiation[108]. As such the role of sandwich culture in the experimental process is often limited to short term studies.

3D culture models: Significant progress has been made using 3D *in vitro* hepatic models with benefits in terms of maturity of hepatocytes, long term viability, and more precise representations of the microenvironment of the in vivo liver[109]. In vitro liver modelling studies with human cells have allowed investigation of liver development, liver disease modelling, liver regeneration, and therapeutic transplantation. Given the complex 3D structure and functional regionalization of the liver, 3D liver models including organoids offer the advantage of more closely recapitulating spatial organization, important cell-cell and cell-ECM contacts that stimulate proliferation, differentiation, liver specific expression, and responsiveness to exogenous stimuli[110]. These advantages have been further emphasised by the use of coculture in such systems, allowing the inclusion of key non-parenchymal cells[97].

Human-induced pluripotent stem cells (hIPSCs) offer an effectively unlimited source of genetically diverse cell lines that can be generated from both healthy and diseased livers. Furthermore, these cells are amenable to genetic modification using the CRISPR technology in order to facilitate disease modelling[111]. hIPSCs have further expanded the opportunity for 3D in vitro culture systems by the development of hepatic organoids from hIPSCs[112]. Thus, reports describe the design of organoids involving multiple cell types by co-differentiating hepatocytes and cholangiocytes [113], or hepatocytes with other supporting cell types including stellate-like and Kupffer-like cells[114]. However, an important limitation is the relative immaturity of hepatocyte-like cells generated from hIPSCs. This is demonstrated by continued alphafetoprotein (α FP), low albumin expression, and distinctive CYP expression and function[115]. The problem of functional maturity has been partially addressed by modifying culture conditions, including the medium composition (e.g., inclusion of specific growth factors, hormones)[115]. More recent approaches to circumvent the disadvantages of hIPSCs have involved the use of primary cells to form organoids.



Thus, Huch et al[116] generated human liver organoids from primary ductal Epidermal Cell Adhesion Molecule positive cells grown in a defined human liver media allowing culture with stable function for over 6 mo, and Justin *et al*[117] describe the formation of biliary organoids from primary cholangiocytes.

In spite of these advances, 3D ex vivo cultures do carry their own drawback including difficulties in controlling cellular distribution, innervation, and vascularization-with the latter aspect of particular importance given the central role of the liver sinusoids to hepatic function.

Repopulation of decellularised livers: Repopulation of decellularised liver scaffolds with cells has offered a further refinement to the in vitro investigation of liver regeneration as well as potential therapeutic opportunities[118].

Earlier approaches to generating functional livers include hepatocyte transplantation in humans[119], as well as in animals[120]. In the latter case, microcarriers and biodegradable polymer scaffolds have been described, resulting in albumin production and clearance of bilirubin and urea metabolites [120]. These efforts have laid the ground for three-dimensional scaffolds[121] which are either biological membranes[122], collagen sponges[123], or synthetic hydrogels[124], and which enable the production of hepatic organoids. In another approach to liver regeneration, chimeric murine models have been developed, whereby mouse liver is extensively repopulated with human hepatocytes, thus permitting the study of liver disease (e.g., viral hepatitis) within humanised organs in vivo[125].

More recently, techniques in whole liver decellularization and repopulation have moved the field further, although significant challenges remain. In general terms, the process involves removal of the liver's cellular and immunogenic components, thus creating a scaffold which retains the ultrastructure and properties of the ECM[126]. This is usually followed by static cultivation of cells (e.g., hepatocytes) and their subsequent infusion into the scaffold.

In one of the first such studies, Uygun et al[127] demonstrated that ischaemic rat livers can be decellularised whilst preserving structural and basement-membranebased components of the ECM, as well as the microvasculature. The investigators achieved decellularisation by portal vein perfusion using sodium dodecylsulfate (an anionic detergent), and repopulation with primary rat hepatocytes via the same route. Recellularised grafts were implanted in rats for 8 h, and after explantation underwent ex vivo blood perfusion for 24 h, demonstrating ongoing hepatocyte metabolic activity. Others have demonstrated that implanting repopulated ECM liver scaffolds into rats which had undergone extended hepatectomy improved liver function and extended their mean lifespan from 16 to 72 h[128]. In the last 10 years, a variety of animal models, decellularisation techniques, repopulation routes and cell sources have been described, with promising outcomes in terms of vascular repopulation[118,129,130], hepatocyte survival[131] as well as formation of biliary duct-like structures and activation of liver detoxification enzymes [132]. One of the commonest sources of liver scaffolds is the rat[118,127,132-137] repopulated with rat hepatocytes (although cholangiocytes[136] and lineages from pluripotent stem cells, mesenchymal cells, and fibroblasts have also been described [137] usually *via* the portal vein. With regards to human tissue, Verstegen et al[138] demonstrated that decellularised human livers can be repopulated with human umbilical vein endothelial cells, leading to re-endothelialisation of the vascular tree. Table 4 presents further examples of the different approaches to liver decellularisation-repopulation developed thus far.

The main challenges in producing a viable whole organ from liver decellularisationrepopulation techniques include heterogeneity of cell engraftment, thrombosis (partly related to incomplete or suboptimally functional endothelium as well as microvascular injury[121,130]), the re-creation of an intact and functional biliary tree, as well as attaining the specific distribution of liver cell types seen in the native healthy organ. Mesenchymal and pluripotent stem cells for repopulation are currently considered attractive research avenues[121] as they may lead to more clinically applicable models.

Human models

The study of human liver regeneration is limited to observational data in the context of clinical pathology and applied therapies, and thus contrasts to the directed experimental approaches possible in animal models. Moreover, access to human liver tissue during the regenerative process is not possible as liver biopsy can only be justified by clinical need, given the risks of the procedure including a measurable mortality [139]. The available observational data comes from a combination of clinical findings, serum biomarker measurements, and imaging. Clinical observations and blood bio markers are subject to difficult interpretation because of the confounding effects of the hetero-



Table 4 Ex	kampi	es of liver (accellularization-repopulatio	on studies		
First author	Yr	Liver scaffold source	Cell source & type	Repopulation route	Outcomes	Ref.
Uygun	2010	Rat	Rat hepatocytes	Portal vein	Recellularised liver grafts implanted in rats, perfused in vivo for 8 h, explanted and assessed after 24 h, demonstrating hepatocyte survival, albumin secretion, urea synthesis and cytochrome P450 expression.	Uygun 2010 [127]
Zhou	2011	Mouse	Human foetal hepatocytes	Portal vein	Recellularised liver matrix implanted in mice, achieving hepatocyte survival after 6 wk, with albumin secretion and cytochrome P450 expression.	Zhou 2011 [<mark>131</mark>]
Ко	2014	Pig	Murine endothelial cells, after scaffold conjugation with rat anti-mouse CD31 antibodies	Portal veinHepatic arteryInferior vena cava	Recellularised liver grafts implanted in pigs, demonstrating good blood flow and patency throughout vascular network over 24 h after transplantation.	Ko 2015 [<mark>130</mark>]
Navarro- Tableros	2015	Rat	Human liver stem-like cells	Portal vein	Loss of embryonic markers, expression of albumin, lactate dehydrogenase and cytochrome P450 subtypes. Production of urea and nitrogen.	Navarro- Tableros 2015[<mark>133</mark>]
Ogiso	2016	Rat	Mouse hepatocytes	Biliary tree; Portal vein	(1) > 80% of cells seeded <i>via</i> biliary tree entered the parenchyma; (2) Approximate 20% of cells seeded <i>via</i> portal vein entered the parenchyma; and (3) Increased gene expression of foetal hepatocyte albumin, glucose 6-phosphatase, transferrin, cytokeratin 19, and gamma-glutamyl transpeptidase, activation of liver detoxification enzymes, formation of biliary duct-like structures.	Ogiso 2016 [132] [PMID 27767181]
Verstegen	2017	Human	Human umbilical vein endothelial cells.	-	Re-endothelialisation of vascular tree, demonstrated by luminal vimentin and von Willebrand Factor/F8 staining.	Verstegen 2017[<mark>138</mark>]
Butter	2018	Rat	Rat hepatocytes	Hepatic artery and portal vein	In vitro demonstration of hepatocyte spread to all liver lobes, with proliferation, and production of aminotransferases, lactate dehydrogenase and albumin.	Butter 2018 [<mark>134</mark>]
Chen	2018	Rat	Rat hepatocytes	Portal vein	None (description of materials and methods).	Chen 2018 [<mark>135</mark>]
Chen	2019	Rat	Rat cholangiocytes Rat hepatocytes	Common bile duct; Portal vein	In vitro viability and function demonstrated by albumin and urea secretion, and gene expression of functional proteins.	Chen 2019 [<mark>136</mark>]
Harper	2020	Rat	Rat bone marrow cells	Portal vein	Stem cells engrafted in portal, sinusoidal and hepatic vein compartments, achieving expression of endothelial cell surface markers for up to 30 d.	Harper 2020 [<mark>118</mark>]
Takeishi	2020	Rat	Human hepatocytes, biliary epithelial cells, and vascular endothelial cells derived from pluripotent stem cells, mesenchymal cells, and fibroblasts.	Biliary tree; Portal vein; Central veins	Auxiliary grafts implanted in rats, achieving in vivo functionality for 4 d.	Takeishi 2020[<mark>137</mark>]

geneity of the study population, diverse pathologies, and varied clinical scenarios even within a defined patient group. Although there are reports of serum biomarkers such as aFP and micro RNAs correlating with liver regeneration, their clinical applicability remains to be established. Combining clinical and serological measurements, scores such as the Acute Liver Failure Study Group index has allowed the identification of patients likely to require liver transplant[140-142].

In this context, the relatively non-invasive nature of modern imaging techniques has provided the main means of assessing liver growth and function, as markers of regeneration. Although liver function correlates well with liver volume in uncompromised livers, this relationship is less clear in patients with pre-existing parenchymal liver[143,144]. Estimation of remnant liver function instead of remnant liver volume is a better predictor of clinical outcome after liver resection in patients with decreased liver function[145]. In order to avoid PHLF, clinicians must ensure that the future remnant liver (FRL) will be sufficient to sustain life. Traditionally, this functionality assessment is made by pre-operatively measuring the volume of the FRL as a surrogate measure of functionality[144]. Volumetry, however, assumes liver parenchymal homogeneity and normal underlying liver function, which are not always present in patients undergoing extensive hepatic resections. This lack of

homogeneity in hepatic function can cause a discrepancy between FRL volume (FRL-V%) and FRL function (FRL-F)[146] which is especially important in patients who present with pre-existing liver disease or who have previously received chemotherapy that resulted in steatotic or microvascular liver changes [146]. As such, FRL-V% cut-off values may not accurately predict the quality of the FRL in some patients, with implication on the development of PHLF and associated mortality. The radiological modalities most used to predict the FLR are outlined below.

Standard liver volumes can be calculated from the patient's body surface area or mass using the formulas originally proposed by Vauthey et al[144]. However, these formulas are limited by subject demographics (healthy individuals) and by their modest correlation to liver sizes calculated by more advanced forms of volumetry [147]. CT volumetry of the liver was first performed on cadavers by Heymsfield *et al* [148] in 1979 and was shown to be accurate within 5% of water displacement volumetry. CT is more commonly used due to its greater accessibility, higher spatial resolution, and short acquisition time. MRI, conversely, offers multiple contrast mechanisms and the ability to assess vascular and biliary anatomy in addition to parenchymal pathology. Additionally, MRI also minimises the risk of contrast induced nephrotoxicity and eliminates concerns of radiation exposure[149]. Liver segmentation has emerged as the preferred technique CT volumetry can be used to calculate the volume of the FLR and is widely used to exclude patients from liver resection or to select patients who will benefit from a procedure to increase the volume of the future remnant, such as PVE[150]. However, the outcomes of previous reports correlating the findings of CT volumetric analysis of the future remnant with post-resectional outcome, have not been consistent and the role alternative imaging modalities has been examined[150,151].

Hepatobiliary scintigraphy using 99mTc-iminodiacetic acid analogues, such as 99mTc-mebrofenin, can be used to measure segmental liver function. 99mTcmebrofenin is excreted into the bile by adenosine triphosphate-dependent export pumps the multidrug-resistance-associated proteins 1 and 2 without undergoing biotransformation during transit through the hepatocytes [146,152,153]. Previous reports in the literature have shown that 99mTc-mebrofenin hepatobiliary scintigraphy (HBS) can provide clinicians with information on FRL-F instead of volumetric information alone^[153]. HBS provides visual and quantitative information of global and regional liver function as well as excretory function (intrahepatic and extrahepatic bile transport). 99mTc-mebrofenin is intravenously injected and consequently excreted in bile by the hepatocytes without undergoing biotransformation. As such, the clearance measurement of Technetium-99m mebrofenin using scintigraphy can quantify hepatic function [146,153]. FRL-F assessment using HBS has been proven to be superior to volumetry in the prediction of PHLF and PHLF(M), making HBS the imaging modality of choice prior to proceeding with major hepatectomy. Reports in the literature have illustrated that an HBS cut-off value of 2.7%/min/m² can outperform volumetry cut-off values in the prediction and prevention of PHLF and PHLF(M) by identifying high-risk patients with borderline predicted remnant liver function, and consequent selection for pre-operative PVE or other hypertrophic strategies (e.g., ALPPS[154,155]. Certain hepatobiliary units have already implemented HBS in favor of CT volumetry before hepatic resection based on emerging evidence in the literature.

SECTION 2: EARLY EVENTS POST LIVER INJURY AND TRIGGERS TO LIVER REGENERATION

The PH model in rodents has allowed the examination of immediate events which occur within minutes of liver resection and provides an insight into the mechanisms that trigger the process of liver regeneration. These early events relate to vascular portal flow, tissue hypoxia, haemostatic mechanisms, and changes in extracellular matrix integrity.

Vascular events

Following PH, the increased portal blood flow through the remnant liver exerts a heightened shear stress on the LSECs[156]. Shear stress on LSEC induces numerous physiological changes[157] including microscopically visible ones such as increased sinusoidal diameter and changes to LSEC fenestrae and sieve plates[158,159]. Shear stress also induces biochemical responses including the release of vascular endothelial cell growth factor (VEGF) from LSEC[160], the secretion of VEGF and HGF from



hepatic stellate cells[161], and the LSEC production of nitric oxide (NO) by Nitric Oxide synthase (NOS), which increases hepatocyte sensitivity to HGF[162,163]. The physiological importance of NO is suggested by the finding that inhibition of NOS severely impairs liver regeneration in mice after PH[59].

Shear stress also induces the hepatocyte priming cytokine interleukin 6 (IL6) in LSEC[164], as well as expression in of liver regeneration associated WNT, VEGF, and epithelial cell adhesion molecules in hepatic progenitor cells^[165].

Another consequence of increased portal flow through the remnant liver is increased exposure to lipopolysaccharide (LPS), which is derived from gut bacteria, and which translocates from the gut into portal blood. PH increases the concentration of LPS in the remnant liver not only because of diversion of more portal blood to the remnant liver, but also because the rise in portal pressure increases intestinal permeability, allowing greater LPS translocation [166,167]. In the sinusoids, LPS binds Toll like receptors (TLR) on Kupffer cells, resulting in the secretion of the hepatocyte priming cytokines IL6 and tumour necrosis factor alpha TNF α [168], in a signalling pathway that is dependent on myeloid differentiation factor 88[169].

The increased expression of liver regeneration promoting biochemicals is not confined to the liver. Following PH in the rat, increased expression of VEGF, HGF, and hypoxia inducible factor (HIF) is also observed portal vein drained tissues such as the spleen and small intestine, whereupon portal VEGF concentrations exceed those of the systemic circulation. The mechanism stimulating this extrahepatic expression of growth factors from portal drained tissues is unclear but may also be related to portal pressure changes[170].

Hypoxia

Following PH, increased portal flow brings about a reflex arterial vasoconstriction (the arterial buffer response), which can result in hypoxia in the remnant liver, given the low partial pressure of oxygen (pO2) in portal venous blood[171].

An important outcome of hypoxia is the induction of HIF, which in turn leads to the activation of multiple genes involved in tissue adaptations to hypoxia ranging from glycolytic metabolism to angiogenesis[172].

In the liver, PH leads to increased expression of HIF and subsequently VEGF[173]. In elegant experiments, Dirscherl et al[174] show that the hypoxic environment triggers hepatic stellate cell expression of HIF, resulting in increased expression of VEGF, which then elicits a range of responses in LSEC including proliferation and angiogenesis, but also genes associated with matrix remodelling (discussed in a later section), and LSEC expression of the potent hepatocyte mitogen HGF[174], as well as other liver regenerative genes[175]. Thus, the authors suggest that HSC function as hypoxia sensors in the liver, and trigger angiogenesis in liver regeneration, highlighting the complexity of intercellular cross-talk in this process. In addition, hypoxia induced secretion of complex regeneration promoting molecules from stem cells at extra-hepatic locations may also contribute to promotion of liver regeneration [176].

Haemostasis related factors

The injury to liver tissue in PH results in the activation of mechanisms for haemostasis carried out by platelets and the coagulation cascade. The role of platelets is not confined to haemostasis, but also includes functions relating to liver regeneration [177], and studies in animals and humans suggest impaired liver regeneration in individuals with low platelet counts[178,179].

Following PH, platelets migrate to the space of Disse, where they release liver regeneration promoting biochemical including serotonin, VEGF, and HGF from secreted cytoplasmic granules[180]. In addition to growth factor containing vesicles, platelets contain cytoplasmic RNA, which can be transferred to nearby hepatocytes, resulting in gene expression, and promoting hepatocyte proliferation[181]. Finally, platelets may stimulate liver regeneration by activation of immune cells which also have an important role in cellular cross-talk[180].

In addition to the role of platelets, the coagulation cascade both individually and in combination with 'damage associated molecular patterns' (DAMPs) (including mitochondrial DNA and peptides)[182], activates elements of the complement cascade [183]. These include C3a and C5a, which have a role in stimulating pathways involved in the priming of hepatocytes[184], enabling them to respond to growth factors, as discussed in "Priming of hepatocytes".

Other elements of the coagulation cascade may also play key roles in liver regeneration. Thus, Groeneveld et al[185] report that intrahepatic deposition of fibrinogen after PH is a key driver to platelet accumulation in the liver. Fibrinogen



depletion was associated with impaired liver regeneration in a mouse model, and in humans undergoing liver resection, low intrahepatic fibrinogen and low post op serum fibrinogen levels were associated with poor liver function an increased mortality.

ECM changes

Urokinase-type plasminogen activator (uPA) activity increases within one minute of PH in rats[186], resulting in the activation of plasminogen to plasmin, which then activates key metalloproteinases (MMP) such as MMP-9[187,188], which remodel the hepatic ECM, where HGF is present in its inactive form. uPA also activates HGF to its active form[189], releasing it locally in the liver parenchyma and also into the circulation in significant quantities[190]. uPA knockout mice show impaired liver regeneration[60]. As well as HGF, the ECM contains other inactive forms of growth factors including HB-EGF and fibroblast growth factor (FGF)[191]. Moreover, the importance of matrix alteration in the initiation goes beyond the release of growth factor stores in the ECM in that hepatocyte response to key growth factors is ineffective in the presence of intact ECM, and that ECM changes are required for growth factor driven hepatocyte proliferation[192,193].

Thus, the rapid action of uPA following PH provides a mechanism to kick start the liver regenerative process by liberating ECM stored growth factors, until such time as other mechanisms begin to contribute to maintaining the liver regenerative process.

SECTION 3: HEPATIC MITOGENS

Liver regeneration is characterized by hypertrophy and rapid proliferation allowing return to the starting volume of liver even if recovering from a 25% remnant in humans, or a 10% remnant in some rodent models. The proliferation of hepatocytes is controlled by a maelstrom of growth factors with different but overlapping effects. Within this complexity exists a hierarchy of functions, whereby hepatocytes first require to be primed ("Priming of hepatocytes"), after which they become responsive to a range of mitogens referred to as complete ("Complete mitogens") auxiliary ("Auxiliary mitogens"), and complex ("Complex mitogens"). "Intracellular signalling pathways" summarizes the intracellular pathways which transmit the effect of the growth factors described in sections "Priming of hepatocytes" - "Complex mitogens".

Priming of hepatocytes

Hepatocyte transition from G0 to G1: Although proliferation of hepatocytes is stimulated by a wide range of biochemicals in response to injury, most hepatocytes in uninjured liver do not proliferate[194], although there is some heterogeneity in this regard as discussed in "Hepatocyte response heterogeneity after PH" entitled 'hepatocyte regenerative heterogeneity'. The stimulus to proliferation from the multitude of mitogens requires hepatocytes to be 'primed', a complex phenomenon characterized by the induction of > 100 genes[195], which then enables the hepatocytes to respond to these mitogenic stimuli.

Although cell cycle biology is outside the scope of this review, a brief summary of key events is useful to frame the subsequent sections relating to the priming effects of cytokines and proliferative stimulus of mitogens on hepatocytes.

The cell cycle is divided into 2 main phases: mitosis (the actual process of cell division) and interphase (the phase preparing the cell for mitosis). Interphase is further divided into 3 stages, which, in order, consist of the G1 phase (during which the cell synthesises protein and organelles), S phase (during which DNA is replicated) and G2 phase (during which the machinery for mitosis is assembled). Although some cells undergo this cycle continuously, others exit the cycle and enter a stationary phase G0. In order for a cell in G0 to replicate, it first needs to be 'primed' by molecular signals to return to G1, whereupon a different set of signals will determine the speed of replication and how long it continues.

Hepatocytes provide an example of this situation, and are, in the absence of injury, almost entirely in G0[194]. Their proliferation therefore requires priming factors to return them to G1. The priming function is carried out by cytokines $TNF\alpha$ and IL6.

Thus the current working model[184] (illustrated in Figure 1) suggests that the cytokine priming mechanism starts with the activation of nuclear factor-kappa B (NFKB) in Kupffer cells. NFKB activation may be triggered by several stimuli including (1) Binding of TNFα to its receptor; (2) Binding of complement components C3a & C5a to their receptor; or (3) Binding of lipo-polysaccharide to the TLR receptor. Activation





Figure 1 Cytokine priming of hepatocytes. PH induced increase in portal pressure exerts sheer stress on LSEC inducing IL6 secretion. Gut derived LPS, complement components C3 & C5, ICAM1, and LTXa from T lymphocytes all induce IL6 and TNF expression from Kupffer cells. IL6 & TNFa prime hepatocytes after binding to IL6R and TNFaR. LTXa also acts directly on hepatocytes via the TNFaR. LSEC: Liver sinusoidal endothelial cell; IL6: Interleukin 6; IL6R: Interleukin 6 receptor; LPS: Lipopolysaccharide; TLR: Toll-like receptor; TNFα: TNF alpha; TNFαR: TNF alpha receptor; CtR: Complement receptor; ICAM1: Intercellular adhesion molecule 1; ICAM1R: Intercellular adhesion molecule 1 receptor; LTXα: Lymphotoxin alpha; LTXαR: Lymphotoxin alpha receptor.

of NF κ B results in increased expression of both TNF α and IL6. TNF α may stimulate its own further expression in the Kupfer cell in an autocrine manner. IL6 binds IL6R on hepatocytes, producing activation of signal transducer and activator of transcription 3 (STAT3), which results in the transcription of multiple other genes which push hepatocyte from G0 to G1, thus priming the cell to be responsive to circulating growth factors.

Crucially, in vivo, infusion of the powerful complete mitogens EGF and HGF produces only modest hepatocyte proliferation, whereas marked hepatocyte proliferation is observed if EGF and HGF infusion is preceded by the priming effect of a single TNF α injection[196].

Consistent with this model the following events are observed in the minutes after PH: (1) TNFα and IL6 mRNA and protein increase immediately[197,198]; and (2) Activation of the transcription factors NFkB and STAT3[199,200]. Moreover, DNA replication in hepatocytes is blocked by TNFa antibodies[201], TNF receptor (TNFR) [57] and IL6[56] knockout mice show impaired liver regeneration, and liver regeneration in TNFR knockout mice is rescued with IL6 infusion[57].

Of note, highlighting the necessary caution needed before extrapolating between animal models, TNF α levels after PH differ between rats and mice, with higher levels in rats. Also, the model exemplifies the recurring theme of redundancy in the system with the TNFα knockout mice showing normal liver regeneration because of the ability of other ligands to bind the TNFR[202]. Similarly, the activation of STAT3 may be achieved by other cytokines than IL6, such as Stem Cell Factor[203] and Oncostatin [203].

Triggers to cytokine priming: The initial triggers to expression of the priming cytokines TNFα and IL6 after PH are doubtless numerous and not all identified, but at least 5 stimuli have been demonstrated.

Firstly, PH results in an immediate increase in portal venous pressure which causes a sheer stress on liver sinusoidal endothelial cells[156]. This physical stimulus has many consequences [159] which are discussed in more detail in section 2 on early events post hepatectomy, but which include the induction of IL6 expression in LSECs [164], thereby contributing to the priming of hepatocytes.

Secondly, another trigger to cytokine expression after PH is binding of LPS derived from gut bacteria and translocated to portal blood, to the TLR, and producing expression of IL6 and TNFα. The increase in portal pressure resulting from PH increases gut permeability and may therefore result in exposure of the remnant liver to higher concentrations of LPS[204]. Supporting the physiological relevance of this hypothesis, is the observation that rodents with germ free guts have impaired liver regeneration [205]. The effects of LPS on liver regeneration may not be limited to induction of the priming molecules IL6 and TNFα, but also producing an increase in secretion of hepatic mitogens including insulin, epidermal growth factor, and triiodo-



thyronine[204] (see "*Complete mitogens*" and "*Auxiliary mitogens*" on complete and auxiliary mitogens).

Thirdly, it is also known that binding of complement cascade components C3 and C5 to the complement receptors on Kupfer cells also triggers an NF κ B dependant increase in both IL6 and TNF. Thus, complement activation resulting from physical injury to liver in PH may also contribute to the initiation of cytokine priming of hepatocytes. The significance of this mechanism is suggested by the finding that following PH, C3-5 knockout mice show diminished activation of NF κ B and STAT3, decreased expression of TNF α and IL6 impaired liver regeneration[64].

Fourthly, it is observed that mice lacking the receptor intercellular cell adhesion molecule 1 (ICAM1) show diminished TNF α and IL6 expression and impaired liver regeneration after PH. It is thought that leucocytes, attracted to a liver injury site may mediate triggering of ICAM 1 on Kupfer cells, thus providing another stimulus to initiating the cytokine cascade[206].

Fifth, it is known that the TNFR may be activated not only by TNF α , but also by the protein lymphotoxin alpha (LTx α). This is markedly upregulated in intra-hepatic T lymphocytes after PH[207] and may thus allow T cells to contribute to initiation of the cytokine cascade by activation of TNFR on Kupffer cells. Consistent with this, mice lacking both TNF α and LT α show impaired liver regeneration[208]. Moreover, LTx α may act directly on hepatocytes.

Thus, having been primed by the initial injury triggered cytokine cascade, hepatocytes return to the G1 phase of the cell cycle where they are susceptible to stimulation by mitogens including growth factors, hormones and other biochemicals to accelerate the rate of proliferation.

The concept of mitogen hierarchy: A multitude of different hepatocyte mitogens have been identified which originate from a variety of different tissues, different cell types within a given tissue, acting *via* different receptors, or sometimes overlapping in their receptor binding, and producing a variety of different effects on the target hepatocyte. This complexity exemplifies a key feature of liver regeneration biology, which is the existence of high levels of redundancy, presumably an evolutionary outcome enabling the liver to cope meet the wide range of physical, biochemical and infectious injuries it may encounter.

Amongst this complexity however, as arisen the concept of a hierarchy amongst hepatic mitogens, classifying them as 'complete mitogens', 'auxiliary mitogens', and 'complex mitogens' [209]: (1) Complete mitogens cause proliferation of hepatocyte cultures in serum- free media, and, when injected into whole animals, cause liver enlargement and hepatocyte DNA synthesis. Moreover, ablation of both the MET and EGFR pathways leads to complete inhibition of liver regeneration. The complete mitogens are (a) Hepatocyte growth factor which binds to its receptor MET; and (b) Ligands of the EGFR: EGF, transforming growth factor- α (TGF α), heparin- binding EGF- like growth factor and amphiregulin; (2) Auxiliary mitogens do not cause hepatocyte proliferation in culture in serum free media, do not cause hepatocyte DNA synthesis and liver enlargement when injected *in vivo*, and ablation of their signalling pathways delays but does not abolish liver regeneration. The auxiliary mitogens are noradrenaline and the α 1- adrenergic receptor, VEGF and its receptors (VEGFR1 and VEGFR2), bile acids, serotonin, insulin, and growth hormone; and (3) Complex mitogens are the third category and are much less well defined than complete or auxiliary mitogens, with pathways involving multiple overlapping extracellular signals, disruption of which delays but does not abolish liver regeneration. The complex mitogens are the proteins involved in the *Wnt*, β -catenin, Hippo and Yap pathways.

Complete mitogens

Hepatocyte growth factor: HGF was the first complete hepatic mitogen, identified in 1984 with the human homolog cloned in 1989[210]. Thus, HGF produces hepatocyte proliferation in serum free media *in vitro*, and liver enlargement when infused *in vivo*. HGF mediates its effect on hepatocytes by binding to its receptor MET, a receptor tyrosine kinase with wide ranging roles in diverse areas of cell biology including not only cell survival and proliferation[211], but also metabolism[212], growth and development[213]. MET signalling is dependent on the transcription factor CCAAT/enhancer-binding protein beta C/EBP beta[214], and Inhibition of MET signalling results in blocking of mitosis and increased expression of apoptosis genes after PH[215].

Zaishidenq® WJCO | https://www.wjgnet.com

After PH, HGF is mobilised in a biphasic manner, first with the activation and recruitment of ECM bound inactive HGF in the immediate minutes after PH, and then secondly by secretion of newly expressed HGF in a second wave.

Thus, whilst HGF is bound in inactive form in the ECM in resting liver[216], ECM remodelling[187] resulting from PH results in activation of HGF with binding to hepatocytes and released into the circulation [217], which peaks 30-60 min after PH.

Thereafter, peaking at 24 h post PH[218], a second wave of HGF is observed, newly synthesized by LSEC and stellate cells in the liver, but also from extra-hepatic cells and tissues including platelets^[219], lung^[220], kidney, spleen^[221], thyroid, brain, and salivary glands^[221]. In spite of these multiple sites of HGF production, experiments using genetically altered mice showed that inhibiting HGF production specifically in LSECs resulted in impaired liver regeneration, suggesting that extra-hepatic HGF production cannot compensate for depletion of hepatic HGF production[222]. The factors that stimulate HGF expression in the second wave after its release from ECM include noradrenergic signals[223], insulin like growth factors[224].

Epidermal growth factor: Ligands of the EGFR make up the other known complete mitogens. EGF is one member of a family of 7 Ligands which bind a group of 4 receptors (EGFR/ErbB1, HER2/ErbB2, HER3/ErbB3 and HER4/ErbB4)[225]. Of the 7 known ligands, the ones that relate to liver regeneration are epidermal EGF, transforming growth factor-alpha (TGFa), Amphiregulin (AR), and Heparin bound EGF-like growth factor (HB-EGF), with their role illustrated in Figure 2.

EGF is a complete mitogen and produces hepatocyte proliferation in vitro and in vivo when infused[226]. It is produced in many tissues[227], but the most relevant sites of production are the Brunner glands of the duodenum which provide a constant supply of EGF to the liver via the portal vein[228]. EGF production is increased by noradrenaline which is secreted during the physical stress of PH[229]. EGF is also produced in significant quantities in salivary glands, and sialadenectomised rats do show impaired liver regeneration after hepatectomy [230].

TGFa is produced by hepatocytes themselves during liver regeneration^[231] and therefore functions as a mitogen in an autocrine or paracrine way [232]. TGF α knockout mice have no liver regeneration deficiency however, presumably as a result of the considerable redundancy in the EGFR signalling pathway[233].

Amphiregulin, like TGFa, is produced by hepatocytes. Its expression is in part regulated by inflammatory mediators providing a mechanism for its upregulation following PH. Its significance is suggested by the observation that AR knockout mice have impaired liver regeneration[234,235].

HB EGF is produced Kupffer cells and sinusoidal endothelial cells[236]. Its expression seems to be in part determined by the magnitude of liver resection, as it is increased in 2/3 PH but not 1/3 PH. Its physiological significance is emphasised by the fact that HBEGF transgenic mice[237] and HB EGF knockout mice[70] have accelerated and delayed liver regeneration, respectively.

In the midst of these multiple ligand binding events, EGFR activation peaks at 60 minutes post PH[238], and ablation of EGFR by antisense RNA impairs liver regeneration[239].

Auxiliary mitogens

Bile acids: Primary bile acids are synthesized in the liver by a multistep oxidative metabolism of cholesterol and secreted in bile. In the intestine, bile acids emulsify fats thus facilitating their digestion. Bile acids are metabolized by gut bacteria to produce secondary bile acids, and although some are lost through faecal excretion, a significant proportion are reabsorbed in the gut and recycled in the liver, in the entero-hepatic circulation^[240].

Above a certain concentration, bile acids are toxic to liver and may induce apoptosis and necrosis, such that bile salt synthesis is tightly regulated by means of negative feedback loops involving bile acid receptors in the ileum[241]. At non-toxic concentrations however, bile acids play an important part in regulation of liver regeneration. Both the negative feedback controls and liver regenerative roles are mediated by bile acid receptors which comprise the extracellular TGR5 receptor (TGR5) on Kupffer cells [242], and intracellular Farnesoid X receptor (FXR) within hepatocytes[243].

Bile acids were first investigated as candidate factors for controlling liver regeneration in part because of their exclusively hepatic synthesis, offering the potential for a feedback loop hepatostatic mechanism. Thus, dietary bile acid supplementation was found to produce hepatomegaly in mice with non-injured livers, and increase liver regeneration after PH, in an effect that was dependent on the FXR. Conversely, bile acid sequestering agents resulted in impaired liver regeneration[243].



Hadjittofi C et al. Liver regeneration and liver tumour therapies



Figure 2 Summary of ligand binding to epidermal growth factor receptor in liver regeneration. Endocrine EGFR signalling by EGF from Brunner's glands and salivary glands. Paracrine EGFR signalling by HB EGF from LSEC and Kupffer cells, autocrine EGFR signalling by amphiregulin and TGFa from hepatocytes. LSEC: Liver sinusoidal endothelial cell; EGF: Epidermal growth factor; EGFR: Epidermal growth factor receptor; HB EGF: Heparin bound EGF-like growth factor; TGFa: Transforming growth factor a.

Furthermore, genetically engineered bile salt deficient mice also show impaired liver regeneration after PH[244], and rats having undergone PH with biliary fistula also show impaired liver regeneration, which can be rescued by intestinal delivery of bile acids[245].

After PH, serum bile acid concentration increases in blood within minutes, peaks at 24 h, and diminishes again by 48 h. The mechanism of this serum bile acid increase is not fully understood, but may involve neurological pathways activated by PH related changes in portal pressure[246], consistent with the observation that bile acid increase is also seen after portal vein embolization[247].

The binding of bile acids to the FXR stimulates activation of transcription factor Forkhead box M1b (FoxM1b), an injury-induced transcription factor that promotes cell cycle progression[248]. In addition, bile acids also contribute to liver regeneration by binding extra-hepatic FXR situated in the ileum, resulting in expression of fibroblast growth factor (Fgf15/FGF19). Fgf15/FGF19, which then binds it receptor FGFR4[249] on hepatocytes, stimulating cell cycle progression[241]. In comparison to the FXR receptor, the role of bile acid binding to the TGR5 receptor on Kupffer cells is less well understood, but is clearly important as TGR5 knockout mice show impaired liver regeneration after PH, as well as severe hepatic necrosis[250].

Thus, bile acids have an important role in the control of liver regeneration and may contribute to the post liver injury hepatostat.

Noradrenaline: Noradrenaline secretion increases following PH[251] and is produced by the adrenal medulla, sympathetic neurons, as well as by hepatic stellate cells.

Noradrenalin not only stimulates the production of EGF (from Brunner's glands) and HGF from fibroblasts, but also augments their mitogenic effect^[252], and activates the proliferation associated STAT3 pathway [253], whilst reducing the mito-inhibitory effects of TGF β [254]. Thus, α 1 receptor blockade, and also hepatic sympathectomy significantly delays liver regeneration after PH[251]. Noradrenaline may also stimulate liver regeneration by activating WNT and β -catenin pathways via β -adrenergic receptors[255].

Serotonin: Serotonin is a neurotransmitter stored by platelets and which has a role in the control of inflammation. Mice with absent platelets or lacking tryptophan hydroxylase 1 (a key enzyme in serotonin synthesis) show significantly delayed liver regeneration after PH[256], which is rescued by serotonin infusion. Moreover, serotonin agonist produces LSEC fenestration changes, and a VEGF dependent increase in hepatocyte proliferation [257]. Serotonin may also act via the Hippo proliferative pathway [258]. Although serotonin deficient mice show significantly impaired liver regeneration, serotonin exemplifies the need for caution in assuming that the results of one animal model may be extrapolated to others, as it is noted that rats



lacking the serotonin transporter which are unable to store serotonin in platelets do not show any liver regeneration impairment after PH[259].

Insulin: Insulin is produced by the beta cells of the pancreas, and was one of the earliest identified hepatic mitogens, having been found to prevent liver atrophy when infused directly into the liver via the portal stump in dogs having undergone portocaval shunt[11,13]. Although insulin is not a complete mitogen in that it does not induce hepatocyte proliferation in vitro, its presence is essential for hepatocyte survival in culture[260] and is essential for the effects of complete mitogens EGF and HGF in vitro[261].

The paradox that insulin is not a complete mitogen *in vitro*, but able to prevent liver atrophy after portal diversion is not fully understood but may be partly explained by interactions of the insulin receptor with EGFR and MET, thus triggering those proliferative pathways[212].

Growth hormone and insulin-like growth factor: Growth hormone is synthesized in the pituitary gland and has widespread growth-related roles in many tissues[262]. The effects of GH can be mediated directly via the GH receptor, or indirectly by insulin-like growth factor (IGF), which is synthesized by hepatocytes in response to GH and secreted into the circulation bound to IGF binding protein (IGFBP). Whilst hepatocytes do not have IGF receptors[263], Kupffer cells and stellate cells do[264], allowing a possible paracrine role for IGF in the liver. GH may also act directly on hepatocytes by upregulating the EGFR[265] and also stimulating activity of the EGFR in cross-talk with the GHR[266]. Consistent with this, in the rat model, exogenous dietary or infused GH enhances liver regeneration after PH, and mice lacking IGFBP show impaired liver regeneration after PH[63].

In terms of the physiological relative importance of these pathways, Pennisi et al [267] showed that GH lacking mice showed the greatest impairment to liver regeneration, with less marked liver regeneration impairment seen in IGF and IGFBP lacking mice, suggesting that whilst both direct and indirect GH actions impact on liver regeneration, the direct effect of GH is more significant that IGF mediated effect.

Thyroid hormone: The thyroid hormones Triiodothyronine (T3) and thyroxine (T4) are produced in the follicular cells of the thyroid gland, and have extensive roles in carbohydrate, protein, and lipid metabolism, regulation of metabolic rate, oxygen consumption, thermal regulation, muscle function, and roles in tissue growth and development[268].

In terms of liver regeneration, thyroid hormones have been shown to act as incomplete mitogens with impaired liver regeneration seen in thyroid receptor knockout mice[269], and conversely accelerated liver regeneration in T3 treated rats after PH[270].

In terms of molecular mechanisms of action in promoting liver regeneration, the thyroid hormones do not act via the NFKB or STAT3 pathways which are typically activated by the complete mitogens. Rather, thyroid hormones mediate hepatocyte proliferation by a number of pathways including (1) Increase in expression of transcription factors of the E2F family, which accelerates the transition of hepatocytes from G1 to S phase[271]; (2) Increased expression of cell cycle promotion genes Cyclins A, D1, and E, and diminished expression of their inhibitors[272]; (3) Decreased levels of p53 and p73 (tumour suppressor proteins involved in growth arrest and apoptosis) [269]; and (4) Activation of the *Wnt/b-catenin* signalling pathway[273].

VEGF: The VEGF family of growth factors comprises a group of at least 6 isoforms (VEGF A, B, C, D, E, F), which bind to the 3 different receptors (VEGFR 1, 2, 3), with roles in cell proliferation, migration, metabolism, vasodilation, blood vessel formation and remodelling[274]. Though not directly mitogenic on hepatocytes directly, VEGF plays a central role in liver regeneration in several ways, including the orchestration of proliferation of LSECs, and inducing the LSEC population to produce key hepatocyte mitogens including HGF[175,275].

In a rat 70% PH model, Bockhorn et al^[276] showed that blocking VEGF signalling with anti VEGF antibodies almost completely suppressed hepatic proliferation in the first 24 h after surgery, and conversely that exogenous VEGF promoted hepatocyte proliferation, suggesting a physiologically relevant role for VEGF in the early stages of liver regeneration.

Complex mitogens

WNT/β-catenin signalling pathway: The WNT family of genes are named after the



gene responsible for the Wingless-type phenotype in *Drosophila melanogaster*, and int-1 (a target for insertional activation of mouse mammary tumour virus, and a secretory glycolipoprotein-encoding gene which is regarded as the prototype for several mammalian genes)[277]. The resulting glycolipoproteins participate in several fundamental signalling events, which influence cell proliferation and tissue homeostasis[278].

 β -catenin is a protein encoded by the CTNNB1 and is a subunit of the cadherin protein complex which acts as an intracellular signal transducer in the context of WNT signalling, but which also interacts with a variety of transcription factors such as T-cell factor and hypoxia-inducible factor 1α [279]. β -catenin plays an important role in human embryogenesis, including liver development[280]. It is widely expressed in the adult liver and is always active in the pericentral region. Usually bound to a multiprotein degradation complex, it can be activated by several pathways, including WNT.

The WNT/ β -catenin pathway has been shown to be active during liver regeneration, contributing towards mass and functional recovery from a the very early stages of liver injury[281]. β -catenin is in fact detected in rat hepatocyte nuclei within 5 minutes of partial hepatectomy[282]. Upon WNT binding to its receptor (Frizzled), β -catenin translocates to the nucleus, where it promotes the expression of key genes, such as high-level controllers of transcription like *c-myc* and cell cycle regulating genes like cyclin D1[279].

In the normal liver, β -catenin regulates the expression of genes in pericentral hepatocytes and influences hepatic lobular zonation[279] and is involved in cell-cell adhesion[280]. Additionally, it implicated in a variety of diseased liver states, although the exact mechanisms remain incompletely understood. Specifically, β -catenin appears to be involved in the development of NASH, partly by binding to TCF4 and HNF4 α , thus regulating hepatic gluconeogenesis and lipogenesis. In hepatic fibrosis, the literature is currently conflicting regarding the role of WNT/ β -catenin signalling. Nevertheless, evidence is accumulating to show that this signalling pathway is activating during hepatic stellate cell activation and fibrosis, and that WNT blockade is associated with an antifibrotic effect[279].

This pathway has also been identified in hepatic neoplasia. In focal nodular hyperplasia (FNH), glutathione synthetase (the expression of which is regulated by WNT/ β -catenin signalling) stains FNH, which may be of diagnostic value[279]. Other relevant neoplastic processes include hepatocellular adenoma both with and without the presence of CTNNB1 mutations, HCC where mutations may lead to autonomous WNT-mediated activation of β -catenin, and hepatoblastoma, where 90% of tumours are associated with CTNNB1 mutations[279].

Hedgehog signalling pathway: Emerging evidence in the literature has shown the importance of the activation of the Hedgehog (Hh) pathway in the context of liver regeneration. Hh is a protein produced as a 45-kDa precursor that undergoes proteolytic processing in the endoplasmic reticulum[283]. The Hh pathway is a highly complex signalling cascade, which may be summarised in four fundamental components: (1) The ligand Hedgehog; (2) The receptor Patched (Patch); (3) The signal transducer Smoothened (Smo); and (4) The effector transcription factor, Gli. Components of the Hh pathway concentrate in the Primary cilia and a complex Primary cilium trafficking system regulates the interaction of Hh pathway components to enhance, or block, the Hh-initiated signal[284-286].

Previous work in adult rodents has demonstrated that Hh ligand expression increases transiently but significantly following partial hepatectomy[287]. Furthermore, inhibiting Hh pathway induction with a direct pharmacologic antagonist of Smo was found to decrease both recovery of liver volume and overall survival[288]. Evidence in the literature suggests that, mice subjected to portal vein ligation (a procedure commonly done in humans to allow the remnant liver to enlarge prior to hepatectomy) with simultaneous administration of systemic Hh, performed as well as mice submitted to ALPPS, supporting the evidence that Hh signaling plays a major role in promoting liver regeneration[289]. Further evidence suggests that the extracellular matrix of the healthy adult liver, the proteoglycan glypican-3 binds normally to Hh to prevent Hh from binding to Patch in order to constrain activation of the Hh pathway[78].

The evidence above highlights the role of the Hh pathway in post-hepatectomy liver regeneration. Further translational studies are required in order to explore the role of administering a recombinant form of Hh in the pre-operative setting in patients undergoing major hepatectomy who are potentially at risk of liver insufficiency.

Hippo Yap signalling pathway: The Hippo signalling pathway was originally identified in Drosophila melanogaster and its components have mammalian homologs [290]. The Hippo signalling pathway exerts a controlling influence on organ size by regulatory effects on cell proliferation, apoptosis, and stem cell self-renewal[291].

The pathway consists of a series of protein kinases, activation of which results in the phosphorylation of yes-associated protein (YAP), thus preventing its translocation to the nucleus. In the nucleus YAP interacts with a family of transcription enhancer factors. This family of nuclear proteins are involved in the modulation and regulation of multiple genes involved in promoting cell proliferation[292-294].

Multiple signals may activate Hippo signalling, including mechanical stimuli and cell attachment. Thus, in situations of high cell density, activation of Hippo signalling leads to the inhibition of YAP nuclear translocation, and thereby a break on cell proliferation[295]. Conversely, decreased Hippo pathway signalling allows YAP mediated pro-proliferation signals and is associated with an increase in organ size through excessive proliferation and inhibition of apoptosis[292]. Consistent with this, and suggesting its physiological importance, mice with liver specific YAP deletions show significant impairment in liver regeneration[296].

However, the Hippo-YAP signalling pathway is the final step of multiple opposing signalling pathways that contribute to liver regeneration and repair and the conflicting nature of these signals makes the study and understanding of the role of this pathway challenging[209].

Intracellular signalling pathways

The complexity inherent in the multitude of extracellular molecules implicated in the control of liver regeneration is reflected by a similarly complex array of intracellular signalling pathways which transmit the effect of ligand-receptor binding to the nucleus to activate effector genes.

In the same way that there is much redundancy in growth factor function extracellularly, there is also much overlap in the intracellular pathways, probably reflecting an evolutionary mechanism to safeguard against failure of any one individual pathway.

The complexity of each pathway, the activation of different pathways by diverse ligands, and the intracellular cross-talk between pathways makes it difficult to assign quantitative importance to any one pathway. Nevertheless, the section describes the main recognized intracellular pathways relating to liver regeneration. A full account of this area is beyond the scope of this review, but the summary below is intended to give a general overview and an impression of the ramifying complexity of the processes involved.

Figure 3A shows the pathways separately in summary form. Figure 3B shows the overlap in ligand binding, and Figure 3C provides an impression of the cross-talk between the various pathways.

Ras/Raf/MEK/ERK pathway: The Ras/Raf/MEK/ERK Pathway is triggered by binding of ligands to receptor tyrosine kinase receptors, which triggers autophosphorylation of tyrosine residues on the intracellular aspect of the receptor, resulting in the sequential activation of downstream components, ultimately controlling the expression of multiple growth controlling genes including high level 'master genes' such as c-myc, c-fos, and c-jun[297]. The first molecule, RAS, once activated, can activate multiple different signalling intracellular pathways including not only the Raf/MEK/ERK pathway, but also the MEKK/SEK/JNK pathway, and pathways involving NFκB[298,299].

In liver regeneration, the growth factor which activate this pathway include HGF, the EGFR ligand family members, fibroblast growth factor, and VEGF[299].

Phosphatidylinositol 3'-kinase/AKT kinase (also known as protein kinase B)/ mammalian target of rapamycin (PI3K/Akt/mTOR) pathway: The PI3K/ Akt/mTOR is a ubiquitous pathway is involved in the regulation of fundamental physiological processes including transcription, apoptosis, cell cycle progression, and translation [299]. The pathway is activated by binding of ligands to receptor tyrosine kinases or Gprotein-coupled receptors[300], ultimately promoting cell growth, proliferation, survival [301], and malignancy when dysregulated [302]. In the context of liver regeneration, the main growth factors activating the PI3K/Akt/mTOR pathway include TNF- α , IL-6, HGF, EGF, and transforming growth factor (TGF)- α [303] (Figure 3B and C).

Janus Kinase pathway: Activation of the Janus Kinase (JAK) kinase pathway promotes cell proliferation, differentiation, migration, growth. Like RTKs, JAKs activate by




Figure 3 Intracellular signal transduction map. A: Intracellular signal transduction in liver regeneration; B: Ligand overlap and receptor binding redundancy; C: Intracellular cross talk between signalling pathways. HGF: Hepatocyte growth factor; FGF: Fibroblast growth factor; VEGF: Vascular endothelial cell growth factor; EGFR: Epidermal growth factor receptor; IL6: Interleukin 6; TNF: Tumour necrosis factor; LPS: Lipopolysaccharide; RTK: Receptor tyrosine kinase family (including HGF receptor, FGF receptor, VEGF receptor, EGF receptor); GPCR: G protein coupled receptor; IL6R: Interleukin 6 receptor; TNFR: Tumour necrosis factor receptor; TLR: Toll like receptor; RAS/RAF/MEK/ERK: signalling components downstream of receptor tyrosine kinase; PI3K: Phosphatidylinositol 3'-kinase; AKT: Akt kinase

(also known as protein kinase B); mTOR: Mammalian target of rapamycin; JAK: Janus Kinase; STAT3: Signal Transducer And Activator Of Transcription 3; YAP: Yes-associated protein; NFkb: Nuclear factor kappa B.

autophosphorylation. STAT3 is the key downstream messenger which translocates to the nucleus and functions as a high level transcription factor in liver regeneration with resultant induction of gene expression including cytokines[304]. In the context of liver regeneration, this pathway is primarily activated by IL-6 and its receptor, IL-6R.

Interestingly, following PH, a circulating (rather than membrane bound) form of the IL6R cleaved by matrix metalloproteases appears to have a key role in initiating liver regeneration[86].

NF-\kappaB pathway: In priming of hepatocytes, Kupffer cells are induced to secrete TNF- α and IL-6 after stimulation by a variety of stimuli (see section on hepatocyte priming) including complement factors and LPS from the gut. This action of Kupffer cells is mediated by the NF- κ B signaling pathway. Once activated, NF- κ B migrates to the nucleus, where it promotes the further expression of TNF, IL-6, and VEGF[305].

WNT/\beta-Catenin pathway: The *Wnt/\beta-catenin* pathway regulates processes including cell proliferation, and tissue morphology[282]. Upon WNT binding to its receptor (Frizzled), β -catenin translocates to the nucleus, where it promotes the expression of key genes, such as high-level controllers of transcription like *c-myc* and cell cycle regulating genes like cyclin D1[279].

Hippo pathway: The *Hippo* signalling pathway is involved in cell proliferation, apoptosis, and stem cell self-renewal[291], and may have a key role in the ending of hepatocyte proliferation after regeneration[295]. YAP is a key downstream effector of the *Hippo* pathway, which translocates to the nucleus, once activated to promotes expression of target genes. The pathway is activated by numerous factors including organ size, cell attachment, mechanical stress, hormones, growth factors[295], as well as vascular shear stress[306].

Cross-talk between pathways: Mirroring the redundancy and overlap in extracellular growth factors and receptor binding, there exists a complex crosstalk between pathways. Just a few examples of the known positive feedback interactions are shown in Figure 3C, which, albeit incomplete and without including negative feedback interactions, provides an impression of the intricacy of intracellular interactions between pathways[297]. Thus, Ras interacts with phosphatidylinositol 3'-kinase (PI3K) [307]. The NF-κB pathway cross-talks with the PI3K/Akt/mTOR, pathways[308]. β-catenin interacts with the Hippo signalling pathways[195]. YAP cross-talks with PI3K/Akt pathway[309].

SECTION 4: THE CONTRIBUTION OF NON-PARENCHYMAL CELLS TO LIVER REGENERATION

The proliferation of hepatocytes in liver regeneration is in critical ways dependant on the role of non-parenchymal cells (Kupffer cells, stellate cells, and liver sinusoidal endothelial cells). Conversely, proliferating hepatocytes provide many growth factors that elicit non parenchymal cell proliferation, thus suggesting an interdependence allowing proportionate expansion of each cell type to produce liver tissue containing all key constituents. Thus, proliferating hepatocytes produceVEGF and angiopoietins 1 and 2 (LSECs mitogen), TGF α (LSECs and stellate cell mitogen), fibroblast growth factor 1 (FGF1) and FGF2 (HSC and LSEC mitogen), and granulocyte-macrophage colonystimulating factor (GMCSF) (Kupffer cells mitogen)[191,310,311]. In this section we examine in more detail the part played by non-parenchymal cells in liver regeneration.

Liver sinusoidal endothelial cells

LSECs have a key role in the immediate events that trigger the onset of hepatocyte proliferation after PH. Thus, shear stress resulting from portal pressure changes after PH induces the expression of the hepatocyte priming IL6[164].

Zaishideng® WJCO | https://www.wjgnet.com

Once liver regeneration is initiated, hepatocytes form clusters which are initially avascular. The production of VEGF, and angiopoietin 1 & 2 stimulates endothelial cells to migrate into the avascular structures, proliferate, differentiate to a liver sinusoidal phenotype[312]. LSECs also produce VEGF[313], as well as the potent mitogen HGF, and NO[160,162] which increases hepatocyte sensitivity to HGF[163].

In the rat hepatectomy model, liver sinusoidal cell repopulation is not only achieved by resident LSECs, but also by recruitment of LSEC progenitors in the bone marrow which, under the influence of VEGF, undergo proliferation, migration into the bloodstream, and engraftment in the liver, where they contribute a significant proportion of the total HGF production.

Of note, the importance of bone marrow derived LSECs in this model is emphasised by the fact that bone marrow ablation by irradiation abolishes liver regeneration, which can be rescued by exogenous infusion of LSECs[314,315]. Thus, LSECs play a key role, not only in allowing the vasculature of the liver to keep pace with regenerating hepatocytes, but also in providing the very growth factors (such as HGF) that allow hepatocyte to proliferate.

Stellate cells

Stellate cells are situated in the Space of Disse between LSECs and hepatocytes [316], and though representing only approximately 8% of cells in the liver, have multiple long cytoplasmic projections[317] which contact hepatocytes, LSECs, and Kupffer cells, allowing a central role in intercellular signalling in part by production of growth factors such as HGF and VEGF. Stellate cells also exhibit cellular contraction, permitting the control of sinusoidal blood flow [318], and have a key role in the regulation of ECM, both in its production and degradation[319].

After hepatic injury, HSC become activated myofibroblast-like cells. In the midst of liver regeneration, the initiation, perpetuation and resolution of HSC activation only adds further complexity, in processes which are poorly understood[320], but which ultimately result in the laying down of ECM to provide the vital framework on which regeneration may proceed[321].

The triggering of HSC activation is multifactorial, but includes the secretion by hepatocytes of growth factors such as FGF1 & 2, and PDGF, the latter being a potent mitogen and chemo-attractant for HSCs[322], and highlighting an example of interdependent complex paracrine stimulation (with HSCs producing the hepatocyte mitogen HGF and hepatocytes producing the HSC mitogen PDGF).

Following activation, HSC production of HGF increases, in a mechanism dependant on the neurotrophin receptor P75NTR[323], and its downstream mediator Rho[324]. Activated HSCs also produce Noradrenaline, which enhances HGF production by mesenchymal cells[223] and production of EGF from Brunner's glands[229]. HSCs interact directly with LSECs to stabilise and remodel sinusoids[325], via combined actions of PDGF, TGF-B1, FGF, VEGF, and angiopoietin. HSC are the principal cell source of ECM constituent production, and of ECM remodelling control by expression of matrix metalloproteases, thus providing the scaffold in which liver cells can regenerate[326].

The importance of HSC activation in liver regeneration is suggested by the observation that following ablation of HCS activation, liver regeneration is markedly impaired in both the mouse acetaminophen[327] and rat acetyl amino fluorene[328] models of liver injury with much reduced proliferation of hepatocytes and oval cells respectively, and with rescue of liver regeneration by infusion of medium conditioned by HSC[329].

Kupffer cells

Amongst the different resident intrahepatic macrophages, Kupffer cells are the predominant type, and originate from erythromyeloid progenitors in the foetal liver [330]. In the homeostatic situation, Kupffer cells have wide-ranging roles in (1) Clearance of cellular debris in blood[331]; (2) Maintenance of iron homeostasis via phagocytosis of red blood cells[332]; (3) Regulation of cholesterol homeostasis[333]; (4) Antimicrobial defence [334]; and (5) Promotion of immunological tolerance [335].

Kupffer cells have a limited half-life (of approximately 12 d in mice)[336]. Their maintenance is achieved by self-replenishment in the healthy liver[337], but is to an extent dependant on extra-hepatic progenitors in the case of liver injury [338].

There is evidence that replenishment of Kupffer cells following injury may be achieved by engraftment and differentiation of monocyte derived macrophages into a Kupffer cell phenotype, in a manner controlled by HCS and LSEC, highlighting once again the complex cross-talk between the non-parenchymal cell types[339]. The monocyte derived macrophage precursors may originate from within the liver, but



also from the peritoneum[340] and spleen[341].

Kupffer cells are strategically placed with access to the sinusoidal lumen and space of Disse, and carry multiple surface receptors to injury related molecules (the Pattern Recognition Receptors), suggesting a key role for Kupffer cells as sensors of hepatic injury[342].

Thus, Kupffer cells are able to detect, and become activated by: (1) DAMPs, e.g., mitochondrial DNA (mtDNA), and ATP, from damaged hepatocytes; (2) Pathogenassociated molecular patterns (PAMPs)[343]; (3) Hypoxic liver environment[344]; and (4) Extracellular vesicles secreted from various cells containing proinflammatory stimuli[345].

The activation of Kupffer cells by any of the above stimuli results in the secretion a wide range of bioactive molecules including chemokines such as C chemokine 2 which attract inflammatory and immune response cells to the injury site[346], and proinflammatory cytokines such as IL6 and TNF α which prime hepatocytes out of G0 phase an also activate HSCs[347].

Thus, Kupffer cells have a central role in the initiation and orchestration of liver regeneration, and their importance is suggested by the observation that depletion of macrophages[206] or inhibition of monocyte[348] recruitment results in impaired liver regeneration following PH.

SECTION 5: THE 'ALTERNATIVE PATHWAYS' OF LIVER REGENERATION

The liver regenerative response varies not only according to the nature of the injury, but also its magnitude and the status of the underlying liver parenchyma. These different contexts dictate how regenerating cells behave, but also the recruitment of different types of cells to accomplish the task. Current knowledge on the mechanisms of liver regeneration is largely derived from experimental models involving 2/3 PH in rodents[195], where "standard" liver regeneration occurs. This involves the proliferation of hepatocytes and cholangiocytes from homotypic precursors [209] and is addressed in the first subsection below. In the second subsection, the "alternative" liver regeneration pathways, which involve liver progenitor cells (LPCs) and transdifferentiation, will be examined. As seen in other aspects of liver regeneration, the mechanisms outlined below are subject to ongoing scientific scrutiny and are currently incompletely understood.

Hepatocyte response heterogeneity after PH

Liver regeneration after PH is achieved in different ways according to the magnitude of the liver resection. Thus regeneration after 1/3 PH is achieved principally by hypertrophy, with few cell divisions[349]. In contrast, during liver regeneration after resections larger than 1/3PH, although hypertrophy precedes hyperplasia, hyperplasia occurs increasingly as well, such that hypertrophy and hyperplasia contribute equally to liver regeneration in 70% PH. Moreover, during the hyperplastic response, although the majority of hepatocytes entered S phase of the cell cycle, not all undergo actual cell division, and the known significant number of polyploid hepatocytes[350] are shown to undergo division to produce mononuclear hepatocytes.

Thus, hepatocyte behaviour during liver regeneration is not uniform. Hepatocytes within a liver lobule are not equivalent and show functional heterogeneity. The liver lobule may be separated into 3 zones: Zone 1 (periportal) hepatocytes are in the vicinity of the portal triad, zone 3 (perivenular) hepatocytes are situated near the central vein, and zone 2 (pericentral) hepatocytes reside between zones 1 and 2. Metabolically speaking, zone 1 hepatocytes carry out gluconeogenesis and b-oxidation, in contrast to glycolysis, lipogenesis, and detoxification performed by zone 3 hepatocytes[195].

Moreover, there is some evidence that there is heterogeneity in baseline hepatocyte turnover during homeostasis, with a population of cells in zone 3 replenishing the lobule population, albeit slowly[351].

Hepatocyte proliferation heterogeneity is also apparent in the context of the regenerative response. In the context of true proliferative response after PH, lineage experiments have identified a population of hepatocytes in the periportal region zone 1 which appear to have greater proliferative potential. These are referred to as 'hybrid hepatocytes 'in that in addition to hepatocyte markers, they express progenitor cell genes and biliary transcription factors [352]. There appears to be further heterogeneity in proliferative response in that hepatocytes in proximity to LSEC proliferate faster than ones which are more distant[175].



Facultative stem cells and transdifferentiation pathways

The alternative liver regenerative pathways are characterized by deviation from the phenotypic fidelity in which hepatocytes or cholangiocytes proliferate to produce more of the same cell type. In this "alternative" context, liver epithelial cells (i.e., hepatocytes and cholangiocytes) can operate as facultative stem cells for one another in conditions where regeneration of one or other is impaired [209,353], presumably as a rescue mechanism. This mechanism appears beneficial from an evolutionary standpoint, is plausible from a developmental biology perspective given that both hepatocytes and cholangiocytes are derived from hepatoblasts[354], and has been demonstrated in previous studies[81,355-357]. Nevertheless, there is controversy in the field regarding the *in vivo* capability, conditions, and extent to which liver epithelial cells can transdifferentiate and achieve regeneration.

The term "LPCs" is seen in the literature, yet such cells have not been identified on microscopy or tissue dissociation of liver lobules in the resting state and are thought to possibly arise from the transdifferentiation of hepatocytes and cholangiocytes[353] as mentioned above. A central event in this process is the "ductular reaction", which occurs when hepatocyte proliferation is suppressed, thus leading to the expansion of progenitor cells^[209] and which can be observed both in acute and chronic liver disease models, typically after extensive hepatocyte injury.

In conditions such as fulminant hepatic failure, some liver epithelial cells demonstrate an overlapping set of biomarkers (e.g., cholangiocytes may express "hepatocytic" biomarkers such as HNF4, albumin, and HEPPAR3)[353]. The extensive necrosis and apoptosis characteristic of fulminant hepatic failure is thought to pivot liver regeneration towards LPCs, as reflected in elevated α-fetoprotein levels[358] and as demonstrated by histological findings of "regenerative clusters", which consist of atypical ductules lined by cells exhibiting a combination of cholangiocyte and hepatocyte biomarkers[209].

In the context of PH, once a certain resection threshold is exceeded (e.g., > 80%), adequate liver regeneration cannot be achieved by the relatively small number of remaining hepatocytes, and the alternative pathway is thus activated. In this process, biliary epithelial cells (cholangiocytes) de-differentiate into progenitor cells and then re-differentiate into hepatocytes in order to repopulate the liver[195]. As described in the rat models subsection above, Evarts et al[359] demonstrated that administration of 2-AAF to rats which had undergone PH was associated with differentiation of oval cells (a putative hepatic progenitor cell) to hepatocytes. However, this model did not allow for genomic-based cell lineage tagging[209].

More recently, Lu et al[81] induced widespread hepatocyte injury in mice through Mdm 2 deletion, which results in p53 upregulation with p53-induced hepatocyte death and senescence. The authors found that widespread hepatocyte injury was associated with a ductular reaction, whereby hepatocyte progenitor cell populations expanded and where bromodeoxyuridine-positive hepatocyte progenitor cells were often closely associated with bromodeoxyuridine-positive hepatocytes, thus suggesting that (in this context) hepatocytes arise from progenitor cells[81]. In a zebrafish model, Choi et al [355] found that after severe hepatocyte depletion, biliary epithelial cells de-differentiated into hepatoblast-like cells and then differentiated into highly proliferative hepatocytes, thus leading to liver regeneration.

Although the above studies focused on transdifferentiation from cholangiocytes to hepatocytes, the inverse has also been demonstrated in a murine model of Alagille syndrome (a human genetic condition associated with biliary underdevelopment). In this study, Schaub et al[360] found that hepatocytes converted to mature cholangiocytes that were effective in supporting biliary drainage and remained so after cholestasis resolved, in a TGF β signalling mediated process. This persistent phenotypic change is distinct from the reversible conversion of human or murine hepatocytes to progenitor cells seen in other studies[361] and which may more accurately be described as "metaplasia" rather than "transdifferentiation". In their chimeric liver rat model, Michalopoulos et al[362] injected dipeptidyl peptidase IV (DPPIV)-positive rat hepatocytes into DPPIV-negative rats which then underwent partial hepatectomy and bile duct ligation, with or without additional biliary injury by methylene diamiline (DAPM) administration. On animal sacrifice after 30 d, the authors found that ductules exhibited the DPPIV marker, and that this was enhanced 36-fold in rats with additional DAPM-mediated biliary toxicity.

Evidence in support of liver epithelial cell transdifferentiation for regeneration is accumulating, yet several areas of controversy remain to be resolved. In addition to the findings described above, self-renewing facultative stem cells have been located in peribiliary glands and liver progenitor cells of bipotential differentiation capacity have been located in association with the canals of Hering. However, their role in liver



regeneration has been disputed [195]. Also, it remains unclear whether all mature hepatocytes are capable of dedifferentiation to progenitor cells, or whether this is only possible in a subset of cells[195]. Finally, various signalling pathways (e.g., YAP, Rho kinase, TGF- β , glycogen synthase kinase 3) have been implicated in hepatocyte dedifferentiation in animal models[195], yet their role with respect to human hepatocytes remains to be clarified.

SECTION 6: THE INFLUENCE OF UNDERLYING LIVER DISEASE ON LIVER REGENERATION

The processes and mechanisms of liver regeneration are not only influenced by the magnitude and type of injury, but also by the status of the underlying liver parenchyma prior to injury. Although the PH model has contributed much information relating to events relating to regeneration of normal liver, significant differences come to light when the underlying liver is diseased. This section describes the ways in which liver regeneration is altered in the instances of age-related liver impairment, acute liver injury, hepatic steatosis, fibrosis, and cirrhosis.

Age-related liver impairment

Although the adult liver retains regenerative capacity throughout life, this is reduced in old age[363], through several suggested mechanisms. FoxM1B is a transcription factor expressed during embryonal development and also in liver regeneration[364]. Its expression is diminished in aged mice, whose liver regeneration can be rescued through its transgenic overexpression [365]. Age-related liver regeneration impairment may also be mediated by changes in the expression and function of cell cycle affecting genes such as CCAAT/enhancer-binding protein (C/EBP)a, which is an inhibitor of Cyclin D[366]. Budding uninhibited by benzimidazole-related 1 (BubR1) is involved in the control of mitosis and is found to be diminished in old age. Genetically manipulated mice expressing low levels of BubR1 show impaired liver regeneration [367].

In addition to these mechanisms, Conboy et al[368] and Liu et al[369] have demonstrated in parabiotic experiments that the blood or plasma of young mice partially rescues the liver regenerative compromise seen in old mice, suggesting the presence of currently unidentified circulating factors, in work reminiscent of the early experimental approaches used to demonstrate the existence of portal mitogens.

Acute liver injury

Severe acute liver injury may result from a variety of insults including viral infection (e.g., Hepatitis A, B, C), poisoning (e.g., paracetamol) or auto-immune disease. Though different in nature, these diverse insults nevertheless have the common feature of causing significant necrosis and apoptosis, in the midst of which regeneration must happen.

Specific models provide some mechanistic information. For example, the mouse model of paracetamol injury suggests that beyond a threshold of injury, regeneration fails, and that this is associated with failure of β catenin activation, consistent with the correlation of β -catenin activation and regeneration seen in patients[370].

In the setting of widespread necrosis, the contribution of non-parenchymal cells may be particularly important. Macrophages are essential in clearing toxic cellular debris[371], and thus mice deficient in CSF1 which promotes the maturation of macrophages have impaired liver regeneration[372], and CSF1 serum levels correlate with recovery from paracetamol liver injury[373].

Hepatic steatosis

Hepatic steatosis is known to be detrimental to liver regeneration not only in experimental models[374], but also in the clinical setting[375]. The mechanism is not fully understood but may include cell cycle machinery defects in steatosis[376]. Down regulation of the EGFR pathway may also contribute, and EGFR overexpression has been shown to rescue liver regeneration in a mouse model of PH in steatosis[377]. Steatosis may also compromise liver regeneration by inhibition of NFKB[378]. Finally, failure to activate growth arrest and DNA damage-inducible protein GADD34 in fatty liver may partly contribute to impaired liver regeneration, which can be rescued by transgenic overexpression of growth arrest and DNA damage-inducible protein GADD34 in a mouse experimental model[379].



Hepatic fibrosis and cirrhosis

In the setting of overwhelming injury where the capacity of hepatocytes to proliferate is overwhelmed, oval cell transdifferentiation provides a mechanism to assist cellular repopulation. Such injury is, however, also associated with activation of stellate cells to myofibroblasts, which secrete ECM[380]. Excessive ECM secretion is nonetheless harmful because it inhibits the ductular reaction[381] and hepatocyte proliferation [382]. Moreover, excessive fibrosis impairs portal flow, leading to arterialisation of the liver, and senescence of hepatocytes and cholangiocytes[383]. The distortion of the macro and micro-anatomy of the liver results in major compromise to liver function leaving affected patients in an extremely precarious state characterized by rapid and severe decompensation, which can be triggered by relatively minor physiologically stresses.

SECTION 7: MECHANISMS UNDERLYING CESSATION OF LIVER REGENERATION

In the rodent PH model, liver regeneration proceeds until the liver have returned to its pre-PH weight, approximately 10 d later, at which point regeneration ceases. The mechanisms resulting in termination of liver regeneration have received less attention than those driving it, but are equally important, not only from the perspective of the study of liver regeneration biology but also in terms of the light they shed on other pathologies including liver malignancy. Nevertheless, several regeneration termination pathways have been identified, relating to TGF^β, the activins, the ECM, and glypican-3 (GPC3).

TGF β

TGFβ is a multifunction cytokine with wide ranging roles in growth and development. It exists in 3 isoforms resulting from differential protein cleaving and binds to 3 different TGF^β receptors. Binding to the TGF^βR results in autophosphorylation, and activation of SMAD, which translocates to the nucleus, delivering an inhibitory signal to cell proliferation[384]. Although TGFβ does inhibits hepatocyte proliferation in vitro [385], other experimental results in vivo cast some doubt on its role in termination of liver regeneration in that liver specific TGFBR knockout mice terminate liver regeneration appropriately[68]. However, this result does not necessarily rule out TGF β as a significant factor in liver regeneration termination: given the redundancy seen in the processes that drive regeneration, it seems likely that similar redundancy exists in its termination, such that ablation of one mechanism may readily be rescued by other pathways.

Activins

The activins are a family of proteins which are similar in structure to the TGF β family, which also transduce signals *via* receptors that activate SMAD, and convey a growth inhibitory effect. Activins are upregulated during the liver regeneration [386], and blocking their action pharmacologically results in excessive regeneration and hepatomegaly following PH in rats[387].

ECM and integrin linked kinase

The ECM is thought to convey a growth controlling influence on liver cells in a mechanism whereby integrin proteins in intact ECM bind hepatocyte cell membrane Integrin linked kinase (ILK) receptors, which deliver a growth inhibitory signal [388]. Consistent with this, the growth response of hepatocytes to mitogens in vitro is much reduced when grown in the presence of ECM in comparison to plastic[389]. Moreover, ILK knockout mice show not only hepatomegaly in the native state[390], but also an exaggerated regeneration after PH[391]. Thus, it may be that the activation of matrix metalloproteases that occurs early after liver injury[188] results in degradation of the controlling influence of integrins, and that this is gradually recovered during regeneration as new ECM is laid down.

GPC3

GPC3 is a heparan sulphate proteoglycan found on the cell surface of many tissues which conveys a growth inhibitory effect[392]. It is not detectable in quiescent liver but is expressed coinciding with the end of regeneration after PH in rats[393]. Moreover, loss of function mutations of GPC3 results in organ overgrowth[394], and transgenic



over-expression of GPC3 delays liver regeneration after PH[393].

C/EBP

C/EBP is one member of a family of transcription factors with a role in producing cell cycle arrest. Although complete ablation of C/EBP is fatal in mice[395], altering the function of the protein by mutation results in partial loss of function with mice exhibiting excessive regeneration and hepatomegaly after PH and CCl4 injury[396], thus suggesting that the native protein has a role in the control of liver regeneration.

Cyclin E1 and E2

The cyclins are a group of proteins which impact on the progression of the cell through the cell cycle. Amongst these cyclins E1 and E2 influence the advancement of the cell from the G1 phase (during which the cell synthesizes protein and organelles), to S phase (during which DNA is replicated)[397]. Cyclin E1 and E2 have opposing roles, with Cyclin E1 promoting entry into S phase[398], and Cyclin E2 halting it[399]. Thus, mice with ablated Cyclin E2 show increased DNA synthesis and hepatomegaly after PH suggesting a role for Cyclin E2[398,400].

Of note the hepatomegaly seen after CyclinE2 ablation is not due to cell division, but hypertrophy, providing a possible mechanism to explain to the observation that liver growth after 30% PH is hypertrophic rather than the hyperplasia seen in 70% PH.

Hippo/YAP pathway

The Hippo/Yap signalling pathway is conserved in a wide range of organisms and associated with growth suppression[401]. YAP is key downstream effector of Hippo and its activation when dephosphorylated leads to massive liver overgrowth [402], suggesting a possible regulatory role in control of liver regeneration.

Micro RNAs

Micro RNAs are short RNA molecules which bind to messenger RNA and thus affect expression of the gene product by interfering with translation of mRNA to protein [403]. Several micro RNAs have been identified which target the mRNAs of key regeneration promoting proteins, and thus may play a part in controlling liver regeneration. Thus miR-23b targets the growth inhibiting SMAD protein such that the observed downregulation of miR-23b following PH may provide a mechanism for slowing liver regeneration[404]. miR-34a targets several mRNAs including that which codes for the HGF receptor MET, again providing a potential mechanism for limiting hepatocyte proliferation[405].

SECTION 8: LIVER REGENERATION: IMPLICATIONS FOR THERAPY OF LIVER TUMOURS

The complexity of liver regeneration biology, combined with our currently limited understanding has to date much restricted specific clinical interventions to enhance liver regeneration. Moreover, the processes involve such fundamental biochemical pathways that attempts to manipulate these would require very careful assessment, for fear of unintended consequences in the liver and other organ systems. Nevertheless, current knowledge allows clinicians to anticipate what scenarios or treatments may compromise liver regeneration and provides guiding principles which may allow planning treatment strategies to optimise liver regeneration potential.

Liver tumours, be they primary or metastatic, may be treated by chemotherapy in systemic or locoregional delivery methods, or by surgical intervention with resection or local ablation techniques. Radiotherapy, although used to an extent, has a much lesser role and evidence base. In this section, we discuss how the biology of liver regeneration affects treatment choices and delivery, not only in terms of chemotherapy and surgery, but also in relation to wider organ system physiology that relates to liver regeneration.

Chemotherapy and liver regeneration

The principal scenario in which chemotherapy impacts on liver regeneration is the use of neoadjuvant chemotherapy prior to a planned liver resection, or in the instance of downsizing an initially unresectable lesion, most usually in the context of colorectal liver metastases [406]. Chemotherapy affects not only individual cell types within liver parenchyma, but also key extra-hepatic tissues pertinent to the liver regenerative



process such as the bone marrow, and in some cases affects key liver regenerative pathways.

The main toxicities inflicted on the liver by chemotherapy are steatosis, steatohepatitis, and sinusoidal obstruction syndrome (SOS). Steatosis is the excessive deposition of fat within hepatocytes. This may trigger an inflammatory reaction leading to steatohepatitis, and in turn to fibrosis and cirrhosis. Sinusoidal obstruction syndrome is a separate entity with direct toxicity to LSEC leading to occlusive phenomena in the sinusoids[407].

In the context of treatment of (colorectal liver metastases) CRLM, 5FU is associated with steatosis [408], probably as a result of impaired oxidation of fatty acids [409]. 5FU also triggers the activation of pro-inflammatory genes, which may contribute to the evolution of steatohepatitis[410]. Chemotherapy associated Steatohepatitis (CASH) is also particularly associated with regimens containing irinotecan[411] which also inhibits fatty acid oxidation, but elicits steatohepatitis by activation of ERK[412], which if inhibited, leads to a reduction in steatohepatitis. SOS is associated with oxaliplatin regimens. Although not fully understood, a contributing mechanism is the increased expression of matrix metalloproteases, resulting in lifting of LSEC from the basement membrane and allowing infiltration of red blood cells into the space of Disse, thus causing an occlusive phenomenon in the lumen of the sinusoid, and an inflammatory reaction with stellate cell activation and perisinusoidal fibrosis[413].

In addition to these directly hepatotoxic effects, the above agents are myelosuppressive, and may thus compromise the bone marrow constituents that play important roles in liver regeneration, including (1) LSEC progenitors which are important in repopulating LSEC after PH and a key source of the important mitogen HGF[314,315]; (2) Macrophage progenitors which have a key role in clearing cell debris in preparation for regeneration[206]; and (3) Megakaryocytes which replenish platelets, with their important role in delivering liver regenerative signals[178,179].

5FU, oxaliplatin, and irinotecan are frequently used in combination with biological agents targeting specific proliferative pathways within tumours, which overlap with biochemical pathways promoting liver regeneration. Thus, the anti-EGFR antibody cetuximab blocks binding of EGFR ligands including the key mitogen EGF, acting via the RAS-RAF-MEK pathway. Another example is the anti-angiogenic antibody bevacizumab which delivers antitumor effects by blocking VEGF-binding to its receptor, but thereby also interfering with LSEC repopulation in the liver[406]. Thus, the commonly used chemotherapeutic agents and antibodies used in the treatment of CRLM have a multitude of liver regeneration compromising properties. The challenge for clinicians is to find the optimal balance between the oncological benefits and liver regenerative toxicity.

Clinical trials provide some guidance in this regard. Thus 5FU, oxaliplatin and irinotecan-based regimens have been shown to produce CRLM response rates allowing increased potential for curative resection[414,415], with additional benefits attributed to the use of anti EGFR[416] and anti VEGF antibodies[417]. Interestingly, whilst the use of EGFR antibody regimens is established as a means to downsize tumours to increase operability, their perioperative use in primarily resectable CRLM is detrimental^[418].

In terms of duration of chemotherapy, the theoretical ideal is the delivery of the oncological hit to the tumour whilst minimising liver toxicity. In this regard Karoui et al[419] found that patient receiving fewer than 6 chemotherapy cycles experienced significantly fewer post liver resection complications than those who had received more than 6 cycles (19% vs 54% complication rate) although there was no impact on mortality rates. Similarly, Nguyen et al [420]showed a greater incidence of postoperative liver failure in patients undergoing more than 10 cycles of chemotherapy. Moreover, the general practice of leaving an interval of 4-6 wk between chemotherapy and surgery is intended to allow reversible inflammatory changes and bone marrow to recover[421].

The prospect of specific interventions to minimise chemotherapy related injury is the subject of research but has not yet reached widespread clinical application. Nevertheless, there have been reports that bevacizumab may reduce SOS in oxaliplatin regimens[422], and S-adenosylmethionine (SAMe) may have a protective effect in chemotherapy-induced liver injury [423], with SAMe infusion associated with lower serum concentrations of aspartate transaminase and alanine transaminase during chemotherapy treatment[424].

Liver volume manipulation

In the context of a healthy underlying liver parenchyma, up to 75% of the liver may be resected. However, this is the very limit and liver surgeons are often faced with the



problem of some degree of parenchymal pathology or dysfunction, which may present a higher requirement in terms of the remnant liver volume. As discussed in section 1.4, pure volume assessments are gradually being complemented by liver functional assessments, principally by hepatobiliary scintigraphy [146,152].

In instances where future remnant liver volume/function is deemed insufficient, PVE has become an established technique to produce atrophy of the tumour bearing liver parenchyma, and compensatory growth of the future remnant liver, allowing a safer hepatectomy [425]. In the context of resections for CRLM, this technique increases resection rates by 10%-20% [426].

In the instance where PVE fails to produce sufficient hypertrophy, perhaps caused by the development of collateral intrahepatic portal vessels between the embolized and non-embolized parts of the liver, further growth of the future remnant liver may be achieved by parenchymal section to interrupt the collateral vessels.

This approach, as a salvage manoeuvre after PVE is a modification of the originally described ALPPS technique^[427] that combined single stage portal vein embolization and parenchymal transection. Whilst undoubtedly producing significant additional growth, the technique remains debated owing to questions regarding the functionality of the rapidly expanded liver, and high post-operative mortality in some series[428].

Liver disease limitations to liver resection

The continuum of steatosis, steatohepatitis, fibrosis, and cirrhosis presents a major challenge to liver resection. In terms of steatosis, the observed experimental and clinical compromise to liver regeneration as a result of steatosis translates to a diminished tolerance to liver resection[429]. Thus, studies examining the outcome of hepatic resection in patients with steatotic livers suggest more marked abnormalities in postoperative liver dysfunction, more morbidity and increased complication rates [143,430], with steatosis identified as an independent predictor of complications[431]. Increased mortality is identified in some studies[430], and although this has not been a universal finding, the presumption is that this relates to careful patient selection.

The tolerance of the liver to resection progressively decreases with more severe underlying liver disease. Thus, cirrhosis is associated with increased mortality rates for all abdominal surgery including[432] liver resection[433].

Interventions to mitigate the risk of liver surgery in the context of underlying liver disease are limited, and practice has focused on patient selection to avoid prohibitively hazardous resections. In the context of steatosis, the response of the steatotic liver to ischaemic insult[434] has motivated research in the concept of ischaemic preconditioning, whereby a short period of ischaemia prior to liver resection is applied with a view to improving subsequent perfusion [435]. Although benefits have been shown in rodent models and in the clinical setting [436], results have not been universal [437], and the practice not widely accepted.

The role of the gut

One of the key biochemical triggers which initiates liver regeneration is LPS from the gut, which translocates into portal blood following liver resection in part as a result of the rise in portal pressure[204] (see "*Priming of hepatocytes*"). The evidence supporting the importance of LPS is that germ free mice show impaired liver regeneration, which can be rescued by exogenous LPS administration[205].

Though the presence of LPS in the blood is important, there appears to be a delicate balance as excessive translocation of LPS into portal blood is detrimental to liver regeneration. Thus, in the rat model after 90% PH, gut mucosal permeability is disrupted with loss of tight junctions, resulting in high levels of portal blood LPS, severe inflammatory changes in the liver with necrosis, associated with high mortality. By decontaminating the gut with gentamycin, gut permeability, portal blood LPS, and liver necrosis is much improved, and associated with a significant improvement in survival from 24% to 56% [27].

No clinical trials have been carried out to test the potential benefit of this animal experimental result in humans, where its applicability could be investigated in patients undergoing major liver resection. Additional considerations would come into play, including the risk of *clostridium difficile* sepsis associated with alteration of bowel flora. Another potential approach, also uninvestigated, could be the use of bowel preparation similar to that used prior to colorectal surgery.

Multivisceral resections

The data from early experimental result suggested that liver regeneration was significantly dependent on portal blood growth factors derived from the upper



WJCO | https://www.wjgnet.com

gastrointestinal tract (stomach, duodenum, pancreas) as portal flow separation experiments comparing isolated portal flow of distal stomach, duodenum, pancreas and spleen to portal flow of small intestine showed that the grafts supplied with intestinal portal flow atrophied, in contrast to those supplied with portal blood from the upper gastrointestinal tract[9,10].

On this basis, there might be a similar risk in humans in cases where liver resection is combined with a significant resection of the upper gastro-intestinal tract, such as Whipples resection (resection of distal stomach, duodenum and head of pancreas) or total pancreatectomy (resection of distal stomach, duodenum, pancreas, and spleen). However, existing case series describing such multivisceral resections do not report problems relating to liver regeneration[438-440].

Thus it may be that the situation in the original animal experiments cannot be extrapolated to humans owing to different physiology, although interpretation of the human literature has to be taken with the caveats of selective reporting (bias against reporting poor outcomes with liver failure), case selection of surgical candidates (likely good performance status individuals for such major resections), and likely small liver resections which might not manifest with post-operative liver failure.

Bile metabolism considerations

Bile salts are important auxiliary mitogens, with rodent models showing impaired liver regeneration with bile acid sequestering agents [243], bile-salt-deficient transgenic mice^[244], and in rats with external biliary fistula^[245].

This observation in rodent models is mirrored in clinical practice, with a randomised trial comparing patients undergoing major liver resection with and without cystic duct biliary drainage showing significantly lower bile salt concentrations in the drained group as well as lesser liver regeneration assessed by volumetric CT[441]. Moreover, in the context of portal vein embolisation, increased systemic bile salt levels predicted hypertrophy of the non-embolised lobe[442].

Thus, the presence of bile in the intestine following liver resection appears to be important and would argue against the use of external biliary drains after liver resection, or in circumstances requiring such drains, to consider enteric bile recycling via nasogastric or nasoenteric tube.

Management of post-hepatectomy portal hypertension

The rise in portal venous pressure following liver resection appears to contribute to providing important triggers to liver regeneration in the form of induction of cytokines, hepatic mitogens and angiogenic growth factors, as described in "Vascular events" section. However, an excessive increase in portal pressure is thought to be detrimental in eliciting a reduced arterial inflow as a result of the arterial buffer response, and a subsequent hypoxia[171], which is hyptothesised to contribute to post liver resection liver failure^[443], in combination with direct mechanical injury occurs to the liver sinusoids[444].

Thus, a number of investigators have examined a variety of interventions to decrease portal venous pressure by surgical (splenectomy, splenic artery ligation, porto-systemic shunt), interventional radiological (pre-operative splenic artery ligation), and pharmacological means (non-selective beta-blockers, terlipressin), with successful reductions in portal pressure and improvement in small-for-size syndrome [444].

Hypoxia

Intrahepatic hypoxia is one of the stimuli which may contribute to the early triggers of liver regeneration[175], via a number of mechanisms including the hypoxia induced secretion of complex regeneration promoting molecules from stem cells[176]. Given the critical necessity of maintaining normoxia in other tissues, manipulating pO2 for hepatic regeneration benefits seems an unlikely strategy, however, pharmacological manipulation of HIF has been used to treat renal anaemia[445], and in vitro studies suggest that such agents could produce angiogenesis in the liver, as well as a hepatocyte cytoprotective effect[174].

Adrenergic stimulation

The finding that surgical denervation of the liver or pharmacological alpha adrenergic blockade significantly impairs liver regeneration, is consistent with known modulatory effects of catecholamines on secretion of hepatic mitogens (including EGF), as well as the finding that catecholamines increase hepatocyte sensitivity to EGF[251]. Moreover, hepatic stellate cells and hepatic progenitor cells are innervated by the



WJCO | https://www.wjgnet.com

sympathetic and parasympathetic nervous system[446].

Hepatic sympathectomy is effectively carried out in humans in the context of hilar cholangiocarcinoma resections (when the entire hepatic hilum is skeletonized) and in liver transplantation, when the implanted graft is totally denervated, however there is no obvious clinical evidence that this has a detrimental effect on liver regeneration.

The absence of reported clinical compromise to liver regeneration following sympathectomy in humans may reflect differences in autonomic supply [447], compensatory effects from adrenal catecholamine secretion, or perhaps relatively rapid re-innervation, as seen in some animal models[448]. Finally, the effect of catecholamine stimulation is complicated by the fact that inhibition of alpha adrenergic signals may have a beneficial effect in minimising hepatic stellate cell activation and therefore reduce detrimental excessive fibrosis[446].

The role of platelets and fibrinogen

Platelets clearly have an important role in promoting liver regeneration with animals and humans studies showing impaired liver regeneration in individuals thrombocytopenia[178,179], and the association between thrombocytopenia and post-operative mortality after major liver resection [449]. Moreover, fibrinogen deposition in the liver appears key in driving platelet accumulation in the liver, with post hepatectomy hypofibrinogenaemia being associated with liver dysfunction and mortality. Moreover, studies in rodents suggest platelets are the key source of the auxiliary hepatic mitogen serotonin in liver regeneration, and that serotonin infusion rescues impaired liver regeneration observed in thrombocytopenic mice[256].

Correction of platelet count and fibrinogen levels in patients post hepatectomy has not been investigated in humans, but could potentially have beneficial effects, although thromboembolic risks would have to be taken into account. Serotonin infusion has also not been investigated in humans in the context of liver regeneration, but has been carried out in other clinical contexts to stimulate prolactin[450], and as a desired effect in amitriptyline treatment of depression[451].

Non parenchymal cell modulation

In the setting of liver regeneration with significant hepatic necrosis, the role of nonparenchymal cells may be particularly important. Macrophages clear toxic cellular debris, and mice deficient in macrophage colony stimulating factor 1 (CSF1) which promotes the maturation of macrophages have impaired liver regeneration[372]. Moreover, CSF1 correlates with recovery from paracetamol liver injury in patients with liver failure[373]. In addition to clearing metabolic debris, macrophages may in part act by stimulating the hepatic ductular reaction [452] and limiting fibrosis [453].

Although macrophage colony-stimulating factor-1 has not been used in humans in the context of liver failure and regeneration, it does have a multitude of other clinical applications[454].

Similarly, in rodent models at least, the bone marrow is a significant source of LSEC progenitor cells, which emigrate from the marrow into the bloodstream from which they engraft into regenerating liver, where they produce a significant quantities of the complete hepatic mitogen HGF, in a VEGF driven process[314,315]. Specific LSEC progenitor stimulation therefore offers a theoretical therapeutic opportunity but has not been investigated.

Modifying the proliferative response.

The characterization of the multiple cytokines, hormones and growth factors involved in liver regeneration has motivated investigation of means to modulate the hepatic proliferative response. Such approaches have involved the infusion of specific mitogens, or the use growth enhancing progenitor cells or their secretome and are to date at experimental animal model stage.

These wide-ranging experimental approaches are beyond the scope of this review, but a few examples provide an insight into potential avenues for the future. In a murine model of 85% PH, Cataldegirmen et al[455] investigated the potential therapeutic opportunity of blocking the Receptor for advanced glycation end-products (RAGE), which is upregulated in massive hepatectomy and associated with cell stress when binding its ligands. Blockage of RAGE pharmacologically or by transgenic means resulted in significant improvement in survival post massive PH[455]. In a mouse model of partial biliary ligation, Mangieri et al[456] report improved liver regeneration produced by infusion of the complete mitogen HGF. Similarly, but targeting the other complete mitogen pathway of the EGFR ligands, Zimmers et al[377] demonstrated improved liver regeneration after plasmid delivery of EGF receptor.



WJCO | https://www.wjgnet.com

In addition to the focus on individual hepatic mitogens, investigation has also been carried out in the infusion of whole cells[457], ranging from primary human hepatocytes[458], pluripotent stem cell derived hepatocyte-like cells[459], and mesenchymal stem cells[460]. Cell based therapies have met several obstacles including sourcing, immuno-compatibility, and potential malignant transformation, thus motivating research into potential for using the secretome of stem cells, thus obviating the difficulties presented by the whole cell therapies [176,461-463].

CONCLUSION

Liver regeneration is highly complex, and current understanding is based largely on animal and in vitro models. The likelihood that not all hepatic mitogens have been identified, the multitude of known ones, the complexity and incomplete understanding of their associated biochemical pathways, the equally complex and poorly understood cross talk between cell types, and our even poorer understanding of the factors that cease liver regeneration all suggest that a comprehensive working understanding of the process is improbable in the foreseeable future. Consequently, specific interventions to influence liver regeneration in the clinical setting are commensurately limited, though allow clinicians to at least optimise conditions for liver regeneration to occur. The implications of this in relation to the treatment of liver tumours are most notably applicable in the context of liver resection for malignancy, where assessment and optimisation of remnant liver function not only increases the proportion of patients eligible for treatment, but also improves patient safety. The increasingly sophisticated in vitro organoid models, and potential opportunities presented by repopulation of decellularised scaffolds may allow the creation of constructs that allow not only deeper understanding, but also novel therapeutic options.

REFERENCES

- Mortensen KE, Revhaug A. Liver regeneration in surgical animal models a historical perspective and clinical implications. Eur Surg Res 2011; 46: 1-18 [PMID: 21135558 DOI: 10.1159/000321361]
- 2 Eck NVK. Concerning ligation of the vena porta (in Russian). Voen Med Zh 1877; 130 (English translation: Child CG: Eck's fistula. Surg Gynecol Obstet 1953; 96: 375-376
- 3 Child CG 3rd, Barr D, Holswade GR, Harrison CS. Liver regeneration following portacaval transposition in dogs. Ann Surg 1953; 138: 600-608 [PMID: 13092790 DOI: 10.1097/00000658-195310000-00013]
- Mccredie JA, Doggart JR, Welbourn RB. Total arterialization of the liver. Br J Surg 1957; 45: 83-100 [PMID: 13510666 DOI: 10.1002/bjs.18004518919]
- Hahn M, Massen O, Nencki M, Pavlov J. Die Eck'sche Fistel zwischen der unteren Hohlvene und 5 der Pfortader und Folgen für den Organismus. Arch Exp Pathol Pharmakol 1893; 189: 161-210 [DOI: 10.1007/BF01995065]
- Rous P, Larimore LD. Relation of the portal blood to liver maintenance: A demonstration of liver 6 atrophy conditional on compensation. J Exp Med 1920; 31: 609-632 [PMID: 19868417 DOI: 10.1084/jem.31.5.609]
- 7 Marchioro TL, Porter KA, Brown BI, Faris TD, Herrmann TJ, Sudweeks A, Starzl TE. The specific influence of nonhepatic splanchnic venous blood flow on the liver. Surg Forum 1965; 16: 280-282 [PMID: 5835158]
- Marchioro TL, Porter KA, Brown BI, Otte JB, Starzl TE. The effect of partial portacaval 8 transposition on the canine liver. Surgery 1967; 61: 723-732 [PMID: 4290396]
- Marchioro TL, Porter KA, Dickinson TC, Faris TD, Starzi TE. Physiologic requirements for auxiliary liver homotransplantation. Surg Gynecol Obstet 1965; 121: 17-31 [PMID: 14314095]
- 10 Starzl TE, Francavilla A, Halgrimson CG, Francavilla FR, Porter KA, Brown TH, Putnam CW. The origin, hormonal nature, and action of hepatotrophic substances in portal venous blood. Surg Gynecol Obstet 1973; 137: 179-199 [PMID: 4353133]
- Duguay LR, Orloff MJ. Role of the pancreas in regulation of liver regeneration in dogs. Surg Forum 11 1977; 28: 387-390 [PMID: 617477]
- 12 Starzl TE, Francavilla A, Porter KA, Benichou J. The effect upon the liver of evisceration with or without hormone replacement. Surg Gynecol Obstet 1978; 146: 524-532 [PMID: 205003]
- Francavilla A, Starzl TE, Porter K, Foglieni CS, Michalopoulos GK, Carrieri G, Trejo J, Azzarone 13 A, Barone M, Zeng QH. Screening for candidate hepatic growth factors by selective portal infusion after canine Eck's fistula. Hepatology 1991; 14: 665-670 [PMID: 1916668 DOI: 10.1016/0270-9139(91)90055-z]



- 14 Sigel B, Dunn MR, Butterfield J. Effect of partial hepatectomy and ECK fistula on autotransplanted liver tissue. Evidence for a humoral mechanism in liver regeneration. Surg Forum 1963; 14: 72-74 [PMID: 14064594]
- 15 Bucher NL, Scott JF, Aub JC. Regeneration of the liver in parabiotic rats. Cancer Res 1951; 11: 457-465 [PMID: 14839642]
- 16 Wenneker AS, Sussman N. Regeneration of liver tissue following partial hepatectomy in parabiotic rats. Proc Soc Exp Biol Med 1951; 76: 683-686 [PMID: 14844311 DOI: 10.3181/00379727-76-18594
- 17 Bhushan B, Chavan H, Borude P, Xie Y, Du K, McGill MR, Lebofsky M, Jaeschke H, Krishnamurthy P, Apte U. Dual Role of Epidermal Growth Factor Receptor in Liver Injury and Regeneration after Acetaminophen Overdose in Mice. Toxicol Sci 2017; 155: 363-378 [PMID: 28123000 DOI: 10.1093/toxsci/kfw213]
- 18 Bhushan B, Stoops JW, Mars WM, Orr A, Bowen WC, Paranjpe S, Michalopoulos GK. TCPOBOP-Induced Hepatomegaly and Hepatocyte Proliferation are Attenuated by Combined Disruption of MET and EGFR Signaling. Hepatology 2019; 69: 1702-1718 [PMID: 29888801 DOI: 10.1002/hep.30109]
- 19 Wei W, Dirsch O, Mclean AL, Zafarnia S, Schwier M, Dahmen U. Rodent models and imaging techniques to study liver regeneration. Eur Surg Res 2015; 54: 97-113 [PMID: 25402256 DOI: 10.1159/000368573]
- 20 Higgins G, M, Anderson RM. Experimental pathology of the liver – Restoration of the liver of the white rat following partial surgical removal. Arch Pathol 1931; 12: 186-202
- 21 Moolten FL, Bucher NL. Regeneration of rat liver: transfer of humoral agent by cross circulation. Science 1967; 158: 272-274 [PMID: 6053886 DOI: 10.1126/science.158.3798.272]
- 22 Couinaud C. [Liver lobes and segments: notes on the anatomical architecture and surgery of the liver]. Presse Med 1954; 62: 709-712 [PMID: 13177441]
- 23 Kogure K, Ishizaki M, Nemoto M, Kuwano H, Makuuchi M. A comparative study of the anatomy of rat and human livers. J Hepatobiliary Pancreat Surg 1999; 6: 171-175 [PMID: 10398905 DOI: 10.1007/s005340050101
- 24 Andersen KJ, Knudsen AR, Kannerup AS, Sasanuma H, Nyengaard JR, Hamilton-Dutoit S, Erlandsen EJ, Jørgensen B, Mortensen FV. The natural history of liver regeneration in rats: description of an animal model for liver regeneration studies. Int J Surg 2013; 11: 903-908 [PMID: 23899538 DOI: 10.1016/j.ijsu.2013.07.009]
- Madrahimov N, Dirsch O, Broelsch C, Dahmen U. Marginal hepatectomy in the rat: from anatomy 25 to surgery. Ann Surg 2006; 244: 89-98 [PMID: 16794393 DOI: 10.1097/01.sla.0000218093.12408.0f
- 26 Ninomiya M, Shirabe K, Terashi T, Ijichi H, Yonemura Y, Harada N, Soejima Y, Taketomi A, Shimada M, Maehara Y. Deceleration of regenerative response improves the outcome of rat with massive hepatectomy. Am J Transplant 2010; 10: 1580-1587 [PMID: 20642684 DOI: 10.1111/j.1600-6143.2010.03150.x]
- 27 Ren W, Wang X, Zhang A, Li C, Chen G, Ge X, Pan K, Dong JH. Selective bowel decontamination improves the survival of 90% hepatectomy in rats. J Surg Res 2015; 195: 454-464 [PMID: 25770738 DOI: 10.1016/j.jss.2015.01.024]
- 28 Mariotti V, Strazzabosco M, Fabris L, Calvisi DF. Animal models of biliary injury and altered bile acid metabolism. Biochim Biophys Acta Mol Basis Dis 2018; 1864: 1254-1261 [PMID: 28709963 DOI: 10.1016/j.bbadis.2017.06.027]
- 29 Rozga J, Jeppsson B, Bengmark S. Portal branch ligation in the rat. Reevaluation of a model. Am J Pathol 1986; 125: 300-308 [PMID: 3789089]
- 30 Khan RA, Khan MR, Sahreen S. CCl4-induced hepatotoxicity: protective effect of rutin on p53, CYP2E1 and the antioxidative status in rat. BMC Complement Altern Med 2012; 12: 178 [PMID: 23043521 DOI: 10.1186/1472-6882-12-178]
- Vidhya A, Renjugopal V, Indira M. Impact of thiamine supplementation in the reversal of ethanol 31 induced toxicity in rats. Indian J Physiol Pharmacol 2013; 57: 406-417 [PMID: 24968580]
- Adeyanju AA, Asejeje FO, Molehin OR, Owoeye O, Olatoye EO, Ekpo EN. Protective role of 32 protocatechuic acid in carbon tetrachloride-induced oxidative stress via modulation of proinflammatory cytokines levels in brain and liver of Wistar rats. J Basic Clin Physiol Pharmacol 2021 [PMID: 33735948 DOI: 10.1515/jbcpp-2020-0202]
- 33 Delgado-Venegas CS, Martínez-Hernández SL, Cervantes-García D, Montes de Oca-Luna R, de Jesús Loera-Arias M, Mata-Martínez MG, Ventura-Juárez J, Muñoz-Ortega MH. Modulating effects of the probiotic *Lactococcus lactis* on the hepatic fibrotic process induced by CCl₄ in Wistar rats. Exp Ther Med 2021; 21: 339 [PMID: 33732312 DOI: 10.3892/etm.2021.9770]
- 34 Tong M, Zheng Q, Liu M, Chen L, Lin YH, Tang SG, Zhu YM. 5-methoxytryptophan alleviates liver fibrosis by modulating FOXO3a/miR-21/ATG5 signaling pathway mediated autophagy. Cell Cycle 2021; 20: 676-688 [PMID: 33734029 DOI: 10.1080/15384101.2021.1897241]
- 35 Evarts RP, Nagy P, Marsden E, Thorgeirsson SS. A precursor-product relationship exists between oval cells and hepatocytes in rat liver. Carcinogenesis 1987; 8: 1737-1740 [PMID: 3664968 DOI: 10.1093/carcin/8.11.1737
- 36 Trautwein C, Will M, Kubicka S, Rakemann T, Flemming P, Manns MP. 2-acetaminofluorene blocks cell cycle progression after hepatectomy by p21 induction and lack of cyclin E expression. Oncogene 1999; 18: 6443-6453 [PMID: 10597246 DOI: 10.1038/sj.onc.1203045]



- 37 Rahman TM, Hodgson HJ. Animal models of acute hepatic failure. Int J Exp Pathol 2000; 81: 145-157 [PMID: 10762442 DOI: 10.1046/j.1365-2613.2000.00144.x]
- 38 Peeling J, Shoemaker L, Gauthier T, Benarroch A, Sutherland GR, Minuk GY. Cerebral metabolic and histological effects of thioacetamide-induced liver failure. Am J Physiol 1993; 265: G572-G578 [PMID: 8214078 DOI: 10.1152/ajpgi.1993.265.3.G572]
- 39 Martins PN, Theruvath TP, Neuhaus P. Rodent models of partial hepatectomies. Liver Int 2008; 28: 3-11 [PMID: 18028319 DOI: 10.1111/j.1478-3231.2007.01628.x]
- 40 Sakamoto T, Ezure T, Lunz J, Murase N, Tsuji H, Fung JJ, Demetris AJ. Concanavalin A simultaneously primes liver hematopoietic and epithelial progenitor cells for parallel expansion during liver regeneration after partial hepatectomy in mice. Hepatology 2000; 32: 256-267 [PMID: 10915732 DOI: 10.1053/jhep.2000.9406]
- 41 Kollmar O, Corsten M, Scheuer C, Vollmar B, Schilling MK, Menger MD. Tumour growth following portal branch ligation in an experimental model of liver metastases. Br J Surg 2010; 97: 917-926 [PMID: 20474002 DOI: 10.1002/bjs.7003]
- 42 Heinrich S, Georgiev P, Weber A, Vergopoulos A, Graf R, Clavien PA. Partial bile duct ligation in mice: a novel model of acute cholestasis. Surgery 2011; 149: 445-451 [PMID: 20817234 DOI: 10.1016/j.surg.2010.07.046]
- Bruccoleri A, Gallucci R, Germolec DR, Blackshear P, Simeonova P, Thurman RG, Luster MI. 43 Induction of early-immediate genes by tumor necrosis factor alpha contribute to liver repair following chemical-induced hepatotoxicity. Hepatology 1997; 25: 133-141 [PMID: 8985279 DOI: 10.1002/hep.510250125]
- 44 Koniaris LG, Zimmers-Koniaris T, Hsiao EC, Chavin K, Sitzmann JV, Farber JM. Cytokineresponsive gene-2/IFN-inducible protein-10 expression in multiple models of liver and bile duct injury suggests a role in tissue regeneration. J Immunol 2001; 167: 399-406 [PMID: 11418676 DOI: 10.4049/jimmunol.167.1.399]
- Gardner CR, Laskin JD, Dambach DM, Sacco M, Durham SK, Bruno MK, Cohen SD, Gordon 45 MK, Gerecke DR, Zhou P, Laskin DL. Reduced hepatotoxicity of acetaminophen in mice lacking inducible nitric oxide synthase: potential role of tumor necrosis factor-alpha and interleukin-10. Toxicol Appl Pharmacol 2002; 184: 27-36 [PMID: 12392966]
- 46 Diehl AM. Effect of ethanol on tumor necrosis factor signaling during liver regeneration. Clin Biochem 1999; 32: 571-578 [PMID: 10614720 DOI: 10.1016/s0009-9120(99)00057-0]
- 47 Erhardt A, Tiegs G. Tolerance induction in response to liver inflammation. Dig Dis 2010; 28: 86-92 [PMID: 20460895 DOI: 10.1159/000282069]
- 48 Imajo K, Yoneda M, Kessoku T, Ogawa Y, Maeda S, Sumida Y, Hyogo H, Eguchi Y, Wada K, Nakajima A. Rodent models of nonalcoholic fatty liver disease/nonalcoholic steatohepatitis. Int J Mol Sci 2013; 14: 21833-21857 [PMID: 24192824 DOI: 10.3390/ijms141121833]
- Tsuchida T, Lee YA, Fujiwara N, Ybanez M, Allen B, Martins S, Fiel MI, Goossens N, Chou HI, 49 Hoshida Y, Friedman SL. A simple diet- and chemical-induced murine NASH model with rapid progression of steatohepatitis, fibrosis and liver cancer. J Hepatol 2018; 69: 385-395 [PMID: 29572095 DOI: 10.1016/j.jhep.2018.03.011]
- Nikfarjam M, Malcontenti-Wilson C, Fanartzis M, Daruwalla J, Christophi C. A model of partial 50 hepatectomy in mice. J Invest Surg 2004; 17: 291-294 [PMID: 15385262 DOI: 10.1080/08941930490502871
- Palmes D, Spiegel HU. Animal models of liver regeneration. Biomaterials 2004; 25: 1601-1611 51 [PMID: 14697862 DOI: 10.1016/s0142-9612(03)00508-8]
- 52 Preisegger KH, Factor VM, Fuchsbichler A, Stumptner C, Denk H, Thorgeirsson SS. Atypical ductular proliferation and its inhibition by transforming growth factor beta1 in the 3,5diethoxycarbonyl-1,4-dihydrocollidine mouse model for chronic alcoholic liver disease. Lab Invest 1999; 79: 103-109 [PMID: 10068199]
- Akhurst B, Croager EJ, Farley-Roche CA, Ong JK, Dumble ML, Knight B, Yeoh GC. A modified 53 choline-deficient, ethionine-supplemented diet protocol effectively induces oval cells in mouse liver. Hepatology 2001; 34: 519-522 [PMID: 11526537 DOI: 10.1053/jhep.2001.26751]
- 54 Tumanov AV, Koroleva EP, Christiansen PA, Khan MA, Ruddy MJ, Burnette B, Papa S, Franzoso G, Nedospasov SA, Fu YX, Anders RA. T cell-derived lymphotoxin regulates liver regeneration. Gastroenterology 2009; 136: 694-704.e4 [PMID: 18952083 DOI: 10.1053/j.gastro.2008.09.015]
- Webber EM, Wu JC, Wang L, Merlino G, Fausto N. Overexpression of transforming growth factor-55 alpha causes liver enlargement and increased hepatocyte proliferation in transgenic mice. Am J Pathol 1994; 145: 398-408 [PMID: 8053497]
- Cressman DE, Greenbaum LE, DeAngelis RA, Ciliberto G, Furth EE, Poli V, Taub R. Liver failure 56 and defective hepatocyte regeneration in interleukin-6-deficient mice. Science 1996; 274: 1379-1383 [PMID: 8910279 DOI: 10.1126/science.274.5291.1379]
- 57 Yamada Y, Kirillova I, Peschon JJ, Fausto N. Initiation of liver growth by tumor necrosis factor: deficient liver regeneration in mice lacking type I tumor necrosis factor receptor. Proc Natl Acad Sci USA 1997; 94: 1441-1446 [PMID: 9037072 DOI: 10.1073/pnas.94.4.1441]
- 58 Greenbaum LE, Li W, Cressman DE, Peng Y, Ciliberto G, Poli V, Taub R. CCAAT enhancerbinding protein beta is required for normal hepatocyte proliferation in mice after partial hepatectomy. J Clin Invest 1998; 102: 996-1007 [PMID: 9727068 DOI: 10.1172/jci3135]
- 59 Rai RM, Lee FY, Rosen A, Yang SQ, Lin HZ, Koteish A, Liew FY, Zaragoza C, Lowenstein C, Diehl AM. Impaired liver regeneration in inducible nitric oxide synthasedeficient mice. Proc Natl



Acad Sci USA 1998; 95: 13829-13834 [PMID: 9811886 DOI: 10.1073/pnas.95.23.13829]

- 60 Roselli HT, Su M, Washington K, Kerins DM, Vaughan DE, Russell WE. Liver regeneration is transiently impaired in urokinase-deficient mice. Am J Physiol 1998; 275: G1472-G1479 [PMID: 9843786 DOI: 10.1152/ajpgi.1998.275.6.G1472]
- 61 Yamada Y, Webber EM, Kirillova I, Peschon JJ, Fausto N. Analysis of liver regeneration in mice lacking type 1 or type 2 tumor necrosis factor receptor: requirement for type 1 but not type 2 receptor. Hepatology 1998; 28: 959-970 [PMID: 9755232 DOI: 10.1002/hep.510280410]
- Anderson SP, Yoon L, Richard EB, Dunn CS, Cattley RC, Corton JC. Delayed liver regeneration in 62 peroxisome proliferator-activated receptor-alpha-null mice. Hepatology 2002; 36: 544-554 [PMID: 12198646 DOI: 10.1053/jhep.2002.35276]
- Leu JI, Crissey MA, Craig LE, Taub R. Impaired hepatocyte DNA synthetic response 63 posthepatectomy in insulin-like growth factor binding protein 1-deficient mice with defects in C/EBP beta and mitogen-activated protein kinase/extracellular signal-regulated kinase regulation. Mol Cell Biol 2003; 23: 1251-1259 [PMID: 12556485 DOI: 10.1128/mcb.23.4.1251-1259.2003]
- 64 Strey CW, Markiewski M, Mastellos D, Tudoran R, Spruce LA, Greenbaum LE, Lambris JD. The proinflammatory mediators C3a and C5a are essential for liver regeneration. J Exp Med 2003; 198: 913-923 [PMID: 12975457 DOI: 10.1084/jem.20030374]
- Borowiak M, Garratt AN, Wüstefeld T, Strehle M, Trautwein C, Birchmeier C. Met provides essential signals for liver regeneration. Proc Natl Acad Sci USA 2004; 101: 10608-10613 [PMID: 15249655 DOI: 10.1073/pnas.0403412101]
- 66 Mohammed FF, Smookler DS, Taylor SE, Fingleton B, Kassiri Z, Sanchez OH, English JL, Matrisian LM, Au B, Yeh WC, Khokha R. Abnormal TNF activity in Timp3-/- mice leads to chronic hepatic inflammation and failure of liver regeneration. Nat Genet 2004; 36: 969-977 [PMID: 15322543 DOI: 10.1038/ng1413]
- Nakamura K, Nonaka H, Saito H, Tanaka M, Miyajima A. Hepatocyte proliferation and tissue 67 remodeling is impaired after liver injury in oncostatin M receptor knockout mice. Hepatology 2004; 39: 635-644 [PMID: 14999682 DOI: 10.1002/hep.20086]
- Oe S, Lemmer ER, Conner EA, Factor VM, Levéen P, Larsson J, Karlsson S, Thorgeirsson SS. 68 Intact signaling by transforming growth factor beta is not required for termination of liver regeneration in mice. Hepatology 2004; 40: 1098-1105 [PMID: 15389868 DOI: 10.1002/hep.20426]
- Duffield JS, Forbes SJ, Constandinou CM, Clay S, Partolina M, Vuthoori S, Wu S, Lang R, Iredale 69 JP. Selective depletion of macrophages reveals distinct, opposing roles during liver injury and repair. J Clin Invest 2005; 115: 56-65 [PMID: 15630444 DOI: 10.1172/jci22675]
- 70 Mitchell C, Nivison M, Jackson LF, Fox R, Lee DC, Campbell JS, Fausto N. Heparin-binding epidermal growth factor-like growth factor links hepatocyte priming with cell cycle progression during liver regeneration. J Biol Chem 2005; 280: 2562-2568 [PMID: 15536070 DOI: 10.1074/jbc.M412372200
- Oliver JR, Mara TW, Cherian MG. Impaired hepatic regeneration in metallothionein-I/II knockout 71 mice after partial hepatectomy. Exp Biol Med (Maywood) 2005; 230: 61-67 [PMID: 15618127 DOI: 10.1177/153537020523000108
- 72 Seki E, Tsutsui H, Iimuro Y, Naka T, Son G, Akira S, Kishimoto T, Nakanishi K, Fujimoto J. Contribution of Toll-like receptor/myeloid differentiation factor 88 signaling to murine liver regeneration. Hepatology 2005; 41: 443-450 [PMID: 15723296 DOI: 10.1002/hep.20603]
- Fernández MA, Albor C, Ingelmo-Torres M, Nixon SJ, Ferguson C, Kurzchalia T, Tebar F, Enrich 73 C, Parton RG, Pol A. Caveolin-1 is essential for liver regeneration. Science 2006; 313: 1628-1632 [PMID: 16973879 DOI: 10.1126/science.1130773]
- 74 Olle EW, Ren X, McClintock SD, Warner RL, Deogracias MP, Johnson KJ, Colletti LM. Matrix metalloproteinase-9 is an important factor in hepatic regeneration after partial hepatectomy in mice. Hepatology 2006; 44: 540-549 [PMID: 16941692 DOI: 10.1002/hep.21314]
- 75 Mayoral R, Fernández-Martínez A, Roy R, Boscá L, Martín-Sanz P. Dispensability and dynamics of caveolin-1 during liver regeneration and in isolated hepatic cells. Hepatology 2007; 46: 813-822 [PMID: 17654701 DOI: 10.1002/hep.21746]
- 76 Liu B, Bell AW, Paranjpe S, Bowen WC, Khillan JS, Luo JH, Mars WM, Michalopoulos GK. Suppression of liver regeneration and hepatocyte proliferation in hepatocyte-targeted glypican 3 transgenic mice. Hepatology 2010; 52: 1060-1067 [PMID: 20812357 DOI: 10.1002/hep.23794]
- 77 Borude P, Edwards G, Walesky C, Li F, Ma X, Kong B, Guo GL, Apte U. Hepatocyte-specific deletion of farnesoid X receptor delays but does not inhibit liver regeneration after partial hepatectomy in mice. Hepatology 2012; 56: 2344-2352 [PMID: 22730081 DOI: 10.1002/hep.25918]
- 78 Bhave VS, Mars W, Donthamsetty S, Zhang X, Tan L, Luo J, Bowen WC, Michalopoulos GK. Regulation of liver growth by glypican 3, CD81, hedgehog, and Hhex. Am J Pathol 2013; 183: 153-159 [PMID: 23665349 DOI: 10.1016/j.ajpath.2013.03.013]
- 79 Kong B, Huang J, Zhu Y, Li G, Williams J, Shen S, Aleksunes LM, Richardson JR, Apte U, Rudnick DA, Guo GL. Fibroblast growth factor 15 deficiency impairs liver regeneration in mice. Am J Physiol Gastrointest Liver Physiol 2014; 306: G893-G902 [PMID: 24699334 DOI: 10.1152/ajpgi.00337.2013]
- 80 Yang J, Mowry LE, Nejak-Bowen KN, Okabe H, Diegel CR, Lang RA, Williams BO, Monga SP. β -catenin signaling in murine liver zonation and regeneration: a Wnt-Wnt situation! Hepatology 2014; 60: 964-976 [PMID: 24700412 DOI: 10.1002/hep.27082]



- 81 Lu WY, Bird TG, Boulter L, Tsuchiya A, Cole AM, Hay T, Guest RV, Wojtacha D, Man TY, Mackinnon A, Ridgway RA, Kendall T, Williams MJ, Jamieson T, Raven A, Hay DC, Iredale JP, Clarke AR, Sansom OJ, Forbes SJ. Hepatic progenitor cells of biliary origin with liver repopulation capacity. Nat Cell Biol 2015; 17: 971-983 [PMID: 26192438 DOI: 10.1038/ncb3203]
- 82 Swiderska-Syn M, Xie G, Michelotti GA, Jewell ML, Premont RT, Syn WK, Diehl AM. Hedgehog regulates yes-associated protein 1 in regenerating mouse liver. Hepatology 2016; 64: 232-244 [PMID: 26970079 DOI: 10.1002/hep.28542]
- 83 Tsagianni A, Mars WM, Bhushan B, Bowen WC, Orr A, Stoops J, Paranjpe S, Tseng GC, Liu S, Michalopoulos GK. Combined Systemic Disruption of MET and Epidermal Growth Factor Receptor Signaling Causes Liver Failure in Normal Mice. Am J Pathol 2018; 188: 2223-2235 [PMID: 30031724 DOI: 10.1016/j.ajpath.2018.06.009]
- 84 Sivertsen Åsrud K, Pedersen L, Aesoy R, Muwonge H, Aasebø E, Nitschke Pettersen IK, Herfindal L, Dobie R, Jenkins S, Berge RK, Henderson NC, Selheim F, Døskeland SO, Bakke M. Mice depleted for Exchange Proteins Directly Activated by cAMP (Epac) exhibit irregular liver regeneration in response to partial hepatectomy. Sci Rep 2019; 9: 13789 [PMID: 31551444 DOI: 10.1038/s41598-019-50219-8
- 85 Fortier M, Cadoux M, Boussetta N, Pham S, Donné R, Couty JP, Desdouets C, Celton-Morizur S. Hepatospecific ablation of p38a MAPK governs liver regeneration through modulation of inflammatory response to CCl₄-induced acute injury. Sci Rep 2019; 9: 14614 [PMID: 31601995 DOI: 10.1038/s41598-019-51175-z]
- Fazel Modares N, Polz R, Haghighi F, Lamertz L, Behnke K, Zhuang Y, Kordes C, Häussinger D, 86 Sorg UR, Pfeffer K, Floss DM, Moll JM, Piekorz RP, Ahmadian MR, Lang PA, Scheller J. IL-6 Trans-signaling Controls Liver Regeneration After Partial Hepatectomy. Hepatology 2019; 70: 2075-2091 [PMID: 31100194 DOI: 10.1002/hep.30774]
- 87 Zhou Y, Xu M, Liu P, Liang B, Qian M, Wang H, Song X, Nyshadham P, Che L, Calvisi DF, Li F, Lin S, Chen X. Mammalian Target of Rapamycin Complex 2 Signaling Is Required for Liver Regeneration in a Cholestatic Liver Injury Murine Model. Am J Pathol 2020; 190: 1414-1426 [PMID: 32275903 DOI: 10.1016/j.ajpath.2020.03.010]
- 88 Laschinger M, Wang Y, Holzmann G, Wang B, Stöß C, Lu M, Brugger M, Schneider A, Knolle P, Wohlleber D, Schulze S, Steiger K, Tsujikawa K, Altmayr F, Friess H, Hartmann D, Hüser N, Holzmann B. The CGRP receptor component RAMP1 Links sensory innervation with YAP activity in the regenerating liver. FASEB J 2020; 34: 8125-8138 [PMID: 32329113 DOI: 10.1096/fi.201903200R
- Seguin A, Jia X, Earl AM, Li L, Wallace J, Qiu A, Bradlev T, Shrestha R, Troadec MB, Hockin M, 89 Titen S, Warner DE, Dowdle PT, Wohlfahrt ME, Hillas E, Firpo MA, Phillips JD, Kaplan J, Paw BH, Barasch J, Ward DM. The mitochondrial metal transporters mitoferrin1 and mitoferrin2 are required for liver regeneration and cell proliferation in mice. J Biol Chem 2020; 295: 11002-11020 [PMID: 32518166 DOI: 10.1074/jbc.RA120.013229]
- 90 Xue Y, Bhushan B, Mars WM, Bowen W, Tao J, Orr A, Stoops J, Yu Y, Luo J, Duncan AW, Michalopoulos GK. Phosphorylated Ezrin (Thr567) Regulates Hippo Pathway and Yes-Associated Protein (Yap) in Liver. Am J Pathol 2020; 190: 1427-1437 [PMID: 32289287 DOI: 10.1016/j.ajpath.2020.03.014]
- 91 Goessling W, Sadler KC. Zebrafish: an important tool for liver disease research. Gastroenterology 2015; 149: 1361-1377 [PMID: 26319012 DOI: 10.1053/j.gastro.2015.08.034]
- 92 Howe K, Clark MD, Torroja CF, Torrance J, Berthelot C, Muffato M, Collins JE, Humphray S, McLaren K, Matthews L, McLaren S, Sealy I, Caccamo M, Churcher C, Scott C, Barrett JC, Koch R, Rauch GJ, White S, Chow W, Kilian B, Quintais LT, Guerra-Assunção JA, Zhou Y, Gu Y, Yen J, Vogel JH, Eyre T, Redmond S, Banerjee R, Chi J, Fu B, Langley E, Maguire SF, Laird GK, Lloyd D, Kenyon E, Donaldson S, Sehra H, Almeida-King J, Loveland J, Trevanion S, Jones M, Quail M, Willey D, Hunt A, Burton J, Sims S, McLay K, Plumb B, Davis J, Clee C, Oliver K, Clark R, Riddle C, Elliot D, Threadgold G, Harden G, Ware D, Begum S, Mortimore B, Kerry G, Heath P, Phillimore B, Tracey A, Corby N, Dunn M, Johnson C, Wood J, Clark S, Pelan S, Griffiths G, Smith M, Glithero R, Howden P, Barker N, Lloyd C, Stevens C, Harley J, Holt K, Panagiotidis G, Lovell J, Beasley H, Henderson C, Gordon D, Auger K, Wright D, Collins J, Raisen C, Dyer L, Leung K, Robertson L, Ambridge K, Leongamornlert D, McGuire S, Gilderthorp R, Griffiths C, Manthravadi D, Nichol S, Barker G, Whitehead S, Kay M, Brown J, Murnane C, Gray E, Humphries M, Sycamore N, Barker D, Saunders D, Wallis J, Babbage A, Hammond S, Mashreghi-Mohammadi M, Barr L, Martin S, Wray P, Ellington A, Matthews N, Ellwood M, Woodmansey R, Clark G, Cooper J, Tromans A, Grafham D, Skuce C, Pandian R, Andrews R, Harrison E, Kimberley A, Garnett J, Fosker N, Hall R, Garner P, Kelly D, Bird C, Palmer S, Gehring I, Berger A, Dooley CM, Ersan-Ürün Z, Eser C, Geiger H, Geisler M, Karotki L, Kirn A, Konantz J, Konantz M, Oberländer M, Rudolph-Geiger S, Teucke M, Lanz C, Raddatz G, Osoegawa K, Zhu B, Rapp A, Widaa S, Langford C, Yang F, Schuster SC, Carter NP, Harrow J, Ning Z, Herrero J, Searle SM, Enright A, Geisler R, Plasterk RH, Lee C, Westerfield M, de Jong PJ, Zon LI, Postlethwait JH, Nüsslein-Volhard C, Hubbard TJ, Roest Crollius H, Rogers J, Stemple DL. The zebrafish reference genome sequence and its relationship to the human genome. Nature 2013; 496: 498-503 [PMID: 23594743 DOI: 10.1038/nature12111]
- Wilkins BJ, Pack M. Zebrafish models of human liver development and disease. Compr Physiol 93 2013; 3: 1213-1230 [PMID: 23897685 DOI: 10.1002/cphy.c120021]



- 94 Wang S, Miller SR, Ober EA, Sadler KC. Making It New Again: Insight Into Liver Development, Regeneration, and Disease From Zebrafish Research. Curr Top Dev Biol 2017; 124: 161-195 [PMID: 28335859 DOI: 10.1016/bs.ctdb.2016.11.012]
- 95 Ko S, Russell JO, Tian J, Gao C, Kobayashi M, Feng R, Yuan X, Shao C, Ding H, Poddar M, Singh S, Locker J, Weng HL, Monga SP, Shin D. Hdac1 Regulates Differentiation of Bipotent Liver Progenitor Cells During Regeneration via Sox9b and Cdk8. Gastroenterology 2019; 156: 187-202.e14 [PMID: 30267710 DOI: 10.1053/j.gastro.2018.09.039]
- 96 Chu J, Sadler KC. New school in liver development: lessons from zebrafish. Hepatology 2009; 50: 1656-1663 [PMID: 19693947 DOI: 10.1002/hep.23157]
- 97 Godoy P, Hewitt NJ, Albrecht U, Andersen ME, Ansari N, Bhattacharya S, Bode JG, Bolleyn J, Borner C, Böttger J, Braeuning A, Budinsky RA, Burkhardt B, Cameron NR, Camussi G, Cho CS, Choi YJ, Craig Rowlands J, Dahmen U, Damm G, Dirsch O, Donato MT, Dong J, Dooley S, Drasdo D, Eakins R, Ferreira KS, Fonsato V, Fraczek J, Gebhardt R, Gibson A, Glanemann M, Goldring CE, Gómez-Lechón MJ, Groothuis GM, Gustavsson L, Guyot C, Hallifax D, Hammad S, Hayward A, Häussinger D, Hellerbrand C, Hewitt P, Hoehme S, Holzhütter HG, Houston JB, Hrach J, Ito K, Jaeschke H, Keitel V, Kelm JM, Kevin Park B, Kordes C, Kullak-Ublick GA, LeCluyse EL, Lu P, Luebke-Wheeler J, Lutz A, Maltman DJ, Matz-Soja M, McMullen P, Merfort I, Messner S, Meyer C, Mwinyi J, Naisbitt DJ, Nussler AK, Olinga P, Pampaloni F, Pi J, Pluta L, Przyborski SA, Ramachandran A, Rogiers V, Rowe C, Schelcher C, Schmich K, Schwarz M, Singh B, Stelzer EH, Stieger B, Stöber R, Sugiyama Y, Tetta C, Thasler WE, Vanhaecke T, Vinken M, Weiss TS, Widera A, Woods CG, Xu JJ, Yarborough KM, Hengstler JG. Recent advances in 2D and 3D in vitro systems using primary hepatocytes, alternative hepatocyte sources and non-parenchymal liver cells and their use in investigating mechanisms of hepatotoxicity, cell signaling and ADME. Arch Toxicol 2013; 87: 1315-1530 [PMID: 23974980 DOI: 10.1007/s00204-013-1078-5]
- Ramboer E, Vanhaecke T, Rogiers V, Vinken M. Immortalized Human Hepatic Cell Lines for In Vitro Testing and Research Purposes. Methods Mol Biol 2015; 1250: 53-76 [PMID: 26272134 DOI: 10.1007/978-1-4939-2074-7 4]
- 99 Michalopoulos G, Cianciulli HD, Novotny AR, Kligerman AD, Strom SC, Jirtle RL. Liver regeneration studies with rat hepatocytes in primary culture. Cancer Res 1982; 42: 4673-4682 [PMID: 6215120]
- 100 Jiang J, Wolters JE, van Breda SG, Kleinjans JC, de Kok TM. Development of novel tools for the in vitro investigation of drug-induced liver injury. Expert Opin Drug Metab Toxicol 2015; 11: 1523-1537 [PMID: 26155718 DOI: 10.1517/17425255.2015.1065814]
- Hart SN, Li Y, Nakamoto K, Subileau EA, Steen D, Zhong XB. A comparison of whole genome 101 gene expression profiles of HepaRG cells and HepG2 cells to primary human hepatocytes and human liver tissues. Drug Metab Dispos 2010; 38: 988-994 [PMID: 20228232 DOI: 10.1124/dmd.109.031831
- Gómez-Lechón MJ, Tolosa L, Conde I, Donato MT. Competency of different cell models to predict 102 human hepatotoxic drugs. Expert Opin Drug Metab Toxicol 2014; 10: 1553-1568 [PMID: 25297626 DOI: 10.1517/17425255.2014.967680]
- Smith CM, Nolan CK, Edwards MA, Hatfield JB, Stewart TW, Ferguson SS, Lecluyse EL, Sahi J. 103 A comprehensive evaluation of metabolic activity and intrinsic clearance in suspensions and monolayer cultures of cryopreserved primary human hepatocytes. J Pharm Sci 2012; 101: 3989-4002 [PMID: 22806329 DOI: 10.1002/jps.23262]
- 104 Tuschl G, Lauer B, Mueller SO. Primary hepatocytes as a model to analyze species-specific toxicity and drug metabolism. Expert Opin Drug Metab Toxicol 2008; 4: 855-870 [PMID: 18624675 DOI: 10.1517/17425255.4.7.855
- Bell CC, Dankers ACA, Lauschke VM, Sison-Young R, Jenkins R, Rowe C, Goldring CE, Park K, 105 Regan SL, Walker T, Schofield C, Baze A, Foster AJ, Williams DP, van de Ven AWM, Jacobs F, Houdt JV, Lähteenmäki T, Snoeys J, Juhila S, Richert L, Ingelman-Sundberg M. Comparison of Hepatic 2D Sandwich Cultures and 3D Spheroids for Long-term Toxicity Applications: A Multicenter Study. Toxicol Sci 2018; 162: 655-666 [PMID: 29329425 DOI: 10.1093/toxsci/kfx289]
- Liu X, Brouwer KL, Gan LS, Brouwer KR, Stieger B, Meier PJ, Audus KL, LeCluyse EL. Partial 106 maintenance of taurocholate uptake by adult rat hepatocytes cultured in a collagen sandwich configuration. Pharm Res 1998; 15: 1533-1539 [PMID: 9794494 DOI: 10.1023/a:1011994831139]
- 107 LeCluyse EL, Audus KL, Hochman JH. Formation of extensive canalicular networks by rat hepatocytes cultured in collagen-sandwich configuration. Am J Physiol 1994; 266: C1764-C1774 [PMID: 8023906 DOI: 10.1152/ajpcell.1994.266.6.C1764]
- 108 Du Y, Han R, Wen F, Ng San San S, Xia L, Wohland T, Leo HL, Yu H. Synthetic sandwich culture of 3D hepatocyte monolayer. Biomaterials 2008; 29: 290-301 [PMID: 17964646 DOI: 10.1016/j.biomaterials.2007.09.016]
- 109 Underhill GH, Khetani SR. Emerging trends in modeling human liver disease in vitro. APL Bioeng 2019; 3: 040902 [PMID: 31893256 DOI: 10.1063/1.5119090]
- 110 Edmondson R, Broglie JJ, Adcock AF, Yang L. Three-dimensional cell culture systems and their applications in drug discovery and cell-based biosensors. Assay Drug Dev Technol 2014; 12: 207-218 [PMID: 24831787 DOI: 10.1089/adt.2014.573]
- Hsia GSP, Esposito J, da Rocha LA, Ramos SLG, Okamoto OK. Clinical Application of Human 111 Induced Pluripotent Stem Cell-Derived Organoids as an Alternative to Organ Transplantation. Stem Cells Int 2021; 2021: 6632160 [PMID: 33679987 DOI: 10.1155/2021/6632160]



- 112 Ramli MNB, Lim YS, Koe CT, Demircioglu D, Tng W, Gonzales KAU, Tan CP, Szczerbinska I, Liang H, Soe EL, Lu Z, Ariyachet C, Yu KM, Koh SH, Yaw LP, Jumat NHB, Lim JSY, Wright G, Shabbir A, Dan YY, Ng HH, Chan YS. Human Pluripotent Stem Cell-Derived Organoids as Models of Liver Disease. Gastroenterology 2020; 159: 1471-1486.e12 [PMID: 32553762 DOI: 10.1053/j.gastro.2020.06.010
- Wang B, Jakus AE, Baptista PM, Soker S, Soto-Gutierrez A, Abecassis MM, Shah RN, Wertheim 113 JA. Functional Maturation of Induced Pluripotent Stem Cell Hepatocytes in Extracellular Matrix-A Comparative Analysis of Bioartificial Liver Microenvironments. Stem Cells Transl Med 2016; 5: 1257-1267 [PMID: 27421950 DOI: 10.5966/sctm.2015-0235]
- 114 Ouchi R, Togo S, Kimura M, Shinozawa T, Koido M, Koike H, Thompson W, Karns RA, Mayhew CN, McGrath PS, McCauley HA, Zhang RR, Lewis K, Hakozaki S, Ferguson A, Saiki N, Yoneyama Y, Takeuchi I, Mabuchi Y, Akazawa C, Yoshikawa HY, Wells JM, Takebe T. Modeling Steatohepatitis in Humans with Pluripotent Stem Cell-Derived Organoids. Cell Metab 2019; 30: 374-384.e6 [PMID: 31155493 DOI: 10.1016/j.cmet.2019.05.007]
- 115 Olgasi C, Cucci A, Follenzi A. iPSC-Derived Liver Organoids: A Journey from Drug Screening, to Disease Modeling, Arriving to Regenerative Medicine. Int J Mol Sci 2020; 21 [PMID: 32867371 DOI: 10.3390/ijms21176215]
- Huch M, Gehart H, van Boxtel R, Hamer K, Blokzijl F, Verstegen MM, Ellis E, van Wenum M, 116 Fuchs SA, de Ligt J, van de Wetering M, Sasaki N, Boers SJ, Kemperman H, de Jonge J, Ijzermans JN, Nieuwenhuis EE, Hoekstra R, Strom S, Vries RR, van der Laan LJ, Cuppen E, Clevers H. Longterm culture of genome-stable bipotent stem cells from adult human liver. Cell 2015; 160: 299-312 [PMID: 25533785 DOI: 10.1016/j.cell.2014.11.050]
- 117 Justin AW, Saeb-Parsy K, Markaki AE, Vallier L, Sampaziotis F. Advances in the generation of bioengineered bile ducts. Biochim Biophys Acta Mol Basis Dis 2018; 1864: 1532-1538 [PMID: 29097260 DOI: 10.1016/j.bbadis.2017.10.034]
- 118 Harper S, Hoff M, Skepper J, Davies S, Huguet E. Portal venous repopulation of decellularised rat liver scaffolds with syngeneic bone marrow stem cells. J Tissue Eng Regen Med 2020; 14: 1502-1512 [PMID: 32808475 DOI: 10.1002/term.3117]
- 119 Mito M, Kusano M, Kawaura Y. Hepatocyte transplantation in man. Transplant Proc 1992; 24: 3052-3053 [PMID: 1466053]
- 120 Davis MW, Vacanti JP. Toward development of an implantable tissue engineered liver. Biomaterials 1996; 17: 365-372 [PMID: 8745334 DOI: 10.1016/0142-9612(96)85575-x]
- 121 Meng F, Assiri A, Dhar D, Broering D. Whole liver engineering: A promising approach to develop functional liver surrogates. Liver Int 2017; 37: 1759-1772 [PMID: 28393454 DOI: 10.1111/liv.13444]
- 122 Li J, Pan J, Zhang L, Guo X, Yu Y. Culture of primary rat hepatocytes within porous chitosan scaffolds. J Biomed Mater Res A 2003; 67: 938-943 [PMID: 14613242 DOI: 10.1002/jbm.a.10076]
- 123 Harada K, Mitaka T, Miyamoto S, Sugimoto S, Ikeda S, Takeda H, Mochizuki Y, Hirata K. Rapid formation of hepatic organoid in collagen sponge by rat small hepatocytes and hepatic nonparenchymal cells. J Hepatol 2003; 39: 716-723 [PMID: 14568252 DOI: 10.1016/s0168-8278(03)00412-4
- Liu Tsang V, Chen AA, Cho LM, Jadin KD, Sah RL, DeLong S, West JL, Bhatia SN. Fabrication 124 of 3D hepatic tissues by additive photopatterning of cellular hydrogels. FASEB J 2007; 21: 790-801 [PMID: 17197384 DOI: 10.1096/fj.06-7117com]
- 125 Douglas DN, Kneteman NM. Mice with Chimeric Human Livers and Their Applications. Methods Mol Biol 2019; 1911: 459-479 [PMID: 30593646 DOI: 10.1007/978-1-4939-8976-8 32]
- Hillebrandt KH, Everwien H, Haep N, Keshi E, Pratschke J, Sauer IM. Strategies based on organ 126 decellularization and recellularization. Transpl Int 2019; 32: 571-585 [PMID: 31099920 DOI: 10.1111/tri.13462
- 127 Uygun BE, Soto-Gutierrez A, Yagi H, Izamis ML, Guzzardi MA, Shulman C, Milwid J, Kobayashi N, Tilles A, Berthiaume F, Hertl M, Nahmias Y, Yarmush ML, Uygun K. Organ reengineering through development of a transplantable recellularized liver graft using decellularized liver matrix. Nat Med 2010; 16: 814-820 [PMID: 20543851 DOI: 10.1038/nm.2170]
- 128 Bao J, Shi Y, Sun H, Yin X, Yang R, Li L, Chen X, Bu H. Construction of a portal implantable functional tissue-engineered liver using perfusion-decellularized matrix and hepatocytes in rats. Cell Transplant 2011; 20: 753-766 [PMID: 21054928 DOI: 10.3727/096368910X536572]
- Huppert SS, Campbell KM. Emerging advancements in liver regeneration and organogenesis as 129 tools for liver replacement. Curr Opin Organ Transplant 2016; 21: 581-587 [PMID: 27755169 DOI: 10.1097/mot.00000000000365]
- 130 Ko IK, Peng L, Peloso A, Smith CJ, Dhal A, Deegan DB, Zimmerman C, Clouse C, Zhao W, Shupe TD, Soker S, Yoo JJ, Atala A. Bioengineered transplantable porcine livers with re-endothelialized vasculature. Biomaterials 2015; 40: 72-79 [PMID: 25433603 DOI: 10.1016/j.biomaterials.2014.11.027
- Zhou P, Lessa N, Estrada DC, Severson EB, Lingala S, Zern MA, Nolta JA, Wu J. Decellularized 131 liver matrix as a carrier for the transplantation of human fetal and primary hepatocytes in mice. Liver Transpl 2011; 17: 418-427 [PMID: 21445925 DOI: 10.1002/lt.22270]
- Ogiso S, Yasuchika K, Fukumitsu K, Ishii T, Kojima H, Miyauchi Y, Yamaoka R, Komori J, 132 Katayama H, Kawai T, Yoshitoshi EY, Kita S, Yasuda K, Uemoto S. Efficient recellularisation of decellularised whole-liver grafts using biliary tree and foetal hepatocytes. Sci Rep 2016; 6: 35887



[PMID: 27767181 DOI: 10.1038/srep35887]

- Navarro-Tableros V, Herrera Sanchez MB, Figliolini F, Romagnoli R, Tetta C, Camussi G. 133 Recellularization of rat liver scaffolds by human liver stem cells. Tissue Eng Part A 2015; 21: 1929-1939 [PMID: 25794768 DOI: 10.1089/ten.TEA.2014.0573]
- 134 Butter A, Aliyev K, Hillebrandt KH, Raschzok N, Kluge M, Seiffert N, Tang P, Napierala H, Muhamma AI, Reutzel-Selke A, Andreou A, Pratschke J, Sauer IM, Struecker B. Evolution of graft morphology and function after recellularization of decellularized rat livers. J Tissue Eng Regen Med 2018; 12: e807-e816 [PMID: 27957815 DOI: 10.1002/term.2383]
- Chen Y, Geerts S, Jaramillo M, Uygun BE. Preparation of Decellularized Liver Scaffolds and 135 Recellularized Liver Grafts. Methods Mol Biol 2018; 1577: 255-270 [PMID: 28735385 DOI: 10.1007/7651_2017_56]
- 136 Chen Y, Devalliere J, Bulutoglu B, Yarmush ML, Uygun BE. Repopulation of intrahepatic bile ducts in engineered rat liver grafts. Technology (Singap World Sci) 2019; 7: 46-55 [PMID: 31388515 DOI: 10.1142/S2339547819500043]
- Takeishi K, Collin de l'Hortet A, Wang Y, Handa K, Guzman-Lepe J, Matsubara K, Morita K, Jang 137 S, Haep N, Florentino RM, Yuan F, Fukumitsu K, Tobita K, Sun W, Franks J, Delgado ER, Shapiro EM, Fraunhoffer NA, Duncan AW, Yagi H, Mashimo T, Fox IJ, Soto-Gutierrez A. Assembly and Function of a Bioengineered Human Liver for Transplantation Generated Solely from Induced Pluripotent Stem Cells. Cell Rep 2020; 31: 107711 [PMID: 32492423 DOI: 10.1016/j.celrep.2020.107711]
- Verstegen MMA, Willemse J, van den Hoek S, Kremers GJ, Luider TM, van Huizen NA, 138 Willemssen FEJA, Metselaar HJ, IJzermans JNM, van der Laan LJW, de Jonge J. Decellularization of Whole Human Liver Grafts Using Controlled Perfusion for Transplantable Organ Bioscaffolds. Stem Cells Dev 2017; 26: 1304-1315 [PMID: 28665233 DOI: 10.1089/scd.2017.0095]
- Neuberger J, Patel J, Caldwell H, Davies S, Hebditch V, Hollywood C, Hubscher S, Karkhanis S, 139 Lester W, Roslund N, West R, Wyatt JI, Heydtmann M. Guidelines on the use of liver biopsy in clinical practice from the British Society of Gastroenterology, the Royal College of Radiologists and the Royal College of Pathology. Gut 2020; 69: 1382-1403 [PMID: 32467090 DOI: 10.1136/gutinl-2020-321299
- 140 Schiødt FV, Ostapowicz G, Murray N, Satyanarana R, Zaman A, Munoz S, Lee WM. Alphafetoprotein and prognosis in acute liver failure. Liver Transpl 2006; 12: 1776-1781 [PMID: 17133565 DOI: 10.1002/Lt.20886]
- John K, Hadem J, Krech T, Wahl K, Manns MP, Dooley S, Batkai S, Thum T, Schulze-Osthoff K, 141 Bantel H. MicroRNAs play a role in spontaneous recovery from acute liver failure. Hepatology 2014; 60: 1346-1355 [PMID: 24913549 DOI: 10.1002/hep.27250]
- 142 Rutherford A, King LY, Hynan LS, Vedvyas C, Lin W, Lee WM, Chung RT; ALF Study Group. Development of an accurate index for predicting outcomes of patients with acute liver failure. Gastroenterology 2012; 143: 1237-1243 [PMID: 22885329 DOI: 10.1053/j.gastro.2012.07.113]
- Belghiti J, Hiramatsu K, Benoist S, Massault P, Sauvanet A, Farges O. Seven hundred forty-seven 143 hepatectomies in the 1990s: an update to evaluate the actual risk of liver resection. J Am Coll Surg 2000; 191: 38-46 [PMID: 10898182 DOI: 10.1016/s1072-7515(00)00261-1]
- Vauthey JN, Chaoui A, Do KA, Bilimoria MM, Fenstermacher MJ, Charnsangavej C, Hicks M, 144 Alsfasser G, Lauwers G, Hawkins IF, Caridi J. Standardized measurement of the future liver remnant prior to extended liver resection: methodology and clinical associations. Surgery 2000; 127: 512-519 [PMID: 10819059 DOI: 10.1067/msy.2000.105294]
- 145 Schindl MJ, Redhead DN, Fearon KC, Garden OJ, Wigmore SJ; Edinburgh Liver Surgery and Transplantation Experimental Research Group (eLISTER). The value of residual liver volume as a predictor of hepatic dysfunction and infection after major liver resection. Gut 2005; 54: 289-296 [PMID: 15647196 DOI: 10.1136/gut.2004.046524]
- Bennink RJ, Dinant S, Erdogan D, Heijnen BH, Straatsburg IH, Van Vliet AK, Van Gulik TM. 146 Preoperative assessment of postoperative remnant liver function using hepatobiliary scintigraphy. J Nuclear Med 2004; 45: 965-971
- 147 Martel G, Cieslak KP, Huang R, van Lienden KP, Wiggers JK, Belblidia A, Dagenais M, Lapointe R, van Gulik TM, Vandenbroucke-Menu F. Comparison of techniques for volumetric analysis of the future liver remnant: implications for major hepatic resections. HPB (Oxford) 2015; 17: 1051-1057 [PMID: 26373675 DOI: 10.1111/hpb.12480]
- 148 Heymsfield SB, Fulenwider T, Nordlinger B, Barlow R, Sones P, Kutner M. Accurate measurement of liver, kidney, and spleen volume and mass by computerized axial tomography. Ann Intern Med 1979; 90: 185-187 [PMID: 443650 DOI: 10.7326/0003-4819-90-2-185]
- 149 Fulcher AS, Szucs RA, Bassignani MJ, Marcos A. Right lobe living donor liver transplantation: preoperative evaluation of the donor with MR imaging. AJR Am J Roentgenol 2001; 176: 1483-1491 [PMID: 11373218 DOI: 10.2214/ajr.176.6.1761483]
- Gotra A, Sivakumaran L, Chartrand G, Vu KN, Vandenbroucke-Menu F, Kauffmann C, Kadoury S, 150 Gallix B, de Guise JA, Tang A. Liver segmentation: indications, techniques and future directions. Insights Imaging 2017; 8: 377-392 [PMID: 28616760 DOI: 10.1007/s13244-017-0558-1]
- 151 Shoup M, Gonen M, D'Angelica M, Jarnagin WR, DeMatteo RP, Schwartz LH, Tuorto S, Blumgart LH, Fong Y. Volumetric analysis predicts hepatic dysfunction in patients undergoing major liver resection. J Gastrointest Surg 2003; 7: 325-330 [PMID: 12654556 DOI: 10.1016/s1091-255x(02)00370-0



- 152 Dinant S, de Graaf W, Verwer BJ, Bennink RJ, van Lienden KP, Gouma DJ, van Vliet AK, van Gulik TM. Risk assessment of posthepatectomy liver failure using hepatobiliary scintigraphy and CT volumetry. J Nucl Med 2007; 48: 685-692 [PMID: 17475954 DOI: 10.2967/jnumed.106.038430]
- 153 Erdogan D, Heijnen BH, Bennink RJ, Kok M, Dinant S, Straatsburg IH, Gouma DJ, van Gulik TM. Preoperative assessment of liver function: a comparison of 99mTc-Mebrofenin scintigraphy with indocyanine green clearance test. Liver Int 2004; 24: 117-123 [PMID: 15078475 DOI: 10.1111/j.1478-3231.2004.00901.x
- 154 Olthof PB, Coelen RJS, Bennink RJ, Heger M, Lam MF, Besselink MG, Busch OR, van Lienden KP, van Gulik TM. 99m Tc-mebrofenin hepatobiliary scintigraphy predicts liver failure following major liver resection for perihilar cholangiocarcinoma. HPB (Oxford) 2017; 19: 850-858 [PMID: 28687148 DOI: 10.1016/j.hpb.2017.05.007]
- de Graaf W, van Lienden KP, Dinant S, Roelofs JJ, Busch OR, Gouma DJ, Bennink RJ, van Gulik 155 TM. Assessment of future remnant liver function using hepatobiliary scintigraphy in patients undergoing major liver resection. J Gastrointest Surg 2010; 14: 369-378 [PMID: 19937195 DOI: 10.1007/s11605-009-1085-2]
- 156 Sato Y, Koyama S, Tsukada K, Hatakeyama K. Acute portal hypertension reflecting shear stress as a trigger of liver regeneration following partial hepatectomy. Surg Today 1997; 27: 518-526 [PMID: 9306545 DOI: 10.1007/bf02385805]
- Song Z, Gupta K, Ng IC, Xing J, Yang YA, Yu H. Mechanosensing in liver regeneration. Semin Cell Dev Biol 2017; 71: 153-167 [PMID: 28768152 DOI: 10.1016/j.semcdb.2017.07.041]
- 158 Sato Y, Tsukada K, Hatakeyama K. Role of shear stress and immune responses in liver regeneration after a partial hepatectomy. Surg Today 1999; 29: 1-9 [PMID: 9934824 DOI: 10.1007/bf02482962]
- 159 Morsiani E, Aleotti A, Ricci D. Haemodynamic and ultrastructural observations on the rat liver after two-thirds partial hepatectomy. J Anat 1998; 192: 507-515 [PMID: 9723978 DOI: 10.1046/j.1469-7580.1998.19240507.x]
- 160 Carnovale CE, Ronco MT. Role of nitric oxide in liver regeneration. Ann Hepatol 2012; 11: 636-647 [PMID: 22947523]
- Fausto N, Campbell JS, Riehle KJ. Liver regeneration. J Hepatol 2012; 57: 692-694 [PMID: 161 22613006 DOI: 10.1016/j.jhep.2012.04.016]
- 162 Schoen JM, Wang HH, Minuk GY, Lautt WW. Shear stress-induced nitric oxide release triggers the liver regeneration cascade. Nitric Oxide 2001; 5: 453-464 [PMID: 11587560 DOI: 10.1006/niox.2001.0373]
- Poisson J, Lemoinne S, Boulanger C, Durand F, Moreau R, Valla D, Rautou PE. Liver sinusoidal 163 endothelial cells: Physiology and role in liver diseases. J Hepatol 2017; 66: 212-227 [PMID: 27423426 DOI: 10.1016/j.jhep.2016.07.009]
- 164 Kawai M, Naruse K, Komatsu S, Kobayashi S, Nagino M, Nimura Y, Sokabe M. Mechanical stressdependent secretion of interleukin 6 by endothelial cells after portal vein embolization: clinical and experimental studies. J Hepatol 2002; 37: 240-246 [PMID: 12127429 DOI: 10.1016/s0168-8278(02)00171-x
- 165 Nishii K, Brodin E, Renshaw T, Weesner R, Moran E, Soker S, Sparks JL. Shear stress upregulates regeneration-related immediate early genes in liver progenitors in 3D ECM-like microenvironments. J Cell Physiol 2018; 233: 4272-4281 [PMID: 29052842 DOI: 10.1002/jcp.26246]
- 166 Abu Rmilah A, Zhou W, Nelson E, Lin L, Amiot B, Nyberg SL. Understanding the marvels behind liver regeneration. Wiley Interdiscip Rev Dev Biol 2019; 8: e340 [PMID: 30924280 DOI: 10.1002/wdev.340]
- Arab JP, Martin-Mateos RM, Shah VH. Gut-liver axis, cirrhosis and portal hypertension: the 167 chicken and the egg. Hepatol Int 2018; 12: 24-33 [PMID: 28550391 DOI: 10.1007/s12072-017-9798-x]
- 168 Ozaki M. Cellular and molecular mechanisms of liver regeneration: Proliferation, growth, death and protection of hepatocytes. Semin Cell Dev Biol 2020; 100: 62-73 [PMID: 31669133 DOI: 10.1016/j.semcdb.2019.10.007
- Campbell JS, Riehle KJ, Brooling JT, Bauer RL, Mitchell C, Fausto N. Proinflammatory cytokine 169 production in liver regeneration is Myd88-dependent, but independent of Cd14, Tlr2, and Tlr4. J Immunol 2006; 176: 2522-2528 [PMID: 16456013 DOI: 10.4049/jimmunol.176.4.2522]
- 170 Yamamoto C, Yagi S, Hori T, Iida T, Taniguchi K, Isaji S, Uemoto S. Significance of portal venous VEGF during liver regeneration after hepatectomy. J Surg Res 2010; 159: e37-e43 [PMID: 19394640 DOI: 10.1016/j.jss.2008.11.007]
- Eipel C, Abshagen K, Vollmar B. Regulation of hepatic blood flow: the hepatic arterial buffer 171 response revisited. World J Gastroenterol 2010; 16: 6046-6057 [PMID: 21182219 DOI: 10.3748/wjg.v16.i48.6046]
- 172 Schödel J, Oikonomopoulos S, Ragoussis J, Pugh CW, Ratcliffe PJ, Mole DR. High-resolution genome-wide mapping of HIF-binding sites by ChIP-seq. Blood 2011; 117: e207-e217 [PMID: 21447827 DOI: 10.1182/blood-2010-10-314427]
- Maeno H, Ono T, Dhar DK, Sato T, Yamanoi A, Nagasue N. Expression of hypoxia inducible 173 factor-1alpha during liver regeneration induced by partial hepatectomy in rats. Liver Int 2005; 25: 1002-1009 [PMID: 16162160 DOI: 10.1111/j.1478-3231.2005.01144.x]
- 174 Dirscherl K, Schläpfer M, Roth Z'graggen B, Wenger RH, Booy C, Flury-Frei R, Fatzer R, Aloman C, Bartosch B, Parent R, Kurtcuoglu V, de Zélicourt D, Spahn DR, Beck Schimmer B, Schadde E. Hypoxia sensing by hepatic stellate cells leads to VEGF-dependent angiogenesis and may contribute



to accelerated liver regeneration. Sci Rep 2020; 10: 4392 [PMID: 32152325 DOI: 10.1038/s41598-020-60709-9]

- 175 Ding BS, Nolan DJ, Butler JM, James D, Babazadeh AO, Rosenwaks Z, Mittal V, Kobayashi H, Shido K, Lyden D, Sato TN, Rabbany SY, Rafii S. Inductive angiocrine signals from sinusoidal endothelium are required for liver regeneration. Nature 2010; 468: 310-315 [PMID: 21068842 DOI: 10.1038/nature09493]
- Lee SC, Jeong HJ, Lee SK, Kim SJ. Hypoxic Conditioned Medium From Human Adipose-Derived 176 Stem Cells Promotes Mouse Liver Regeneration Through JAK/STAT3 Signaling. Stem Cells Transl Med 2016; 5: 816-825 [PMID: 27102647 DOI: 10.5966/sctm.2015-0191]
- 177 Starlinger P, Luyendyk JP, Groeneveld DJ. Hemostasis and Liver Regeneration. Semin Thromb Hemost 2020; 46: 735-742 [PMID: 32906177 DOI: 10.1055/s-0040-1715450]
- Matsuo R, Nakano Y, Ohkohchi N. Platelet administration via the portal vein promotes liver 178 regeneration in rats after 70% hepatectomy. Ann Surg 2011; 253: 759-763 [PMID: 21475016 DOI: 10.1097/SLA.0b013e318211caf8
- Han S, Park HW, Song JH, Gwak MS, Lee WJ, Kim G, Lee SK, Ko JS. Association Between 179 Intraoperative Platelet Transfusion and Early Graft Regeneration in Living Donor Liver Transplantation. Ann Surg 2016; 264: 1065-1072 [PMID: 26720430 DOI: 10.1097/sla.0000000000001526]
- 180 Lisman T, Porte RJ. Mechanisms of platelet-mediated liver regeneration. Blood 2016; 128: 625-629 [PMID: 27297793 DOI: 10.1182/blood-2016-04-692665]
- 181 Kirschbaum M, Karimian G, Adelmeijer J, Giepmans BN, Porte RJ, Lisman T. Horizontal RNA transfer mediates platelet-induced hepatocyte proliferation. Blood 2015; 126: 798-806 [PMID: 26056167 DOI: 10.1182/blood-2014-09-600312]
- 182 Zhang Q, Raoof M, Chen Y, Sumi Y, Sursal T, Junger W, Brohi K, Itagaki K, Hauser CJ. Circulating mitochondrial DAMPs cause inflammatory responses to injury. Nature 2010; 464: 104-107 [PMID: 20203610 DOI: 10.1038/nature08780]
- 183 Satyam A, Graef ER, Lapchak PH, Tsokos MG, Dalle Lucca JJ, Tsokos GC. Complement and coagulation cascades in trauma. Acute Med Surg 2019; 6: 329-335 [PMID: 31592318 DOI: 10.1002/ams2.426
- Fausto N, Campbell JS, Riehle KJ. Liver regeneration. Hepatology 2006; 43: S45-S53 [PMID: 184 16447274 DOI: 10.1002/hep.20969]
- 185 Groeneveld D, Pereyra D, Veldhuis Z, Adelmeijer J, Ottens P, Kopec AK, Starlinger P, Lisman T, Luyendyk JP. Intrahepatic fibrin(ogen) deposition drives liver regeneration after partial hepatectomy in mice and humans. Blood 2019; 133: 1245-1256 [PMID: 30655274 DOI: 10.1182/blood-2018-08-869057
- Mars WM, Liu ML, Kitson RP, Goldfarb RH, Gabauer MK, Michalopoulos GK. Immediate early 186 detection of urokinase receptor after partial hepatectomy and its implications for initiation of liver regeneration. Hepatology 1995; 21: 1695-1701 [PMID: 7768515]
- 187 Kim TH, Mars WM, Stolz DB, Petersen BE, Michalopoulos GK. Extracellular matrix remodeling at the early stages of liver regeneration in the rat. Hepatology 1997; 26: 896-904 [PMID: 9328311 DOI: 10.1002/hep.510260415]
- Kim TH, Mars WM, Stolz DB, Michalopoulos GK. Expression and activation of pro-MMP-2 and 188 pro-MMP-9 during rat liver regeneration. Hepatology 2000; 31: 75-82 [PMID: 10613731 DOI: 10.1002/hep.510310114]
- Mars WM, Kim TH, Stolz DB, Liu ML, Michalopoulos GK. Presence of urokinase in serum-free 189 primary rat hepatocyte cultures and its role in activating hepatocyte growth factor. Cancer Res 1996; 56: 2837-2843 [PMID: 8665523]
- 190 Lindroos PM, Zarnegar R, Michalopoulos GK. Hepatocyte growth factor (hepatopoietin A) rapidly increases in plasma before DNA synthesis and liver regeneration stimulated by partial hepatectomy and carbon tetrachloride administration. Hepatology 1991; 13: 743-750 [PMID: 1826282]
- Michalopoulos GK. Principles of liver regeneration and growth homeostasis. Compr Physiol 2013; 191 3: 485-513 [PMID: 23720294 DOI: 10.1002/cphy.c120014]
- Roos F, Ryan AM, Chamow SM, Bennett GL, Schwall RH. Induction of liver growth in normal 192 mice by infusion of hepatocyte growth factor/scatter factor. Am J Physiol 1995; 268: G380-G386 [PMID: 7864135 DOI: 10.1152/ajpgi.1995.268.2.G380]
- 193 Liu ML, Mars WM, Zarnegar R, Michalopoulos GK. Collagenase pretreatment and the mitogenic effects of hepatocyte growth factor and transforming growth factor-alpha in adult rat liver. Hepatology 1994; 19: 1521-1527 [PMID: 8188184]
- Michalopoulos GK. Hepatostat: Liver regeneration and normal liver tissue maintenance. 194 Hepatology 2017; 65: 1384-1392 [PMID: 27997988 DOI: 10.1002/hep.28988]
- 195 Gilgenkrantz H, Collin de l'Hortet A. Understanding Liver Regeneration: From Mechanisms to Regenerative Medicine. Am J Pathol 2018; 188: 1316-1327 [PMID: 29673755 DOI: 10.1016/j.aipath.2018.03.008
- 196 Webber EM, Bruix J, Pierce RH, Fausto N. Tumor necrosis factor primes hepatocytes for DNA replication in the rat. Hepatology 1998; 28: 1226-1234 [PMID: 9794905 DOI: 10.1002/hep.510280509]
- 197 Trautwein C, Rakemann T, Niehof M, Rose-John S, Manns MP. Acute-phase response factor, increased binding, and target gene transcription during liver regeneration. Gastroenterology 1996; 110: 1854-1862 [PMID: 8964411 DOI: 10.1053/gast.1996.v110.pm8964411]



- 198 Iwai M, Cui TX, Kitamura H, Saito M, Shimazu T. Increased secretion of tumour necrosis factor and interleukin 6 from isolated, perfused liver of rats after partial hepatectomy. Cytokine 2001; 13: 60-64 [PMID: 11145844 DOI: 10.1006/cyto.2000.0797]
- 199 FitzGerald MJ, Webber EM, Donovan JR, Fausto N. Rapid DNA binding by nuclear factor kappa B in hepatocytes at the start of liver regeneration. Cell Growth Differ 1995; 6: 417-427 [PMID: 77948091
- 200 Cressman DE, Diamond RH, Taub R. Rapid activation of the Stat3 transcription complex in liver regeneration. Hepatology 1995; 21: 1443-1449 [PMID: 7737651]
- 201 Akerman P, Cote P, Yang SQ, McClain C, Nelson S, Bagby GJ, Diehl AM. Antibodies to tumor necrosis factor-alpha inhibit liver regeneration after partial hepatectomy. Am J Physiol 1992; 263: G579-G585 [PMID: 1415718 DOI: 10.1152/ajpgi.1992.263.4.G579]
- 202 Hayashi H, Nagaki M, Imose M, Osawa Y, Kimura K, Takai S, Imao M, Naiki T, Kato T, Moriwaki H. Normal liver regeneration and liver cell apoptosis after partial hepatectomy in tumor necrosis factor-alpha-deficient mice. Liver Int 2005; 25: 162-170 [PMID: 15698414 DOI: 10.1111/j.1478-3231.2005.01029.x]
- Ren X, Hogaboam C, Carpenter A, Colletti L. Stem cell factor restores hepatocyte proliferation in 203 IL-6 knockout mice following 70% hepatectomy. J Clin Invest 2003; 112: 1407-1418 [PMID: 14597766 DOI: 10.1172/jci17391]
- 204 Cornell RP. Gut-derived endotoxin elicits hepatotrophic factor secretion for liver regeneration. Am J Physiol 1985; 249: R551-R562 [PMID: 2865902 DOI: 10.1152/ajpregu.1985.249.5.R551]
- Cornell RP, Liljequist BL, Bartizal KF. Depressed liver regeneration after partial hepatectomy of 205 germ-free, athymic and lipopolysaccharide-resistant mice. Hepatology 1990; 11: 916-922 [PMID: 2194922 DOI: 10.1002/hep.1840110603]
- Selzner N, Selzner M, Odermatt B, Tian Y, Van Rooijen N, Clavien PA. ICAM-1 triggers liver 206 regeneration through leukocyte recruitment and Kupffer cell-dependent release of TNF-alpha/IL-6 in mice. Gastroenterology 2003; 124: 692-700 [PMID: 12612908 DOI: 10.1053/gast.2003.50098]
- Anders RA, Subudhi SK, Wang J, Pfeffer K, Fu YX. Contribution of the lymphotoxin beta receptor 207 to liver regeneration. J Immunol 2005; 175: 1295-1300 [PMID: 16002734 DOI: 10.4049/iimmunol.175.2.1295
- 208 Knight B, Yeoh GC. TNF/LTalpha double knockout mice display abnormal inflammatory and regenerative responses to acute and chronic liver injury. Cell Tissue Res 2005; 319: 61-70 [PMID: 15592751 DOI: 10.1007/s00441-004-1003-6]
- 209 Michalopoulos GK, Bhushan B. Liver regeneration: biological and pathological mechanisms and implications. Nat Rev Gastroenterol Hepatol 2021; 18: 40-55 [PMID: 32764740 DOI: 10.1038/s41575-020-0342-4]
- Nakamura T, Nishizawa T, Hagiya M, Seki T, Shimonishi M, Sugimura A, Tashiro K, Shimizu S. 210 Molecular cloning and expression of human hepatocyte growth factor. Nature 1989; 342: 440-443 [PMID: 2531289 DOI: 10.1038/342440a0]
- Wang X, DeFrances MC, Dai Y, Pediaditakis P, Johnson C, Bell A, Michalopoulos GK, Zarnegar R. 211 A mechanism of cell survival: sequestration of Fas by the HGF receptor Met. Mol Cell 2002; 9: 411-421 [PMID: 11864613 DOI: 10.1016/s1097-2765(02)00439-2]
- 212 Fafalios A, Ma J, Tan X, Stoops J, Luo J, Defrances MC, Zarnegar R. A hepatocyte growth factor receptor (Met)-insulin receptor hybrid governs hepatic glucose metabolism. Nat Med 2011; 17: 1577-1584 [PMID: 22081023 DOI: 10.1038/nm.2531]
- Schmidt C, Bladt F, Goedecke S, Brinkmann V, Zschiesche W, Sharpe M, Gherardi E, Birchmeier 213 C. Scatter factor/hepatocyte growth factor is essential for liver development. Nature 1995; 373: 699-702 [PMID: 7854452 DOI: 10.1038/373699a0]
- Wang B, Gao C, Ponder KP. C/EBPbeta contributes to hepatocyte growth factor-induced replication 214 of rodent hepatocytes. J Hepatol 2005; 43: 294-302 [PMID: 15922473 DOI: 10.1016/j.jhep.2005.02.029]
- 215 Paranjpe S, Bowen WC, Bell AW, Nejak-Bowen K, Luo JH, Michalopoulos GK. Cell cycle effects resulting from inhibition of hepatocyte growth factor and its receptor c-Met in regenerating rat livers by RNA interference. Hepatology 2007; 45: 1471-1477 [PMID: 17427161 DOI: 10.1002/hep.21570]
- 216 Schuppan D, Schmid M, Somasundaram R, Ackermann R, Ruehl M, Nakamura T, Riecken EO. Collagens in the liver extracellular matrix bind hepatocyte growth factor. Gastroenterology 1998; 114: 139-152 [PMID: 9428228 DOI: 10.1016/s0016-5085(98)70642-0]
- 217 Pediaditakis P, Lopez-Talavera JC, Petersen B, Monga SP, Michalopoulos GK. The processing and utilization of hepatocyte growth factor/scatter factor following partial hepatectomy in the rat. Hepatology 2001; 34: 688-693 [PMID: 11584364 DOI: 10.1053/jhep.2001.27811]
- Zarnegar R, DeFrances MC, Kost DP, Lindroos P, Michalopoulos GK. Expression of hepatocyte 218 growth factor mRNA in regenerating rat liver after partial hepatectomy. Biochem Biophys Res Commun 1991; 177: 559-565 [PMID: 1828343 DOI: 10.1016/0006-291x(91)92020-k]
- 219 Nakamura T, Teramoto H, Ichihara A. Purification and characterization of a growth factor from rat platelets for mature parenchymal hepatocytes in primary cultures. Proc Natl Acad Sci USA 1986; 83: 6489-6493 [PMID: 3529086 DOI: 10.1073/pnas.83.17.6489]
- Yanagita K, Nagaike M, Ishibashi H, Niho Y, Matsumoto K, Nakamura T. Lung may have an 220 endocrine function producing hepatocyte growth factor in response to injury of distal organs. Biochem Biophys Res Commun 1992; 182: 802-809 [PMID: 1531175 DOI:



10.1016/0006-291x(92)91803-x

- 221 Kono S, Nagaike M, Matsumoto K, Nakamura T. Marked induction of hepatocyte growth factor mRNA in intact kidney and spleen in response to injury of distant organs. Biochem Biophys Res Commun 1992; 186: 991-998 [PMID: 1379811 DOI: 10.1016/0006-291x(92)90844-b]
- 222 Zhang XJ, Olsavszky V, Yin Y, Wang B, Engleitner T, Öllinger R, Schledzewski K, Koch PS, Rad R, Schmid RM, Friess H, Goerdt S, Hüser N, Géraud C, von Figura G, Hartmann D. Angiocrine Hepatocyte Growth Factor Signaling Controls Physiological Organ and Body Size and Dynamic Hepatocyte Proliferation to Prevent Liver Damage during Regeneration. Am J Pathol 2020; 190: 358-371 [PMID: 31783007 DOI: 10.1016/j.ajpath.2019.10.009]
- 223 Broten J, Michalopoulos G, Petersen B, Cruise J. Adrenergic stimulation of hepatocyte growth factor expression. Biochem Biophys Res Commun 1999; 262: 76-79 [PMID: 10448071 DOI: 10.1006/bbrc.1999.1183]
- 224 Skrtic S, Wallenius V, Ekberg S, Brenzel A, Gressner AM, Jansson JO. Insulin-like growth factors stimulate expression of hepatocyte growth factor but not transforming growth factor beta1 in cultured hepatic stellate cells. Endocrinology 1997; 138: 4683-4689 [PMID: 9348194 DOI: 10.1210/endo.138.11.5540]
- 225 Burgess AW. EGFR family: structure physiology signalling and therapeutic targets. Growth Factors 2008; 26: 263-274 [PMID: 18800267 DOI: 10.1080/08977190802312844]
- 226 McGowan JA, Strain AJ, Bucher NL. DNA synthesis in primary cultures of adult rat hepatocytes in a defined medium: effects of epidermal growth factor, insulin, glucagon, and cyclic-AMP. J Cell Physiol 1981; 108: 353-363 [PMID: 6270165 DOI: 10.1002/jcp.1041080309]
- 227 Bucher NL, Patel U, Cohen S. Hormonal factors concerned with liver regeneration. Ciba Found Symp 1977; 95-107 [PMID: 306914 DOI: 10.1002/9780470720363.ch5]
- 228 St Hilaire RJ, Jones AL. Epidermal growth factor: its biologic and metabolic effects with emphasis on the hepatocyte. *Hepatology* 1982; 2: 601-613 [PMID: 6981576 DOI: 10.1002/hep.1840020515]
- 229 Olsen PS, Poulsen SS, Kirkegaard P. Adrenergic effects on secretion of epidermal growth factor from Brunner's glands. Gut 1985; 26: 920-927 [PMID: 2863199 DOI: 10.1136/gut.26.9.920]
- Jones DE Jr, Tran-Patterson R, Cui DM, Davin D, Estell KP, Miller DM. Epidermal growth factor 230 secreted from the salivary gland is necessary for liver regeneration. Am J Physiol 1995; 268: G872-G878 [PMID: 7762671 DOI: 10.1152/ajpgi.1995.268.5.G872]
- 231 Webber EM, FitzGerald MJ, Brown PI, Bartlett MH, Fausto N. Transforming growth factor-alpha expression during liver regeneration after partial hepatectomy and toxic injury, and potential interactions between transforming growth factor-alpha and hepatocyte growth factor. Hepatology 1993; 18: 1422-1431 [PMID: 8244268]
- Mead JE, Fausto N. Transforming growth factor alpha may be a physiological regulator of liver 232 regeneration by means of an autocrine mechanism. Proc Natl Acad Sci USA 1989; 86: 1558-1562 [PMID: 2922399 DOI: 10.1073/pnas.86.5.1558]
- Russell WE, Kaufmann WK, Sitaric S, Luetteke NC, Lee DC. Liver regeneration and 233 hepatocarcinogenesis in transforming growth factor-alpha-targeted mice. Mol Carcinog 1996; 15: 183-189 [PMID: 8597531 DOI: 10.1002/(sici)1098-2744(199603)15:3<183::Aid-mc4>3.0.Co;2-j]
- 234 Berasain C, García-Trevijano ER, Castillo J, Erroba E, Lee DC, Prieto J, Avila MA, Amphiregulin: an early trigger of liver regeneration in mice. Gastroenterology 2005; 128: 424-432 [PMID: 15685553 DOI: 10.1053/j.gastro.2004.11.006]
- 235 Michalopoulos GK, Khan Z. Liver regeneration, growth factors, and amphiregulin. Gastroenterology 2005; 128: 503-506 [PMID: 15685562 DOI: 10.1053/j.gastro.2004.12.039]
- 236 Kiso S, Kawata S, Tamura S, Higashiyama S, Ito N, Tsushima H, Taniguchi N, Matsuzawa Y. Role of heparin-binding epidermal growth factor-like growth factor as a hepatotrophic factor in rat liver regeneration after partial hepatectomy. Hepatology 1995; 22: 1584-1590 [PMID: 7590679]
- 237 Kiso S, Kawata S, Tamura S, Inui Y, Yoshida Y, Sawai Y, Umeki S, Ito N, Yamada A, Miyagawa J, Higashiyama S, Iwawaki T, Saito M, Taniguchi N, Matsuzawa Y, Kohno K. Liver regeneration in heparin-binding EGF-like growth factor transgenic mice after partial hepatectomy. Gastroenterology 2003; 124: 701-707 [PMID: 12612909 DOI: 10.1053/gast.2003.50097]
- 238 Stolz DB, Mars WM, Petersen BE, Kim TH, Michalopoulos GK. Growth factor signal transduction immediately after two-thirds partial hepatectomy in the rat. Cancer Res 1999; 59: 3954-3960 [PMID: 10463591]
- 239 Paranjpe S, Bowen WC, Tseng GC, Luo JH, Orr A, Michalopoulos GK. RNA interference against hepatic epidermal growth factor receptor has suppressive effects on liver regeneration in rats. Am J Pathol 2010; 176: 2669-2681 [PMID: 20395437 DOI: 10.2353/ajpath.2010.090605]
- 240 Russell DW. The enzymes, regulation, and genetics of bile acid synthesis. Annu Rev Biochem 2003; 72: 137-174 [PMID: 12543708 DOI: 10.1146/annurev.biochem.72.121801.161712]
- Schaap FG, Trauner M, Jansen PL. Bile acid receptors as targets for drug development. Nat Rev 241 Gastroenterol Hepatol 2014; 11: 55-67 [PMID: 23982684 DOI: 10.1038/nrgastro.2013.151]
- Keitel V, Kubitz R, Häussinger D. Endocrine and paracrine role of bile acids. World J Gastroenterol 242 2008; 14: 5620-5629 [PMID: 18837077 DOI: 10.3748/wjg.14.5620]
- 243 Evans RM, Mangelsdorf DJ. Nuclear Receptors, RXR, and the Big Bang. Cell 2014; 157: 255-266 [PMID: 24679540 DOI: 10.1016/j.cell.2014.03.012]
- 244 Meng Z, Liu N, Fu X, Wang X, Wang YD, Chen WD, Zhang L, Forman BM, Huang W. Insufficient bile acid signaling impairs liver repair in CYP27(-/-) mice. J Hepatol 2011; 55: 885-895 [PMID: 21334403 DOI: 10.1016/j.jhep.2010.12.037]



- 245 Naugler WE. Bile acid flux is necessary for normal liver regeneration. PLoS One 2014; 9: e97426 [PMID: 24841254 DOI: 10.1371/journal.pone.0097426]
- 246 Doignon I, Julien B, Serrière-Lanneau V, Garcin I, Alonso G, Nicou A, Monnet F, Gigou M, Humbert L, Rainteau D, Azoulay D, Castaing D, Gillon MC, Samuel D, Duclos-Vallée JC, Tordjmann T. Immediate neuroendocrine signaling after partial hepatectomy through acute portal hyperpressure and cholestasis. J Hepatol 2011; 54: 481-488 [PMID: 21163545 DOI: 10.1016/j.jhep.2010.07.012]
- 247 Hoekstra LT, Rietkerk M, van Lienden KP, van den Esschert JW, Schaap FG, van Gulik TM. Bile salts predict liver regeneration in rabbit model of portal vein embolization. J Surg Res 2012; 178: 773-778 [PMID: 22763217 DOI: 10.1016/j.jss.2012.06.038]
- 248 Chen WD, Wang YD, Zhang L, Shiah S, Wang M, Yang F, Yu D, Forman BM, Huang W. Farnesoid X receptor alleviates age-related proliferation defects in regenerating mouse livers by activating forkhead box m1b transcription. Hepatology 2010; 51: 953-962 [PMID: 19998409 DOI: 10.1002/hep.23390]
- 249 Baier P, Wolf-Vorbeck G, Hempel S, Hopt UT, von Dobschuetz E. Effect of liver regeneration after partial hepatectomy and ischemia-reperfusion on expression of growth factor receptors. World J Gastroenterol 2006; 12: 3835-3840 [PMID: 16804967 DOI: 10.3748/wjg.v12.i24.3835]
- 250 Jourdainne V, Péan N, Doignon I, Humbert L, Rainteau D, Tordjmann T. The Bile Acid Receptor TGR5 and Liver Regeneration. Dig Dis 2015; 33: 319-326 [PMID: 26045264 DOI: 10.1159/000371668
- 251 Cruise JL, Knechtle SJ, Bollinger RR, Kuhn C, Michalopoulos G. Alpha 1-adrenergic effects and liver regeneration. Hepatology 1987; 7: 1189-1194 [PMID: 2824312 DOI: 10.1002/hep.1840070604]
- 252 Cruise JL, Houck KA, Michalopoulos GK. Induction of DNA synthesis in cultured rat hepatocytes through stimulation of alpha 1 adrenoreceptor by norepinephrine. Science 1985; 227: 749-751 [PMID: 2982212 DOI: 10.1126/science.2982212]
- 253 Han C, Bowen WC, Michalopoulos GK, Wu T. Alpha-1 adrenergic receptor transactivates signal transducer and activator of transcription-3 (Stat3) through activation of Src and epidermal growth factor receptor (EGFR) in hepatocytes. J Cell Physiol 2008; 216: 486-497 [PMID: 18314882 DOI: 10.1002/jcp.21420
- 254 Houck KA, Cruise JL, Michalopoulos G. Norepinephrine modulates the growth-inhibitory effect of transforming growth factor-beta in primary rat hepatocyte cultures. J Cell Physiol 1988; 135: 551-555 [PMID: 3165094 DOI: 10.1002/jcp.1041350327]
- 255 Soeda J, Mouralidarane A, Ray S, Novelli M, Thomas S, Roskams T, Diehl AM, Oben JA. The β adrenoceptor agonist isoproterenol rescues acetaminophen-injured livers through increasing progenitor numbers by Wnt in mice. Hepatology 2014; 60: 1023-1034 [PMID: 24923719 DOI: 10.1002/hep.27266
- Lesurtel M, Graf R, Aleil B, Walther DJ, Tian Y, Jochum W, Gachet C, Bader M, Clavien PA. 256 Platelet-derived serotonin mediates liver regeneration. Science 2006; 312: 104-107 [PMID: 16601191 DOI: 10.1126/science.1123842]
- 257 Furrer K, Rickenbacher A, Tian Y, Jochum W, Bittermann AG, Käch A, Humar B, Graf R, Moritz W, Clavien PA. Serotonin reverts age-related capillarization and failure of regeneration in the liver through a VEGF-dependent pathway. Proc Natl Acad Sci USA 2011; 108: 2945-2950 [PMID: 21282654 DOI: 10.1073/pnas.1012531108]
- Fang Y, Liu C, Shu B, Zhai M, Deng C, He C, Luo M, Han T, Zheng W, Zhang J, Liu S. Axis of 258 serotonin -pERK-YAP in liver regeneration. Life Sci 2018; 209: 490-497 [PMID: 30142376 DOI: 10.1016/j.lfs.2018.08.047]
- Matondo RB, Punt C, Homberg J, Toussaint MJ, Kisjes R, Korporaal SJ, Akkerman JW, Cuppen E, 259 de Bruin A. Deletion of the serotonin transporter in rats disturbs serotonin homeostasis without impairing liver regeneration. Am J Physiol Gastrointest Liver Physiol 2009; 296: G963-G968 [PMID: 19246633 DOI: 10.1152/ajpgi.90709.2008]
- Michalopoulos G, Pitot HC. Primary culture of parenchymal liver cells on collagen membranes. 260 Morphological and biochemical observations. Exp Cell Res 1975; 94: 70-78 [PMID: 243 DOI: 10.1016/0014-4827(75)90532-7
- Block GD, Locker J, Bowen WC, Petersen BE, Katyal S, Strom SC, Riley T, Howard TA, 261 Michalopoulos GK. Population expansion, clonal growth, and specific differentiation patterns in primary cultures of hepatocytes induced by HGF/SF, EGF and TGF alpha in a chemically defined (HGM) medium. J Cell Biol 1996; 132: 1133-1149 [PMID: 8601590 DOI: 10.1083/jcb.132.6.1133]
- Lu M, Flanagan JU, Langley RJ, Hay MP, Perry JK. Targeting growth hormone function: strategies 262 and therapeutic applications. Signal Transduct Target Ther 2019; 4: 3 [PMID: 30775002 DOI: 10.1038/s41392-019-0036-y]
- McElduff A, Poronnik P, Baxter RC, Williams P. A comparison of the insulin and insulin-like 263 growth factor I receptors from rat brain and liver. Endocrinology 1988; 122: 1933-1939 [PMID: 2966060 DOI: 10.1210/endo-122-5-1933]
- Zindy F, Lamas E, Schmidt S, Kirn A, Brechot C. Expression of insulin-like growth factor II (IGF-264 II) and IGF-II, IGF-I and insulin receptors mRNAs in isolated non-parenchymal rat liver cells. J Hepatol 1992; 14: 30-34 [PMID: 1310705 DOI: 10.1016/0168-8278(92)90127-b]
- Thompson BJ, Shang CA, Waters MJ. Identification of genes induced by growth hormone in rat 265 liver using cDNA arrays. Endocrinology 2000; 141: 4321-4324 [PMID: 11089569 DOI:



10.1210/endo.141.11.7874]

- 266 Yamauchi T, Ueki K, Tobe K, Tamemoto H, Sekine N, Wada M, Honjo M, Takahashi M, Takahashi T, Hirai H, Tushima T, Akanuma Y, Fujita T, Komuro I, Yazaki Y, Kadowaki T. Tyrosine phosphorylation of the EGF receptor by the kinase Jak2 is induced by growth hormone. Nature 1997; 390: 91-96 [PMID: 9363897 DOI: 10.1038/36369]
- Pennisi PA, Kopchick JJ, Thorgeirsson S, LeRoith D, Yakar S. Role of growth hormone (GH) in 267 liver regeneration. Endocrinology 2004; 145: 4748-4755 [PMID: 15242989 DOI: 10.1210/en.2004-0655]
- 268 Tanase DM, Gosav EM, Neculae E, Costea CF, Ciocoiu M, Hurjui LL, Tarniceriu CC, Floria M. Hypothyroidism-Induced Nonalcoholic Fatty Liver Disease (HIN): Mechanisms and Emerging Therapeutic Options. Int J Mol Sci 2020; 21 [PMID: 32824723 DOI: 10.3390/ijms21165927]
- López-Fontal R, Zeini M, Través PG, Gómez-Ferrería M, Aranda A, Sáez GT, Cerdá C, Martín-269 Sanz P, Hortelano S, Boscá L. Mice lacking thyroid hormone receptor Beta show enhanced apoptosis and delayed liver commitment for proliferation after partial hepatectomy. PLoS One 2010; 5: e8710 [PMID: 20090848 DOI: 10.1371/journal.pone.0008710]
- 270 Bockhorn M, Frilling A, Benko T, Best J, Sheu SY, Trippler M, Schlaak JF, Broelsch CE. Triiodothyronine as a stimulator of liver regeneration after partial and subtotal hepatectomy. Eur Surg Res 2007; 39: 58-63 [PMID: 17213727 DOI: 10.1159/000098443]
- Pibiri M, Ledda-Columbano GM, Cossu C, Simbula G, Menegazzi M, Shinozuka H, Columbano A. 271 Cyclin D1 is an early target in hepatocyte proliferation induced by thyroid hormone (T3). FASEB J 2001; 15: 1006-1013 [PMID: 11292661 DOI: 10.1096/fj.00-0416com]
- 272 Alisi A, Demori I, Spagnuolo S, Pierantozzi E, Fugassa E, Leoni S. Thyroid status affects rat liver regeneration after partial hepatectomy by regulating cell cycle and apoptosis. Cell Physiol Biochem 2005; 15: 69-76 [PMID: 15665517 DOI: 10.1159/000083639]
- 273 Alvarado TF, Puliga E, Preziosi M, Poddar M, Singh S, Columbano A, Nejak-Bowen K, Monga SP. Thyroid Hormone Receptor β Agonist Induces β -Catenin-Dependent Hepatocyte Proliferation in Mice: Implications in Hepatic Regeneration. Gene Expr 2016; 17: 19-34 [PMID: 27226410 DOI: 10.3727/105221616X6916311
- 274 Shaik F, Cuthbert GA, Homer-Vanniasinkam S, Muench SP, Ponnambalam S, Harrison MA. Structural Basis for Vascular Endothelial Growth Factor Receptor Activation and Implications for Disease Therapy. Biomolecules 2020; 10 [PMID: 33333800 DOI: 10.3390/biom10121673]
- 275 Jia C. Advances in the regulation of liver regeneration. Expert Rev Gastroenterol Hepatol 2011; 5: 105-121 [PMID: 21309676 DOI: 10.1586/egh.10.87]
- Bockhorn M, Goralski M, Prokofiev D, Dammann P, Grünewald P, Trippler M, Biglarnia A, 276 Kamler M, Niehues EM, Frilling A, Broelsch CE, Schlaak JF. VEGF is important for early liver regeneration after partial hepatectomy. J Surg Res 2007; 138: 291-299 [PMID: 17275844 DOI: 10.1016/j.jss.2006.07.027]
- Nusse R, Brown A, Papkoff J, Scambler P, Shackleford G, McMahon A, Moon R, Varmus H. A 277 new nomenclature for int-1 and related genes: the Wnt gene family. Cell 1991; 64: 231 [PMID: 1846319 DOI: 10.1016/0092-8674(91)90633-a]
- MacDonald BT, Tamai K, He X. Wnt/beta-catenin signaling: components, mechanisms, and 278 diseases. Dev Cell 2009; 17: 9-26 [PMID: 19619488 DOI: 10.1016/j.devcel.2009.06.016]
- 279 Monga SP. β-Catenin Signaling and Roles in Liver Homeostasis, Injury, and Tumorigenesis. Gastroenterology 2015; 148: 1294-1310 [PMID: 25747274 DOI: 10.1053/j.gastro.2015.02.056]
- 280 Perugorria MJ, Olaizola P, Labiano I, Esparza-Baquer A, Marzioni M, Marin JJG, Bujanda L, Banales JM. Wnt-β-catenin signalling in liver development, health and disease. Nat Rev Gastroenterol Hepatol 2019; 16: 121-136 [PMID: 30451972 DOI: 10.1038/s41575-018-0075-9]
- 281 Hu S, Monga SP. Wnt/-Catenin Signaling and Liver Regeneration: Circuit, Biology, and Opportunities. Gene Expr 2021; 20: 189-199 [PMID: 33472727 DOI: 10.3727/105221621X16111780348794
- 282 Monga SP, Pediaditakis P, Mule K, Stolz DB, Michalopoulos GK. Changes in WNT/beta-catenin pathway during regulated growth in rat liver regeneration. Hepatology 2001; 33: 1098-1109 [PMID: 1343237 DOI: 10.1053/jhep.2001.23786]
- 283 Chen Y, Sasai N, Ma G, Yue T, Jia J, Briscoe J, Jiang J. Sonic Hedgehog dependent phosphorylation by CK1a and GRK2 is required for ciliary accumulation and activation of smoothened. PLoS Biol 2011; 9: e1001083 [PMID: 21695114 DOI: 10.1371/journal.pbio.1001083]
- 284 Roy S. Cilia and Hedgehog: when and how was their marriage solemnized? Differentiation 2012; 83: S43-S48 [PMID: 22154138 DOI: 10.1016/j.diff.2011.11.010]
- 285 Ramsbottom SA, Pownall ME. Regulation of Hedgehog Signalling Inside and Outside the Cell. J Dev Biol 2016; 4: 23 [PMID: 27547735 DOI: 10.3390/jdb4030023]
- 286 Merchant JL, Saqui-Salces M. Inhibition of Hedgehog signaling in the gastrointestinal tract: targeting the cancer microenvironment. Cancer Treat Rev 2014; 40: 12-21 [PMID: 24007940 DOI: 10.1016/j.ctrv.2013.08.003]
- Ochoa B, Syn WK, Delgado I, Karaca GF, Jung Y, Wang J, Zubiaga AM, Fresnedo O, Omenetti A, 287 Zdanowicz M, Choi SS, Diehl AM. Hedgehog signaling is critical for normal liver regeneration after partial hepatectomy in mice. Hepatology 2010; 51: 1712-1723 [PMID: 20432255 DOI: 10.1002/hep.23525]
- 288 Swiderska-Syn M, Syn WK, Xie G, Krüger L, Machado MV, Karaca G, Michelotti GA, Choi SS, Premont RT, Diehl AM. Myofibroblastic cells function as progenitors to regenerate murine livers



after partial hepatectomy. Gut 2014; 63: 1333-1344 [PMID: 24173292 DOI: 10.1136/gutjnl-2013-305962]

- Langiewicz M, Schlegel A, Saponara E, Linecker M, Borger P, Graf R, Humar B, Clavien PA. 289 Hedgehog pathway mediates early acceleration of liver regeneration induced by a novel two-staged hepatectomy in mice. J Hepatol 2017; 66: 560-570 [PMID: 27771454 DOI: 10.1016/j.jhep.2016.10.014]
- Patel SH, Camargo FD, Yimlamai D. Hippo Signaling in the Liver Regulates Organ Size, Cell Fate, 290 and Carcinogenesis. Gastroenterology 2017; 152: 533-545 [PMID: 28003097 DOI: 10.1053/j.gastro.2016.10.047]
- 291 Zheng Y, Pan D. The Hippo Signaling Pathway in Development and Disease. Dev Cell 2019; 50: 264-282 [PMID: 31386861 DOI: 10.1016/j.devcel.2019.06.003]
- 292 Grijalva JL, Huizenga M, Mueller K, Rodriguez S, Brazzo J, Camargo F, Sadri-Vakili G, Vakili K. Dynamic alterations in Hippo signaling pathway and YAP activation during liver regeneration. Am J Physiol Gastrointest Liver Physiol 2014; 307: G196-G204 [PMID: 24875096 DOI: 10.1152/ajpgi.00077.2014]
- 293 Chen L, Loh PG, Song H. Structural and functional insights into the TEAD-YAP complex in the Hippo signaling pathway. Protein Cell 2010; 1: 1073-1083 [PMID: 21213102 DOI: 10.1007/s13238-010-0138-3
- 294 Miesfeld JB, Link BA. Establishment of transgenic lines to monitor and manipulate Yap/Taz-Tead activity in zebrafish reveals both evolutionarily conserved and divergent functions of the Hippo pathway. Mech Dev 2014; 133: 177-188 [PMID: 24560909 DOI: 10.1016/j.mod.2014.02.003]
- 295 Meng Z, Moroishi T, Guan KL. Mechanisms of Hippo pathway regulation. Genes Dev 2016; 30: 1-17 [PMID: 26728553 DOI: 10.1101/gad.274027.115]
- Oh SH, Swiderska-Syn M, Jewell ML, Premont RT, Diehl AM. Liver regeneration requires Yap1-296 TGFβ-dependent epithelial-mesenchymal transition in hepatocytes. J Hepatol 2018; 69: 359-367 [PMID: 29758331 DOI: 10.1016/j.jhep.2018.05.008]
- 297 Yagi S, Hirata M, Miyachi Y, Uemoto S. Liver Regeneration after Hepatectomy and Partial Liver Transplantation. Int J Mol Sci 2020; 21 [PMID: 33182515 DOI: 10.3390/ijms21218414]
- 298 Wellbrock C, Karasarides M, Marais R. The RAF proteins take centre stage. Nat Rev Mol Cell Biol 2004; 5: 875-885 [PMID: 15520807 DOI: 10.1038/nrm1498]
- 299 Asati V, Mahapatra DK, Bharti SK. PI3K/Akt/mTOR and Ras/Raf/MEK/ERK signaling pathways inhibitors as anticancer agents: Structural and pharmacological perspectives. Eur J Med Chem 2016; 109: 314-341 [PMID: 26807863 DOI: 10.1016/j.ejmech.2016.01.012]
- Ersahin T, Tuncbag N, Cetin-Atalay R. The PI3K/AKT/mTOR interactive pathway. Mol Biosyst 300 2015; 11: 1946-1954 [PMID: 25924008 DOI: 10.1039/c5mb00101c]
- 301 Wullschleger S, Loewith R, Hall MN. TOR signaling in growth and metabolism. Cell 2006; 124: 471-484 [PMID: 16469695 DOI: 10.1016/j.cell.2006.01.016]
- 302 Fresno Vara JA, Casado E, de Castro J, Cejas P, Belda-Iniesta C, González-Barón M. PI3K/Akt signalling pathway and cancer. Cancer Treat Rev 2004; 30: 193-204 [PMID: 15023437 DOI: 10.1016/j.ctrv.2003.07.007]
- 303 Jackson LN, Larson SD, Silva SR, Rychahou PG, Chen LA, Qiu S, Rajaraman S, Evers BM. PI3K/Akt activation is critical for early hepatic regeneration after partial hepatectomy. Am J Physiol Gastrointest Liver Physiol 2008; 294: G1401-G1410 [PMID: 18388186 DOI: 10.1152/ajpgi.00062.2008]
- 304 Gao B, Wang H, Lafdil F, Feng D. STAT proteins - key regulators of anti-viral responses, inflammation, and tumorigenesis in the liver. J Hepatol 2012; 57: 430-441 [PMID: 22504331 DOI: 10.1016/j.jhep.2012.01.029]
- 305 Solt LA, May MJ. The IkappaB kinase complex: master regulator of NF-kappaB signaling. Immunol Res 2008; 42: 3-18 [PMID: 18626576 DOI: 10.1007/s12026-008-8025-1]
- 306 Moya IM, Halder G. Hippo-YAP/TAZ signalling in organ regeneration and regenerative medicine. Nat Rev Mol Cell Biol 2019; 20: 211-226 [PMID: 30546055 DOI: 10.1038/s41580-018-0086-y]
- 307 Simanshu DK, Nissley DV, McCormick F. RAS Proteins and Their Regulators in Human Disease. Cell 2017; 170: 17-33 [PMID: 28666118 DOI: 10.1016/j.cell.2017.06.009]
- Taniguchi K, Karin M. NF-kB, inflammation, immunity and cancer: coming of age. Nat Rev 308 Immunol 2018; 18: 309-324 [PMID: 29379212 DOI: 10.1038/nri.2017.142]
- 309 Yimlamai D, Fowl BH, Camargo FD. Emerging evidence on the role of the Hippo/YAP pathway in liver physiology and cancer. J Hepatol 2015; 63: 1491-1501 [PMID: 26226451 DOI: 10.1016/j.jhep.2015.07.008]
- Michalopoulos GK, DeFrances MC. Liver regeneration. Science 1997; 276: 60-66 [PMID: 310 9082986 DOI: 10.1126/science.276.5309.60]
- Michalopoulos GK. Liver regeneration. J Cell Physiol 2007; 213: 286-300 [PMID: 17559071 DOI: 311 10.1002/jcp.21172]
- Ross MA, Sander CM, Kleeb TB, Watkins SC, Stolz DB. Spatiotemporal expression of 312 angiogenesis growth factor receptors during the revascularization of regenerating rat liver. Hepatology 2001; 34: 1135-1148 [PMID: 11732003 DOI: 10.1053/jhep.2001.29624]
- 313 LeCouter J, Moritz DR, Li B, Phillips GL, Liang XH, Gerber HP, Hillan KJ, Ferrara N. Angiogenesis-independent endothelial protection of liver: role of VEGFR-1. Science 2003; 299: 890-893 [PMID: 12574630 DOI: 10.1126/science.1079562]
- 314 Wang L, Wang X, Wang L, Chiu JD, van de Ven G, Gaarde WA, Deleve LD. Hepatic vascular



endothelial growth factor regulates recruitment of rat liver sinusoidal endothelial cell progenitor cells. Gastroenterology 2012; 143: 1555-1563.e2 [PMID: 22902870 DOI: 10.1053/j.gastro.2012.08.008]

- 315 Wang L, Wang X, Xie G, Wang L, Hill CK, DeLeve LD. Liver sinusoidal endothelial cell progenitor cells promote liver regeneration in rats. J Clin Invest 2012; 122: 1567-1573 [PMID: 22406533 DOI: 10.1172/JCI58789]
- 316 Yin C, Evason KJ, Asahina K, Stainier DY. Hepatic stellate cells in liver development, regeneration, and cancer. J Clin Invest 2013; 123: 1902-1910 [PMID: 23635788 DOI: 10.1172/JCI66369]
- 317 Kandilis AN, Koskinas J, Tiniakos DG, Nikiteas N, Perrea DN. Liver regeneration: focus on cell types and topographic differences. Eur Surg Res 2010; 44: 1-12 [PMID: 19864910 DOI: 10.1159/000252770]
- Reynaert H, Urbain D, Geerts A. Regulation of sinusoidal perfusion in portal hypertension. Anat 318 Rec (Hoboken) 2008; 291: 693-698 [PMID: 18484616 DOI: 10.1002/ar.20669]
- 319 Hellerbrand C. Hepatic stellate cells--the pericytes in the liver. Pflugers Arch 2013; 465: 775-778 [PMID: 23292551 DOI: 10.1007/s00424-012-1209-5]
- 320 Tsuchida T, Friedman SL. Mechanisms of hepatic stellate cell activation. Nat Rev Gastroenterol Hepatol 2017; 14: 397-411 [PMID: 28487545 DOI: 10.1038/nrgastro.2017.38]
- Bansal MB. Hepatic stellate cells: fibrogenic, regenerative or both? Hepatol Int 2016; 10: 902-908 321 [PMID: 27578210 DOI: 10.1007/s12072-016-9758-x]
- 322 Friedman SL. Liver fibrosis -- from bench to bedside. J Hepatol 2003: 38 Suppl 1: S38-S53 [PMID: 12591185 DOI: 10.1016/s0168-8278(02)00429-4]
- 323 Passino MA, Adams RA, Sikorski SL, Akassoglou K. Regulation of hepatic stellate cell differentiation by the neurotrophin receptor p75NTR. Science 2007; 315: 1853-1856 [PMID: 17395831 DOI: 10.1126/science.1137603]
- Kato M, Iwamoto H, Higashi N, Sugimoto R, Uchimura K, Tada S, Sakai H, Nakamuta M, Nawata 324 H. Role of Rho small GTP binding protein in the regulation of actin cytoskeleton in hepatic stellate cells. J Hepatol 1999; 31: 91-99 [PMID: 10424288 DOI: 10.1016/s0168-8278(99)80168-8]
- 325 Fernández M, Semela D, Bruix J, Colle I, Pinzani M, Bosch J. Angiogenesis in liver disease. J Hepatol 2009; 50: 604-620 [PMID: 19157625 DOI: 10.1016/j.jhep.2008.12.011]
- Schachtrup C, Le Moan N, Passino MA, Akassoglou K. Hepatic stellate cells and astrocytes: Stars 326 of scar formation and tissue repair. Cell Cycle 2011; 10: 1764-1771 [PMID: 21555919 DOI: 10.4161/cc.10.11.15828
- 327 Shen K, Chang W, Gao X, Wang H, Niu W, Song L, Qin X. Depletion of activated hepatic stellate cell correlates with severe liver damage and abnormal liver regeneration in acetaminophen-induced liver injury. Acta Biochim Biophys Sin (Shanghai) 2011; 43: 307-315 [PMID: 21335335 DOI: 10.1093/abbs/gmr005]
- Pintilie DG, Shupe TD, Oh SH, Salganik SV, Darwiche H, Petersen BE. Hepatic stellate cells' 328 involvement in progenitor-mediated liver regeneration. Lab Invest 2010; 90: 1199-1208 [PMID: 20440274 DOI: 10.1038/labinvest.2010.88]
- Chang W, Song L, Chang X, Ji M, Wang H, Qin X, Niu W. Early activated hepatic stellate cell-329 derived paracrine molecules modulate acute liver injury and regeneration. Lab Invest 2017; 97: 318-328 [PMID: 27991908 DOI: 10.1038/labinvest.2016.130]
- 330 Gomez Perdiguero E, Klapproth K, Schulz C, Busch K, Azzoni E, Crozet L, Garner H, Trouillet C, de Bruijn MF, Geissmann F, Rodewald HR. Tissue-resident macrophages originate from yolk-sacderived erythro-myeloid progenitors. Nature 2015; 518: 547-551 [PMID: 25470051 DOI: 10.1038/nature13989
- 331 Willekens FL, Werre JM, Kruijt JK, Roerdinkholder-Stoelwinder B, Groenen-Döpp YA, van den Bos AG, Bosman GJ, van Berkel TJ. Liver Kupffer cells rapidly remove red blood cell-derived vesicles from the circulation by scavenger receptors. Blood 2005; 105: 2141-2145 [PMID: 15550489 DOI: 10.1182/blood-2004-04-1578]
- Theurl I, Hilgendorf I, Nairz M, Tymoszuk P, Haschka D, Asshoff M, He S, Gerhardt LM, 332 Holderried TA, Seifert M, Sopper S, Fenn AM, Anzai A, Rattik S, McAlpine C, Theurl M, Wieghofer P, Iwamoto Y, Weber GF, Harder NK, Chousterman BG, Arvedson TL, McKee M, Wang F, Lutz OM, Rezoagli E, Babitt JL, Berra L, Prinz M, Nahrendorf M, Weiss G, Weissleder R, Lin HY, Swirski FK. On-demand erythrocyte disposal and iron recycling requires transient macrophages in the liver. Nat Med 2016; 22: 945-951 [PMID: 27428900 DOI: 10.1038/nm.4146]
- 333 Wang Y, van der Tuin S, Tjeerdema N, van Dam AD, Rensen SS, Hendrikx T, Berbée JF, Atanasovska B, Fu J, Hoekstra M, Bekkering S, Riksen NP, Buurman WA, Greve JW, Hofker MH, Shiri-Sverdlov R, Meijer OC, Smit JW, Havekes LM, van Dijk KW, Rensen PC. Plasma cholesteryl ester transfer protein is predominantly derived from Kupffer cells. Hepatology 2015; 62: 1710-1722 [PMID: 26174697 DOI: 10.1002/hep.27985]
- 334 Helmy KY, Katschke KJ Jr, Gorgani NN, Kljavin NM, Elliott JM, Diehl L, Scales SJ, Ghilardi N, van Lookeren Campagne M. CRIg: a macrophage complement receptor required for phagocytosis of circulating pathogens. Cell 2006; 124: 915-927 [PMID: 16530040 DOI: 10.1016/j.cell.2005.12.039]
- You Q, Cheng L, Kedl RM, Ju C. Mechanism of T cell tolerance induction by murine hepatic 335 Kupffer cells. Hepatology 2008; 48: 978-990 [PMID: 18712788 DOI: 10.1002/hep.22395]
- 336 Wacker HH, Radzun HJ, Parwaresch MR. Kinetics of Kupffer cells as shown by parabiosis and combined autoradiographic/immunohistochemical analysis. Virchows Arch B Cell Pathol Incl Mol Pathol 1986; 51: 71-78 [PMID: 2873680 DOI: 10.1007/bf02899017]



- Sieweke MH, Allen JE. Beyond stem cells: self-renewal of differentiated macrophages. Science 337 2013: 342: 1242974 [PMID: 24264994 DOI: 10.1126/science.1242974]
- 338 Yona S, Kim KW, Wolf Y, Mildner A, Varol D, Breker M, Strauss-Avali D, Viukov S, Guilliams M, Misharin A, Hume DA, Perlman H, Malissen B, Zelzer E, Jung S. Fate mapping reveals origins and dynamics of monocytes and tissue macrophages under homeostasis. Immunity 2013; 38: 79-91 [PMID: 23273845 DOI: 10.1016/j.immuni.2012.12.001]
- 339 Bonnardel J, T'Jonck W, Gaublomme D, Browaeys R, Scott CL, Martens L, Vanneste B, De Prijck S, Nedospasov SA, Kremer A, Van Hamme E, Borghgraef P, Toussaint W, De Bleser P, Mannaerts I, Beschin A, van Grunsven LA, Lambrecht BN, Taghon T, Lippens S, Elewaut D, Saeys Y, Guilliams M. Stellate Cells, Hepatocytes, and Endothelial Cells Imprint the Kupffer Cell Identity on Monocytes Colonizing the Liver Macrophage Niche. Immunity 2019; 51: 638-654.e9 [PMID: 31561945 DOI: 10.1016/j.immuni.2019.08.017]
- 340 Wang J, Kubes P. A Reservoir of Mature Cavity Macrophages that Can Rapidly Invade Visceral Organs to Affect Tissue Repair. Cell 2016; 165: 668-678 [PMID: 27062926 DOI: 10.1016/j.cell.2016.03.009
- 341 Swirski FK, Nahrendorf M, Etzrodt M, Wildgruber M, Cortez-Retamozo V, Panizzi P, Figueiredo JL, Kohler RH, Chudnovskiy A, Waterman P, Aikawa E, Mempel TR, Libby P, Weissleder R, Pittet MJ. Identification of splenic reservoir monocytes and their deployment to inflammatory sites. Science 2009; 325: 612-616 [PMID: 19644120 DOI: 10.1126/science.1175202]
- 342 Guillot A, Tacke F. Liver Macrophages: Old Dogmas and New Insights. Hepatol Commun 2019; 3: 730-743 [PMID: 31168508 DOI: 10.1002/hep4.1356]
- Shim YR, Jeong WI. Recent advances of sterile inflammation and inter-organ cross-talk in alcoholic 343 liver disease. Exp Mol Med 2020; 52: 772-780 [PMID: 32457490 DOI: 10.1038/s12276-020-0438-5
- 344 Koh MY, Gagea M, Sargis T, Lemos R Jr, Grandjean G, Charbono A, Bekiaris V, Sedy J, Kiriakova G, Liu X, Roberts LR, Ware C, Powis G. A new HIF-1a/RANTES-driven pathway to hepatocellular carcinoma mediated by germline haploinsufficiency of SART1/HAF in mice. Hepatology 2016; 63: 1576-1591 [PMID: 26799785 DOI: 10.1002/hep.28468]
- Lee JH, Shim YR, Seo W, Kim MH, Choi WM, Kim HH, Kim YE, Yang K, Ryu T, Jeong JM, Choi 345 HG, Eun HS, Kim SH, Mun H, Yoon JH, Jeong WI. Mitochondrial Double-Stranded RNA in Exosome Promotes Interleukin-17 Production Through Toll-Like Receptor 3 in Alcohol-associated Liver Injury. Hepatology 2020; 72: 609-625 [PMID: 31849082 DOI: 10.1002/hep.31041]
- 346 Marra F, Tacke F. Roles for chemokines in liver disease. Gastroenterology 2014; 147: 577-594.e1 [PMID: 25066692 DOI: 10.1053/j.gastro.2014.06.043]
- Wen Y, Lambrecht J, Ju C, Tacke F. Hepatic macrophages in liver homeostasis and diseases-347 diversity, plasticity and therapeutic opportunities. Cell Mol Immunol 2021; 18: 45-56 [PMID: 33041338 DOI: 10.1038/s41423-020-00558-8]
- 348 Wyler SL, D'Ingillo SL, Lamb CL, Mitchell KA. Monocyte chemoattractant protein-1 is not required for liver regeneration after partial hepatectomy. J Inflamm (Lond) 2016; 13: 28 [PMID: 27555804 DOI: 10.1186/s12950-016-0136-1]
- 349 Miyaoka Y, Ebato K, Kato H, Arakawa S, Shimizu S, Miyajima A. Hypertrophy and unconventional cell division of hepatocytes underlie liver regeneration. Curr Biol 2012; 22: 1166-1175 [PMID: 22658593 DOI: 10.1016/j.cub.2012.05.016]
- 350 Donne R, Saroul-Aïnama M, Cordier P, Celton-Morizur S, Desdouets C. Polyploidy in liver development, homeostasis and disease. Nat Rev Gastroenterol Hepatol 2020; 17: 391-405 [PMID: 32242122 DOI: 10.1038/s41575-020-0284-x]
- Wang B, Zhao L, Fish M, Logan CY, Nusse R. Self-renewing diploid Axin2(+) cells fuel 351 homeostatic renewal of the liver. Nature 2015; 524: 180-185 [PMID: 26245375 DOI: 10.1038/nature14863
- 352 Font-Burgada J, Shalapour S, Ramaswamy S, Hsueh B, Rossell D, Umemura A, Taniguchi K, Nakagawa H, Valasek MA, Ye L, Kopp JL, Sander M, Carter H, Deisseroth K, Verma IM, Karin M. Hybrid Periportal Hepatocytes Regenerate the Injured Liver without Giving Rise to Cancer. Cell 2015; 162: 766-779 [PMID: 26276631 DOI: 10.1016/j.cell.2015.07.026]
- 353 Michalopoulos GK, Khan Z. Liver Stem Cells: Experimental Findings and Implications for Human Liver Disease. Gastroenterology 2015; 149: 876-882 [PMID: 26278502 DOI: 10.1053/j.gastro.2015.08.004]
- 354 Zong Y, Stanger BZ. Molecular mechanisms of liver and bile duct development. Wiley Interdiscip Rev Dev Biol 2012; 1: 643-655 [PMID: 23799566 DOI: 10.1002/wdev.47]
- 355 Choi TY, Ninov N, Stainier DY, Shin D. Extensive conversion of hepatic biliary epithelial cells to hepatocytes after near total loss of hepatocytes in zebrafish. Gastroenterology 2014; 146: 776-788 [PMID: 24148620 DOI: 10.1053/j.gastro.2013.10.019]
- Rodrigo-Torres D, Affò S, Coll M, Morales-Ibanez O, Millán C, Blaya D, Alvarez-Guaita A, 356 Rentero C, Lozano JJ, Maestro MA, Solar M, Arroyo V, Caballería J, van Grunsven LA, Enrich C, Ginès P, Bataller R, Sancho-Bru P. The biliary epithelium gives rise to liver progenitor cells. Hepatology 2014; 60: 1367-1377 [PMID: 24700364 DOI: 10.1002/hep.27078]
- 357 Verfaillie CM. Biliary cells to the rescue of Prometheus. Gastroenterology 2014; 146: 611-614 [PMID: 24462990 DOI: 10.1053/j.gastro.2014.01.039]
- Kiseleva YV, Antonyan SZ, Zharikova TS, Tupikin KA, Kalinin DV, Zharikov YO. Molecular 358 pathways of liver regeneration: A comprehensive review. World J Hepatol 2021; 13: 270-290



[PMID: 33815672 DOI: 10.4254/wjh.v13.i3.270]

- 359 Evarts RP, Hu Z, Omori N, Omori M, Marsden ER, Thorgeirsson SS. Precursor-product relationship between oval cells and hepatocytes: comparison between tritiated thymidine and bromodeoxyuridine as tracers. Carcinogenesis 1996; 17: 2143-2151 [PMID: 8895481 DOI: 10.1093/carcin/17.10.2143]
- Schaub JR, Huppert KA, Kurial SNT, Hsu BY, Cast AE, Donnelly B, Karns RA, Chen F, Rezvani 360 M, Luu HY, Mattis AN, Rougemont AL, Rosenthal P, Huppert SS, Willenbring H. De novo formation of the biliary system by TGFβ-mediated hepatocyte transdifferentiation. *Nature* 2018; 557: 247-251 [PMID: 29720662 DOI: 10.1038/s41586-018-0075-5]
- 361 Tarlow BD, Pelz C, Naugler WE, Wakefield L, Wilson EM, Finegold MJ, Grompe M. Bipotential adult liver progenitors are derived from chronically injured mature hepatocytes. Cell Stem Cell 2014; 15: 605-618 [PMID: 25312494 DOI: 10.1016/j.stem.2014.09.008]
- Michalopoulos GK, Barua L, Bowen WC. Transdifferentiation of rat hepatocytes into biliary cells 362 after bile duct ligation and toxic biliary injury. Hepatology 2005; 41: 535-544 [PMID: 15726663 DOI: 10.1002/hep.20600]
- Pibiri M. Liver regeneration in aged mice: new insights. Aging (Albany NY) 2018; 10: 1801-1824 363 [PMID: 30157472 DOI: 10.18632/aging.101524]
- 364 Le Lay J, Kaestner KH. The Fox genes in the liver: from organogenesis to functional integration. Physiol Rev 2010; 90: 1-22 [PMID: 20086072 DOI: 10.1152/physrev.00018.2009]
- 365 Wang X, Quail E, Hung NJ, Tan Y, Ye H, Costa RH. Increased levels of forkhead box M1B transcription factor in transgenic mouse hepatocytes prevent age-related proliferation defects in regenerating liver. Proc Natl Acad Sci USA 2001; 98: 11468-11473 [PMID: 11572993 DOI: 10.1073/pnas.201360898
- Iakova P, Awad SS, Timchenko NA. Aging reduces proliferative capacities of liver by switching 366 pathways of C/EBPalpha growth arrest. Cell 2003; 113: 495-506 [PMID: 12757710 DOI: 10.1016/s0092-8674(03)00318-0]
- 367 Ikawa-Yoshida A, Matsumoto T, Okano S, Aoyagi Y, Matsubara Y, Furuyama T, Nakatsu Y, Tsuzuki T, Onimaru M, Ohkusa T, Nomura M, Maehara Y. BubR1 Insufficiency Impairs Liver Regeneration in Aged Mice after Hepatectomy through Intercalated Disc Abnormality. Sci Rep 2016; 6: 32399 [PMID: 27561386 DOI: 10.1038/srep32399]
- 368 Conboy IM, Conboy MJ, Wagers AJ, Girma ER, Weissman IL, Rando TA. Rejuvenation of aged progenitor cells by exposure to a young systemic environment. Nature 2005; 433: 760-764 [PMID: 15716955 DOI: 10.1038/nature03260]
- Liu A, Guo E, Yang J, Yang Y, Liu S, Jiang X, Hu Q, Dirsch O, Dahmen U, Zhang C, Gewirtz DA, 369 Fang H. Young plasma reverses age-dependent alterations in hepatic function through the restoration of autophagy. Aging Cell 2018; 17 [PMID: 29210183 DOI: 10.1111/acel.12708]
- Apte U, Singh S, Zeng G, Cieply B, Virji MA, Wu T, Monga SP. Beta-catenin activation promotes 370 liver regeneration after acetaminophen-induced injury. Am J Pathol 2009; 175: 1056-1065 [PMID: 19679878 DOI: 10.2353/ajpath.2009.080976]
- Holt MP, Cheng L, Ju C. Identification and characterization of infiltrating macrophages in 371 acetaminophen-induced liver injury. J Leukoc Biol 2008; 84: 1410-1421 [PMID: 18713872 DOI: 10.1189/jlb.0308173
- 372 Amemiya H, Kono H, Fujii H. Liver regeneration is impaired in macrophage colony stimulating factor deficient mice after partial hepatectomy: the role of M-CSF-induced macrophages. J Surg Res 2011; 165: 59-67 [PMID: 20031174 DOI: 10.1016/j.jss.2009.08.008]
- Stutchfield BM, Antoine DJ, Mackinnon AC, Gow DJ, Bain CC, Hawley CA, Hughes MJ, Francis 373 B, Wojtacha D, Man TY, Dear JW, Devey LR, Mowat AM, Pollard JW, Park BK, Jenkins SJ, Simpson KJ, Hume DA, Wigmore SJ, Forbes SJ. CSF1 Restores Innate Immunity After Liver Injury in Mice and Serum Levels Indicate Outcomes of Patients With Acute Liver Failure. Gastroenterology 2015; 149: 1896-1909.e14 [PMID: 26344055 DOI: 10.1053/j.gastro.2015.08.053]
- 374 Veteläinen R, van Vliet AK, van Gulik TM. Severe steatosis increases hepatocellular injury and impairs liver regeneration in a rat model of partial hepatectomy. Ann Surg 2007; 245: 44-50 [PMID: 17197964 DOI: 10.1097/01.sla.0000225253.84501.0e]
- Truant S, Bouras AF, Petrovai G, Buob D, Ernst O, Boleslawski E, Hebbar M, Pruvot FR. 375 Volumetric gain of the liver after major hepatectomy in obese patients: a case-matched study in 84 patients. Ann Surg 2013; 258: 696-702; discussion 702 [PMID: 23979277 DOI: 10.1097/SLA.0b013e3182a61a22
- 376 Selzner M, Clavien PA. Failure of regeneration of the steatotic rat liver: disruption at two different levels in the regeneration pathway. Hepatology 2000; 31: 35-42 [PMID: 10613725 DOI: 10.1002/hep.510310108
- 377 Zimmers TA, Jin X, Zhang Z, Jiang Y, Koniaris LG. Epidermal growth factor receptor restoration rescues the fatty liver regeneration in mice. Am J Physiol Endocrinol Metab 2017; 313: E440-E449 [PMID: 28655714 DOI: 10.1152/ajpendo.00032.2017]
- DeAngelis RA, Markiewski MM, Taub R, Lambris JD. A high-fat diet impairs liver regeneration in 378 C57BL/6 mice through overexpression of the NF-kappaB inhibitor, IkappaBalpha. Hepatology 2005; 42: 1148-1157 [PMID: 16231352 DOI: 10.1002/hep.20879]
- 379 Inaba Y, Furutani T, Kimura K, Watanabe H, Haga S, Kido Y, Matsumoto M, Yamamoto Y, Harada K, Kaneko S, Oyadomari S, Ozaki M, Kasuga M, Inoue H. Growth arrest and DNA damageinducible 34 regulates liver regeneration in hepatic steatosis in mice. Hepatology 2015; 61: 1343-



1356 [PMID: 25420998 DOI: 10.1002/hep.27619]

- Henderson NC, Forbes SJ. Hepatic fibrogenesis: from within and outwith. Toxicology 2008; 254: 380 130-135 [PMID: 18824072 DOI: 10.1016/j.tox.2008.08.017]
- Kallis YN, Robson AJ, Fallowfield JA, Thomas HC, Alison MR, Wright NA, Goldin RD, Iredale 381 JP, Forbes SJ. Remodelling of extracellular matrix is a requirement for the hepatic progenitor cell response. Gut 2011; 60: 525-533 [PMID: 21106552 DOI: 10.1136/gut.2010.224436]
- 382 Issa R, Zhou X, Trim N, Millward-Sadler H, Krane S, Benyon C, Iredale J. Mutation in collagen-1 that confers resistance to the action of collagenase results in failure of recovery from CCl4-induced liver fibrosis, persistence of activated hepatic stellate cells, and diminished hepatocyte regeneration. FASEB J 2003; 17: 47-49 [PMID: 12475903 DOI: 10.1096/fj.02-0494fje]
- Marshall A, Rushbrook S, Davies SE, Morris LS, Scott IS, Vowler SL, Coleman N, Alexander G. 383 Relation between hepatocyte G1 arrest, impaired hepatic regeneration, and fibrosis in chronic hepatitis C virus infection. Gastroenterology 2005; 128: 33-42 [PMID: 15633121 DOI: 10.1053/i.gastro.2004.09.076
- Bierie B, Moses HL. Transforming growth factor beta (TGF-beta) and inflammation in cancer. 384 Cytokine Growth Factor Rev 2010; 21: 49-59 [PMID: 20018551 DOI: 10.1016/j.cytogfr.2009.11.008
- 385 Nakamura T, Tomita Y, Hirai R, Yamaoka K, Kaji K, Ichihara A. Inhibitory effect of transforming growth factor-beta on DNA synthesis of adult rat hepatocytes in primary culture. Biochem Biophys Res Commun 1985; 133: 1042-1050 [PMID: 3910043 DOI: 10.1016/0006-291x(85)91241-0]
- Yasuda H, Mine T, Shibata H, Eto Y, Hasegawa Y, Takeuchi T, Asano S, Kojima I. Activin A: an 386 autocrine inhibitor of initiation of DNA synthesis in rat hepatocytes. J Clin Invest 1993; 92: 1491-1496 [PMID: 8376601 DOI: 10.1172/jci116727]
- Kogure K, Omata W, Kanzaki M, Zhang YQ, Yasuda H, Mine T, Kojima I. A single intraportal 387 administration of follistatin accelerates liver regeneration in partially hepatectomized rats. Gastroenterology 1995; 108: 1136-1142 [PMID: 7698581 DOI: 10.1016/0016-5085(95)90212-0]
- Gkretsi V, Bowen WC, Yang Y, Wu C, Michalopoulos GK. Integrin-linked kinase is involved in 388 matrix-induced hepatocyte differentiation. Biochem Biophys Res Commun 2007; 353: 638-643 [PMID: 17194454 DOI: 10.1016/j.bbrc.2006.12.091]
- 389 Rana B. Mischoulon D. Xie Y. Bucher NL. Farmer SR. Cell-extracellular matrix interactions can regulate the switch between growth and differentiation in rat hepatocytes: reciprocal expression of C/EBP alpha and immediate-early growth response transcription factors. Mol Cell Biol 1994; 14: 5858-5869 [PMID: 8065319 DOI: 10.1128/mcb.14.9.5858]
- 390 Gkretsi V, Apte U, Mars WM, Bowen WC, Luo JH, Yang Y, Yu YP, Orr A, St-Arnaud R, Dedhar S, Kaestner KH, Wu C, Michalopoulos GK. Liver-specific ablation of integrin-linked kinase in mice results in abnormal histology, enhanced cell proliferation, and hepatomegaly. Hepatology 2008; 48: 1932-1941 [PMID: 18846549 DOI: 10.1002/hep.22537]
- 391 Apte U, Gkretsi V, Bowen WC, Mars WM, Luo JH, Donthamsetty S, Orr A, Monga SP, Wu C, Michalopoulos GK. Enhanced liver regeneration following changes induced by hepatocyte-specific genetic ablation of integrin-linked kinase. Hepatology 2009; 50: 844-851 [PMID: 19575460 DOI: 10.1002/hep.23059]
- 392 Song HH, Filmus J. The role of glypicans in mammalian development. Biochim Biophys Acta 2002; 1573: 241-246 [PMID: 12417406 DOI: 10.1016/s0304-4165(02)00390-2]
- 393 Liu B, Paranjpe S, Bowen WC, Bell AW, Luo JH, Yu YP, Mars WM, Michalopoulos GK. Investigation of the role of glypican 3 in liver regeneration and hepatocyte proliferation. Am J Pathol 2009; 175: 717-724 [PMID: 19574424 DOI: 10.2353/ajpath.2009.081129]
- Cano-Gauci DF, Song HH, Yang H, McKerlie C, Choo B, Shi W, Pullano R, Piscione TD, Grisaru 394 S, Soon S, Sedlackova L, Tanswell AK, Mak TW, Yeger H, Lockwood GA, Rosenblum ND, Filmus J. Glypican-3-deficient mice exhibit developmental overgrowth and some of the abnormalities typical of Simpson-Golabi-Behmel syndrome. J Cell Biol 1999; 146: 255-264 [PMID: 10402475 DOI: 10.1083/jcb.146.1.255]
- Wang ND, Finegold MJ, Bradley A, Ou CN, Abdelsayed SV, Wilde MD, Taylor LR, Wilson DR, 395 Darlington GJ. Impaired energy homeostasis in C/EBP alpha knockout mice. Science 1995; 269: 1108-1112 [PMID: 7652557 DOI: 10.1126/science.7652557]
- 396 Jin J, Hong IH, Lewis K, Iakova P, Breaux M, Jiang Y, Sullivan E, Jawanmardi N, Timchenko L, Timchenko NA. Cooperation of C/EBP family proteins and chromatin remodeling proteins is essential for termination of liver regeneration. Hepatology 2015; 61: 315-325 [PMID: 25043739 DOI: 10.1002/hep.27295]
- Geng Y, Yu O, Sicinska E, Das M, Schneider JE, Bhattacharva S, Rideout WM, Bronson RT, 397 Gardner H, Sicinski P. Cyclin E ablation in the mouse. Cell 2003; 114: 431-443 [PMID: 12941272 DOI: 10.1016/s0092-8674(03)00645-7]
- 398 Hu W, Nevzorova YA, Haas U, Moro N, Sicinski P, Geng Y, Barbacid M, Trautwein C, Liedtke C. Concurrent deletion of cyclin E1 and cyclin-dependent kinase 2 in hepatocytes inhibits DNA replication and liver regeneration in mice. Hepatology 2014; 59: 651-660 [PMID: 23787781 DOI: 10.1002/hep.26584]
- 399 Lauper N, Beck AR, Cariou S, Richman L, Hofmann K, Reith W, Slingerland JM, Amati B. Cyclin E2: a novel CDK2 partner in the late G1 and S phases of the mammalian cell cycle. Oncogene 1998; 17: 2637-2643 [PMID: 9840927 DOI: 10.1038/sj.onc.1202477]
- Nevzorova YA, Tschaharganeh D, Gassler N, Geng Y, Weiskirchen R, Sicinski P, Trautwein C, 400



Liedtke C. Aberrant cell cycle progression and endoreplication in regenerating livers of mice that lack a single E-type cyclin. Gastroenterology 2009; 137: 691-703, 703.e1 [PMID: 19445941 DOI: 10.1053/j.gastro.2009.05.003]

- 401 Dong J, Feldmann G, Huang J, Wu S, Zhang N, Comerford SA, Gayyed MF, Anders RA, Maitra A, Pan D. Elucidation of a universal size-control mechanism in Drosophila and mammals. Cell 2007; 130: 1120-1133 [PMID: 17889654 DOI: 10.1016/j.cell.2007.07.019]
- Avruch J, Zhou D, Fitamant J, Bardeesy N. Mst1/2 signalling to Yap: gatekeeper for liver size and 402 tumour development. Br J Cancer 2011; 104: 24-32 [PMID: 21102585 DOI: 10.1038/sj.bjc.6606011]
- 403 Bartel DP. MicroRNAs: target recognition and regulatory functions. Cell 2009; 136: 215-233 [PMID: 19167326 DOI: 10.1016/j.cell.2009.01.002]
- 404 Yuan B, Dong R, Shi D, Zhou Y, Zhao Y, Miao M, Jiao B. Down-regulation of miR-23b may contribute to activation of the TGF-\u00b31/Smad3 signalling pathway during the termination stage of liver regeneration. FEBS Lett 2011; 585: 927-934 [PMID: 21354414 DOI: 10.1016/j.febslet.2011.02.031]
- 405 Chen H, Sun Y, Dong R, Yang S, Pan C, Xiang D, Miao M, Jiao B. Mir-34a is upregulated during liver regeneration in rats and is associated with the suppression of hepatocyte proliferation. PLoS One 2011; 6: e20238 [PMID: 21655280 DOI: 10.1371/journal.pone.0020238]
- 406 Martin J, Petrillo A, Smyth EC, Shaida N, Khwaja S, Cheow HK, Duckworth A, Heister P, Praseedom R, Jah A, Balakrishnan A, Harper S, Liau S, Kosmoliaptsis V, Huguet E. Colorectal liver metastases: Current management and future perspectives. World J Clin Oncol 2020; 11: 761-808 [PMID: 33200074 DOI: 10.5306/wjco.v11.i10.761]
- 407 Gangi A, Lu SC. Chemotherapy-associated liver injury in colorectal cancer. Therap Adv Gastroenterol 2020; 13: 1756284820924194 [PMID: 32547639 DOI: 10.1177/1756284820924194]
- 408 Pawlik TM, Olino K, Gleisner AL, Torbenson M, Schulick R, Choti MA. Preoperative chemotherapy for colorectal liver metastases: impact on hepatic histology and postoperative outcome. J Gastrointest Surg 2007; 11: 860-868 [PMID: 17492335 DOI: 10.1007/s11605-007-0149-4
- 409 Zeng J, Deng S, Wang Y, Li P, Tang L, Pang Y. Specific Inhibition of Acyl-CoA Oxidase-1 by an Acetylenic Acid Improves Hepatic Lipid and Reactive Oxygen Species (ROS) Metabolism in Rats Fed a High Fat Diet. J Biol Chem 2017; 292: 3800-3809 [PMID: 28077576 DOI: 10.1074/jbc.M116.763532
- 410 Sommer J, Mahli A, Freese K, Schiergens TS, Kuecuekoktay FS, Teufel A, Thasler WE, Müller M, Bosserhoff AK, Hellerbrand C. Analysis of molecular mechanisms of 5-fluorouracil-induced steatosis and inflammation in vitro and in mice. Oncotarget 2017; 8: 13059-13072 [PMID: 28055957 DOI: 10.18632/oncotarget.14371]
- 411 Vauthey JN, Pawlik TM, Ribero D, Wu TT, Zorzi D, Hoff PM, Xiong HQ, Eng C, Lauwers GY, Mino-Kenudson M, Risio M, Muratore A, Capussotti L, Curley SA, Abdalla EK. Chemotherapy regimen predicts steatohepatitis and an increase in 90-day mortality after surgery for hepatic colorectal metastases. J Clin Oncol 2006; 24: 2065-2072 [PMID: 16648507 DOI: 10.1200/JCO.2005.05.3074
- 412 Mahli A, Saugspier M, Koch A, Sommer J, Dietrich P, Lee S, Thasler R, Schulze-Luehrmann J, Luehrmann A, Thasler WE, Müller M, Bosserhoff A, Hellerbrand C. ERK activation and autophagy impairment are central mediators of irinotecan-induced steatohepatitis. Gut 2018; 67: 746-756 [PMID: 28053052 DOI: 10.1136/gutjnl-2016-312485]
- 413 Aloia T, Sebagh M, Plasse M, Karam V, Lévi F, Giacchetti S, Azoulay D, Bismuth H, Castaing D, Adam R. Liver histology and surgical outcomes after preoperative chemotherapy with fluorouracil plus oxaliplatin in colorectal cancer liver metastases. J Clin Oncol 2006; 24: 4983-4990 [PMID: 17075116 DOI: 10.1200/jco.2006.05.8156]
- 414 de Gramont A, Figer A, Seymour M, Homerin M, Hmissi A, Cassidy J, Boni C, Cortes-Funes H, Cervantes A, Freyer G, Papamichael D, Le Bail N, Louvet C, Hendler D, de Braud F, Wilson C, Morvan F, Bonetti A. Leucovorin and fluorouracil with or without oxaliplatin as first-line treatment in advanced colorectal cancer. J Clin Oncol 2000; 18: 2938-2947 [PMID: 10944126 DOI: 10.1200/jco.2000.18.16.2938
- Douillard JY, Cunningham D, Roth AD, Navarro M, James RD, Karasek P, Jandik P, Iveson T, 415 Carmichael J, Alakl M, Gruia G, Awad L, Rougier P. Irinotecan combined with fluorouracil compared with fluorouracil alone as first-line treatment for metastatic colorectal cancer: a multicentre randomised trial. Lancet 2000; 355: 1041-1047 [PMID: 10744089 DOI: 10.1016/s0140-6736(00)02034-1
- 416 Petrelli F, Barni S; Anti-EGFR agents for liver metastases. Resectability and outcome with anti-EGFR agents in patients with KRAS wild-type colorectal liver-limited metastases: a meta-analysis. Int J Colorectal Dis 2012; 27: 997-1004 [PMID: 22358385 DOI: 10.1007/s00384-012-1438-2]
- 417 Gruenberger T. Bridgewater J. Chau I. García Alfonso P. Rivoire M. Mudan S. Lasserre S. Hermann F, Waterkamp D, Adam R. Bevacizumab plus mFOLFOX-6 or FOLFOXIRI in patients with initially unresectable liver metastases from colorectal cancer: the OLIVIA multinational randomised phase II trial. Ann Oncol 2015; 26: 702-708 [PMID: 25538173 DOI: 10.1093/annonc/mdu580]
- Bridgewater JA, Pugh SA, Maishman T, Eminton Z, Mellor J, Whitehead A, Stanton L, Radford M, 418 Corkhill A, Griffiths GO, Falk S, Valle JW, O'Reilly D, Siriwardena AK, Hornbuckle J, Rees M,



Iveson TJ, Hickish T, Garden OJ, Cunningham D, Maughan TS, Primrose JN; New EPOC investigators. Systemic chemotherapy with or without cetuximab in patients with resectable colorectal liver metastasis (New EPOC): long-term results of a multicentre, randomised, controlled, phase 3 trial. Lancet Oncol 2020; 21: 398-411 [PMID: 32014119 DOI: 10.1016/S1470-2045(19)30798-3]

- Karoui M, Penna C, Amin-Hashem M, Mitry E, Benoist S, Franc B, Rougier P, Nordlinger B. 419 Influence of preoperative chemotherapy on the risk of major hepatectomy for colorectal liver metastases. Ann Surg 2006; 243: 1-7 [PMID: 16371728 DOI: 10.1097/01.sla.0000193603.26265.c3]
- 420 Nguyen-Khac E, Lobry C, Chatelain D, Fuks D, Joly JP, Brevet M, Tramier B, Mouly C, Hautefeuille V, Chauffert B, Regimbeau JM. A Reappraisal of Chemotherapy-Induced Liver Injury in Colorectal Liver Metastases before the Era of Antiangiogenics. Int J Hepatol 2013; 2013: 314868 [PMID: 23533786 DOI: 10.1155/2013/314868]
- Vigano L, De Rosa G, Toso C, Andres A, Ferrero A, Roth A, Sperti E, Majno P, Rubbia-Brandt L. 421 Reversibility of chemotherapy-related liver injury. J Hepatol 2017; 67: 84-91 [PMID: 28284915 DOI: 10.1016/j.jhep.2017.02.031]
- 422 Ribero D, Wang H, Donadon M, Zorzi D, Thomas MB, Eng C, Chang DZ, Curley SA, Abdalla EK, Ellis LM, Vauthey JN. Bevacizumab improves pathologic response and protects against hepatic injury in patients treated with oxaliplatin-based chemotherapy for colorectal liver metastases. Cancer 2007; 110: 2761-2767 [PMID: 17960603 DOI: 10.1002/cncr.23099]
- Santini D, Vincenzi B, Massacesi C, Picardi A, Gentilucci UV, Esposito V, Liuzzi G, La Cesa A, 423 Rocci L, Marcucci F, Montesarchio V, Groeger AM, Bonsignori M, Tonini G. Sadenosylmethionine (AdoMet) supplementation for treatment of chemotherapy-induced liver injury. Anticancer Res 2003; 23: 5173-5179 [PMID: 14981985]
- 424 Vincenzi B, Daniele S, Frezza AM, Berti P, Vespasiani U, Picardi A, Tonini G. The role of Sadenosylmethionine in preventing oxaliplatin-induced liver toxicity: a retrospective analysis in metastatic colorectal cancer patients treated with bevacizumab plus oxaliplatin-based regimen. Support Care Cancer 2012; 20: 135-139 [PMID: 21229271 DOI: 10.1007/s00520-010-1078-4]
- 425 Makuuchi M, Thai BL, Takayasu K, Takayama T, Kosuge T, Gunvén P, Yamazaki S, Hasegawa H, Ozaki H. Preoperative portal embolization to increase safety of major hepatectomy for hilar bile duct carcinoma: a preliminary report. Surgery 1990; 107: 521-527
- 426 van Lienden KP, van den Esschert JW, de Graaf W, Bipat S, Lameris JS, van Gulik TM, van Delden OM. Portal vein embolization before liver resection: a systematic review. Cardiovasc Intervent Radiol 2013; 36: 25-34 [PMID: 22806245 DOI: 10.1007/s00270-012-0440-y]
- 427 Schnitzbauer AA, Lang SA, Goessmann H, Nadalin S, Baumgart J, Farkas SA, Fichtner-Feigl S, Lorf T, Goralcyk A, Hörbelt R, Kroemer A, Loss M, Rümmele P, Scherer MN, Padberg W, Königsrainer A, Lang H, Obed A, Schlitt HJ. Right portal vein ligation combined with in situ splitting induces rapid left lateral liver lobe hypertrophy enabling 2-staged extended right hepatic resection in small-for-size settings. Ann Surg 2012; 255: 405-414 [PMID: 22330038 DOI: 10.1097/SLA.0b013e31824856f5
- Wigmore SJ. ALPPS: The argument against. Eur J Surg Oncol 2017; 43: 249-251 [PMID: 428 28007323 DOI: 10.1016/j.ejso.2016.11.009]
- 429 Veteläinen R, van Vliet A, Gouma DJ, van Gulik TM. Steatosis as a risk factor in liver surgery. Ann Surg 2007; 245: 20-30 [PMID: 17197961 DOI: 10.1097/01.sla.0000225113.88433.cf]
- Behrns KE, Tsiotos GG, DeSouza NF, Krishna MK, Ludwig J, Nagorney DM. Hepatic steatosis as 430 a potential risk factor for major hepatic resection. J Gastrointest Surg 1998; 2: 292-298 [PMID: 9841987 DOI: 10.1016/s1091-255x(98)80025-5]
- 431 Kooby DA, Fong Y, Suriawinata A, Gonen M, Allen PJ, Klimstra DS, DeMatteo RP, D'Angelica M, Blumgart LH, Jarnagin WR. Impact of steatosis on perioperative outcome following hepatic resection. J Gastrointest Surg 2003; 7: 1034-1044 [PMID: 14675713 DOI: 10.1016/j.gassur.2003.09.012]
- 432 Bhangui P, Laurent A, Amathieu R, Azoulay D. Assessment of risk for non-hepatic surgery in cirrhotic patients. J Hepatol 2012; 57: 874-884 [PMID: 22634123 DOI: 10.1016/j.jhep.2012.03.037]
- Berzigotti A, Reig M, Abraldes JG, Bosch J, Bruix J. Portal hypertension and the outcome of 433 surgery for hepatocellular carcinoma in compensated cirrhosis: a systematic review and metaanalysis. Hepatology 2015; 61: 526-536 [PMID: 25212123 DOI: 10.1002/hep.27431]
- 434 Selzner M, Rüdiger HA, Sindram D, Madden J, Clavien PA. Mechanisms of ischemic injury are different in the steatotic and normal rat liver. Hepatology 2000; 32: 1280-1288 [PMID: 11093735 DOI: 10.1053/jhep.2000.20528]
- 435 Serafín A, Roselló-Catafau J, Prats N, Xaus C, Gelpí E, Peralta C. Ischemic preconditioning increases the tolerance of Fatty liver to hepatic ischemia-reperfusion injury in the rat. Am J Pathol 2002; 161: 587-601 [PMID: 12163383 DOI: 10.1016/s0002-9440(10)64214-9]
- Clavien PA, Harvey PR, Strasberg SM. Preservation and reperfusion injuries in liver allografts. An 436 overview and synthesis of current studies. Transplantation 1992; 53: 957-978 [PMID: 1585489 DOI: 10.1097/00007890-199205000-00001]
- 437 Clavien PA, Selzner M, Rüdiger HA, Graf R, Kadry Z, Rousson V, Jochum W. A prospective randomized study in 100 consecutive patients undergoing major liver resection with vs without ischemic preconditioning. Ann Surg 2003; 238: 843-50; discussion 851 [PMID: 14631221 DOI: 10.1097/01.sla.0000098620.27623.7d]



- Hemming AW, Magliocca JF, Fujita S, Kayler LK, Hochwald S, Zendejas I, Kim RD. Combined 438 resection of the liver and pancreas for malignancy. J Am Coll Surg 2010; 210: 808-814, 814 [PMID: 20421055 DOI: 10.1016/j.jamcollsurg.2009.12.007]
- 439 Klein F, Puhl G, Guckelberger O, Pelzer U, Pullankavumkal JR, Guel S, Neuhaus P, Bahra M. The impact of simultaneous liver resection for occult liver metastases of pancreatic adenocarcinoma. Gastroenterol Res Pract 2012; 2012: 939350 [PMID: 23258977 DOI: 10.1155/2012/939350]
- 440 Singh A, Singh T, Chaudhary A. Synchronous resection of solitary liver metastases with pancreaticoduodenectomy. JOP 2010; 11: 434-438 [PMID: 20818110]
- 441 Otao R. Beppu T, Isiko T, Mima K, Okabe H, Hayashi H, Masuda T, Chikamoto A, Takamori H, Baba H. External biliary drainage and liver regeneration after major hepatectomy. Br J Surg 2012; 99: 1569-1574 [PMID: 23027074 DOI: 10.1002/bjs.8906]
- 442 Hayashi H, Beppu T, Sugita H, Horino K, Komori H, Masuda T, Okabe H, Takamori H, Baba H. Increase in the serum bile acid level predicts the effective hypertrophy of the nonembolized hepatic lobe after right portal vein embolization. World J Surg 2009; 33: 1933-1940 [PMID: 19551429 DOI: 10.1007/s00268-009-0111-6]
- 443 Rahbari NN, Garden OJ, Padbury R, Brooke-Smith M, Crawford M, Adam R, Koch M, Makuuchi M, Dematteo RP, Christophi C, Banting S, Usatoff V, Nagino M, Maddern G, Hugh TJ, Vauthey JN, Greig P, Rees M, Yokoyama Y, Fan ST, Nimura Y, Figueras J, Capussotti L, Büchler MW, Weitz J. Posthepatectomy liver failure: a definition and grading by the International Study Group of Liver Surgery (ISGLS). Surgery 2011; 149: 713-724 [PMID: 21236455 DOI: 10.1016/j.surg.2010.10.001]
- 444 Riddiough GE, Christophi C, Jones RM, Muralidharan V, Perini MV. A systematic review of small for size syndrome after major hepatectomy and liver transplantation. HPB (Oxford) 2020; 22: 487-496 [PMID: 31786053 DOI: 10.1016/j.hpb.2019.10.2445]
- 445 Fishbane S, El-Shahawy MA, Pecoits-Filho R, Van BP, Houser MT, Frison L, Little DJ, Guzman NJ, Pergola PE. Roxadustat for Treating Anemia in Patients with CKD Not on Dialysis: Results from a Randomized Phase 3 Study. J Am Soc Nephrol 2021; 32: 737-755 [PMID: 33568383 DOI: 10.1681/ASN.2020081150
- Roskams T, Cassiman D, De Vos R, Libbrecht L. Neuroregulation of the neuroendocrine 446 compartment of the liver. Anat Rec A Discov Mol Cell Evol Biol 2004; 280: 910-923 [PMID: 15382010 DOI: 10.1002/ar.a.20096]
- 447 **Kiba T**. The role of the autonomic nervous system in liver regeneration and apoptosis--recent developments. Digestion 2002; 66: 79-88 [PMID: 12428066 DOI: 10.1159/000065594]
- Takahashi T, Kakita A, Sakamoto I, Takahashi Y, Hayashi K, Tadokoro F, Yamashina S. 448 Immunohistochemical and electron microscopic study of extrinsic hepatic reinnervation following orthotopic liver transplantation in rats. Liver 2001; 21: 300-308 [PMID: 11589766 DOI: 10.1034/i.1600-0676.2001.210502.x
- 449 Golriz M, Ghamarnejad O, Khajeh E, Sabagh M, Mieth M, Hoffmann K, Ulrich A, Hackert T, Weiss KH, Schirmacher P, Büchler MW, Mehrabi A. Preoperative Thrombocytopenia May Predict Poor Surgical Outcome after Extended Hepatectomy. Can J Gastroenterol Hepatol 2018; 2018: 1275720 [PMID: 30515369 DOI: 10.1155/2018/1275720]
- 450 MacIndoe JH, Turkington RW. Stimulation of human prolactin secretion by intravenous infusion of L-tryptophan. J Clin Invest 1973; 52: 1972-1978 [PMID: 4541674 DOI: 10.1172/jci107381]
- 451 Banki CM, Vojnik M. Effect of intravenous infusion of amitriptyline on total blood serotonin content. Eur J Clin Pharmacol 1978; 13: 259-262 [PMID: 668782 DOI: 10.1007/bf00716360]
- 452 Bird TG, Lu WY, Boulter L, Gordon-Keylock S, Ridgway RA, Williams MJ, Taube J, Thomas JA, Wojtacha D, Gambardella A, Sansom OJ, Iredale JP, Forbes SJ. Bone marrow injection stimulates hepatic ductular reactions in the absence of injury via macrophage-mediated TWEAK signaling. Proc Natl Acad Sci USA 2013; 110: 6542-6547 [PMID: 23576749 DOI: 10.1073/pnas.1302168110]
- 453 Thomas JA, Pope C, Wojtacha D, Robson AJ, Gordon-Walker TT, Hartland S, Ramachandran P, Van Deemter M, Hume DA, Iredale JP, Forbes SJ. Macrophage therapy for murine liver fibrosis recruits host effector cells improving fibrosis, regeneration, and function. Hepatology 2011; 53: 2003-2015 [PMID: 21433043 DOI: 10.1002/hep.24315]
- 454 Hume DA, MacDonald KP. Therapeutic applications of macrophage colony-stimulating factor-1 (CSF-1) and antagonists of CSF-1 receptor (CSF-1R) signaling. Blood 2012; 119: 1810-1820 [PMID: 22186992 DOI: 10.1182/blood-2011-09-379214]
- 455 Cataldegirmen G, Zeng S, Feirt N, Ippagunta N, Dun H, Qu W, Lu Y, Rong LL, Hofmann MA, Kislinger T, Pachydaki SI, Jenkins DG, Weinberg A, Lefkowitch J, Rogiers X, Yan SF, Schmidt AM, Emond JC. RAGE limits regeneration after massive liver injury by coordinated suppression of TNF-alpha and NF-kappaB. J Exp Med 2005; 201: 473-484 [PMID: 15699076 DOI: 10.1084/jem.20040934]
- 456 Mangieri CW, McCartt JC, Strode MA, Lowry JE, Balakrishna PM. Perioperative hepatocyte growth factor (HGF) infusions improve hepatic regeneration following portal branch ligation (PBL) in rodents. Surg Endosc 2017; 31: 2789-2797 [PMID: 27752816 DOI: 10.1007/s00464-016-5288-9]
- Alwahsh SM, Rashidi H, Hay DC. Liver cell therapy: is this the end of the beginning? Cell Mol Life 457 Sci 2018; 75: 1307-1324 [PMID: 29181772 DOI: 10.1007/s00018-017-2713-8]
- 458 Muraca M, Gerunda G, Neri D, Vilei MT, Granato A, Feltracco P, Meroni M, Giron G, Burlina AB. Hepatocyte transplantation as a treatment for glycogen storage disease type 1a. Lancet 2002; 359: 317-318 [PMID: 11830200 DOI: 10.1016/s0140-6736(02)07529-3]
- 459 Liu H, Kim Y, Sharkis S, Marchionni L, Jang YY. In vivo liver regeneration potential of human



induced pluripotent stem cells from diverse origins. Sci Transl Med 2011; 3: 82ra39 [PMID: 21562231 DOI: 10.1126/scitranslmed.3002376]

- 460 Liu WH, Song FQ, Ren LN, Guo WQ, Wang T, Feng YX, Tang LJ, Li K. The multiple functional roles of mesenchymal stem cells in participating in treating liver diseases. J Cell Mol Med 2015; 19: 511-520 [PMID: 25534251 DOI: 10.1111/jcmm.12482]
- 461 Lee SC, Jeong HJ, Lee SK, Kim SJ. Lipopolysaccharide preconditioning of adipose-derived stem cells improves liver-regenerating activity of the secretome. Stem Cell Res Ther 2015; 6: 75 [PMID: 25890074 DOI: 10.1186/s13287-015-0072-7]
- Lee SC, Kim JO, Kim SJ. Secretome from human adipose-derived stem cells protects mouse liver 462 from hepatic ischemia-reperfusion injury. Surgery 2015; 157: 934-943 [PMID: 25704431 DOI: 10.1016/j.surg.2014.12.016]
- 463 Lee SK, Lee SC, Kim SJ. A novel cell-free strategy for promoting mouse liver regeneration: utilization of a conditioned medium from adipose-derived stem cells. Hepatol Int 2015; 9: 310-320 [PMID: 25788187 DOI: 10.1007/s12072-014-9599-4]



W J C O World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1157-1168

DOI: 10.5306/wico.v12.i12.1157

ISSN 2218-4333 (online)

REVIEW

Primary vascular tumours of the kidney

Ayo O Omiyale

ORCID number: Ayo O Omiyale 0000-0001-9955-9207.

Author contributions: Omiyale AO reviewed the literature and wrote the manuscript.

Conflict-of-interest statement: The author declares no conflict of interest for this article.

Country/Territory of origin: United Kingdom

Specialty type: Oncology

Provenance and peer review: Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): 0 Grade B (Very good): B Grade C (Good): 0 Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works

Ayo O Omiyale, Department of Cellular Pathology, Imperial College Healthcare NHS Trust, Charing Cross Hospital, London W6 8RF, United Kingdom

Corresponding author: Ayo O Omiyale, MBChB, MPH, Doctor, Department of Cellular Pathology, Imperial College Healthcare NHS Trust, Charing Cross Hospital, Fulham Palace Road, London W6 8RF, United Kingdom. ayodeji.omiyale@nhs.net

Abstract

Primary vascular tumours of the kidney are rare and may pose diagnostic difficulties because of their similar clinical, morphological, and immunohistochemical features. This article summarizes the clinical and pathological features of primary renal angiosarcoma and anastomosing haemangioma of the kidney including epidemiology, genetics, and prognosis. Renal anastomosing haemangiomas are benign neoplasms characterized by anastomosing capillary-sized vascular channels. These tumours are rare, with about 75 cases reported in the literature. Most anastomosing haemangiomas are found incidentally on ultrasound, computed tomography, or magnetic resonance imaging. Common symptoms include abdominal pain, haematuria, and abdominal mass. Renal anastomosing haemangiomas are characterized by recurrent mutations in GNAQ and GNA14 genes. The prognosis of anastomosing haemangioma is excellent.

Primary renal angiosarcomas are malignant tumours showing endothelial differentiation. To date, 76 cases have been described in the literature. Primary renal angiosarcomas are frequently symptomatic. The clinical features of renal angiosarcomas are similar to those of renal anastomosing haemangiomas, including abdominal pain, haematuria, and abdominal mass. Angiogenesis-related genes and vascular-specific receptor tyrosine kinases such as KDR, TIE1, SNRK, TEK, and FLT1 are upregulated in angiosarcomas. Primary renal angiosarcomas are highly aggressive neoplasms with a poor prognosis despite surgical treatment, chemotherapy, radiotherapy, or targeted therapy.

Key Words: Kidney; Renal tumours; Angiosarcoma; Haemangioma; Anastomosing haemangioma of the kidney; Vascular tumours

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: Primary vascular tumours of the kidney are extremely rare. This article


on different terms, provided the original work is properly cited and the use is non-commercial. See: htt ps://creativecommons.org/Licens es/by-nc/4.0/

Received: April 28, 2021 Peer-review started: April 28, 2021 First decision: June 16, 2021 Revised: July 1, 2021 Accepted: November 25, 2021 Article in press: November 25, 2021 Published online: December 24, 2021

P-Reviewer: Salvadori M S-Editor: Liu M L-Editor: A P-Editor: Liu M



summarizes the clinical and pathological features of primary renal angiosarcoma and anastomosing haemangioma of the kidney.

Citation: Omiyale AO. Primary vascular tumours of the kidney. World J Clin Oncol 2021; 12(12): 1157-1168

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1157.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1157

INTRODUCTION

Although vascular tumours are relatively common in the skin and soft tissue, they are extremely rare in the kidney, ranging from benign to malignant neoplasms that may be diagnostically challenging because of the overlapping clinical, morphological and immunohistochemical features.

These tumours include renal angiosarcomas and renal haemangiomas. Various subtypes of haemangioma have been described in the kidney including cavernous, capillary, and anastomosing haemangiomas^[1-4]. However, the most common subtype is anastomosing haemangioma[1,2,5].

This article provides an overview of the clinical and pathological features of anastomosing haemangioma of the kidney and primary angiosarcoma of the kidney, and discusses the epidemiology, genetics, and prognosis.

ANASTOMOSING HAEMANGIOMA OF THE KIDNEY

Renal anastomosing haemangiomas are benign neoplasms characterized by anastomosing capillary-sized vascular channels. These tumours are exceptionally rare with about 75 anastomosing haemangiomas reported in the literature[5-9]. These tumours occur in a wide age range from 10 to 83 years (mean, 49 years) with a male-to-female ratio of 2:1[10].

The aetiology and risk factors for renal anastomosing haemangiomas are unknown. Some cases have been reported in the setting of end stage renal disease[11,12].

The vast majority of anastomosing haemangiomas are found incidentally on radiological evaluation for other purposes. Common symptoms include abdominal pain, haematuria, and abdominal mass[5,10].

The imaging findings are non-specific. On computed tomography, these tumours are often circumscribed, hyperdense, and heterogeneous due to fatty or non-enhancing hypodense areas and show post-contrast enhancement^[13].

Renal anastomosing haemangiomas are characterized by recurrent mutations in GNAQ and GNA14 genes[14,15]. GNAQ gene encodes guanine nucleotide-binding protein G (q) subunit alpha (Gaq protein) that activates signalling pathways that regulates cell proliferation, survival, development, and function of blood vessels[14-16

Grossly, anastomosing haemangiomas are typically small ranging from 0.1 cm to 12 cm (mean, 2.2 cm) [5,10,17]. These tumours are often well-demarcated spongy mahogany brown masses^[5]. They are usually unilateral and solitary tumours; however, a few cases of bilateral^[18] and multifocal^[11] tumours have been described.

Histologically, anastomosing haemangiomas consist of anastomosing capillarysized blood vessels, reminiscent of splenic sinusoids. The blood vessels are lined by bland endothelial cells. Typically, these tumours lack endothelial cell multilayering, papillary tufting, cytologic atypia, necrosis, and prominent mitotic figures. They may show extramedullary haematopoiesis, hyaline globules, and mild lymphocytic infiltrate[1,2,5,10,18].

Rarely, renal anastomosing haemangiomas may infiltrate perinephric fat, renal sinus fat[2,12], and the renal vein or its segmental branches[2,12,18,19]. The neoplastic cells are immunoreactive for CD31, CD34, ERG, FLI1, and factor VIII-related antigen (now rarely used)[5].

Renal anastomosing haemangiomas may co-exist with other renal neoplasms such as metanephric adenoma, papillary adenoma, papillary renal cell carcinoma, acquired cystic disease-associated renal cell carcinoma, and clear cell renal cell carcinoma[11,12].



Most patients with renal anastomosing haemangioma, described in the literature, were treated with radical nephrectomy, reflecting a tendency to overtreat these patients, probably because of inaccurate preoperative diagnosis (Table 1).

The prognosis of anastomosing haemangioma is excellent with no evidence of recurrence, metastasis or tumour-related death at an average follow-up of 24.8 mo (range, < 1-156 mo) (Table 1).

PRIMARY ANGIOSARCOMA OF THE KIDNEY

Renal angiosarcomas are malignant tumours showing endothelial differentiation. These tumours are very rare, with about 76 cases described in the literature as case series and reports[20-24]. There is a male-to-female ratio of 6:1, with patient age ranging from 24 years to 95 years (median, 62 years).

Although angiosarcomas arising in other anatomical sites have been associated with risk factors such as exposure to thorium dioxide, arsenic-based pesticides, polyvinyl chloride, and radiation therapy particularly for breast, endometrial and prostate cancers[25-27], no specific aetiology or risk factors have been established for primary angiosarcoma of the kidney[20,28].

The clinical features of renal angiosarcomas are identical to those of renal anastomosing haemangiomas. Common symptoms include abdominal pain, haematuria, abdominal mass, and weight loss. A rare case of spontaneous tumour rupture with retroperitoneal haematoma has been described[29].

Computed tomography imaging shows large masses with heterogeneous enhancement and hypervascularity[30].

Angiogenesis-related genes and vascular-specific receptor tyrosine kinases such as KDR, TIE1, SNRK, TEK, and FLT1, are upregulated in angiosarcomas[31]. High-level MYC gene amplifications are seen in most radiation-induced and chronic lymphoedema-associated angiosarcomas[32]. A subset of cases is characterized by PLCG1, KDR, and PTPRB mutations[33,34].

Some primary angiosarcomas, typically in young adults, have recurrent CIC gene rearrangements, with or without concurrent CIC mutations, and are characterized by upregulation of CIC-target genes including *ETV1*, *ETV4*, and *ETV5*[34]. Angiosarcomas with CIC gene abnormalities are associated with an inferior disease-free survival[34].

Primary renal angiosarcomas are usually large ranging from 3.6 cm to 30 cm (mean, 13 cm). Typically, they are ill-defined haemorrhagic spongy masses with necrosis[1, 20].

Microscopically, these tumours range from well-formed vasoformative areas to areas with solid morphology showing sparse vasoformation. These patterns are often mixed within the same tumour. Vasoformative areas are composed of small to medium-sized anastomosing blood vessels, lined by epithelioid and/or spindled endothelial cells showing nuclear pleomorphism, endothelial papillary tufting, multilayering, intraluminal budding, and hobnailing[1,20,28]. Solid areas consist of sheets of malignant epithelioid and/or spindled cells with subtle vasoformation, cytologic atypia, and mitotic figures. Necrosis may be present. Angiosarcomas usually have a haemorrhagic background and extravasated red blood cells are seen within the tumour[1,20,28].

Epithelioid angiosarcomas are composed of sheets of large atypical polygonal or epithelioid cells with nuclear pleomorphism, high nuclear to cytoplasmic ratio, prominent central nucleoli, mitotic figures, and moderate amounts of cytoplasm. Epithelioid angiosarcomas may be mistaken for carcinoma, melanoma, or lymphoma [20,28,35-37].

The neoplastic cells are positive for CD31, ERG, FLI1, CD34, and factor VIII-related antigen[20,28,38,39]. Epithelioid angiosarcomas may be positive for epithelial markers including CK7, Cam5.2, AE1/AE3, and EMA, which may lead to a misdiagnosis of carcinoma[20].

Similar to angiosarcomas arising at other locations, renal angiosarcomas have a tendency for widespread metastasis at diagnosis or afterwards in the course of the disease. Approximately 66% of patients develop metastases, most commonly to the lung and liver. Other sites of metastasis include bone, lymph nodes, peritoneum, small bowel, soft tissue, and skin. Currently, there are no specific standardized treatment guidelines for primary renal angiosarcomas. These tumours are treated with radical nephrectomy, chemotherapy, radiotherapy, or targeted therapy (Table 2).

Zaishidena® WJCO | https://www.wjgnet.com

Table 1 Treatment, follow-up, and outcome of patients with anastomosing haemangioma of the kidney					
Ref.	Treatment	Follow-up (mo)	Outcome		
Bean et al[14]	Nephrectomy	9	NED		
Bean et al[14]	Nephrectomy	84	NED		
Bean et al[14]	Nephrectomy	107	NED		
Memmedoğlu and Musayev[41]	Nephrectomy	12	NED		
Memmedoğlu and Musayev[41]	Nephrectomy	12	NED		
Tahir and Folwell[42]	Nephrectomy	1	NED		
Pantelides <i>et al</i> [6]	Nephrectomy	6	NED		
Downes et al[7]	Nephrectomy	NA	NA		
Downes et al[7]	Biopsy	NA	NA		
Chandran et al[8]	Nephrectomy	NA	NA		
Cha et al[9]	Nephrectomy	5	NED		
Montgomery and Epstein[2]	Nephrectomy	12	NED		
Montgomery and Epstein[2]	Nephrectomy	36	NED		
Montgomery and Epstein[2]	Nephrectomy	NA	NA		
Montgomery and Epstein[2]	Excision	8	NED		
Heidegger et al[43]	Nephrectomy	156	NED		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Kryvenko <i>et al</i> [12]	Nephrectomy	NA	NA		
Al-Maghrabi and Al-Rashed[44]	Partial nephrectomy	12	NED		
Caballes <i>et al</i> [17]	Nephrectomy	18	NED		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner <i>et al</i> [11]	Nephrectomy	NA	NA		
Büttner et al[11]	Nephrectomy	NA	NA		
Lee <i>et a</i> [4 5]	Nephrectomy	NA	NA		
Zhao et al[<mark>46</mark>]	Nephrectomy	12	NED		
Kryvenko <i>et al</i> [18]	Nephrectomy	7	NED		



Kryvenko et al[18]	Nephrectomy	6	NED
Kryvenko <i>et al</i> [18]	Nephrectomy	3	NED
Kryvenko <i>et al</i> [18]	Nephrectomy	122	NED
Tao <i>et al</i> [47]	Nephrectomy	21	NED
Abboudi <i>et al</i> [48]	Nephrectomy	<1	NED
Silva et al[49]	Resection	NA	NA
Berker et al[50]	Partial nephrectomy	10	NED
Berker et al[50]	Nephrectomy	4	NED
O'Neill <i>et al</i> [13]	NA	NA	NA
O'Neill <i>et al</i> [13]	NA	NA	NA
O'Neill <i>et al</i> [13]	NA	NA	NA
O'Neill et al[13]	NA	NA	NA
O'Neill <i>et al</i> [13]	NA	NA	NA
O'Neill et al[13]	NA	NA	NA
O'Neill et al[13]	NA	NA	NA
Brown et al[1]	Nephrectomy	72	NED
Brown et al[1]	Nephrectomy	24	NED
Brown et al[1]	Partial nephrectomy	NA	NA
Brown et al[1]	Nephrectomy	24	NED
Brown et al[1]	Nephrectomy	NA	NA
Perdiki <i>et al</i> [51]	Partial nephrectomy	25	NED
Perdiki <i>et al</i> [51]	Nephrectomy	14	NED
Wetherell <i>et al</i> [52]	Nephrectomy	1	DFUD
Manohar et al[53]	Nephrectomy	24	NED
Omiyale <i>et al</i> [19]	Nephrectomy	10	NED
Johnstone <i>et al</i> [<mark>54</mark>]	Nephrectomy	NA	NA
Mehta et al[4]	Nephrectomy	3	NED
Mehta et al[4]	Nephrectomy	12	NED
Mehta et al[4]	Nephrectomy	3	NED
Tran and Pernicone[55]	Nephrectomy	NA	NA
Zhang et al[56]	Partial nephrectomy	16	NED
Cheon et al[57]	Nephrectomy	6	NED
Chou et al[58]	Nephrectomy	8	NED
Chou et al[58]	Nephrectomy	14	NED

DFUD: Died from unrelated disease; NED: No evidence of disease; NA: Not available.

Primary renal angiosarcomas are highly aggressive neoplasms with 76.3% of patients dying of tumour within 1 mo to 24 mo (mean, 7.5 mo), despite surgical and adjuvant therapy (Table 2). Poor prognostic factors for angiosarcomas occurring at other anatomic sites include age > 69 years, tumour size \ge 5 cm, regional disease (vs localized disease), non-surgical treatment, and distant metastasis[40].

CONCLUSION

Primary vascular tumours of the kidney are rare neoplasms. Unlike primary renal



Baisbideng® WJCO https://www.wjgnet.com

Table 2 Treatment, follow-up, and outcome of patients with primary angiosarcoma of the kidney					
Ref.	Treatment	Follow-up (mo)	Outcome		
Costero-Barrios et al[59]	Nephrectomy, Chemo, RT	12	AWD		
Peters <i>et al</i> [60]	Nephrectomy	2	DOD		
Singh et al[<mark>35</mark>]	NA	NA	NA		
Kern <i>et al</i> [61]	Nephrectomy	3	DOD		
Kern <i>et al</i> [61]	Nephrectomy	1.5	DOD		
Aydogdu <i>et al</i> [62]	Nephrectomy	NA	NA		
Akkad et al[63]	Nephrectomy	30	NED		
Witczak <i>et al</i> [64]	nephrectomy	NA	NA		
Chaabouni et al[38]	Nephrectomy	1	DOD		
Johnson <i>et al</i> [65]	Rapid deterioration	NA	DOD		
Zenico <i>et al</i> [66]	Nephrectomy	4	DOD		
Nguyen <i>et al</i> [67]	Nephrectomy, Chemo	18	DOD		
Terris <i>et al</i> [68]	Nephrectomy, RT	10	DOD		
Matter et al[69]	Nephrectomy, Chemo, RT	18	DOD		
Yoshida et al[70]	Nephrectomy, Recombinant IL-2	13	DOD		
Pauli and Strutton[71]	Nephrectomy, RT	2	DOD		
Martínez-Piñeiro et al[72]	Nephrectomy, S	4	DOD		
Bernstein et al ^[73]	NA	NA	NA		
Liu et al[<mark>36</mark>]	Nephrectomy, RT	6	NED		
Yau et al[74]	Nephrectomy, Chemo, RT	3	DOD		
Carnero López et al[75]	Nephrectomy, Chemo	5	DOD		
Kazaz et al <mark>[76]</mark>	Nephrectomy, Chemo	NA	NA		
Souza et al[77]	Nephrectomy	1	DFUD		
Detorakis et al[78]	Nephrectomy, Chemo	11	DOD		
Komoto et al[79]	Nephrectomy	9.2	DOD		
Boni et al[80]	Nephrectomy, Chemo	15	DOD		
Chang et al[81]	Nephrectomy, Chemo, RT	NA	NA		
Iannaci <i>et al</i> [<mark>82</mark>]	Nephrectomy	NA	DOD		
Subramanian et al[83]	Nephrectomy	NA	NA		
Waqas et al[84]	Nephrectomy, Chemo	NA	NA		
Gourley et al[85]	Nephrectomy	NA	DOD		
Su[<mark>86</mark>]	Nephrectomy, Chemo	NA	DOD		
López Cubillana et al[87]	Nephrectomy, Chemo	5	DOD		
Juan et al[<mark>88</mark>]	Nephrectomy, Chemo, RT	9	DOD		
Prince[21]	Nephrectomy, RT	NA	A and W		
Sesar et al[22]	Nephroureterectomy	NA	NA		
Testa et al[23]	Nephrectomy	27	DFUD		
Xuan[24]	Nephrectomy	NA	NA		
Brown <i>et al</i> [1]	NA	6	DOD		
Brown <i>et al</i> [1]	NA	11	DOD		
Brown <i>et al</i> [1]	NA	1	DOD		



Brown <i>et al</i> [1]	NA	NA	NA
Brown <i>et al</i> [1]	NA	1	DOD
Brown <i>et al</i> [1]	Nephrectomy	NA	NA
Brown <i>et al</i> [1]	Nephrectomy	NA	NA
Brown <i>et al</i> [1]	Nephrectomy	2	DFUD
Hiratsuka et al[89]	Nephrectomy	29	NED
Adjiman et al[90]	Nephrectomy	NA	DOD
Limmer <i>et al</i> [91]	Nephrectomy	1	DOD
Darlington <i>et al</i> [92]	Nephrectomy, Chemo	12	NED
Allred et al[93]	Nephrectomy, Chemo	3	DOD
Fukunaga et al[94]	Nephrectomy	13	DOD
Desai <i>et al</i> [95]	Nephrectomy, Chemo	4	DOD
Sabharwal et al[96]	Nephrectomy, Chemo	>1	NA
Aksoy et al[29]	Nephrectomy, S	3	DOD
Heo et al[97]	Nephrectomy	NA	NA
Mordkin et al[98]	Nephrectomy, Chemo, S	NA	NA
Berretta <i>et al</i> [99]	Nephrectomy, Chemo	8	DOD
Lodhi et al[100]	Chemo	NA	AWD
Cason et al[101]	Nephrectomy, RT	10	DOD
Askari <i>et al</i> [102]	Nephrectomy	4	DOD
Guan <i>et al</i> [103]	Nephrectomy, Chemo	4	DOD
Papadimitriou et al[104]	Nephrectomy	NA	A and W
Celebi et al[105]	Nephrectomy, Chemo, TKI, VEGF Inhibitor	13	DOD
Rüb et al[106]	Nephrectomy, Chemo	12	AWD
Zhang et al[107]	Nephrectomy	NA	NA
Tsuda et al[108]	Nephrectomy	21	DOD
Grapsa et al[109]	NA	NA	NA
Li et al[37]	NA	NA	NA
Qayyum et al[110]	Palliative (patient's decision)	NA	NA
Leggio <i>et al</i> [30]	Nephrectomy	8	DOD
Garmendia et al[111]	Nephrectomy	NA	NA
Sanyal <i>et al</i> [112]	Nephrectomy, RT	24	DOD
Cerilli <i>et al</i> [113]	Nephrectomy, RT	6	DOD
Douard et al[114]	Nephrectomy	3	DOD
Yamamoto et al[115]	Nephrectomy, RT	19	NED

RT: Radiotherapy; Chemo: Chemotherapy; TKI: Tyrosine kinase inhibitor; S: Splenectomy; DOD: Died of disease; AWD: Alive with disease; A and W: Alive and well; DFUD: Died from unrelated disease; NED: No evidence of disease; NA: Not available.

angiosarcoma, the prognosis of renal anastomosing haemangioma is excellent with no evidence of recurrence or metastasis. These tumours share similar clinical, morphological and immunohistochemical features, and must be distinguished from each other. Features that favour angiosarcomas include the presence of malignant spindled and/or epithelioid cells with a variable degree of vasoformation, cytologic atypia, prominent mitotic figures, endothelial multilayering, papillary tufting, and necrosis.

 Kaishideng®
 WJCO
 https://www.wjgnet.com

December 24, 2021 Volume 12 Issue 12

REFERENCES

- Brown JG, Folpe AL, Rao P, Lazar AJ, Paner GP, Gupta R, Parakh R, Cheville JC, Amin MB. Primary vascular tumors and tumor-like lesions of the kidney: a clinicopathologic analysis of 25 cases. Am J Surg Pathol 2010; 34: 942-949 [PMID: 20534992 DOI: 10.1097/PAS.0b013e3181e4f32a]
- 2 Montgomery E, Epstein JI. Anastomosing hemangioma of the genitourinary tract: a lesion mimicking angiosarcoma. Am J Surg Pathol 2009; 33: 1364-1369 [PMID: 19606014 DOI: 10.1097/PAS.0b013e3181ad30a7]
- Sethi S, Agarwal V, Chopra P. Cavernous hemangioma of the kidney: A report of two cases and review of the literature. Urol Ann 2012; 4: 187-190 [PMID: 23248529 DOI: 10.4103/0974-7796.102674]
- Mehta V, Ananthanarayanan V, Antic T, Krausz T, Milner J, Venkataraman G, Picken MM. Primary benign vascular tumors and tumorlike lesions of the kidney: a clinicopathologic analysis of 15 cases. Virchows Arch 2012; 461: 669-676 [PMID: 23090628 DOI: 10.1007/s00428-012-1333-9]
- 5 Omiyale AO. Anastomosing hemangioma of the kidney: a literature review of a rare morphological variant of hemangioma. Ann Transl Med 2015; 3: 151 [PMID: 26244138 DOI: 10.3978/j.issn.2305-5839.2015.06.16
- Pantelides NM, Agrawal S, Mawson I, Hazell S, Gibbons N. An Anastomosing Haemangioma: A Rare Vascular Tumour Presenting as a Solid Renal Mass. Br J Med Surg Urol 2012; 5: 84-86 [DOI: 10.1016/j.bjmsu.2011.01.003
- Downes MR, Dickson BC, Cheung CC. Anastomosing haemangioma of kidney: morphologic 7 features and diagnostic considerations of an unusual vasoformative tumour. Diagn Histopathol 2014; **20**: 208-212 [DOI: 10.1016/j.mpdhp.2014.03.002]
- Chandran N, Kannan MS, Veeramani M. Renal anastomosing hemangioma: a diagnosis to ponder. 8 Indian J Transplant 2019; 13: 59-61 [DOI: 10.4103/ijot.ijot 63 18]
- 9 Cha JS, Jeong YB, Kim HJ. Anastomosing hemangioma mimicking renal cell carcinoma. Korean J Urol Oncol 2016; 14: 88-92 [DOI: 10.22465/kjuo.2016.14.2.88]
- 10 Omiyale AO, Carton J. Clinicopathological and genetic features of anastomosing haemangioma of the kidney: a narrative review. AME Med J 2021; 6: 30 [DOI: 10.21037/amj-20-181]
- 11 Büttner M, Kufer V, Brunner K, Hartmann A, Amann K, Agaimy A. Benign mesenchymal tumours and tumour-like lesions in end-stage renal disease. Histopathology 2013; 62: 229-236 [PMID: 23020314 DOI: 10.1111/j.1365-2559.2012.04349.x]
- Kryvenko ON, Haley SL, Smith SC, Shen SS, Paluru S, Gupta NS, Jorda M, Epstein JI, Amin MB, 12 Truong LD. Haemangiomas in kidneys with end-stage renal disease: a novel clinicopathological association. Histopathology 2014; 65: 309-318 [PMID: 24548339 DOI: 10.1111/his.12394]
- 13 O'Neill AC, Craig JW, Silverman SG, Alencar RO. Anastomosing hemangiomas: locations of occurrence, imaging features, and diagnosis with percutaneous biopsy. Abdom Radiol (NY) 2016; 41: 1325-1332 [PMID: 26960722 DOI: 10.1007/s00261-016-0690-2]
- 14 Bean GR, Joseph NM, Gill RM, Folpe AL, Horvai AE, Umetsu SE. Recurrent GNAQ mutations in anastomosing hemangiomas. Mod Pathol 2017; 30: 722-727 [PMID: 28084343 DOI: 10.1038/modpathol.2016.234]
- 15 Bean GR, Joseph NM, Folpe AL, Horvai AE, Umetsu SE. Recurrent GNA14 mutations in anastomosing haemangiomas. Histopathology 2018; 73: 354-357 [PMID: 29574926 DOI: 10.1111/his.13519
- 16 Urtatiz O, Van Raamsdonk CD. Gnaq and Gna11 in the Endothelin Signaling Pathway and Melanoma. Front Genet 2016; 7: 59 [PMID: 27148356 DOI: 10.3389/fgene.2016.00059]
- Caballes AB, Abelardo AD, Farolan MJ, Veloso JAD. Pediatric Anastomosing Hemangioma: Case 17 Report and Review of Renal Vascular Tumors in Children. Pediatr Dev Pathol 2019; 22: 269-275 [PMID: 30369288 DOI: 10.1177/1093526618809230]
- Kryvenko ON, Gupta NS, Meier FA, Lee MW, Epstein JI. Anastomosing hemangioma of the 18 genitourinary system: Eight cases in the kidney and ovary with immunohistochemical and ultrastructural analysis. Am J Clin Pathol 2011; 136: 450-457 [PMID: 21846922 DOI: 10.1309/AJCPJPW34QCQYTMT]
- Omiyale AO, Golash A, Mann A, Kyriakidis D, Kalyanasundaram K. Anastomosing Haemangioma 19 of the Kidney Involving a Segmental Branch of the Renal Vein. Case Rep Surg 2015; 2015: 927286 [PMID: 26435872 DOI: 10.1155/2015/927286]
- 20 Omiyale AO, Carton J. Clinical and Pathologic Features of Primary Angiosarcoma of the Kidney. Curr Urol Rep 2018; 19: 4 [PMID: 29383452 DOI: 10.1007/s11934-018-0755-6]
- Prince CL. Primary angio-endothelioma of the kidney: report of a case and brief review. J Urol 21 1942; 47: 787-789 [DOI: 10.1016/S0022-5347%2817%2970875-0]
- Sesar P, Ulamec M, Šoša D, Trnski D, Tomas D. Primary renal angiosarcoma. Acta Clin Croat 22 2012; 51: 182
- 23 Testa G, Talamona G, Tufano A, Marino-Marsilia G. Primary renal angiosarcoma: a case report. Acta Urologica Italica 1998; 12: 225-227
- Xuan Y. Primary renal angiosarcoma: one case report and literature review. Chin J Clin Oncol 2008; 24 5: 229-230 [DOI: 10.1007/s11805-008-0229-6]
- Penel N, Grosjean J, Robin YM, Vanseymortier L, Clisant S, Adenis A. Frequency of certain 25 established risk factors in soft tissue sarcomas in adults: a prospective descriptive study of 658 cases.



Sarcoma 2008; 2008: 459386 [PMID: 18497869 DOI: 10.1155/2008/459386]

- 26 Kulaga A, Yilmaz A, Wilkin RP, Trpkov K. Epithelioid angiosarcoma of the bladder after irradiation for endometrioid adenocarcinoma. Virchows Arch 2007; 450: 245-246 [PMID: 17149614 DOI: 10.1007/s00428-006-0336-91
- 27 Chandan VS, Wolsh L. Postirradiation angiosarcoma of the prostate. Arch Pathol Lab Med 2003; 127: 876-878 [PMID: 12823047 DOI: 10.5858/2003-127-876-PAOTP]
- 28 Omiyale AO. Clinicopathological features of primary angiosarcoma of the kidney: a review of 62 cases. Transl Androl Urol 2015; 4: 464-473 [PMID: 26816844 DOI: 10.3978/j.issn.2223-4683.2015.05.04]
- 29 Aksoy Y, Gürsan N, Ozbey I, Biçgi O, Keleş M. Spontaneous rupture of a renal angiosarcoma. Urol Int 2002; 68: 60-62 [PMID: 11803270 DOI: 10.1159/000048419]
- 30 Leggio L, Addolorato G, Abenavoli L, Ferrulli A, D'Angelo C, Mirijello A, Vonghia L, Schinzari G, Arena V, Perrone L, Citterio F, Bonomo L, Rapaccini GL, Capelli A, Barone C, Gasbarrini G. Primary renal angiosarcoma: a rare malignancy. A case report and review of the literature. Urol Oncol 2006; 24: 307-312 [PMID: 16818182 DOI: 10.1016/j.urolonc.2005.10.002]
- 31 Antonescu CR, Yoshida A, Guo T, Chang NE, Zhang L, Agaram NP, Qin LX, Brennan MF, Singer S, Maki RG. KDR activating mutations in human angiosarcomas are sensitive to specific kinase inhibitors. Cancer Res 2009; 69: 7175-7179 [PMID: 19723655 DOI: 10.1158/0008-5472.CAN-09-2068
- Guo T, Zhang L, Chang NE, Singer S, Maki RG, Antonescu CR. Consistent MYC and FLT4 gene 32 amplification in radiation-induced angiosarcoma but not in other radiation-associated atypical vascular lesions. Genes Chromosomes Cancer 2011; 50: 25-33 [PMID: 20949568 DOI: 10.1002/gcc.20827]
- 33 Behjati S, Tarpey PS, Sheldon H, Martincorena I, Van Loo P, Gundem G, Wedge DC, Ramakrishna M, Cooke SL, Pillay N, Vollan HKM, Papaemmanuil E, Koss H, Bunney TD, Hardy C, Joseph OR, Martin S, Mudie L, Butler A, Teague JW, Patil M, Steers G, Cao Y, Gumbs C, Ingram D, Lazar AJ, Little L, Mahadeshwar H, Protopopov A, Al Sannaa GA, Seth S, Song X, Tang J, Zhang J, Ravi V, Torres KE, Khatri B, Halai D, Roxanis I, Baumhoer D, Tirabosco R, Amary MF, Boshoff C, McDermott U, Katan M, Stratton MR, Futreal PA, Flanagan AM, Harris A, Campbell PJ. Recurrent PTPRB and PLCG1 mutations in angiosarcoma. Nat Genet 2014; 46: 376-379 [PMID: 24633157 DOI: 10.1038/ng.2921]
- 34 Huang SC, Zhang L, Sung YS, Chen CL, Kao YC, Agaram NP, Singer S, Tap WD, D'Angelo S, Antonescu CR, Recurrent CIC Gene Abnormalities in Angiosarcomas: A Molecular Study of 120 Cases With Concurrent Investigation of PLCG1, KDR, MYC, and FLT4 Gene Alterations. Am J Surg Pathol 2016; 40: 645-655 [PMID: 26735859 DOI: 10.1097/PAS.000000000000882]
- 35 Singh C, Xie L, Schmechel SC, Manivel JC, Pambuccian SE. Epithelioid angiosarcoma of the kidney: a diagnostic dilemma in fine-needle aspiration cytology. Diagn Cytopathol 2012; 40 Suppl 2: E131-E139 [PMID: 21698784 DOI: 10.1002/dc.21762]
- Liu H, Huang X, Chen H, Wang X, Chen L. Epithelioid angiosarcoma of the kidney: A case report 36 and literature review. Oncol Lett 2014; 8: 1155-1158 [PMID: 25120677 DOI: 10.3892/ol.2014.2292]
- Li N, Li W, Li Z. Primary renal epithelioid angiosarcoma with transitional cell carcinoma in renal 37 pelvis. Zhonghua Waike Zazhi 1997; 35: 294-295 [PMID: 10374567]
- Chaabouni A, Rebai N, Chabchoub K, Fourati M, Bouacida M, Slimen MH, Bahloul A, Mhiri MN. 38 Primary renal angiosarcoma: Case report and literature review. Can Urol Assoc J 2013; 7: E430-E432 [PMID: 23826057 DOI: 10.5489/cuaj.1396]
- 39 Iacovelli R, Orlando V, Palazzo A, Cortesi E. Clinical and pathological features of primary renal angiosarcoma. Can Urol Assoc J 2014; 8: E223-E226 [PMID: 24839487 DOI: 10.5489/cuaj.1585]
- 40 Zhang C, Xu G, Liu Z, Xu Y, Lin F, Baklaushev VP, Chekhonin VP, Peltzer K, Mao M, Wang X, Wang G, Zhang C. Epidemiology, tumor characteristics and survival in patients with angiosarcoma in the United States: a population-based study of 4537 cases. Jpn J Clin Oncol 2019; 49: 1092-1099 [PMID: 31612920 DOI: 10.1093/jjco/hyz113]
- Memmedoğlu A, Musayev J. Spontaneous rupture of the kidney in the patients with synchronous 41 renal hemangioma and nephrogenic hypertension. Turk J Urol 2015; 41: 231-234 [PMID: 26623154 DOI: 10.5152/tud.2015.48264]
- Tahir M, Folwell A. Anastomosing haemangioma of kidney: a rare subtype of vascular tumour of 42 the kidney mimicking angiosarcoma. ANZ J Surg 2016; 86: 838-839 [PMID: 25041271 DOI: 10.1111/ans.12779]
- Heidegger I, Pichler R, Schäfer G, Zelger B, Aigner F, Bektic J, Horninger W. Long-term follow up 43 of renal anastomosing hemangioma mimicking renal angiosarcoma. Int J Urol 2014; 21: 836-838 [PMID: 24650180 DOI: 10.1111/iju.12433]
- 44 Al-Maghrabi HA, Al Rashed AS. Challenging Pitfalls and Mimickers in Diagnosing Anastomosing Capillary Hemangioma of the Kidney: Case Report and Literature Review. Am J Case Rep 2017; 18: 255-262 [PMID: 28286335 DOI: 10.12659/ajcr.902939]
- Lee HS, Koh BH, Kim JW, Kim YS, Rhim HC, Cho OK, Hahm CK, Woo YN, Park MH. 45 Radiologic findings of renal hemangioma: report of three cases. Korean J Radiol 2000; 1: 60-63 [PMID: 11752931 DOI: 10.3348/kjr.2000.1.1.60]
- 46 Zhao M, Li C, Zheng J, Sun K. Anastomosing hemangioma of the kidney: a case report of a rare subtype of hemangioma mimicking angiosarcoma and review of the literature. Int J Clin Exp Pathol 2013; 6: 757-765 [PMID: 23573324]



- 47 Tao LL, Dai Y, Yin W, Chen J. A case report of a renal anastomosing hemangioma and a literature review: an unusual variant histologically mimicking angiosarcoma. Diagn Pathol 2014; 9: 159 [PMID: 25102914 DOI: 10.1186/s13000-014-0159-y]
- 48 Abboudi H, Tschobotko B, Carr C, DasGupta R. Bilateral Renal Anastomosing Hemangiomas: A Tale of Two Kidneys. J Endourol Case Rep 2017; 3: 176-178 [PMID: 29279869 DOI: 10.1089/cren.2017.0018]
- 49 Silva MA, Fonseca EKUN, Yamauchi FI, Baroni RH. Anastomosing hemangioma simulating renal cell carcinoma. Int Braz J Urol 2017; 43: 987-989 [PMID: 28727378 DOI: 10.1590/S1677-5538.IBJU.2016.0653]
- Berker NK, Bayram A, Tas S, Bakir B, Caliskan Y, Ozcan F, Kilicaslan I, Ozluk Y. Comparison of 50 Renal Anastomosing Hemangiomas in End-Stage and Non-End-Stage Kidneys: A Meta-Analysis With a Report of 2 Cases. Int J Surg Pathol 2017; 25: 488-496 [PMID: 28436289 DOI: 10.1177/1066896917706025
- Perdiki M, Datseri G, Liapis G, Chondros N, Anastasiou I, Tzardi M, Delladetsima JK, Drakos E. 51 Anastomosing hemangioma: report of two renal cases and analysis of the literature. *Diagn Pathol* 2017; 12: 14 [PMID: 28118845 DOI: 10.1186/s13000-017-0597-4]
- 52 Wetherell DR, Skene A, Manya K, Manecksha RP, Chan Y, Bolton DM. Anastomosing haemangioma of the kidney: a rare morphological variant of haemangioma characteristic of genitourinary tract location. Pathology 2013; 45: 193-196 [PMID: 23250038 DOI: 10.1097/PAT.0b013e32835c782b]
- 53 Manohar V, Krishnamurthy S, Ranganathan J, Pai VD. A case of giant anastomosing hemangioma of the kidney with extramedullary hematopoiesis: A great mimicker. Indian J Pathol Microbiol 2020; 63: 292-294 [PMID: 32317537 DOI: 10.4103/IJPM.IJPM_434_18]
- Johnstone KJ, Strutton GM, Perry-Keene JL, Hazratwala K, Delahunt B. Multifocal anastomosing 54 haemangioma of the kidney with intravascular growth and sinus fat invasion: a rare benign mimic of angiosarcoma. Pathology 2020; 52: 394-396 [PMID: 32111399 DOI: 10.1016/j.pathol.2020.01.681]
- 55 Tran TA, Pernicone P. Anastomosing hemangioma with fatty changes of the genitourinary tract: a lesion mimicking angiomyolipoma. Cent European J Urol 2012; 65: 40-42 [PMID: 24578924 DOI: 10.5173/ceju.2012.01.art13
- 56 Zhang W, Wang Q, Liu YL, Yu WJ, Liu Y, Zhao H, Zhuang J, Jiang YX, Li YJ. Anastomosing hemangioma arising from the kidney: a case of slow progression in four years and review of literature. Int J Clin Exp Pathol 2015; 8: 2208-2213 [PMID: 25973131]
- Cheon PM, Rebello R, Naqvi A, Popovic S, Bonert M, Kapoor A. Anastomosing hemangioma of 57 the kidney: radiologic and pathologic distinctions of a kidney cancer mimic. Curr Oncol 2018; 25: e220-e223 [PMID: 29962849 DOI: 10.3747/co.25.3927]
- Chou S, Subramanian V, Lau HM, Achan A. Renal Anastomosing Hemangiomas With a Diverse 58 Morphologic Spectrum: Report of Two Cases and Review of Literature. Int J Surg Pathol 2014; 22: 369-373 [PMID: 23816823 DOI: 10.1177/1066896913492850]
- Costero-Barrios CB, Oros-Ovalle C. Primary renal angiosarcoma. Gac Med Mex 2004; 140: 463-59 466 [PMID: 15456157]
- 60 Peters HJ, Nuri M, Münzenmaier R. Hemangioendothelioma of the kidney: a case report and review of the literature. J Urol 1974; 112: 723-726 [PMID: 4548091 DOI: 10.1016/s0022-5347(17)59836-5]
- Kern SB, Gott L, Faulkner J 2nd. Occurrence of primary renal angiosarcoma in brothers. Arch 61 Pathol Lab Med 1995; 119: 75-78 [PMID: 7802558]
- Aydogdu I, Turhan O, Sari R, Ates M, Türk M. Coincidental acute myeloblastic leukemia in a 62 patient with renal angiosarcoma. Haematologia (Budap) 1999; 29: 313-317 [PMID: 10438072]
- 63 Akkad T, Tsankov A, Pelzer A, Peschel R, Bartsch G, Steiner H. Early diagnosis and straight forward surgery of an asymptomatic primary angiosarcoma of the kidney led to long-term survival. Int J Urol 2006; 13: 1112-1114 [PMID: 16903939 DOI: 10.1111/j.1442-2042.2006.01490.x]
- Witczak W, Szubstarski F, Szymański C, Rajca Z, Staszczyk S. Renal hemangiosarcoma. Pol Tyg 64 Lek 1993; 48: 483-484 [PMID: 8170817]
- Johnson VV, Gaertner EM, Crothers BA. Fine-needle aspiration of renal angiosarcoma. Arch Pathol 65 Lab Med 2002; 126: 478-480 [PMID: 11900578 DOI: 10.5858/2002-126-0478-FNAORA]
- Zenico T, Saccomanni M, Salomone U, Bercovich E. Primary renal angiosarcoma: case report and 66 review of world literature. *Tumori* 2011; 97: e6-e9 [PMID: 21989448 DOI: 10.1700/950.10412]
- 67 Nguyen T, Auquier MA, Renard C, Cordonnier C, Saint F, Remond A. Hemoptysis and spontaneous rupture of a primary renal angiosarcoma: a case report. J Radiol 2010; 91: 1313-1317 [PMID: 21242919 DOI: 10.1016/s0221-0363(10)70201-8]
- 68 Terris D, Plaine L, Steinfeld A. Renal angiosarcoma. Am J Kidney Dis 1986; 8: 131-133 [PMID: 3740062 DOI: 10.1016/s0272-6386(86)80126-3]
- 69 Matter LE, Flury R, Hailemariam S, Hauri D, Sulser T, Angiosarcoma of the kidney. Case report and review of the literature. Urologe A 1999; 38: 65-68 [PMID: 10081105 DOI: 10.1007/s001200050248]
- Yoshida K, Ito F, Nakazawa H, Maeda Y, Tomoe H, Aiba M. A case of primary renal 70 angiosarcoma. Rare Tumors 2009; 1: e28 [PMID: 21139907 DOI: 10.4081/rt.2009.e28]
- Pauli JL, Strutton G. Primary renal angiosarcoma. Pathology 2005; 37: 187-189 [PMID: 16028856 71 DOI: 10.1080/003130205000589791
- Martínez-Piñeiro L, López-Ferrer P, Picazo ML, Martínez-Piñeiro JA. Primary renal angiosarcoma. 72



Case report and review of the literature. Scand J Urol Nephrol 1995; 29: 103-108 [PMID: 7618041 DOI: 10.3109/00365599509180547]

- 73 Bernstein ML, Leclerc JM, Bunin G, Brisson L, Robison L, Shuster J, Byrne T, Gregory D, Hill G, Dougherty G. A population-based study of neuroblastoma incidence, survival, and mortality in North America. J Clin Oncol 1992; 10: 323-329 [PMID: 1732433 DOI: 10.1200/JCO.1992.10.2.323]
- 74 Yau T, Leong CH, Chan WK, Chan JK, Liang RH, Epstein RJ. A case of mixed adult Wilms' tumour and angiosarcoma responsive to carboplatin, etoposide and vincristine (CEO). Cancer Chemother Pharmacol 2008; 61: 717-720 [PMID: 17571263 DOI: 10.1007/s00280-007-0529-1]
- Carnero López B, Fernández Pérez I, Carrasco Alvarez JA, Lázaro Quintela ME, López Jato C, 75 Jorge Fernández M, Gentil González M, Vázquez Tuñas L, Castellanos Díez J. Renal primary angiosarcoma. Clin Transl Oncol 2007; 9: 806-810 [PMID: 18158986 DOI: 10.1007/s12094-007-0144-y
- 76 Kazaz IO, Ersoz S, Colak F, Teoman AS, Kazaz SN, Karaguzel E, Kutlu O. Primary renal angiosarcoma: A case report and a short review of literature. Indian J Pathol Microbiol 2020; 63: S44-S46 [PMID: 32108626 DOI: 10.4103/IJPM.IJPM_66_19]
- Souza OE, Etchebehere RM, Lima MA, Monti PR. Primary renal angiosarcoma. Int Braz J Urol 77 2006; **32**: 448-450 [PMID: 16953913 DOI: 10.1590/s1677-55382006000400011]
- 78 Detorakis EE, Chryssou E, Raissaki M, Androulidakis E, Heretis I, Haniotis V, Karantanas A. Primary renal angiosarcoma: radiologic-pathologic correlation and literature review. Tumori 2013; 99: e111-e116 [PMID: 24158078 DOI: 10.1700/1334.14817]
- Komoto H, Kitajima K, Kawanaka Y, Yoshimura N, Kunimoto R, Yokoyama H, Shinkai Y, 79 Kaizuka Y, Yamamoto S, Kihara T, Kimura N, Hirota S, Yamakado K. CT Findings of Primary Renal Angiosarcoma. Case Rep Oncol 2021; 14: 212-216 [PMID: 33776706 DOI: 10.1159/000512015]
- 80 Boni A, Cochetti G, Sidoni A, Bellezza G, Lepri E, Giglio A, Turco M, Vermandois JAR, Zingaro MD, Cirocchi R, Mearini E. Primary Angiosarcoma of the Kidney: Case Report and Comprehensive Literature Review. Open Med (Wars) 2019; 14: 443-455 [PMID: 31410364 DOI: 10.1515/med-2019-0048
- Chang CW, Chien CC, Juan YS, Chueh KS. Primary renal angiosarcoma mimicking urothelial 81 carcinoma - A case report and literature reviews. Urol Case Rep 2021; 34: 101407 [PMID: 33145170 DOI: 10.1016/j.eucr.2020.101407]
- 82 Iannaci G, Crispino M, Cifarelli P, Montella M, Panarese I, Ronchi A, Russo R, Tremiterra G, Luise R, Sapere P. Epithelioid angiosarcoma arising in schwannoma of the kidney: report of the first case and review of the literature. World J Surg Oncol 2016; 14: 29 [PMID: 26842370 DOI: 10.1186/s12957-016-0789-5
- 83 Subramanian H, Parepalli D, Srinivas BH, Gochhait D, Ks S. Primary Renal Angiosarcoma With Lymph Node Metastasis- A Rare Intriguing Malignancy With a Grim Outcome. Urology 2021; 153: 14-16 [PMID: 33621584 DOI: 10.1016/j.urology.2021.02.016]
- Waqas M, Rahim W, Shohab D, Khawaja MA, Ali Z, Mamoon N. Primary Renal Epithelioid Angiosarcoma. J Coll Physicians Surg Pak 2018; 28: S66-S68 [PMID: 29482713 DOI: 10.29271/jcpsp.2018.03.S66
- 85 Gourley E, Digman G, Nicolas M, Kaushik D. Primary renal angiosarcoma. BMJ Case Rep 2018; 2018 [PMID: 30061122 DOI: 10.1136/bcr-2017-222672]
- Su YC. Angiosarcoma of kidney with calcification: A rare case report. Kaohsiung J Med Sci 2017; 33: 367-368 [PMID: 28738979 DOI: 10.1016/j.kjms.2017.03.009]
- 87 López Cubillana P, Martínez Barba E, Server Pastor G, Prieto González A, Ferri Níguez B, Cao Avellaneda E, López López AI, Nicolás Torroalba JA, Rigabert Montiel M, Tornero Ruiz J, García Hernández JA, Gómez Gómez G, Martínez Pertusa P, Guardiola Mas A, Pérez Albacete M. Fatal evolution of a renal angiosarcoma. Arch Esp Urol 2004; 57: 425-426 [PMID: 15270285]
- 88 Juan CJ, Yu CY, Hsu HH, Chian CP, Huang GS, Fan HC, Lin CC, Ching Jiunn Wu JW, Hsiao HS, Chen CY. Visceral and non-visceral angiosarcoma: imaging features and clinical correlation. Chin J Radiol 2000; 25: 183-189
- 89 Hiratsuka Y, Nishimura H, Kajiwara I, Matsuoka H, Kawamura K. Renal angiosarcoma: a case report. Int J Urol 1997; 4: 90-93 [PMID: 9179675 DOI: 10.1111/j.1442-2042.1997.tb00148.x]
- 90 Adjiman S, Zerbib M, Flam T, Brochard M, Deslignères S, Boissonnas A, Debré B, Steg A. Genitourinary tumors and HIV1 infection. Eur Urol 1990; 18: 61-63 [PMID: 2401309 DOI: 10.1159/000463869]
- 91 Limmer S, Wagner T, Leipprand E, Arnholdt H. Primary renal hemangiosarcoma. Case report and review of the literature. Pathologe 2001; 22: 343-348 [PMID: 11572116 DOI: 10.1007/s002920100481
- 92 Darlington D, Anitha FS. Primary Renal Angiosarcoma Mimicking Renal Cell Carcinoma: A Case Report. Cureus 2019; 11: e3841 [PMID: 30891382 DOI: 10.7759/cureus.3841]
- 93 Allred CD, Cathey WJ, McDivitt RW. Primary fenal angiosarcoma: a case report. Hum Pathol 1981; 12: 665-668 [PMID: 7196879 DOI: 10.1016/s0046-8177(81)80054-8]
- Fukunaga M. Angiosarcoma of the kidney with minute clear cell carcinomas: a case report. Pathol 94 Res Pract 2009; 205: 347-351 [PMID: 19147300 DOI: 10.1016/j.prp.2008.11.005]
- Desai MB, Chess Q, Naidich JB, Weiner R. Primary renal angiosarcoma mimicking a renal cell 95 carcinoma. Urol Radiol 1989; 11: 30-32 [PMID: 2734970 DOI: 10.1007/BF02926469]
- Sabharwal S, John NT, Kumar RM, Kekre NS. Primary renal angiosarcoma. Indian J Urol 2013; 96



29: 145-147 [PMID: 23956520 DOI: 10.4103/0970-1591.114040]

- 97 Heo SH, Shin SS, Kang TW, Kim GE. Primary renal angiosarcoma with extensive hemorrhage: CT and MRI findings. Int Braz J Urol 2019; 45: 402-405 [PMID: 30735338 DOI: 10.1590/S1677-5538.IBJU.2018.0375]
- 98 Mordkin RM, Dahut WL, Lynch JH. Renal angiosarcoma: a rare primary genitourinary malignancy. South Med J 1997; 90: 1159-1160 [PMID: 9386065 DOI: 10.1097/00007611-199711000-00023]
- Berretta M, Rupolo M, Buonadonna A, Canzonieri V, Brollo A, Morra A, Berretta S, Bearz A, 99 Tirelli U, Frustaci S. Metastatic angiosarcoma of the kidney: a case report with treatment approach and review of the literature. J Chemother 2006; 18: 221-224 [PMID: 16736893 DOI: 10.1179/joc.2006.18.2.2211
- 100 Lodhi HT, Inayat F, Munir A, Ilyas G. Primary renal angiosarcoma: a diagnostic and therapeutic challenge. BMJ Case Rep 2018; 2018 [PMID: 30244222 DOI: 10.1136/bcr-2018-225484]
- Cason JD, Waisman J, Plaine L. Angiosarcoma of kidney. Urology 1987; 30: 281-283 [PMID: 101 3629777 DOI: 10.1016/0090-4295(87)90258-5]
- 102 Askari A, Novick A, Braun W, Steinmuller D. Late ureteral obstruction and hematuria from de novo angiosarcoma in a renal transplant patient. J Urol 1980; 124: 717-719 [PMID: 7005464 DOI: 10.1016/s0022-5347(17)55625-6]
- Guan H, Zhang L, Zhang Q, Qi W, Xie S, Hou J, Wang H. Primary angiosarcoma arising in an 103 angiomyolipoma of the kidney: case report and literature review. Diagn Pathol 2018; 13: 53 [PMID: 30111336 DOI: 10.1186/s13000-018-0730-z]
- 104 Papadimitriou VD, Stamatiou KN, Takos DM, Adamopoulos VM, Heretis IE, Sofras FA. Angiosarcoma of kidney: a case report and review of literature. Urol J 2009; 6: 223-225 [PMID: 197112811
- 105 Celebi F, Pilanci KN, Saglam S, Balci NC. Primary renal angiosarcoma with progressive clinical course despite surgical and adjuvant treatment: A case report. Oncol Lett 2015; 9: 1937-1939 [PMID: 25789072 DOI: 10.3892/ol.2015.2902]
- 106 Rüb J, Bauer S, Pastor J, Noldus J, Palisaar RJ. Primary renal angiosarcoma. Uncommon manifestation of a rare malignancy. Urologe A 2015; 54: 849-853 [PMID: 25805159 DOI: 10.1007/s00120-015-3809-0]
- Zhang HM, Yan Y, Luo M, Xu YF, Peng B, Zheng JH. Primary angiosarcoma of the kidney: case 107 analysis and literature review. Int J Clin Exp Pathol 2014; 7: 3555-3562 [PMID: 25120734]
- 108 Tsuda N, Chowdhury PR, Hayashi T, Anami M, Iseki M, Koga S, Matsuya F, Kanetake H, Saito Y, Horita Y. Primary renal angiosarcoma: a case report and review of the literature. Pathol Int 1997; 47: 778-783 [PMID: 9413038 DOI: 10.1111/j.1440-1827.1997.tb04457.x]
- Grapsa D, Sakellariou S, Politi E. Fine-needle aspiration cytology of primary renal angiosarcoma 109 with histopathologic and immunocytochemical correlation: a case report. Diagn Cytopathol 2014; 42: 872-876 [PMID: 24166896 DOI: 10.1002/dc.23051]
- 110 Qayyum S, Parikh JG, Zafar N. Primary renal angiosarcoma with extensive necrosis: a difficult diagnosis. Case Rep Pathol 2014; 2014: 416170 [PMID: 25133004 DOI: 10.1155/2014/416170]
- 111 Garmendia JC, López García JA, Acinas García O, Garrido Rivas C, Sanroma Ortueta C, Arocena Lanz F. Angiosarcoma of the kidney. Actas Urol Esp 1989; 13: 223-224 [PMID: 2763892]
- 112 Sanyal B, Mehrotra ML, Gupta S, Pant GC. Radiotherapy in renal angiosarcoma. J Indian Med Assoc 1979; 72: 85-86 [PMID: 574526]
- 113 Cerilli LA, Huffman HT, Anand A. Primary renal angiosarcoma: a case report with immunohistochemical, ultrastructural, and cytogenetic features and review of the literature. Arch Pathol Lab Med 1998; 122: 929-935 [PMID: 9786357]
- 114 Douard A, Pasticier G, Deminière C, Wallerand H, Ferrière JM, Bernhard JC. Primary angiosarcoma of the kidney: case report and literature review. Prog Urol 2012; 22: 438-441 [PMID: 22657265 DOI: 10.1016/j.purol.2011.11.006]
- Yamamoto Y, Izaki H, Harada A, Taue R, Kishimoto T, Tanimoto S, Fukumori T, Takahashi M, 115 Nishitani MA, Kanayama HO, Sano N, Uema K. A case of renal capsular hemangiosarcoma. Hinyokika Kiyo 2006; 52: 215-217 [PMID: 16617877]



World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1169-1181

DOI: 10.5306/wjco.v12.i12.1169

ISSN 2218-4333 (online)

MINIREVIEWS

Detection of circulating tumour cells in colorectal cancer: Emerging techniques and clinical implications

Alka Yadav, Ashok Kumar, Mohammad Haris Siddiqui

ORCID number: Alka Yadav 0000-0002-7162-7586; Ashok Kumar 0000-0002-6222-7337; Mohammad Haris Siddiqui 0000-0002-6547-7905.

Author contributions: Yadav A wrote the paper; Kumar A conceptualized, corrected, and finalized the manuscript; Siddiqui MH helped in language polishing, editing, and correction in revising the manuscript.

Conflict-of-interest statement: The authors declare no conflicts of interest for this article.

Country/Territory of origin: India

Specialty type: Oncology

Provenance and peer review: Invited article; Externally peer reviewed.

Peer-review report's scientific quality classification

Grade A (Excellent): 0 Grade B (Very good): 0 Grade C (Good): C Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution

Alka Yadav, Ashok Kumar, Department of Surgical Gastroenterology, Sanjay Gandhi Post Graduate Institute of Medical Sciences, Lucknow 226014, Uttar Pradesh, India

Mohammad Haris Siddiqui, Department of Bioengineering, Integral University, Lucknow 226026, India

Corresponding author: Ashok Kumar, FRCS, MS, Professor, Department of Surgical Gastroenterology, Sanjay Gandhi Post Graduate Institute of Medical Sciences, Rae Bareli Road, Lucknow 226014, Uttar Pradesh, India. doc.ashokgupta@gmail.com

Abstract

Despite several advances in oncological management of colorectal cancer, morbidity and mortality are still high and devastating. The diagnostic evaluation by endoscopy is cumbersome, which is uncomfortable to many. Because of the intra- and inter-tumour heterogeneity and changing tumour dynamics, which is continuous in nature, the diagnostic biopsy and assessment of the pathological sample are difficult and also not adequate. Late manifestation of the disease and delayed diagnosis may lead to relapse or metastases. One of the keys to improving the outcome is early detection of cancer, ease of technology to detect with uniformity, and its therapeutic implications, which are yet to come. "Liquid biopsy" is currently the most recent area of interest in oncology, which may provide important tools regarding the characterization of the primary tumour and its metastasis as cancer cells shed into the bloodstream even at the early stages of the disease. By using this approach, clinicians may be able to find out information about the tumour at a given time. Any of the following three types of sampling of biological material can be used in the "liquid biopsy". These are circulating tumour cells (CTCs), circulating tumour DNA, and exosomes. The most commonly studied amongst the three is CTCs. CTCs with their different applications and prognostic value has been found useful in colorectal cancer detection and therapeutics. In this review, we will discuss various markers for CTCs, the core tools/techniques for detection, and also important findings of clinical studies in colorectal cancer and its clinical implications.

Key Words: Circulating tumour cells; Colorectal cancer; Tools and techniques; Clinical implications

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.



NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt p://creativecommons.org/License s/by-nc/4.0/

Received: April 1, 2021 Peer-review started: April 1, 2021 First decision: July 6, 2021 Revised: July 15, 2021 Accepted: November 15, 2021 Article in press: November 15, 2021 Published online: December 24, 2021

P-Reviewer: Guadagni S S-Editor: Fan JR L-Editor: Wang TQ P-Editor: Fan JR



Core Tip: Circulating tumour cells (CTCs) in the blood have been found to be mainly associated with the stage of the disease and serve as a prognostic marker for survival in colorectal cancer. Some studies have also reported its role in the diagnosis and treatment monitoring. By focusing molecular research on rare CTCs, targeting cellular markers of CTCs, and discovering new cellular markers may improve the management of colorectal cancer and play a role in prevention of metastatic disease. Patients at high risk might benefit from additional individualized treatment which can be investigated in future clinical trials.

Citation: Yadav A, Kumar A, Siddiqui MH. Detection of circulating tumour cells in colorectal cancer: Emerging techniques and clinical implications. World J Clin Oncol 2021; 12(12): 1169-1181

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1169.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1169

INTRODUCTION

Colorectal cancer (CRC) is one of the most commonly diagnosed cancers, which stands second and third in women and men, respectively, across the globe with more than 1.2 \times 10⁶ new cases and 608700 mortalities annually[1]. It develops due to genetic and epigenetic alterations in human genome and environmental factors. Mode of presentation of CRC can be inherited, familial, and sporadic. Inherited CRC accounts for 5%-10% of all cases, for example, Lynch syndrome, familial adenomatous polyposis, and Peutz-Jeghers syndrome. Among all the CRCs, familial CRC accounts for 20%-30% and sporadic cases approximately 70% of all CRCs which are associated with somatic mutations^[2]. There are many invasive and non-invasive diagnostic and prognostic tools with varying sensitivity and specificity, and each has its limitation. There is a need for new tools which may be simpler, non-invasive, cheaper, reproducible, and easily available with high sensitivity and specificity. "Liquid biopsy" is currently the most recent area of interest in oncology, which may provide important tools regarding the characterization of the primary tumour as well as metastasis because tumour cells shed into the bloodstream at the early stages of the disease. In "Liquid biopsy", one of the three types of sampling of biological material can be used, which are-circulating tumour cells (CTCs), circulating tumour DNA, and exosomes. CTCs are one of the main components of liquid biopsy, where subsets of tumour cells can disseminate from the primary tumour and intravasate to the circulatory system. CTCs are non-invasive and safe in comparison to traditional tissue biopsy, and can be used for monitoring of tumour progression and tumour response to therapy in real time. CTCs in peripheral blood serve as a source of valuable tumour markers. The present review will describe the main areas of the ongoing investigation on CTCs with particular emphasis on different tools and techniques used for CTC capturing and analysis, and also currently available data of clinical relevance of CTCs.

CTCs

A tumour cell contains millions of cells maintaining genetic mutations driving them to grow, divide, and invade the local tissues. Some cells separate from the edges of a tumour and are released into the bloodstream or lymphatic system. These cells are CTCs. CTCs can also be defined as cells spreading into vasculature by a primary tumour and they keep circulating in the bloodstream of cancer patients[3]. It was Ashworth (1869) who reported the CTCs for the first time and described the presence of tumour cells with resemblance to the cells from the primary tumour, in the blood of a patient with metastatic breast carcinoma. Later, in 1955, evidence of the presence of CTCs in the blood of a patient with primary and metastatic carcinoma was found by immunohistochemistry. In 1990, Moss and Sanders in their study found evidence for CTCs in seven out of ten disseminated neuroblastoma patients by immunostaining. In CRC, CTCs were first reported in 1993 with the help of conventional cytology and cytokeratin staining. Tumour cells were isolated from 42 patients who underwent

resection with the help of density gradient centrifugation, immune histological evidence for CTCs was reported in 4 out of 42 patients. Above mentioned studies have showed that tumour cells could be detected by traditional immunochemistry techniques; however, their results were based on small sample size and single-center studies.

Some studies have also reported CTC circulation in the body fluids before metastasizing to other parts of the body even in the early stages of the disease [4,5]. Wang et al[6] analysed the prognostic role of CTCs, highlighting the importance of CTC count before and after chemotherapy. They found that the presence of CTCs during chemotherapy is an unfavorable but independent factor and may play a role in deciding overall survival (OS) and survival without disease progression [progression free survival (PFS)] in advanced CRC cases. From this study, it was clear that CTCs in peripheral blood can be used as useful tumour markers. Characterization and early detection of CTCs have been reported to play an important role as a prognostic and predictive factor in different types of solid tumours[7,8]. Many epithelial cancers, including breast, prostate and lung cancers, have also been found to be associated with CTCs[9,10].

Early diagnosis, prediction of prognosis, assessment of recurrent risk, individualized treatment, and treatment with curative intent have focused research in the field of CTCs[11]. CTCs have faced difficulties for years because of their very low number (1-10 cells per 10 mL of blood) in many studies, and they have a short half-life which ranges from 1 to 2.4 h in blood[12,13], hence posing difficulty in further study. Their detection, quantification, and characterization of molecular features are also difficult. At present, there are several limitations to available CTC isolation techniques. Moreover, only a very small number of CTCs possess metastatic property[14]. Hence, it is very important to characterize them exactly so as to differentiate the nonmetastatic CTCs from metastatic ones. There are several techniques which are described here for isolation and detection of CTCs effectively.

CIRCULATING TUMOUR CELL ISOLATION AND DETECTION METHODS

Basic principles

Investigation of CTCs can provide helpful clinical information. However, as described earlier, blood stream harbors very few CTCs and every single CTC is surrounded by 10⁶-10⁷ mononuclear white blood cells (WBCs). To isolate CTCs and detect their characteristics, it is crucial to isolate them from whole blood cells.

Although there are several methods described for isolation of CTCs (Table 1), there are only two basic approaches. The first one is isolation methods based on the detection of specific surface markers for CTCs, which is also termed as "labeldependent methods" (or cell surface markers), and the second method is based on physical or biological properties of CTCs, termed as "label-independent methods". These approaches are not based on antibodies or other markers for labeling the cells of interest, but they enrich them by use of the difference of physical properties.

Label-dependent methods

In these methods, CTC isolation is based on specific markers. The majority of labeldependent methods use specific epithelial tissue marker-epithelial cell adhesion molecule (EpCAM). EpCAM is the most commonly used method of capturing CTCs because its expression is virtually universal in the cells of epithelial origin and is absent in blood cells. Cell capture with conjugated antibodies followed by purification of captured cells *via* the magnetic field was initially used to enrich CTCs from the blood of patients with prostate or breast cancer. The CellSearch system (Veridex) is a commercial platform which is based on this feature; CTCs are characterized as a population of EpCAM-captured cells that are confirmed to be negative for CD45 and positive for cytokeratins^[15]. Other markers are also used, like human epidermal growth factor receptor 2 (HER2), mucin 1 (MUC1), and cytokeratins[16,17].

The CellSearch CTCs system (Veridex) is commonly used, and in today's scenario it is the gold standard and the only FDA-approved method for CTC detection. It was approved in 2004 for extracting CTCs in metastatic breast cancer^[18] and later in colorectal^[19] and prostate cancers^[20]. Equipment cost ranges from 600000-800000

The CellSearch CTCs Kit is generally used for the enumerization of CTCs of epithelial origin (CD45-, EpCAM+, and cytokeratins 8+, 18+, and/or 19+) from whole blood sample and works on the principle of anti-EpCAM immunomagnetic



Tab	Table 1 Techniques for circulating tumour cell isolation, markers, and their limitations						
No.	Name	Property	Markers	Limitations	Ref.		
1	CellSearch	Isolation by anti-EpCAM antibody coated immunomagnetic beads	EpCAM, CKs, CD45, DAPI	Only suitable for cancer of epithelial origin but not for that undergoing the EMT	[48]		
				Cells are not viable after detection			
2	AdnaTest	Separation by way of anti-EpCAM and anti-MUC1 antibody coated immunomagnetic beads	EpCAM, MUC1, mucin-1, HER2	Possible false-positive finding due to expression of a selection marker being present in other cells other than CTCs	[49]		
				Cells are not viable after detection			
3	MACS	Immunomagnetic CTC enrichment by antibodies against cell surface markers	CK19, EpCAM, Her-2, MUC-1 CK7, CK8, CK18, CK19	Lengthy processing time and low sensitivity	[50]		
4	MagSweeper (Illumina Inc)	Immunomagnetic isolation of CTC by antibodies against EpCAM and cellsurface markers	EpCAM, CD45, DAPI	Less sensitive during the early stages of tumour development	[51]		
		Captured cells are viable with intact RNA					
5	CTC Chip	Utilizes bifurcating traps to capture CTCs, release <i>via</i> flow reversal	EpCAM, CKs, CD45, DAPI	Identification of CTCs is lower than other methods	[<mark>52</mark>]		
6	GEM chip	Geometrically enhanced mixing chip structure that allows enhanced capture of CTC on antibody coated surfaces	EpCAM, DAPI, CD45, cytokeratin	Low sensitivity	[53]		
7	Onco Quick (Greiner BioOne, Frickenhausen, Germany)	Separation of erythrocytes and some leukocytes from CTC. High sensitivity, Quantification	CCNE2, DKFZp762E1312, EMP2	No morphology confirmation; not really capture CTCs	[54]		
8	ISET (Rarecells Diagnostics)	Rapid processing; non-antigen dependent; Filter based approach	CKs, EGFR, VE-cadherin, Ki67	Size-dependent, manual processing	[55]		
9	EPISPOT	Removes leukocytes via CD45 depletion	CD45, CK19, mucin-1,	Problem arises when antigen levels are	[<mark>56</mark>]		
		Can detect viable CTCs	cathepsin-D	lower or binding efficiency is reduced			
10	Ficoll + RT-PCR	Separation of CTC based on size dependent enrichment. High Sensitivity	CK-19, HER2, h-MAM, CEA, maspin, GABA A, B726P	No morphology confirmation	[57]		
11	Cyttel Method	Negative immune-magnetic selection of WBC (CD45 antibody)-High detection rate	CD45	-	[58]		
12	MetaCell	Size-based enrichment and separation for viable CTCs	CK-18, -19, -20, CK-7, EPCAM, MUC1, HER2, EGFR	Lengthy processing time	[59]		

MACS: Magnetic-activated Cell Sorting; CTC: Circulating tumour cell; RT-PCR: Real-time polymerase chain reaction.

enrichment. For the CTC enumeration step, the CTC kit has reagents to stain and fix the cells. The protocol has been described in detail by Coumans and Terstappen[21]; EpCAM is unique for epithelial cells and is expressed in most carcinomas in a very strong manner, while its expression is limited to embryonic stem cells in non-epithelial cells[22]. However, EpCAM is not a universal cancer marker. EpCAM expression is quite absent in squamous carcinoma or down-regulated if cancer cells undergo epithelial-to-mesenchymal transition (EMT); such cancer cells can escape the capturing process. A gene expression study on breast cancer showed that EpCAM was downregulated in mesenchymal lines relative to the epithelial cell lines^[23] and EMTinduced breast cancer cells^[24].

Despite its high specificity and efficiency, some of the disadvantages of the CellSearch system are: (1) It is only suitable for cancer of epithelial origin but not for cancer cells undergoing EMT; (2) CTCs cannot be further analysed in real-time and live-cell conditions, because CTCs cannot be kept alive for a long time; and (3) The use of expensive antibodies leads to high detection cost.

Other label dependent methods

AdnaTest is the second most common method used for CTC detection after the CellSearch. It is a commercially available positive selection method in which immuno-



magnetic beads are coated with a combination of antibodies for the increased capture and enrichment of CTCs. Through gene expression testing of specific tumour markers in the captured cells and comparison of this with their primary and metastatic tumour equivalents, clinicians may analyze the clinical implications of CTCs. Therefore, it has both diagnostic and prognostic value. CTCs captured by magnetic beads coated with antibodies (EpCAM, MUC-1, etc.) are then analyzed by multiplex real-time polymerase chain reaction (RT-PCR) gene panels.

The other techniques which are used for CTC enrichment are as follow: (1) Magnetic-activated cell sorting system: This system works on immunomagnetic CTC enrichment by antibodies against cell surface markers. Magnetic-activated cell sorting offers both positive and negative enrichment for the high-efficient and accurate isolation of CTCs (Clinical value: Prognosis and diagnosis); (2) MagSweeper: This system works on immunomagnetic isolation of CTCs by antibodies against EpCAM and other cell surface markers. It can process large amount of blood (approximately 9 mL/h) and can detect 1-3 CTCs per 1 mL of whole blood (Clinical value: Prognosis); (3) GEM chip: This is geometrically enhanced mixing chip that permits increased identification of CTCs on antibody-coated surfaces (Clinical value: Treatment monitoring and prognosis); (4) Onco cell enrichment and extraction: This platform uses microfluidic chip with internal surfaces functionalized with an antibodies group against bio-tumour-associated and mesenchymal markers (Clinical value: Treatment monitoring, prognosis, and diagnosis); (5) Graphene oxide chip: In this platform, graphene oxides (GO) nanosheets are used to capture antibodies against cell surface markers of CTCs with a high sensitivity (Clinical value: Prognosis); (6) Ephesia (CTCchip): Micromagnetic particles are functionalized with EpCAM antibodies which can be self-assembled in a micro-fluidic platform (Clinical value: Prognosis and diagnosis); (7) Quadrupole magnetic separator: This separator works as negative CTC enrichment after it combines with viscous flow stress and magnetic force for the recovery of unlabelled CTCs (Clinical value: Treatment monitoring, prognosis, and diagnosis); and (8) CTC-iChip: This chip works on lateral displacement, inertial focusing, and magnetophoresis for fast isolation of leukocytes by using anti-CD45 and anti-CD66B antibodies in negative enrichment or EpCAM activated beads for CTC enrichment in positive enrichment of CTCs (Clinical value: Prognosis and diagnosis).

Label independent methods

Many newly studied methods forCTC recognition have been reported[25]. Separation of circulating tumor cells by physical properties, i.e., density gradients and gravity, using microfluidic technology [26,27] have been found to be able to capture CTCs efficiently.

The different tools and techniques described for CTC isolation in this category are as follows: (1) ISET: Filter based isolation and enrichment (Clinical value: Treatment regimen and prognosis); (2) MetaCell system: Size-based enrichment and separation (Clinical value: Diagnosis and prognosis); (3) Parylene filter: Filter based isolation and enrichment (Clinical value: Diagnosis and prognosis); (4) ScreenCellCyto: Filter based size-exclusion separation and enrichment (Clinical value: Diagnosis); (5) Cell sieve: Micofilter based isolation and enrichment (Clinical value: Diagnosis and prognosis); (6) Parsorti technology: Micro fluidic separation of CTC based on their size and deformability (Clinical value: Diagnosis and prognosis); (7) RosetteSep CTC enrichment/CD45 depletion: This is an immuno-density negative selection method for CTCs using tetrameric antibody complexes that identify CD45, CD66, and glycophorin on WBCs and red blood cells (RBCs) (Clinical value: Prognosis); (8) Onco Quick: Isolation of RBCs and some leukocytes from CTCs by using filtration through porous membrane followed by density-gradient centrifugation for better CTC enrichment (Clinical value: Prognosis); (9) Cyttel method: Based on the negative immunomagnetic selection of WBCs (antibody CD45) followed by gradient centrifugation and smearing through slides of isolated CTCs (Clinical value: Prognosis and treatment regimen); (10) AccuCyte-CyteFinder: Automated rapid imaging of single rare cells in CTCs, followed by density-based cell separation method (Clinical value: Prognosis); (11) EPISPOT: Negative enrichment using CD45 depletion (Clinical value: Prognosis); (12) Cyto Track: Use of fluorescently labeled cells against EpCAM and scanned with the help of beam (Clinical value: Prognosis); (13) Fiber optic array scanning technology (FAST) (Clinical value: Prognosis); (14) Image Stream: Immunogenetic sorting of blood followed by flow cytometry and enumeration of CTCs by fluorescent microscopy (Clinical value: Diagnosis); (15) DEPArray: Moving dielectrophoretic cages for cell capture coupled with Sanger sequencing (Clinical value: Tumour monitoring and prognosis); (16) Vortex: CTC extraction using microscale vortices and inertial focusing (Clinical value: Diagnosis, prognosis, and treatment planning); (17) ClearCell FX: CTC



separation based on size using Dean Flow Fractionation (Clinical value: Diagnosis); and (18) qRT-PCR: Separation of CTCs based on size-dependent enrichment using CD45, CK19, and CK20 (Clinical value: Prognosis).

COMPARISON OF CELLSEARCH SYSTEM WITH OTHER TECHNIQUES

The high sensitivity and specificity of CTC detection methods have a great effect in improving patient outcomes. Politaki et al [28] have compared CTC detection rates and prognostic significance in breast cancer patients by comparing three commonly used methods including CellSearch, qRT-PCR, and double immunofluorescence (IF) microscopy. They analyzed early diagnosed (n = 200) and metastatic (n = 164) breast cancer patients before the start of adjuvant or first-line chemotherapy. They compared CellSearch system, qRT-PCR for CK19 mRNA detection, and double IF microscopy by using A45-B/B3 and CD45 antibodies and concluded that patients were more likely to be CTC-positive using the CellSearch (37%) than qRT-PCR (37% vs 18.0%, P < 0.001) or IF (37% vs 16.9%, P < 0.001). In another study[29], CellSearch was compared with Adna Test and RT-PCR in breast cancer, and it was found that multimarker qRT-PCR showed a superior sensitivity for the detection of CTCs in metastatic breast cancer patients compared with the CellSearch system and the AdnaTest. There is limitation of the assessment by PCR as it provides the number of target transcripts based on the actual number of CTCs present in a sample[30] and does not allow the morphological assessment of cells. Two cell-based detection assays, the CellSearch and Onco-Quick (for density gradient centrifugation), on comparison revealed that the CellSearch was a far more accurate and sensitive method to detect and enumerate CTCs[31].

There is one study by Gervasoni *et al*[32], in which they compared the capacity of three methods, multimarker RT-PCR assay, standardized CellSearch method, and dHPLC-based gene mutation analysis, to detect CTCs in the blood of 20 CRC patients (stage I = 5, stage II = 8, stage III = 6, and stage IV = 1). They found CTC positivity in 75% of samples by RT-PCR, 20% by CellSearch method, and only 14.3% of samples were found to be gene mutated with the presence of CTCs by HPLC method. These results show that out of these three methods tested, multimarker RT-PCR assay provides the maximum probability of CTC detection. Future studies, by using the above three distinct methods for follow-up, may provide more information about the prognostic significance of CTCs detected through single method assay vs combination of different assays[32].

CIRCULATING TUMOUR CELLS AND THEIR CLINICAL APPLICATIONS IN COLORECTAL CANCER

CTC characterization and number may be useful in several ways where they can be used both as a prognostic marker for survival as well as prediction of response to cancer treatment[33]. A multivariate analysis[34] demonstrated that CTC count is the strongest prognostic biomarker for patient survival. If the CTC number increases or remains static, the treatment can be deemed to be ineffective, whereas, if CTC number decreases, the treatment may be effective. Several studies have shown that the presence of as few as 3 to 5 CTCs in 7.5 mL of blood is associated with poor PFS and OS rates^[35]. Studies with the CellSearch system and others have shown that high numbers of CTCs are associated with lower DS and OS rates[36]. In a study of 413 metastatic CRC patients being treated with first, second, or third-line therapy, patients with a baseline CTC number of more than 3/7.5 mL had significantly poor median PFS (4.4 mo vs 7.8 mo, P = 0.004) and OS (9.4 mo vs 20.6 mo, P < 0.0001) compared with patients with less than 3 CTCs/7.5 mL[37,38]. CTC evaluation, during treatment, may be used as a prognostic predictive marker to determine progression-free survival (PFS) and OS. The CellSearch system has its own limitation; the method of isolation utilizes EpCAM expression on the cell surface of the tumour, which is expressed in 75% of cancer types. A study by Fang et al[39] (2016) analyzed the expression of cell surface markers CD133, CD54, and CD44 with the help of flow cytometry to analyze the correlation between cellular subpopulations and colorectal liver metastasis. They observed that the expression of cellular subpopulations (CD133+, CD54+, and CD44+) was higher in the peripheral blood of CRC liver metastasis in comparison with those with no metastasis (P < 0.001). In a study by Lalmahomed *et al*[40] (2015) on peripheral blood of 151 CRC patients who underwent liver metastasectomy, CTCs were detected

by the CellSearch system after a density-gradient-based enrichment step. They found that CTCs were detected in 75 samples (43%), out of which 16% had 3 CTCs/7.5 mL of blood. Patients with or without detectable CTCs have an almost similar 1-year recurrence rate (47% vs 48%, respectively). A similar recurrence rate was also reported with low vs high CTC count (< 3 or 3 CTCs/7.5 mL of blood: 50% vs 47%, respectively). In their report, no difference was found in disease-free survival and OS among patients with or without CTCs. A report by Shimada *et al*[41] (2012) found that detecting CEA/CK/CD133 mRNA in tumour drainage blood (RT-PCR method) could act as a prognostic marker in patients with Duke's stages B and C CRC. The findings of the CTC isolation techniques and their clinical significance have been given in detail in Table 2. Hendricks et al [42] (2020) used qRT-PCR for indirect CTC detection, which was already applied in previous studies on CRC patients and found to have prognostic value. An earlier study by Sastre et al[43] (2008) reported that the CellSearch system could identify CTCs in CRC patients and that CTC positive cases were correlated with the stage of the disease (P = 0.005) but there was no significant correlation between CEA levels, tumour locations, grade of differentiation, and lactate dehydrogenase (LDH) levels. A meta-analysis by Katsuno et al[44] (2008) of a total of nine studies found that CTC-positive patients (in blood samples by RT-PCR), correlated with lymph node (LN)-positive patients (50%) vs LN-negative patients (21%).

Guadagni *et al*[45] (2020) have published a couple of studies about the role of CTC based therapeutic decision making in CRC. In the first study[45], they included 62 patients with advanced unresectable rectal cancer and reported that where the patients were selected for the treatment based on CTCs (HPP/target-therapy group, n = 43); the disease control rate was significantly higher (PFS = 8 mo, OS = 20 mo) as compared to those given systemic chemotherapy (n = 19) based on age, co-morbidity, and performance status (PFS = 4 mo, OS = 8 mo). The second study[46] was performed on 106 advanced unresectable CRC patients. The therapy was decided based on CTCs (HAI/targeted, n = 44), age, and co-morbidity performance status (systemic chemotherapy, n = 62). The authors found that the group where treatment was given based on CTCs had longer PFS and median survival (MS) (PFS = 5 mo, MS = 14 mo) as compared to those given therapy based on age and co-morbidity performance status (PFS = 3 mo, MS = 8.5). Finally, they concluded that CTCs can be used to choose therapeutic options in unresectable CRC.

Inherited or acquired resistance in response to specific treatment can be assessed with CTCs which may also work as pharmacodynamic markers. CTCs have enhanced our knowledge and understanding about the primary mechanisms of cancer metastasis. This understanding may be useful in therapeutic manipulation with the help of new targets. CTCs were evaluated in phase I trial based on their count and the expression of insulin-like growth factor-1 receptor (IGF-1R) to find out their therapeutic applications. The CellSearch system was used, either alone or in combination with docetaxel, to count CTCs in patients treated with monoclonal antibodies against IGF-1R. Positive IGF-1R and CTC response was seen in 23 out of 26 patients. These patients responded better in case of combined treatment than in case of the remaining three patients who were negative for IGF-1R. From these findings, it was concluded that CTCs can be used as a potential marker for the selection of chemotherapy[47].

CHALLENGES IN CIRCULATING TUMOUR CELL IDENTIFICATION

CTC interpretation is quite promising but has limitations such as factors like requirement of large volume of blood, small size of the cancer patient population, and the standard value for comparison (*i.e.*, CellSearch, blood sample, other micro-devices, *etc.*). Till now, many reports have enlightened the prospects for cancer patient monitoring, and for few years researchers have focused on CTCs to explore their biological metastatic property and role in cancer treatment monitoring. Among the several important clinical applications for CTC technology is the correlation of CTC count with OS and PFS as a measure of clinical outcome.

The presence of CTCs in the blood sample is also a major challenge. If they are present, their heterogeneity of unknown extent is also present. Because of this nature, it demands an ongoing diversity in the detection and characterization of CTCs using the present available and upcoming methods in the future.

Zaishidena® WJCO | https://www.wjgnet.com

Table 2 Studies showing postoperative isolation of circulating tumour cells in colorectal cancer-markers, techniques, and clinical implications

No.	Technology	Markers	Number of patients	TNM stage	Correlation	Clinical significance	Ref.
1	CellSearch system	ЕрСАМ	164	I-III	With stage	N/A	[<mark>60</mark>]
		ЕрСАМ	24	IV	With therapy response	May be used in monitoring response to therapy	[6 1]
		ЕрСАМ	97	Π	With stage	Correlates with stage	[<mark>62</mark>]
		CD133+, CD54+, CD44+	15 (nmCRC); 95 (mCRC)	I-IV	≥ 5 CTCs were 8 times more likely to develop distant metastasis. CTC counts show good correlation with colorectal neoplasm	Independent prognostic marker for nmCRC	[63]
		hTERT, CK19, CK20, CEA	438	I-III	-	Poor relapse free survival	[<mark>64</mark>]
		hTERT, CK19, CK20, CEA	157	I-III	With stage	Poor relapse free survival and overall survival	[65]
		Survivin, CK20 and CEA	156	I-III	With stages (Duke's) and lymph node metastasis.	Useful as an adjunct in detection of CRC patients	[<mark>66</mark>]
		CD133, CEA, CK20, CK19,	197	II-III	CEA/CK/CD133 expression and stage (Duke's)	Prognostic significance (Duke's stages B and C)	[44]
		hTERT, CK-19, CK-20, CEA, GAPDH and mRNA	72	I-IV	CEA, mRNA: With stage, vascular invasion, and postoperative metastasis	Prognostic and predictive	[67]
2	Flow-cytometry with immunofluorescence	CTCs	18	I-III	With stage and also detected in an early cancer stage.	Predictive	[<mark>68</mark>]
3	Pyrosequencing	KRAS (Codon 12/13)	26	IV	No association	Prognostic	[69]
4	MetaCell separation	CTCs	98	I-IV	CTC-positive in 83%	Prognosis and	[70]
	method				CTC-negative in 17%	predictive	
5	Mag Sweeper	PIK3CA	242	-	Mutational discordance found between CTCs, DTCs, and metastases, and among CTCs; DTCs from this patient propagated <i>in</i> <i>vitro</i> contained a PIK3CA mutation	Investigating new drug therapies	[71]
6	CTC-Chip	EpCAM, HER2, and EGFR	-	-	Efficiency of 87.5%	<i>In situ</i> protein expression, and culture CTCs from the same set of cells	[72]

CRC: Colorectal cancer; DTC: Disseminated tumour cells; GAPDH: Glyceraldehydes 3-phosphate dehydrogenase; nmCRC: Non-metastatic CRC; mCRC: Metastatic CRC.

CONCLUSION

CTCs have become a hot pursuit and in recent years many new CTC detection technologies have emerged. Discoveries of these technologies from laboratory to clinical practice are non-trivial. Only a few systems are available for routine use in the clinical setting, but not freely available. CTC detection is challenging because of the small number of circulating cells but has been found both in metastatic and non-metastatic cancer (Table 3). It has been well correlated with the stage of the disease, prognosis, and survival but has a limited role in therapeutic decision-making. There is a need for the development of newer, cheaper techniques of CTC detection which can be used as an alternative to invasive diagnosis and treatment monitoring. Future research is required as the current literature has limited information on its use in routine clinical practice but the future is promising.

Baichidena® WJCO | https://www.wjgnet.com

No.	Type of CRC	Markers used	Detection method used	Relevance	Clinical implications	Limitations of the study	Ref
1	nmCRC	CEA, CA19-9, CA72-4	Cyttel	Diagnostic/prognostic/predictive	Combination of CTCs and CEA: Diagnostic and prognostic indicators	Small sample size, weak power of the study	[73]
2	mCRC	CK, CD45	Immunomagnetic separation	Prognostic/predictive	The number of CTCs before and during treatment is an independent predictor of PFS and OS in patients with mCRC	The baseline unfavourable CTC was low (26%) and overall CTC yield was less than in other epithelial cells	[74]
3	mCRC	ALDH1, CD44, CD133, MRP5, Survivin	qRT-PCR	Prognostic	Poor prognosis and chemo therapy non- responsiveness	Require further molecular analyses of CTCs for selection of targeted agents	[75]
					Survivin and MRP5 selection of mCRC patients resistant to 5-FU and L-OHP		
4	mCRC	CEA	Cyttel method, immunofluorescence <i>in</i>	Prognostic	PFS, OS	Small sample size	[<mark>6</mark>]
			sin in crainer ceruio seco (ini icri)			Lack of dynamic enumeration of CTCs	
5	mCRC	VEGF, CD133+, CD34+/KDR + EPC, CD-34 ⁻ VEGFR2	Flow cytometry/IHC	Prognostic	Treatment response; PFS, OS	-	[76]
6	nmCRC	CD133, CD166, CD44, EpCAM, ALDH1	Tissue microarray, IHC	Prognostic	No association with poor clinical response; OS	Treatment information was missing (local recurrence, distant metastasis, and postoperative therapy)	[77]
7	nmCRC	CK19, MUC1, CD44, CD133, ALDH1	Flow-cytometry, CellSearch, Cytomorphology, qPCR	Prognostic	May be useful as a therapeutic target; PFS, OS	-	[78]

CRC: Colorectal cancer; nmCRC: Non-metastatic CRC; MCRC: Metastatic CRC; OS: Overall survival; PFS: Progression-free survival; IHC: Immunohistochemistry.

REFERENCES

- Siegel R, Desantis C, Jemal A. Colorectal cancer statistics, 2014. CA Cancer J Clin 2014; 64: 104-117 [PMID: 24639052 DOI: 10.3322/caac.21220]
- Pino MS, Chung DC. The chromosomal instability pathway in colon cancer. *Gastroenterology* 2010; 138: 2059-2072 [PMID: 20420946 DOI: 10.1053/j.gastro.2009.12.065]
- 3 Yap TA, Lorente D, Omlin A, Olmos D, de Bono JS. Circulating tumor cells: a multifunctional biomarker. *Clin Cancer Res* 2014; 20: 2553-2568 [PMID: 24831278 DOI: 10.1158/1078-0432.CCR-13-2664]
- 4 Gold B, Cankovic M, Furtado LV, Meier F, Gocke CD. Do circulating tumor cells, exosomes, and circulating tumor nucleic acids have clinical utility? *J Mol Diagn* 2015; 17: 209-224 [PMID: 25908243 DOI: 10.1016/j.jmoldx.2015.02.001]
- 5 Masuda T, Hayashi N, Iguchi T, Ito S, Eguchi H, Mimori K. Clinical and biological significance of circulating tumor cells in cancer. *Mol Oncol* 2016; 10: 408-417 [PMID: 26899533 DOI:

Table 3 Circulating tumour cells in metastatic vs non-metastatic colorectal cancer

10.1016/i.molonc.2016.01.010

- 6 Wang L, Zhou S, Zhang W, Wang J, Wang M, Hu X, Liu F, Zhang Y, Jiang B, Yuan H. Circulating tumor cells as an independent prognostic factor in advanced colorectal cancer: a retrospective study in 121 patients. Int J Colorectal Dis 2019; 34: 589-597 [PMID: 30627849 DOI: 10.1007/s00384-018-03223-91
- 7 Franken B, de Groot MR, Mastboom WJ, Vermes I, van der Palen J, Tibbe AG, Terstappen LW. Circulating tumor cells, disease recurrence and survival in newly diagnosed breast cancer. Breast Cancer Res 2012; 14: R133 [PMID: 23088337 DOI: 10.1186/bcr3333]
- 8 Banys-Paluchowski M, Schneck H, Blassl C, Schultz S, Meier-Stiegen F, Niederacher D, Krawczyk N, Ruckhaeberle E, Fehm T, Neubauer H. Prognostic Relevance of Circulating Tumor Cells in Molecular Subtypes of Breast Cancer. Geburtshilfe Frauenheilkd 2015; 75: 232-237 [PMID: 25914415 DOI: 10.1055/s-0035-1545788]
- Yu M, Stott S, Toner M, Maheswaran S, Haber DA. Circulating tumor cells: approaches to isolation and characterization. J Cell Biol 2011; 192: 373-382 [PMID: 21300848 DOI: 10.1083/jcb.201010021]
- 10 Lianidou ES, Markou A, Strati A. The Role of CTCs as Tumor Biomarkers. Adv Exp Med Biol 2015; **867**: 341-367 [DOI: 10.1007/978-94-017-7215-0_21]
- Huang MY, Tsai HL, Huang JJ, Wang JY. Clinical Implications and Future Perspectives of 11 Circulating Tumor Cells and Biomarkers in Clinical Outcomes of Colorectal Cancer. Transl Oncol 2016; 9: 340-347 [PMID: 27567958 DOI: 10.1016/j.tranon.2016.06.006]
- Alix-Panabières C, Pantel K. Challenges in circulating tumour cell research. Nat Rev Cancer 2014; 12 14: 623-631 [PMID: 25154812 DOI: 10.1038/nrc3820]
- Haber DA, Velculescu VE. Blood-based analyses of cancer: circulating tumor cells and circulating 13 tumor DNA. Cancer Discov 2014; 4: 650-661 [PMID: 24801577 DOI: 10.1158/2159-8290.CD-13-1014]
- 14 Zhe X, Cher ML, Bonfil RD. Circulating tumor cells: finding the needle in the haystack. Am J Cancer Res 2011; 1: 740-751 [PMID: 22016824]
- 15 Serrano MJ, Ortega FG, Alvarez-Cubero MJ, Nadal R, Sanchez-Rovira P, Salido M, Rodríguez M, García-Puche JL, Delgado-Rodriguez M, Solé F, García MA, Perán M, Rosell R, Marchal JA, Lorente JA. EMT and EGFR in CTCs cytokeratin negative non-metastatic breast cancer. Oncotarget 2014; 5: 7486-7497 [PMID: 25277187 DOI: 10.18632/oncotarget.2217]
- 16 Cheng JP, Yan Y, Wang XY, Lu YL, Yuan YH, Jia J, Ren J. MUC1-positive circulating tumor cells and MUC1 protein predict chemotherapeutic efficacy in the treatment of metastatic breast cancer. Chin J Cancer 2011; 30: 54-61 [PMID: 21192844 DOI: 10.5732/cjc.010.10239]
- Chan CM, Au TC, Chan AT, Ma BB, Tsui NB, Ng SS, Hui EP, Chan LW, Ho WS, Yung BY, Wong 17 SC. Advanced technologies for studying circulating tumor cells at the protein level. Expert Rev Proteomics 2013; 10: 579-589 [PMID: 24206230 DOI: 10.1586/14789450.2013.858021]
- Cristofanilli M, Budd GT, Ellis MJ, Stopeck A, Matera J, Miller MC, Reuben JM, Doyle GV, Allard 18 WJ, Terstappen LW, Hayes DF. Circulating tumor cells, disease progression, and survival in metastatic breast cancer. N Engl J Med 2004; 351: 781-791 [PMID: 15317891 DOI: 10.1056/NEJMoa040766
- Negin BP, Cohen SJ. Circulating tumor cells in colorectal cancer: past, present, and future challenges. 19 Curr Treat Options Oncol 2010; 11: 1-13 [PMID: 20143276 DOI: 10.1007/s11864-010-0115-3]
- Resel Folkersma L, Olivier Gómez C, San José Manso L, Veganzones de Castro S, Galante Romo I, 20 Vidaurreta Lázaro M, de la Orden GV, Arroyo Fernández M, Díaz Rubio E, Silmi Moyano A, Maestro de Las Casas MA. Immunomagnetic quantification of circulating tumoral cells in patients with prostate cancer: clinical and pathological correlation. Arch Esp Urol 2010; 63: 23-31 [PMID: 20157216
- Coumans F, Terstappen L. Detection and Characterization of Circulating Tumor Cells by the 21 CellSearch Approach. Methods Mol Biol 2015; 1347: 263-278 [PMID: 26374323 DOI: 10.1007/978-1-4939-2990-0 18
- Gires O, Stoecklein NH. Dynamic EpCAM expression on circulating and disseminating tumor cells: 22 causes and consequences. Cell Mol Life Sci 2014; 71: 4393-4402 [PMID: 25103341 DOI: 10.1007/s00018-014-1693-11
- Santisteban M, Reiman JM, Asiedu MK, Behrens MD, Nassar A, Kalli KR, Haluska P, Ingle JN, 23 Hartmann LC, Manjili MH, Radisky DC, Ferrone S, Knutson KL. Immune-induced epithelial to mesenchymal transition in vivo generates breast cancer stem cells. Cancer Res 2009; 69: 2887-2895 [PMID: 19276366 DOI: 10.1158/0008-5472.CAN-08-3343]
- Taube JH, Herschkowitz JI, Komurov K, Zhou AY, Gupta S, Yang J, Hartwell K, Onder TT, Gupta 24 PB, Evans KW, Hollier BG, Ram PT, Lander ES, Rosen JM, Weinberg RA, Mani SA. Core epithelial-to-mesenchymal transition interactome gene-expression signature is associated with claudin-low and metaplastic breast cancer subtypes. Proc Natl Acad Sci US A 2010; 107: 15449-15454 [PMID: 20713713 DOI: 10.1073/pnas.1004900107]
- 25 Chen F, Wang S, Fang Y, Zheng L, Zhi X, Cheng B, Chen Y, Zhang C, Shi D, Song H, Cai C, Zhou P, Xiong B. Feasibility of a novel one-stop ISET device to capture CTCs and its clinical application. Oncotarget 2017; 8: 3029-3041 [PMID: 27935872 DOI: 10.18632/oncotarget.13823]
- Huang SB, Wu MH, Lin YH, Hsieh CH, Yang CL, Lin HC, Tseng CP, Lee GB. High-purity and 26 label-free isolation of circulating tumor cells (CTCs) in a microfluidic platform by using opticallyinduced-dielectrophoretic (ODEP) force. Lab Chip 2013; 13: 1371-1383 [PMID: 23389102 DOI: 10.1039/c3lc41256c]



- Alam MK, Koomson E, Zou H, Yi C, Li CW, Xu T, Yang M. Recent advances in microfluidic 27 technology for manipulation and analysis of biological cells (2007-2017). Anal Chim Acta 2018; 1044: 29-65 [PMID: 30442405 DOI: 10.1016/j.aca.2018.06.054]
- 28 Politaki E, Agelaki S, Apostolaki S, Hatzidaki D, Strati A, Koinis F, Perraki M, Saloustrou G, Stoupis G, Kallergi G, Spiliotaki M, Skaltsi T, Lianidou E, Georgoulias V, Mavroudis D. A Comparison of Three Methods for the Detection of Circulating Tumor Cells in Patients with Early and Metastatic Breast Cancer. Cell Physiol Biochem 2017; 44: 594-606 [PMID: 29161698 DOI: 10.1159/000485115]
- 29 Van der Auwera I, Peeters D, Benoy IH, Elst HJ, Van Laere SJ, Prové A, Maes H, Huget P, van Dam P, Vermeulen PB, Dirix LY. Circulating tumour cell detection: a direct comparison between the CellSearch System, the AdnaTest and CK-19/mammaglobin RT-PCR in patients with metastatic breast cancer. Br J Cancer 2010; 102: 276-284 [PMID: 19953098 DOI: 10.1038/sj.bjc.6605472]
- Lianidou ES, Markou A. Circulating tumor cells in breast cancer: detection systems, molecular 30 characterization, and future challenges. Clin Chem 2011; 57: 1242-1255 [PMID: 21784769 DOI: 10.1373/clinchem.2011.165068
- 31 Balic M, Dandachi N, Hofmann G, Samonigg H, Loibner H, Obwaller A, van der Kooi A, Tibbe AG, Doyle GV, Terstappen LW, Bauernhofer T. Comparison of two methods for enumerating circulating tumor cells in carcinoma patients. Cytometry B Clin Cytom 2005; 68: 25-30 [PMID: 16142788 DOI: 10.1002/cvto.b.20065]
- 32 Gervasoni A, Sandri MT, Nascimbeni R, Zorzino L, Cassatella MC, Baglioni L, Panigara S, Gervasi M, Di Lorenzo D, Parolini O. Comparison of three distinct methods for the detection of circulating tumor cells in colorectal cancer patients. Oncol Rep 2011; 25: 1669-1703 [PMID: 21455578 DOI: 10.3892/or.2011.1231]
- Giuliano M, Giordano A, Jackson S, Hess KR, De Giorgi U, Mego M, Handy BC, Ueno NT, Alvarez 33 RH, De Laurentiis M, De Placido S, Valero V, Hortobagyi GN, Reuben JM, Cristofanilli M. Circulating tumor cells as prognostic and predictive markers in metastatic breast cancer patients receiving first-line systemic treatment. Breast Cancer Res 2011; 13: R67 [PMID: 21699723 DOI: 10.1186/bcr2907]
- Bidard FC, Peeters DJ, Fehm T, Nolé F, Gisbert-Criado R, Mavroudis D, Grisanti S, Generali D, Garcia-Saenz JA, Stebbing J, Caldas C, Gazzaniga P, Manso L, Zamarchi R, de Lascoiti AF, De Mattos-Arruda L, Ignatiadis M, Lebofsky R, van Laere SJ, Meier-Stiegen F, Sandri MT, Vidal-Martinez J, Politaki E, Consoli F, Bottini A, Diaz-Rubio E, Krell J, Dawson SJ, Raimondi C, Rutten A, Janni W, Munzone E, Carañana V, Agelaki S, Almici C, Dirix L, Solomayer EF, Zorzino L, Johannes H, Reis-Filho JS, Pantel K, Pierga JY, Michiels S. Clinical validity of circulating tumour cells in patients with metastatic breast cancer: a pooled analysis of individual patient data. Lancet Oncol 2014; 15: 406-414 [PMID: 24636208 DOI: 10.1016/S1470-2045(14)70069-5]
- Krebs MG, Hou JM, Ward TH, Blackhall FH, Dive C. Circulating tumour cells: their utility in cancer 35 management and predicting outcomes. Ther Adv Med Oncol 2010; 2: 351-365 [PMID: 21789147 DOI: 10.1177/17588340103784141
- 36 Graves H, Czerniecki BJ. Circulating tumor cells in breast cancer patients: an evolving role in patient prognosis and disease progression. Patholog Res Int 2011; 2011: 621090 [PMID: 21253472 DOI: 10.4061/2011/621090
- Cohen SJ, Punt CJ, Iannotti N, Saidman BH, Sabbath KD, Gabrail NY, Picus J, Morse MA, Mitchell 37 E, Miller MC, Doyle GV, Tissing H, Terstappen LW, Meropol NJ. Prognostic significance of circulating tumor cells in patients with metastatic colorectal cancer. Ann Oncol 2009; 20: 1223-1229 [PMID: 19282466 DOI: 10.1093/annonc/mdn786]
- Tol J, Koopman M, Miller MC, Tibbe A, Cats A, Creemers GJ, Vos AH, Nagtegaal ID, Terstappen 38 LW, Punt CJ. Circulating tumour cells early predict progression-free and overall survival in advanced colorectal cancer patients treated with chemotherapy and targeted agents. Ann Oncol 2010; 21: 1006-1012 [PMID: 19861577 DOI: 10.1093/annonc/mdp463]
- Fang C, Fan C, Wang C, Huang Q, Meng W, Yu Y, Yang L, Peng Z, Hu J, Li Y, Mo X, Zhou Z. 39 CD133+CD54+CD44+ circulating tumor cells as a biomarker of treatment selection and liver metastasis in patients with colorectal cancer. Oncotarget 2016; 7: 77389-77403 [PMID: 27764803 DOI: 10.18632/oncotarget.12675]
- Lalmahomed ZS, Mostert B, Onstenk W, Kraan J, Ayez N, Gratama JW, Grünhagen D, Verhoef C, 40 Sleijfer S. Prognostic value of circulating tumour cells for early recurrence after resection of colorectal liver metastases. Br J Cancer 2015; 112: 556-561 [PMID: 25562435 DOI: 10.1038/bic.2014.6511
- 41 Shimada R, Iinuma H, Akahane T, Horiuchi A, Watanabe T. Prognostic significance of CTCs and CSCs of tumor drainage vein blood in Dukes' stage B and C colorectal cancer patients. Oncol Rep 2012; 27: 947-953 [PMID: 22267181 DOI: 10.3892/or.2012.1649]
- Hendricks A, Brandt B, Geisen R, Dall K, Röder C, Schafmayer C, Becker T, Hinz S, Sebens S. 42 Isolation and Enumeration of CTC in Colorectal Cancer Patients: Introduction of a Novel Cell Imaging Approach and Comparison to Cellular and Molecular Detection Techniques. Cancers (Basel) 2020; 12 [PMID: 32947903 DOI: 10.3390/cancers12092643]
- Sastre J, Maestro ML, Puente J, Veganzones S, Alfonso R, Rafael S, García-Saenz JA, Vidaurreta M, 43 Martín M, Arroyo M, Sanz-Casla MT, Díaz-Rubio E. Circulating tumor cells in colorectal cancer: correlation with clinical and pathological variables. Ann Oncol 2008; 19: 935-938 [PMID: 18212090 DOI: 10.1093/annonc/mdm583]



- 44 Katsuno H, Zacharakis E, Aziz O, Rao C, Deeba S, Paraskeva P, Ziprin P, Athanasiou T, Darzi A. Does the presence of circulating tumor cells in the venous drainage of curative colorectal cancer resections determine prognosis? Ann Surg Oncol 2008; 15: 3083-3091 [PMID: 18787906 DOI: 10.1245/s10434-008-0131-8]
- 45 Guadagni S, Fiorentini G, De Simone M, Masedu F, Zoras O, Mackay AR, Sarti D, Papasotiriou I, Apostolou P, Catarci M, Clementi M, Ricevuto E, Bruera G. Precision oncotherapy based on liquid biopsies in multidisciplinary treatment of unresectable recurrent rectal cancer: a retrospective cohort study. J Cancer Res Clin Oncol 2020; 146: 205-219 [PMID: 31620896 DOI: 10.1007/s00432-019-03046-3
- Guadagni S, Clementi M, Mackay AR, Ricevuto E, Fiorentini G, Sarti D, Palumbo P, Apostolou P, 46 Papasotiriou I, Masedu F, Valenti M, Giordano AV, Bruera G. Real-life multidisciplinary treatment for unresectable colorectal cancer liver metastases including hepatic artery infusion with chemofiltration and liquid biopsy precision oncotherapy: observational cohort study. J Cancer Res Clin Oncol 2020; 146: 1273-1290 [PMID: 32088781 DOI: 10.1007/s00432-020-03156-3]
- 47 de Bono JS, Attard G, Adjei A, Pollak MN, Fong PC, Haluska P, Roberts L, Melvin C, Repollet M, Chianese D, Connely M, Terstappen LW, Gualberto A. Potential applications for circulating tumor cells expressing the insulin-like growth factor-I receptor. Clin Cancer Res 2007; 13: 3611-3616 [PMID: 17575225 DOI: 10.1158/1078-0432.CCR-07-0268]
- Wang L, Balasubramanian P, Chen AP, Kummar S, Evrard YA, Kinders RJ. Promise and limits of 48 the CellSearch platform for evaluating pharmacodynamics in circulating tumor cells. Semin Oncol 2016; 43: 464-475 [PMID: 27663478 DOI: 10.1053/j.seminoncol.2016.06.004]
- Danila DC, Samoila A, Patel C, Schreiber N, Herkal A, Anand A, Bastos D, Heller G, Fleisher M, 49 Scher HI. Clinical Validity of Detecting Circulating Tumor Cells by AdnaTest Assay Compared With Direct Detection of Tumor mRNA in Stabilized Whole Blood, as a Biomarker Predicting Overall Survival for Metastatic Castration-Resistant Prostate Cancer Patients. Cancer J 2016; 22: 315-320 [PMID: 27749322 DOI: 10.1097/PPO.00000000000220]
- 50 Chinen LT, de Carvalho FM, Rocha BM, Aguiar CM, Abdallah EA, Campanha D, Mingues NB, de Oliveira TB, Maciel MS, Cervantes GM, Dettino AL, Soares FA, Paterlini-Bréchot P, Fanelli MF. Cytokeratin-based CTC counting unrelated to clinical follow up. J Thorac Dis 2013; 5: 593-599 [PMID: 24255771 DOI: 10.3978/j.issn.2072-1439.2013.09.18]
- 51 Ramsköld D, Luo S, Wang YC, Li R, Deng Q, Faridani OR, Daniels GA, Khrebtukova I, Loring JF, Laurent LC, Schroth GP, Sandberg R. Full-length mRNA-Seq from single-cell levels of RNA and individual circulating tumor cells. Nat Biotechnol 2012; 30: 777-782 [PMID: 22820318 DOI: 10.1038/nbt.2282]
- 52 Sequist LV, Nagrath S, Toner M, Haber DA, Lynch TJ. The CTC-chip: an exciting new tool to detect circulating tumor cells in lung cancer patients. J Thorac Oncol 2009; 4: 281-283 [PMID: 19247082 DOI: 10.1097/JTO.0b013e3181989565]
- Sheng W, Ogunwobi OO, Chen T, Zhang J, George TJ, Liu C, Fan ZH. Capture, release and culture 53 of circulating tumor cells from pancreatic cancer patients using an enhanced mixing chip. Lab Chip 2014; 14: 89-98 [PMID: 24220648 DOI: 10.1039/c3lc51017d]
- Königsberg R, Obermayr E, Bises G, Pfeiler G, Gneist M, Wrba F, de Santis M, Zeillinger R, Hudec 54 M, Dittrich C. Detection of EpCAM positive and negative circulating tumor cells in metastatic breast cancer patients. Acta Oncol 2011; 50: 700-710 [PMID: 21261508 DOI: 10.3109/0284186X.2010.549151
- 55 Morris KL, Tugwood JD, Khoja L, Lancashire M, Sloane R, Burt D, Shenjere P. Circulating biomarkers in hepatocellular carcinoma. Cancer Chemother Pharmacol 2014; 74: 323-332 [DOI: 10.1007/s00280-014-2508-7]
- 56 Ramirez JM, Fehm T, Orsini M, Cayrefourcq L, Maudelonde T, Pantel K, Alix-Panabières C. Prognostic relevance of viable circulating tumor cells detected by EPISPOT in metastatic breast cancer patients. Clin Chem 2014; 60: 214-221 [PMID: 24255082 DOI: 10.1373/clinchem.2013.215079]
- Pfitzner C, Schröder I, Scheungraber C, Dogan A, Runnebaum IB, Dürst M, Häfner N. Digital-57 Direct-RT-PCR: a sensitive and specific method for quantification of CTC in patients with cervical carcinoma. Sci Rep 2014; 4: 3970 [PMID: 24496006 DOI: 10.1038/srep03970]
- Zhang Z, Xiao Y, Zhao J, Chen M, Xu Y, Zhong W, Xing J, Wang M. Relationship between 58 circulating tumour cell count and prognosis following chemotherapy in patients with advanced nonsmall-cell lung cancer. Respirology 2016; 21: 519-525 [PMID: 26661896 DOI: 10.1111/resp.12696]
- Bobek V, Gurlich R, Eliasova P, Kolostova K. Circulating tumor cells in pancreatic cancer patients: 59 enrichment and cultivation. World J Gastroenterol 2014; 20: 17163-17170 [PMID: 25493031 DOI: 10.3748/wig.v20.i45.17163
- 60 Maestro LM, Sastre J, Rafael SB, Veganzones SB, Vidaurreta M, Martín M, Olivier C, DE La Orden VB, Garcia-Saenz JA, Alfonso R, Arroyo M, Diaz-Rubio E. Circulating tumor cells in solid tumor in metastatic and localized stages. Anticancer Res 2009; 29: 4839-4843 [PMID: 20032444]
- 61 Das A, Kunkel M, Joudeh J, Dicker DT, Scicchitano A, Allen JE, Sarwani N, Yang Z, Kaifi J, Zhu J, Liao J, El-Deiry WS. Clinico-pathological correlation of serial measurement of circulating tumor cells in 24 metastatic colorectal cancer patients receiving chemotherapy reveals interpatient heterogeneity correlated with CEA levels but independent of KRAS and BRAF mutation. Cancer Biol Ther 2015; 16: 709-713 [PMID: 25806877 DOI: 10.1080/15384047.2015.1030555]
- Chiu TK, Chou WP, Huang SB, Wang HM, Lin YC, Hsieh CH, Wu MH. Application of optically-62



induced-dielectrophoresis in microfluidic system for purification of circulating tumour cells for gene expression analysis- Cancer cell line model. *Sci Rep* 2016; **6**: 32851 [PMID: 27609546 DOI: 10.1038/srep32851]

- 63 Tsai WS, Chen JS, Shao HJ, Wu JC, Lai JM, Lu SH, Hung TF, Chiu YC, You JF, Hsieh PS, Yeh CY, Hung HY, Chiang SF, Lin GP, Tang R, Chang YC. Circulating Tumor Cell Count Correlates with Colorectal Neoplasm Progression and Is a Prognostic Marker for Distant Metastasis in Non-Metastatic Patients. *Sci Rep* 2016; 6: 24517 [PMID: 27075165 DOI: 10.1038/srep24517]
- 64 Zhou J, Hu L, Yu Z, Zheng J, Yang D, Bouvet M, Hoffman RM. Marker expression in circulating cancer cells of pancreatic cancer patients. *J Surg Res* 2011; 171: 631-636 [PMID: 20869080 DOI: 10.1016/j.jss.2010.05.007]
- 65 Wang JY, Lin SR, Wu DC, Lu CY, Yu FJ, Hsieh JS, Cheng TL, Koay LB, Uen YH. Multiple molecular markers as predictors of colorectal cancer in patients with normal perioperative serum carcinoembryonic antigen levels. *Clin Cancer Res* 2007; 13: 2406-2413 [PMID: 17406027 DOI: 10.1158/1078-0432.CCR-06-2054]
- 66 Shen C, Hu L, Xia L, Li Y. Quantitative real-time RT-PCR detection for survivin, CK20 and CEA in peripheral blood of colorectal cancer patients. *Jpn J Clin Oncol* 2008; **38**: 770-776 [PMID: 18845519 DOI: 10.1093/jjco/hyn105]
- 67 Wang JY, Wu CH, Lu CY, Hsieh JS, Wu DC, Huang SY, Lin SR. Molecular detection of circulating tumor cells in the peripheral blood of patients with colorectal cancer using RT-PCR: significance of the prediction of postoperative metastasis. *World J Surg* 2006; **30**: 1007-1013 [PMID: 16736329 DOI: 10.1007/s00268-005-0485-z]
- 68 Lu Y, Liang H, Yu T, Xie J, Chen S, Dong H, Sinko PJ, Lian S, Xu J, Wang J, Yu S, Shao J, Yuan B, Wang L, Jia L. Isolation and characterization of living circulating tumor cells in patients by immunomagnetic negative enrichment coupled with flow cytometry. *Cancer* 2015; **121**: 3036-3045 [PMID: 25945459 DOI: 10.1002/cncr.29444]
- 69 Buim ME, Fanelli MF, Souza VS, Romero J, Abdallah EA, Mello CA, Alves V, Ocea LM, Mingues NB, Barbosa PN, Tyng CJ, Chojniak R, Chinen LT. Detection of KRAS mutations in circulating tumor cells from patients with metastatic colorectal cancer. *Cancer Biol Ther* 2015; 16: 1289-1295 [PMID: 26252055 DOI: 10.1080/15384047.2015.1070991]
- 70 Eliasova P, Pinkas M, Kolostova K, Gurlich R, Bobek V. Circulating tumor cells in different stages of colorectal cancer. *Folia Histochem Cytobiol* 2017; 55: 1-5 [PMID: 28509310 DOI: 10.5603/FHC.a2017.0005]
- 71 Deng G, Krishnakumar S, Powell AA, Zhang H, Mindrinos MN, Telli ML, Davis RW, Jeffrey SS. Single cell mutational analysis of PIK3CA in circulating tumor cells and metastases in breast cancer reveals heterogeneity, discordance, and mutation persistence in cultured disseminated tumor cells from bone marrow. *BMC Cancer* 2014; 14: 456 [PMID: 24947048 DOI: 10.1186/1471-2407-14-456]
- 72 Lee HJ, Cho HY, Oh JH, Namkoong K, Lee JG, Park JM, Lee SS, Huh N, Choi JW. Simultaneous capture and in situ analysis of circulating tumor cells using multiple hybrid nanoparticles. *Biosens Bioelectron* 2013; 47: 508-514 [PMID: 23628845 DOI: 10.1016/j.bios.2013.03.040]
- 73 Yu H, Ma L, Zhu Y, Li W, Ding L, Gao H. Significant diagnostic value of circulating tumour cells in colorectal cancer. *Oncol Lett* 2020; 20: 317-325 [PMID: 32565958 DOI: 10.3892/ol.2020.11537]
- 74 Cohen SJ, Punt CJ, Iannotti N, Saidman BH, Sabbath KD, Gabrail NY, Picus J, Morse M, Mitchell E, Miller MC, Doyle GV, Tissing H, Terstappen LW, Meropol NJ. Relationship of circulating tumor cells to tumor response, progression-free survival, and overall survival in patients with metastatic colorectal cancer. *J Clin Oncol* 2008; **26**: 3213-3221 [PMID: 18591556 DOI: 10.1200/JCO.2007.15.8923]
- 75 Gazzaniga P, Gradilone A, Petracca A, Nicolazzo C, Raimondi C, Iacovelli R, Naso G, Cortesi E. Molecular markers in circulating tumour cells from metastatic colorectal cancer patients. *J Cell Mol Med* 2010; 14: 2073-2077 [PMID: 20597995 DOI: 10.1111/j.1582-4934.2010.01117.x]
- 76 Pohl M, Werner N, Munding J, Tannapfel A, Graeven U, Nickenig G, Schmiegel W, Reinacher-Schick A. Biomarkers of anti-angiogenic therapy in metastatic colorectal cancer (mCRC): original data and review of the literature. *Z Gastroenterol* 2011; **49**: 1398-1406 [PMID: 21964893 DOI: 10.1055/s-0031-1281752]
- 77 Lugli A, Iezzi G, Hostettler I, Muraro MG, Mele V, Tornillo L, Carafa V, Spagnoli G, Terracciano L, Zlobec I. Prognostic impact of the expression of putative cancer stem cell markers CD133, CD166, CD44s, EpCAM, and ALDH1 in colorectal cancer. *Br J Cancer* 2010; 103: 382-390 [PMID: 20606680 DOI: 10.1038/sj.bjc.6605762]
- 78 Bahnassy AA, Salem SE, Mohanad M, Abulezz NZ, Abdellateif MS, Hussein M, Zekri CAN, Zekri AN, Allahloubi NMA. Prognostic significance of circulating tumor cells (CTCs) in Egyptian nonmetastatic colorectal cancer patients: A comparative study for four different techniques of detection (Flowcytometry, CellSearch, Quantitative Real-time PCR and Cytomorphology). *Exp Mol Pathol* 2019; **106**: 90-101 [PMID: 30578762 DOI: 10.1016/j.yexmp.2018.12.006]

World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1182-1192

DOI: 10.5306/wjco.v12.i12.1182

ISSN 2218-4333 (online)

MINIREVIEWS

Immunotherapy combinations and chemotherapy sparing schemes in first line non-small cell lung cancer

María Sereno, Oliver Higuera, Patricia Cruz Castellanos, Sandra Falagan, Xabier Mielgo-Rubio, Juan Carlos Trujillo-Reyes, Felipe Couñago

ORCID number: María Sereno 0000-0002-0207-2118; Oliver Higuera 0000-0002-6807-4511: Patricia Cruz Castellanos 0000-0002-9837-825X; Sandra Falagan 0000-0001-7172-8873; Xabier Mielgo-Rubio 0000-0002-0985-6150; Juan Carlos Trujillo-Reyes 0000-0002-3370-0869; Felipe Couñago 0000-0001-7233-0234.

Author contributions: Sereno M coordinated the literature review, performed the research, and wrote the manuscript; Higuera O, Cruz Castellanos P and Falagan S contributed to the writing of different parts of the manuscript; Mielgo-Rubio X, Trujillo JC and Couñago F critically reviewed the manuscript for intellectual content and all the authors approved the final version.

Conflict-of-interest statement:

Mielgo-Rubio X reports personal fees and non-financial support from ROCHE, personal fees from ASTRA ZENECA, grants, personal fees and non-financial support from BMS, personal fees from MSD, personal fees from ABBOTT, personal fees from KIOWA-KIRIN, outside the submitted work. Rest of authors has nothing to disclose.

Country/Territory of origin: Spain

Specialty type: Oncology

María Sereno, Sandra Falagan, Medical Oncology Department, Hospital Universitario Infanta Sofía, San Sebastián de los Reyes 28702, Madrid, Spain

Oliver Higuera, Patricia Cruz Castellanos, Medical Oncology Department, Hospital Universitario La Paz, Madrid 28046, Spain

Xabier Mielgo-Rubio, Medical Oncology Department, Hospital Universitario Fundación Alcorcó n, Alcorcón 28922, Madrid, Spain

Juan Carlos Trujillo-Reyes, Department of Thoracic Surgery, Hospital de la Santa Creu I Sant Pau, Barcelona 08029, Catalonia, Spain

Juan Carlos Trujillo-Reyes, Department of Surgery, Universitat Autonoma de Barcelona, Barcelona 08029, Catalonia, Spain

Felipe Couñago, Department of Radiation Oncology, Hospital Universitario Quirónsalud Madrid, Pozuelo de Alarcón 28223, Madrid, Spain

Felipe Couñago, Department of Radiation Oncology, Hospital La Luz, Madrid 28003, Spain

Felipe Couñago, Medicine Department, School of Biomedical Sciences, Universidad Europea de Madrid, Villaviciosa de Odón 28670, Madrid, Spain

Corresponding author: María Sereno, MD, PhD, Consultant Physician-Scientist, Medical Oncology Department, Hospital Universitario Infanta Sofia, Paseo Europa 34, San Sebastián de los Reyes 28702, Madrid, Spain. mariasereno75@gmail.com

Abstract

In recent years, studies have explored different combinations of immunotherapy and chemotherapy. The rationale behind these is the improved survival outcomes of new immunologic therapies used in first-line-treatment of advanced non-small cell lung cancer. Moreover, for the most-studied combinations of anti-programed death-1 (PD-1)/programed death ligand-1 (PD-L1) with the addition of platinumbased chemotherapy, recent research is investigating whether combining different immunologic antitumoral mechanisms of action, such as anti-PD-1/PD-L1 and anti-CTLA-4, or anti-PD-L1 and anti-TIGIT, with or without chemotherapy, can improve efficacy outcomes compared with more classical combinations, or compared with standard chemotherapy alone. Here, we present the data of the



Provenance and peer review:

Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): 0 Grade B (Very good): 0 Grade C (Good): C Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt p://creativecommons.org/License s/by-nc/4.0/

Received: April 28, 2021 Peer-review started: April 28, 2021 First decision: July 16, 2021 Revised: July 28, 2021 Accepted: November 24, 2021 Article in press: November 24, 2021 Published online: December 24, 2021

P-Reviewer: Chen LJ S-Editor: Liu M L-Editor: A P-Editor: Liu M



main randomized studies that have evaluated these combinations, focusing on the basic rationale behind the different combinations, and the efficacy and tolerability data available to date.

Key Words: Immunotherapy combinations; Anti- programmed cell death protein; Antiprogrammed cell death ligand 1; anti-TIGIT; anti-CTLA-4; Combo; Non-small cell lung cancer

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: This review presents the results of the main articles reporting on immunotherapy combinations, with or without chemotherapy, focusing principally on efficacy and toxicity data. The convenience of adding shorter chemotherapy regimens vs the standard 4-cycle regimens to immunotherapy doublets is discussed.

Citation: Sereno M, Higuera O, Cruz Castellanos P, Falagan S, Mielgo-Rubio X, Trujillo-Reyes JC, Couñago F. Immunotherapy combinations and chemotherapy sparing schemes in first line non-small cell lung cancer. World J Clin Oncol 2021; 12(12): 1182-1192

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1182.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1182

INTRODUCTION

Lung cancer is the leading cause of cancer-related deaths worldwide. An increased knowledge of lung cancer pathology as well as of tumoral immunity has permitted the discovery of different targeted molecular treatments and immune-therapies involving an important change of the lung cancer paradigm^[1].

Anti-programed death-1/programed death ligand-1 (PD-1/PD-L1) treatment, recently approved in advanced lung cancer treatment, is an important immune checkpoint of anti-tumor activity in our immune system. Cytotoxic T lymphocyteassociated antigen 4 (CTLA-4) is another component of these control immune checkpoints and different inhibitors have been developed to look their role in tumor recognition[2] (Figures 1 and 2).

Non-small cell lung cancer (NSCLC) has a high level of heterogeneity secondary to a different molecular immune subtype and a complex immune environment. This complexity justifies heterogenous responses to immunotherapy (IT) and their combinations^[2].

The administration of mono-IT (m-IT) with anti-PD-L1/PD-1, or a combination of this type of agent with platinum-based chemotherapy (CT), is the standard therapy for the different subtypes of non-small cell lung cancer without treatable mutations. Several studies focusing on combinations of different checkpoint inhibitors, such as anti-PD-1/PD-L1 and anti-CTLA-4, have shown these to be advantageous compared with CT alone [3,4]. The association of these IT combinations with CT has also been analyzed in two phase III studies [Checkmate 9LA study (CM9LA) and CCTG BR.34], with interesting results in favor of the new combinations [5,6]. The CYTISCAPE study, presented recently in ASCO 2020, proposes a combination of anti-PD-L1 and anti-Tcell immunoreceptor with Ig and ITIM domains (TIGIT), a new checkpoint inhibitor present in activated T-cells. This phase II study reported favorable outcomes for this IT doublet (d-IT) in the population with PD-L1 > 1% vs monotherapy with anti-PD-L1[7].

Here, we briefly summarize the most significant aspects of these new combinations.

BIOLOGICAL BASIS OF IT COMBINATIONS

The biological rationale for combining IT is based on the complexity of immune surveillance regulation, the effect of other anti-tumor treatments on tumor immunogenicity, and the non- redundancy of immune checkpoint regulation[8].

Sereno M et al. Immunotherapy combos in first line NSCLC



Figure 1 A schematic diagram showing programed death-1/ programed death ligand-1 interaction with immune cells in cancer. PD-1: Programed death-1; PD-LI: Programed death ligand-1.





Most chemotherapeutic agents are regarded as immunosuppressants, exerted this effect by a direct inhibition or killing of effector cells, as well as indirectly by inducing anergy or immune paralysis[9]. However, combinations of CT and ICI have a proven synergistic effect: CT aids in fast tumor regression, and a reduction in tumor burden and immune checkpoint blockade may prolong this effect, thus inducing a long-lasting anti-tumor response^[10]. In addition, the non-overlapping toxicity profiles of IT and CT render them good candidates for combination strategies.

In addition to immune regulation, angiogenesis is another hallmark of cancer evolution, promoting tumor vascularity to sustain cancer cell metabolism and invasive capacity, and also favors cancer immune escape. Vascular endothelial growth factor, one of the main regulators of angiogenesis, can also induce T-reg cells and inhibit dendritic cell and T-cell maturation, thus inducing an immunosuppressive status, lymphocyte migration to the tumor site, and negatively regulating immune priming [11,12]. Combining anti-angiogenics with immune checkpoint inhibitors could thus be synergic in reducing tumor vascularization and decreasing the immune suppressive microenvironment. It may also have a synergistic effect by boosting tumor-specific



immunity.

Immune checkpoints exert a negative regulatory effect on CD8+ T cells by interacting with their ligands expressed on different cells, such as APC or T cells. Preclinical data indicate that the combination of PD-1 and CTLA-4 receptor blockade may improve antitumor activity. In vitro combinations of nivolumab plus ipilimumab increase IFN-y production 2- to 7-fold over either agent alone in a mixed lymphocyte reaction (Table 1). The combination was also observed to exert an increased antitumor activity in 3 of 5 syngeneic murine cancer models. In a murine melanoma vaccine model, blockade with either CTLA-4 or PD-1 antibodies increased the proportion of CTLA-4 and PD-1-expressing CD4/CD8 tumor infiltrating T effector cells, and dual blockade increased tumor infiltration of T effector cells and decreased intratumoral T regulatory cells, compared with either agent alone^[13].

TIGIT is a novel immune inhibitory receptor. It is a member of the immunoglobulin superfamily and is expressed in a wide variety of human tumors. TIGIT is expressed on the surface of activated T-cell and natural killer (NK)-cell subsets and interacts with high affinity with CD155 [also known as poliovirus receptor (PVR)]. It is also highly correlated with T-cell infiltration and PD-1 expression[14]. Genetic ablation of TIGIT in T cells in mice results in exacerbated T-cell responses in preclinical models of autoimmune diseases and viral infections, demonstrating the role of TIGIT in the inhibition of T-cell responses. Because TIGIT and PD-1 are coordinately expressed by tumor-infiltrating T cells in several human tumors, inhibition of the TIGIT/PVR pathway may complement and potentiate the anti-tumor activity of a PD-L1 pathway inhibitor. In preclinical models, concomitant blockade of both TIGIT and PD-L1/PD-1 pathways demonstrated superior efficacy over the respective single-agent treatments [15].

In conclusion, based on preclinical and clinical studies, there is a strong rationale for combining IT and other agents, and this potentially synergistic effect has been demonstrated with improved outcomes in the clinical setting.

PHASE I-III STUDIES

As explained previously, there is a clear justification for combining different therapeutic strategies. Consequently, several studies have been designed to explore the clinical efficacy of these combinations in the context of advanced NSCLC.

Since pembrolizumab was first approved in first-line therapy of patients with PDL1 > 50%, several studies, which we describe below, have analyzed the combination of anti PD-1/PDL-1 and anti-CTLA-4, with or without the addition of CT[14] (Table 2).

Checkmate 012 (CM012) was the first phase I study to explore the combination of ipilimumab-nivolumab (NI) vs nivolumab (N) in CT-naïve patients with advanced NSCLC. This study assessed three alternative regimens with different doses of I and N, with 6 weekly vs 12 weekly schemes of I, with tolerability as the primary endpoint. This study confirmed the safety of the combination, predicting a high response rate and improved survival (OS)[17].

With the data from CM012, Checkmate 227 (CM227), a phase III study was proposed. This study was comprised of three parts: Part1A that assessed the efficacy of the combination of NI vs N in monotherapy vs standard CT in patients with negative PD-L1 expression, part 1B that compared the combination of NI vs N + CT vs standard CT in PD-L 1 negative patients, and part 2 which evaluated the efficacy of the combination of N + CT, regardless of histology and PD-L1 expression. The study reported a global benefit for the NI doublet in OS, independently of PD-L1 [17.1 ms vs 13.9 ms, hazard ratio (HR) 0.73; 95%CI: 0.64-0.84]. When these data are analyzed in relation to PD-L1, several important observations can be made: in the PD-L1 negative group the benefit in OS in favor of the IT doublet was maintained at 17.2 vs 12.2 mo (HR 0.62, 95%CI: 0.48-0.78). In the PD-L1 positive group ($\geq 1\%$), OS was also better with the combined IT treatment (17.1 ms vs 14.9 ms, HR 0.79; 95%CI: 0.65-0.96). Noteworthily, in this group with 1-49% PD-L1 expression, no differences in OS were found. Also in this study, data of levels of tumor mutational burden (TMB) were reported early on. Hence, in the high TMB group (> 10 mut/Mb), a statistically significant benefit in favor of the combination was reported. Finally, the results of part 2 were communicated, in which the primary final endpoint of OS for N-CT was not reached in patients with non-squamous NSCLC[3].

The CM9LA is a phase III trial with a novel design that adds two cycles of CT to the NI doublet. In the control arm, the CT regimen preferred by the clinician was established, or maintenance with permetrexed in non-squamous histology. With a



Table 1 Main programed death ligand-1 and cytotoxic T lymphocyte-associated antigen 4 pathways

Checkpoint pathways	Anti-PD-L1 pathway	Anti-CTLA-4 pathway
Receptor expression	Activated T-cells	Activated T-cells, B-cells and NK cells
Ligands	B 7.1 (CD80); B7.2 (CD86)	PD-L1 (B7-H1), PD-L2 (B7-DC)
Mechanism of immune modulation	T-cell activation at initial stage; Competition with co- stimulatory receptor CD-28 for ligand binding; Down regulation of helper T cells CD4 activity; Enhancement Tregs- cells immunosuppressive activity	Suppresses activated T cells in tissues and tumor environment; Express in Tregs-cells may enhance immunosuppressive activity; Limits of B-cells and NK- cells activity; PD-L1 interact with CD-80 to down-modulate T-cells activity

NK: Natural killer; PD-L1: Programed death ligand-1; CTLA-4: Cytotoxic T lymphocyte-associated antigen 4.

Table 2 Summary of the main Phase III studies of IT combinations						
		Drugs	Endpoints/outcome	PD-L1/TMB		
Trial	Ph ¹			Population		
				Subgroups		
Checkmate 227	III	NI vs N vs CT ¹	OS PD-L1 > 1% and in non- selected population	PD-L1 ≥ 1%: NI 17 m <i>vs</i> CT 14.9 m, $P = 0.007$		
			NI vs CT	PD-L1 < 1%: NI 17 m <i>vs</i> CT 12.2 m, <i>P</i> =0.007		
			17 m vs 13.9 m			
MYSTIC	III	II DT vs D vs CT ¹	OS D <i>vs</i> CT OS and PFS DT <i>vs</i> CT PD-L1 > 25% population	PD-L1 > 25%: D vs CT ¹		
				16 m vs 12 m, P = 0.04		
				DT <i>vs</i> CT ¹ 11.9 <i>vs</i> 12 m, <i>P</i> = 0.2		
				TMB > 20		
				DT vs CT ¹ 21 m vs 10 m		
Checkmate9LA	III	NI + $CT^1 X2 vs CT^1 X4$	OS in non selected population	NA		
			OS NI-CT 15.6 m <i>vs</i> CT 10.9 m, <i>P</i> = 0.0006			
CCTG BR.34	III	DT+ CT ¹ x 4 vs Durvalumab + tremelimumab (DT)	OS in non selected population	PD-L1 TPS ≥ 50%, DT-CT <i>vs</i> DT (HR 0.64, 95%CI: 0.40-1.04, $P = 0.07$). Plasma TMB < 20 mut/Mb was associated with shorter survival in both treatment groups (HR 1.99, 95%CI: 1.3-3.1)		
			16.6 m DT-CT vs 14 m DT			
CITYSCAPE	Π	TA vs PA	ORR and PFS en PD-L1 > 1% PFS TA 7.7 m <i>vs</i> PA 3.2 m	PD-L1 > 50%		
			ORR TA 37% vs PA 20%	PFS TA NA vs PA 4.1		
				ORR TA 66 % <i>vs</i> PA 24%		

¹Ph: Phase.

NI: Nivolumab + Ipilimumab; N: Nivo; DT: Tremelimumab + Durvalumab; D: Durvalumab; PD-1: Programed death-1; TMB: Tumor mutational burden; HR: Hazard ratio; CI: Confidence interval; TA: Tiragolumab + atezolizumab; PA: Placebo + atezolizumab; NA: Not available; CT: Chemotherapy.

> follow up of 12.7 mo, the experimental arm showed a significant improvement in OS in all subgroups (15.6 ms vs 10.9 ms, HR 0.66; 95%CI: 0.55-0.8)[5].

> The combination of Durvalumab (D) and Tremelimumab (T) has also been explored in different phase I-III studies. One of the first phase III trials to analyze this combination in NSCLC was the ARTIC study. This was comprised of two independent sub-studies: one focused on the PD-L1 > 25% population, comparing D vs standard CT in patients pre-treated with platinum, and another in the population with PD-L1 < 25% with three arms: DT vs D vs T in the same patient profile. In both sub-studies, the data from the D and DT arms were more favorable than CT alone[18]. Continuing with this combination of DT in first-line, the MYSTIC phase III study[4] once again focused

on the population of PDL-1 > 25%. The patients were randomized to receive D vs DT vs platinum-based CT, in this case in first-line of NSCLC. The efficacy data for the mono-IT vs CT showed a non-significant superiority for the former (16 mo vs 12 mo), although in the d-IT the median OS of 11.9 mo was shorter than that obtained with CT. No benefit in favor of IT was found for PFS either. However, in this study TMB was included as a predictive biomarker of response to IT, with the subgroup TMB > 20presenting an OS in favor of the IT doublet (21.9 mo vs 10 mo).

Continuing to focus on the DT combination, CGC BR.34, another phase III study emerged, in parallel with CM9LA, to address the need to explore the efficacy of d-IT. In this work, 4 CT cycles (platinum-pemetrexed) were added to the DT doublet in firstline therapy in a non-squamous population. Maintenance treatment was allowed in both arms. After a six-month follow up, no differences were found in OS, although the analysis of PFS revealed statistically significant differences in favor of the combined treatment of IT-CT (7.7 mo vs 3.2 mo)[6].

CITYSCAPE is a phase II trial that evaluates the possible efficacy of combining Tiragolumab (Ti) (an inhibitor of the immune modulatory receptor TIGIT) with Atezolizumab (A) in first-line of NSCLC with PDL > 1%. Preliminary phase I studies (GO 30103) showed the benefit of combining TIGIT inhibition and anti-PD-L1, revealing a greater response rate in pre-treated patients. The first data from this study were presented in ASCO 2020, and the preliminary analysis was positive in favor of the combined approach (TiA vs A-placebo) (5.6 ms vs 3.9 ms, HR 0.58; 95% CI: 0.38-0.89). Moreover, an exploratory analysis in patients with high PD-L1 expression, or with a tumor proportion score over 50%, obtained a reduction of 70% in risk of disease progression or death from the illness with TiA vs A-P (HR 0.30; 95% CI: 0.15-0.61). This study appears to confirm the effectiveness of combining a dual inhibition of immunotherapeutic mechanisms and their potential benefit in clinical practice, although the final results and the study design of the phase III trial are still pending[7].

WHAT ARE THE BENEFITS OF IT COMBINATIONS TOGETHER WITH SHORT CT CYCLES?

Several preclinical studies have arrived at an interesting rationale to support CT-IT combinations. These combinations are an important challenge in NSCLC treatment, not only in advanced disease, but also in a neoadjuvant setting, where phase III studies such as CM816 have demonstrated a significant benefit of CT-IT pre-surgery compared to classical platinum based CT[19]. Although CT agents are regarded as immunosuppressants by directly inhibiting or killing effector cells, as well as indirectly by inducing anergy or immune paralysis[20], different combinations of CT and ICI have a proven synergistic effect. For example, CT aids in fast tumor regression and a reduction in tumor burden, and immune checkpoint blockade may prolong this effect, thus inducing a long-lasting anti-tumor response^[21]. In addition, the non-overlapping toxicity profiles of IT and CT render them good candidates for combination strategies. On the hand, preclinical and clinical investigations have revealed that CT could enhance the efficacy of IT through various mechanisms. CT can induce immunogenic cell death and increase tumor antigen presentation and immune effector infiltration in tumors. Moreover, CT can modulate the immune microenvironment through mechanisms such as increasing cancer cell immunogenicity, enhancing the cytotoxicity of T cells and NK cells, and fostering the accumulation of IFN-y and tumor necrosis factor (TNF)- α . The latest data show that CT also reduces the number of tumor immune suppressive cells like T regulatory cells, macrophages and neutrophils, and increases the ratio of cytotoxic T-cell/T regulatory cells and PD-L1 expression on tumor cells[22].

Specifically, platinum agents have been shown to partly exert their anti-tumor effect via modulation of the immune system. They attenuate STAT6 signaling by blocking STAT6 phosphorylation, resulting in a downregulation of PD-L2 on DCs and tumor cells. This triggers increased tumor cell recognition by T lymphocytes^[23]. In a study of NSCLC cell lines, cisplatin upregulated the MHC class I chain-related molecule A and B expression, and led to enhanced NK cell-mediated antitumor effect^[24]. Other mechanisms by which platinum agents stimulate the immune system include increasing human leukocyte 1 gene complex expression encoding MHC-I, which is associated with cytotoxic T-cell function[25].

As mentioned in previous sections, a number of clinical trials, such as the one described above, have focused on the administration of combinations of IT in first-line therapy. Overall, it is found that an unblocking of checkpoints of antitumor response



(PD-L1/CTLA-4), using combinations of checkpoint inhibitors, has proved more effective than CT alone or IT in monotherapy. The CM 227 trial, one of the first phase III studies that compared NI vs CT, reported a greater OS in the former treatment, regardless of PD-L1 expression[3].

In spite of the overall positivity of the CM 227 trial, an initial crossing of the curves takes place in the first 3-6 mo of follow up, both for the entire study population and also in patients with PDL-1 > 1%. This means that CT would be superior to IT in a profile of patients with more aggressive disease, who would potentially be more chemosensitive.

Another parallel study to CM227 in the PDL-1 > 25% population, also mentioned previously, is the MYSTIC study[4], which compares DT vs D vs CT. However, the results of this study were more disappointing, reporting no greater superiority for the IT doublet vs CT, although DT was more effective in the subgroup with TMB > 20. The CITYSCAPE study positions a new combination of IT (Ti-A) as another option in firstline therapy, and once again PD-L1 is a good predictor correlated with a greater benefit in favor of the combination.

Therefore, in spite of being a new option in first-line, d-IT was no more beneficial than its comparators in any of the studies analyzed, and which have been described here

Subsequent studies, which aimed at improving the efficacy data of these schemes, incorporated CT regimens of different durations in an attempt to maximize the benefits of these new combinations.

With the goal of compensating for this initial deficit in OS in patients in the CM227 study, the CM9LA study added two cycles of platinum-based CT to the NI combination and achieved a greater benefit in OS for all the subgroups vs treatment with CT alone. However, as also reported previously in other studies on IT, no benefit in OS was found for patient smokers or for those aged over 75[5].

Another study that adds CT to the IT doublet is the aforementioned CCTG BR.34 study. This was designed with the same goal, in other words, to try to obtain better outcomes than the MYSTIC study[6]. In this case, efficacy data after a 16-month follow up found no differences in OS (with a non-significant favorable tendency in the PD-L1 > 50% group). However, differences were found in the response rates (28% vs 14%) and in PFS 7.7 vs 3.2 in favor of the combination with CT. A longer follow-up would probably produce more mature conclusions about the benefits of the IT doublet-CT combination vs the comparator with single IT-CT. In this case, in contrast to the MYSTIC study, there are no data to support a benefit in patients with a high TMB, although, once again, the high expression of PD-L1 would seem to predict a greater benefit.

In the debate about combinations of IT and CT, there is some controversy about the number of platinum cycles to use. Hence, the CM9LA study used 2 cycles while in the CCTG BR34 study, 4 cycles were administered. Although there are no data to attribute the difference in results to the number of CT cycles given, it is interesting that in the study with more CT cycles, in which better results would be expected, the data were, in fact, negative. It is, therefore, unclear whether the 2 first cycles are insufficient to compensate for the initial fall in the survival curve due to patients with an early sensitivity to IT.

On the other hand, there is no doubt about the rise in platinum-based toxicity of the longer CT schemes. Therefore, prospective studies may be necessary to determine whether reduced CT schemes, with greater tolerability, in combination with the IT doublet could be at least as effective as standard 4-cycle schemes. Another important aspect to consider about these new combinations concerns the choice of the best biomarker to predict benefits with the IT combination. Compared with PD-L1 expression, in the MYSTIC study higher levels of TMB predicted a greater benefit of the combo-IT. However, these results were not reproduced in other studies that analyzed this marker. For combinations of CT with anti-TIGIT/anti-PD-L1, data have not yet been published.

All these studies have only a short follow up and the results of longer follow up periods are required before we can confirm potential benefits for these new combinations of different and complementary therapeutic strategies.

TOXICITY

The toxicity of combinations with IT, whether this corresponds to d-IT or IT-CT, is highly variable, with cytostatic toxicity in addition to immune-mediated adverse



effects. The rates of grade 3-4 toxicities for the different studies on the d- IT combination are presented in Table 3.

Combinations of IT with the NI scheme in the CM227 trial[3], and with DT in the MYSTIC trial[4], show global rates of side effects of 76.7% and 60.1%, respectively; and of 54.2% in arm D. In both studies, higher rates were obtained for the control arm of CT, of 81.9% in CM227 and 83% in MYSTIC. For toxicity rates of G3 or higher, rates of 32.8% were recorded for the NI combination, 22.9% for the DT combination, 14.9% for D in monotherapy, and 33.8% and 36% in the CT arm, thus confirming in both studies that CT was more toxic than IT. Side effects of G3 or higher resulted in interruption of treatment in 9.4%-12.3% in the combination with the IT doublet, in 3.4%-4.9% in the CT arm and in 4.3% in arm D in monotherapy. As expected, the most frequent toxicities of CT were classical toxicities, such as pancytopenia, especially anemia and G3 neutropenia in 10%, and other common effects such as vomiting and asthenia. Immune-mediated adverse events were reported in 13.6% of arm D in monotherapy and in 28.3% with the DT combination. The most frequent immune-mediated toxicities corresponded to cutaneous reactions (30%-35%) and endocrine events (23%-25%). As published previously in other trials and studies, toxicity was also proportional to the level of PD-L1 expression. In CM227, patients with a level of expression of PD-L1 < 1%, presented less treatment-related G 3-4 toxicity (27.0 in PD-1 < 1% vs 32% general population).

As expected, combinations of IT-CT were also associated with an increased toxicity. In the CM 9LA trial[3], the global rate of adverse effects of G3 or higher was 47% in the experimental arm with NI-CT vs 38% in the control arm with CT, and slightly higher in the arms with CT of CM227[3] and MYSTIC[4]. In the CCTG BR.34 study[6], the combination of DT-CT presented side effects of G3 or higher in 82%, and in the control arm with DT alone of 70%, with rates of 22.9% reported in the MYSTIC study[4]. It is noteworthy that this toxicity is also higher, if results are extrapolated, than the toxicities of studies using the CT-IT combination such as Keynote 189, where a G3 toxicity of 67.2% was recorded[26].

The most common treatment-related serious adverse events of any grade were more frequent in the experimental group. These included diarrhea, febrile neutropenia, thrombocytopenia and anemia, with rates of 2%-3%. Treatment-related adverse events of severity G 3-4 that led to interruption of treatment were reported in 16%, and were higher than those reported for IT combinations in CM227[3], MYSTIC[4] and Keynote 189[26].

In studies of the doublet IT-CT, the most frequent treatment-related adverse events of any grade, which are commonly associated with CT, were anemia (23%-38%), neutropenia (10%-17%) and thrombocytopenia (5%-10%) in CM 9LA[5]. Grade 3-4 adverse events in the CCTG BR.34 study corresponded to neutropenia (7%-9%), anemia (6%-14%), diarrhea (1%-4%), elevated lipase levels (1%-6%) and febrile neutropenia (3%-4%). In the CM9LA study[3], the commonest immune-mediated adverse events of level G 3-4 were gastrointestinal (6%), cutaneous (4%) and hepatic (4%). Most of these were successfully resolved, with some patients requiring immune modulating medication, including corticosteroids and TNF antagonists. In the CCTG BR 34 trial[4], the incidence of adverse events related to the immune system was similar in all arms (colitis 11%, pneumonitis 6%, endocrinopathy 21%).

The toxicity of the combination of A and Ti was not negligible, (CITYSCAPE)[7], with global rates of side effects of 80.6% vs 72%, respectively, and G3 toxicity rates of 19.1% vs 14.9%, and rates of interruption of treatment of 10.3% vs 7.5% for the control group. Once again, in the experimental arm the most frequent immune-mediated toxicity effects were skin rashes, perfusion reaction, pancreatitis, thyroid alterations and colitis.

The toxicity of the IT combinations is, therefore, greater than that of studies of single IT, at the expense of more immune-mediated effects, as to be expected. Similarly, toxicity increases when CT is added to these IT doublet schemes because of the toxicity associated with platinum regimens. Indirect comparisons of the more classical schemes of m-IT-CT vs d-IT-CT show greater toxicity in the latter, directly proportional to the number of CT cycles. Hence, in the decision-support algorithm to select the best first-line scheme in our patients, it is essential to take into account the toxicity profiles of each alternative treatment.

CONCLUSION

The combination of different checkpoint inhibitors, or checkpoint inhibitors combined



Table 3 Summary of the main toxicities in each of the studies on immunotherapy combinations or immunotherapy ± chemotherapy						
Dhane II III studios	G3-4 toxicity (%)		Treatment discontinuation (%)			
Phase II-III studies	Exp arm	Control arm	Exp arm	Control arm		
CM 227	32.8 NI	33.8 CT	9.4 NI	3.4 CT		
MYSTIC	22.9 DT	36 CT	12.3 DT	4.9 CT		
CM9LA	47 NI-CT	38 CT	16 NI-CT	9 CT		
CCTG-BG-34	82 DT-CT	70 DT	NA	NA		
CITYSCAPE	19 ATir	14 PA	10.3 ATir	7.5 PlA		
KEYNOTE-189	67 P-CT	65.8 CT	13.8 P-CT	7.9 CT		

NI: Nivolumab + ipilimumab; DT: Durvalumab+ tremelimumab; CT: Chemotherapy; Tir: Tirogolumab; A: Atezolizumab; PL: Placebo; P-CT: Pembrolizumab + chemotherapy; NA: Not available.

> with other immune modulating strategies, such as TIGIT inhibition, in the context of first-line therapy in advanced NSCLC produces superior outcomes to classical CT regimens. The addition of CT (of 2 to 4 cycles) to this IT doublet is a new strategy to rescue patients who could initially present a detrimental clinical course after first receiving the IT doublet. It is necessary to continue researching to find more accurate predictive biomarkers of response to optimize the selection of patients who should receive these types of therapeutic strategies.

ACKNOWLEDGEMENTS

To all the coauthors for the literature review and drafting of the manuscript.

REFERENCES

- 1 Niu M, Yi M, Li N, Luo S, Wu K. Predictive biomarkers of anti-PD-1/PD-L1 therapy in NSCLC. Exp Hematol Oncol 2021; 10: 18 [PMID: 33653420 DOI: 10.1186/s40164-021-00211-8]
- Qu J, Mei Q, Liu L, Cheng T, Wang P, Chen L, Zhou J. The progress and challenge of anti-PD-1/PD-2 L1 immunotherapy in treating non-small cell lung cancer. Ther Adv Med Oncol 2021; 13: 1758835921992968 [PMID: 33643442 DOI: 10.1177/1758835921992968]
- 3 Hellmann MD, Paz-Ares L, Bernabe Caro R, Zurawski B, Kim SW, Carcereny Costa E, Park K, Alexandru A, Lupinacci L, de la Mora Jimenez E, Sakai H, Albert I, Vergnenegre A, Peters S, Syrigos K, Barlesi F, Reck M, Borghaei H, Brahmer JR, O'Byrne KJ, Geese WJ, Bhagavatheeswaran P, Rabindran SK, Kasinathan RS, Nathan FE, Ramalingam SS. Nivolumab plus Ipilimumab in Advanced Non-Small-Cell Lung Cancer. N Engl J Med 2019; 381: 2020-2031 [PMID: 31562796 DOI: 10.1056/NEJMoa1910231]
- 4 Rizvi NA, Cho BC, Reinmuth N, Lee KH, Luft A, Ahn MJ, van den Heuvel MM, Cobo M, Vicente D, Smolin A, Moiseyenko V, Antonia SJ, Le Moulec S, Robinet G, Natale R, Schneider J, Shepherd FA, Geater SL, Garon EB, Kim ES, Goldberg SB, Nakagawa K, Raja R, Higgs BW, Boothman AM, Zhao L, Scheuring U, Stockman PK, Chand VK, Peters S; MYSTIC Investigators. Durvalumab With or Without Tremelimumab vs Standard Chemotherapy in First-line Treatment of Metastatic Non-Small Cell Lung Cancer: The MYSTIC Phase 3 Randomized Clinical Trial. JAMA Oncol 2020; 6: 661-674 [PMID: 32271377 DOI: 10.1001/jamaoncol.2020.0237]
- Paz-Ares L, Ciuleanu TE, Cobo M, Schenker M, Zurawski B, Menezes J, Richardet E, Bennouna J, 5 Felip E, Juan-Vidal O, Alexandru A, Sakai H, Lingua A, Salman P, Souquet PJ, De Marchi P, Martin C, Pérol M, Scherpereel A, Lu S, John T, Carbone DP, Meadows-Shropshire S, Agrawal S, Oukessou A, Yan J, Reck M. First-line nivolumab plus ipilimumab combined with two cycles of chemotherapy in patients with non-small-cell lung cancer (CheckMate 9LA): an international, randomised, openlabel, phase 3 trial. Lancet Oncol 2021; 22: 198-211 [PMID: 33476593 DOI: 10.1016/S1470-2045(20)30641-0]
- Leighl NB, Laurie SA, Goss GD, Hughes BGM, Stockler MR, Tsao MS, Kulkarni S, Blais N, Joy AA, Mates M, Rana P, Yadav S, Underhill C, Lee CW, Bradbury PA, Hiltz A, Dancey J, Ding K, Vera Badillo FE; Canadian Cancer Trials Group, Australasian Lung Cancer Trials Group. CCTG BR.34: A randomized trial of durvalumab and tremelimumab +/- platinum-based chemotherapy in patients with metastatic (Stage IV) squamous or nonsquamous non-small cell lung cancer (NSCLC). J *Clin Oncol* 2020; **38**: 9502 [DOI: 10.1200/jco.2020.38.15_suppl.9502]



- Rodriguez-abreu D, Johnson ML, Hussein MA, Cobo M, Patel AJ, Secen NM, Lee KH, Massuti B, 7 Hiret S, Yang JC, Barlesi F, Lee DH, Paz-ares LG, Hsieh RW, Miller K, Patil N, Twomey P, Kapp AV, Meng R, Cho BC. Primary analysis of a randomized, double-blind, phase II study of the anti-TIGIT antibody tiragolumab (tira) plus atezolizumab (atezo) versus placebo plus atezo as first-line (1L) treatment in patients with PD-L1-selected NSCLC (CITYSCAPE). J Clin Oncol 2020; 38: 9503 [DOI: 10.1200/jco.2020.38.15_suppl.9503]
- 8 Chen DS, Mellman I. Elements of cancer immunity and the cancer-immune set point. Nature 2017; 541: 321-330 [PMID: 28102259 DOI: 10.1038/nature21349]
- 9 Zitvogel L, Apetoh L, Ghiringhelli F, Kroemer G. Immunological aspects of cancer chemotherapy. Nat Rev Immunol 2008; 8: 59-73 [PMID: 18097448 DOI: 10.1038/nri2216]
- 10 Champiat S, Ileana E, Giaccone G, Besse B, Mountzios G, Eggermont A, Soria JC. Incorporating immune-checkpoint inhibitors into systemic therapy of NSCLC. J Thorac Oncol 2014; 9: 144-153 [PMID: 24419410 DOI: 10.1097/JTO.000000000000074]
- Ohm JE, Gabrilovich DI, Sempowski GD, Kisseleva E, Parman KS, Nadaf S, Carbone DP. VEGF 11 inhibits T-cell development and may contribute to tumor-induced immune suppression. Blood 2003; 101: 4878-4886 [PMID: 12586633 DOI: 10.1182/blood-2002-07-1956]
- 12 Manegold C, Dingemans AC, Gray JE, Nakagawa K, Nicolson M, Peters S, Reck M, Wu YL, Brustugun OT, Crinò L, Felip E, Fennell D, Garrido P, Huber RM, Marabelle A, Moniuszko M, Mornex F, Novello S, Papotti M, Pérol M, Smit EF, Syrigos K, van Meerbeeck JP, van Zandwijk N, Yang JC, Zhou C, Vokes E. The Potential of Combined Immunotherapy and Antiangiogenesis for the Synergistic Treatment of Advanced NSCLC. J Thorac Oncol 2017; 12: 194-207 [PMID: 27729297 DOI: 10.1016/j.jtho.2016.10.003]
- Curran MA, Montalvo W, Yagita H, Allison JP. PD-1 and CTLA-4 combination blockade expands 13 infiltrating T cells and reduces regulatory T and myeloid cells within B16 melanoma tumors. Proc Natl Acad Sci U S A 2010; 107: 4275-4280 [PMID: 20160101 DOI: 10.1073/pnas.0915174107]
- Yu X, Harden K, Gonzalez LC, Francesco M, Chiang E, Irving B, Tom I, Ivelja S, Refino CJ, Clark 14 H, Eaton D, Grogan JL. The surface protein TIGIT suppresses T cell activation by promoting the generation of mature immunoregulatory dendritic cells. Nat Immunol 2009; 10: 48-57 [PMID: 19011627 DOI: 10.1038/ni.1674]
- Johnston RJ, Comps-Agrar L, Hackney J, Yu X, Huseni M, Yang Y, Park S, Javinal V, Chiu H, 15 Irving B, Eaton DL, Grogan JL. The immunoreceptor TIGIT regulates antitumor and antiviral CD8(+) T cell effector function. Cancer Cell 2014; 26: 923-937 [PMID: 25465800 DOI: 10.1016/j.ccell.2014.10.018]
- 16 Hanna NH, Schneider BJ, Temin S, Baker S Jr, Brahmer J, Ellis PM, Gaspar LE, Haddad RY, Hesketh PJ, Jain D, Jaiyesimi I, Johnson DH, Leighl NB, Phillips T, Riely GJ, Robinson AG, Rosell R, Schiller JH, Singh N, Spigel DR, Stabler JO, Tashbar J, Masters G. Therapy for Stage IV Non-Small-Cell Lung Cancer Without Driver Alterations: ASCO and OH (CCO) Joint Guideline Update. J Clin Oncol 2020; 38: 1608-1632 [PMID: 31990617 DOI: 10.1200/JCO.19.03022]
- Hellmann MD, Rizvi NA, Goldman JW, Gettinger SN, Borghaei H, Brahmer JR, Ready NE, Gerber 17 DE, Chow LQ, Juergens RA, Shepherd FA, Laurie SA, Geese WJ, Agrawal S, Young TC, Li X, Antonia SJ. Nivolumab plus ipilimumab as first-line treatment for advanced non-small-cell lung cancer (CheckMate 012): results of an open-label, phase 1, multicohort study, Lancet Oncol 2017; 18: 31-41 [PMID: 27932067 DOI: 10.1016/S1470-2045(16)30624-6]
- 18 Planchard D, Reinmuth N, Orlov S, Fischer JR, Sugawara S, Mandziuk S, Marquez-Medina D, Novello S, Takeda Y, Soo R, Park K, McCleod M, Geater SL, Powell M, May R, Scheuring U, Stockman P, Kowalski D. ARCTIC: durvalumab with or without tremelimumab as third-line or later treatment of metastatic non-small-cell lung cancer. Ann Oncol 2020; 31: 609-618 [PMID: 32201234 DOI: 10.1016/j.annonc.2020.02.006]
- 19 Spicer J, Wang C, Tanaka F, Saylors GB, Chen K, Liberman M, Vokes EE, Girard N, Lu S, Provencio M, Mitsudomi T, Awad MM, Felip E, Forde PM, Swanson S, Brahmer JR, Kerr K, Dorange C, Cai J, Broderick S. Surgical outcomes from the phase 3 CheckMate 816 trial: Nivolumab (NIVO) + platinum-doublet chemotherapy (chemo) vs chemo alone as neoadjuvant treatment for patients with resectable non-small cell lung cancer (NSCLC). J Clin Oncol 2021; 39: 8503-8503 [DOI: 10.1200/jco.2021.39.15_suppl.8503]
- Zitvogel L, Galluzzi L, Smyth MJ, Kroemer G. Mechanism of action of conventional and targeted 20 anticancer therapies: reinstating immunosurveillance. Immunity 2013; 39: 74-88 [PMID: 23890065 DOI: 10.1016/j.immuni.2013.06.014]
- Mathew M, Enzler T, Shu CA, Rizvi NA. Combining chemotherapy with PD-1 blockade in NSCLC. 21 Pharmacol Ther 2018; 186: 130-137 [PMID: 29352857 DOI: 10.1016/j.pharmthera.2018.01.003]
- 22 Okita R, Yukawa T, Nojima Y, Maeda A, Saisho S, Shimizu K, Nakata M. MHC class I chainrelated molecule A and B expression is upregulated by cisplatin and associated with good prognosis in patients with non-small cell lung cancer. Cancer Immunol Immunother 2016; 65: 499-509 [PMID: 26940474 DOI: 10.1007/s00262-016-1814-9]
- 23 Jackaman C, Majewski D, Fox SA, Nowak AK, Nelson DJ. Chemotherapy broadens the range of tumor antigens seen by cytotoxic CD8(+) T cells in vivo. Cancer Immunol Immunother 2012; 61: 2343-2356 [PMID: 22714286 DOI: 10.1007/s00262-012-1307-4]
- Huang X, Cui S, Shu Y. Cisplatin selectively downregulated the frequency and immunoinhibitory 24 function of myeloid-derived suppressor cells in a murine B16 melanoma model. Immunol Res 2016; 64: 160-170 [PMID: 26590944 DOI: 10.1007/s12026-015-8734-1]



- 25 Liu WM, Fowler DW, Smith P, Dalgleish AG. Pre-treatment with chemotherapy can enhance the antigenicity and immunogenicity of tumours by promoting adaptive immune responses. Br J Cancer 2010; 102: 115-123 [PMID: 19997099 DOI: 10.1038/sj.bjc.6605465]
- 26 Gandhi L, Rodríguez-Abreu D, Gadgeel S, Esteban E, Felip E, De Angelis F, Domine M, Clingan P, Hochmair MJ, Powell SF, Cheng SY, Bischoff HG, Peled N, Grossi F, Jennens RR, Reck M, Hui R, Garon EB, Boyer M, Rubio-Viqueira B, Novello S, Kurata T, Gray JE, Vida J, Wei Z, Yang J, Raftopoulos H, Pietanza MC, Garassino MC; KEYNOTE-189 Investigators. Pembrolizumab plus Chemotherapy in Metastatic Non-Small-Cell Lung Cancer. N Engl J Med 2018; 378: 2078-2092 [PMID: 29658856 DOI: 10.1056/NEJMoa1801005]



World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1193-1201

DOI: 10.5306/wjco.v12.i12.1193

ISSN 2218-4333 (online)

MINIREVIEWS

Role of liver transplantation in the management of colorectal liver metastases: Challenges and opportunities

Panagiotis T Tasoudis, Ioannis A Ziogas, Sophoclis P Alexopoulos, John J Fung, Georgios Tsoulfas

ORCID number: Panagiotis T Tasoudis 0000-0001-7735-9474; Ioannis A Ziogas 0000-0002-6742-6909; Sophoclis P Alexopoulos 0000-0001-8785-7469; John J Fung 0000-0002-3038-0441; Georgios Tsoulfas 0000-0001-5043-7962.

Author contributions: Tasoudis PT and Ziogas IA wrote the paper; Alexopoulos SP, Fung JJ, Tsoulfas G critically revised the manuscript for important intellectual content.

Conflict-of-interest statement: No conflict of interest.

Country/Territory of origin: United States

Specialty type: Transplantation

Provenance and peer review: Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): A Grade B (Very good): B Grade C (Good): 0 Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in

Panagiotis T Tasoudis, School of Health Sciences, University of Thessaly, Larissa 41500, Greece

loannis A Ziogas, Sophoclis P Alexopoulos, Department of Surgery, Division of Hepatobiliary Surgery and Liver Transplantation, Vanderbilt University Medical Center, Nashville, TN 37232, United States

John J Fung, Department of Surgery, University of Chicago Medicine Transplant Institute, Chicago, IL 60637, United States

Georgios Tsoulfas, Department of Transplant Surgery, Aristotle University School of Medicine, Thessaloniki 54622, Greece

Corresponding author: Georgios Tsoulfas, FACS, FICS, MD, PhD, Associate Professor, Department of Transplant Surgery, Aristotle University School of Medicine, 66 Tsimiski Street, Thessaloniki 54622, Greece. tsoulfasg@gmail.com

Abstract

The liver is the most common site of colorectal cancer metastasis. Complete resection of the metastatic tumor is currently the only treatment modality available with a potential for cure. However, only 20% of colorectal liver metastases (CRLM) are considered resectable at the time of presentation. Liver transplantation (LT) has been proposed as an alternative oncologic treatment for patients with unresectable CRLM. This review summarizes the published experiences of LT in the setting of unresectable CRLM from the previous decades and discusses the challenges and future horizons in the field. Contemporary experiences that come mostly from countries with broader access to liver grafts are also explored and their promising findings in terms of overall survival (OS) and disease-free survival (DFS) are outlined along with their study design and methods. The rationale of establishing specific patient selection criteria and the dilemmas around immunosuppressive regimens in patients undergoing LT for CRLM are also highlighted. Additionally, this review describes the findings of studies comparing LT vs chemotherapy alone and LT vs portal vein embolization plus resection for CRLM in terms of OS and DFS. Last but not least, we present current perspectives and ongoing prospective trials that try to elucidate the role of LT for CRLM.

Key Words: Colorectal cancer; Colorectal liver metastases; Liver transplantation;


accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt p://creativecommons.org/License s/by-nc/4.0/

Received: June 23, 2021 Peer-review started: June 23, 2021 First decision: July 27, 2021 Revised: June 27, 2021 Accepted: November 26, 2021 Article in press: November 26, 2021 Published online: December 24, 2021

P-Reviewer: Sahin TT, Sato Y S-Editor: Wu YXI L-Editor: A P-Editor: Wu YXJ



Transplant oncology; Liver cancer; Oslo score

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: Despite the discouraging results of the previous decades, reports from the recent era showed promising results and reemerged the idea of liver transplantation (LT) for colorectal liver metastases (CRLM). Documentation of patient selection criteria and stronger evidence from ongoing prospective trials may reinforce the implementation of LT as an oncologic treatment for CRLM.

Citation: Tasoudis PT, Ziogas IA, Alexopoulos SP, Fung JJ, Tsoulfas G. Role of liver transplantation in the management of colorectal liver metastases: Challenges and opportunities. World J Clin Oncol 2021; 12(12): 1193-1201

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1193.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1193

INTRODUCTION

Colorectal cancer (CRC) is one of the most common cancer entities worldwide, ranking third in terms of incidence, and second in terms of cancer-related death[1]. The overall survival of the patients with CRC depends primarily on cancer staging^[2-4]. The liver is the most common site of CRC metastasis, mainly due to its anatomical association with the portal circulation[3]. Published data indicate that approximately 20% of patients with CRC present with concomitant liver metastasis at the first medical consultation, while another 50% develops liver metastasis within the first 3 years after primary tumor diagnosis[2,3,5,6]. The life expectancy of patients with colorectal liver metastasis (CRLM) who do not receive any type of treatment ranges from 12 to 15 mo, and the 5-year survival is less than 5% [2]. Implementation of chemotherapy as the only treatment modality for CRLM results in a median patient survival of approximately 25 mo[7]. Complete tumor resection is currently the only potentially curative treatment with a 5-year and 10-year overall survival (OS) of 38% and 26%, respectively[8]. Yet, only 20% of patients present with a hepatic lesion that can be managed surgically with a curative intent[9-11]. Additionally, what constitutes a surgically resectable CRLM is a matter of debate among surgeons[12,13]. Recent advances in surgical techniques[14] combined with the emergence of newer chemotherapeutic drugs[15,16] have increased the proportion of CRLM amenable to resection. Unfortunately, disease recurrence is still reported in 40%-70% of patients within the first 3 years after surgical excision[17, 18] due to the presence of micro-metastatic disease, resulting in a median OS of 10-38 mo for patients relegated to palliative chemotherapy [19,20].

This has led to the consideration of liver transplantation (LT) as an oncologic treatment for patients with CRLM isolated to the liver[9,21]. The aim of this review is to delineate the rationale and outcomes of LT in the setting of unresectable CRLM, and to outline the potential benefits, future perspectives, and ethical dilemmas about this treatment modality.

EARLY EXPERIENCE

LT was historically first attempted in patients with malignant liver tumors (including patients with CRLM)[22]. However, poor survival and high recurrence rates quickly led to restriction of LT utilization to early-stage hepatocellular carcinoma (HCC). The experience gained through the years along with advances in surgical technique and neoadjuvant modalities have broadened the spectrum of malignant indications for LT including advanced HCC, hilar cholangiocarcinoma, as well as metastatic liver tumors (e.g., neuroendocrine metastasis)[23]. This motivated some groups in 1990s to re-assess the role of LT for unresectable CRLM.

The first report of LT for CRLM was from Medical University of Vienna, Austria [24]. Mühlbacher et al[24] reported a series of 25 patients who underwent LT for CRLM between 1982-1994 (all patients had lymph node negative disease). In this study, the 1-,



3-, and 5-year post-LT OS was 76% (19/25), 32% (8/25), and 12% (3/25), respectively [19,24]. It should be acknowledged, that after retrospective histological examination of the excisional specimens, lymph node micro-metastases were observed in 15 out of 21 patients who were initially classified as having negative lymph node status. This finding was associated with a decreased post-LT median survival of 28 mo compared to a median survival of 118 mo in patients without micro-metastases[24-26]. Another early experience was published in 1991 by Penn *et al*[27] from University of Cincinnati reporting on 10 patients undergoing LT for CRLM (eight of them due to unresectable tumor and two of them due to chemotherapy adverse effects) with a 70% recurrence rate. Additionally, Pichlmayr et al[28] published another series of patients undergoing LT in Germany during 1972-1995, and amongst the reported cases there were 4 patients who underwent LT for CRLM. Two of these patients died in the early postoperative period (one due to acute graft rejection), while the other two patients died from disease recurrence at 11 mo and 33 mo follow-up[28]. The discouraging results from these studies in addition to the lack of standardized criteria for patient selection led to CRLM being established as a formal contraindication for LT over the next decades.

RECENT ERA

The broader access to deceased donor liver grafts in Norway led a group from Oslo University Hospital to investigate the outcomes of well-selected LT candidates with unresectable liver-only CRLM[9,21,29,30]. The first prospective study (SECA-I) was published in 2013 and included 21 patients who had undergone LT from 2006 to 2011 [30]. Inclusion criteria were total resection of the primary tumor, ECOG score 0 or 1, at least 6 wk of neoadjuvant chemotherapy, and absence of extrahepatic disease[30]. Liver resection prior to LT had been performed in 3 patients. The median follow-up time was 27 (range, 8-60) months and the 1-, 3-, and 5-year OS was 95%, 68%, and 60%, respectively. All patients received sirolimus for immunosuppression and none of them received adjuvant chemotherapy. Disease free survival (DFS) was 35% at 1 year [30]. Another publication from the same group reported a total of 19 recurrences in 21 patients (16 were single-site and 3 were multiple-sites at first presentation)^[29]. Thirteen of the 16 recurrences were isolated to the lung and patients with isolated pulmonary metastases had a 5-year survival of 72% after recurrence was diagnosed [29]. Notably, there was no isolated hepatic recurrence at initial presentation[29]. However, seven patients developed metastasis to the transplanted liver on subsequent follow-up and six out of those seven patients eventually died from metastatic disease.

Although the results from SECA-I were encouraging, the high recurrence rates led to more stringent candidate selection criteria in SECA-II. SECA-II included 15 patients who had undergone LT for unresectable liver-only CRLM from 2012 to 2016[9]. Similar to the SECA-I trial, all patients received sirolimus for immunosuppression and none of them received adjuvant chemotherapy. The stricter selection criteria required that isolated liver-only CRLM was confirmed by 18F-fluorodeoxyglucose (18F-FDG) positron emission tomography (PET)/computed tomography (CT) scan and patients had more than one year time span from diagnosis of CRC to LT. Additionally, at least 10% response on chemotherapy (according to RECIST-criteria) was a prerequisite for inclusion in the SECA-II study[9]. Resection prior to LT was performed in 4 patients. Median follow up was 36 mo. Compared to SECA-I, the more restrictive selection criteria led to improved 1-, 3-, and 5-year OS of 100%, 83%, and 83%, respectively. However, median DFS remained low at 13.7 mo. Overall, 8 patients were reported to have disease recurrence after LT and 6/8 presented with isolated pulmonary metastasis. On follow-up, 13 patients were alive, and 2 patients died 26 mo after LT due to disease recurrence. The main limitations of this study were the small sample and the relative short follow-up time, but the encouraging results drove the investigators from Oslo to conduct an additional enrollment to the SECA-II study using grafts from extended criteria donors[21]. This study (D-arm of SECA-II) included both patients with synchronous CRLM (within 1 year of primary colorectal tumor diagnosis) and those with concomitant resectable pulmonary metastases or with previously resected pulmonary metastases[21]. Moreover, the investigators did not consider response to chemotherapy as a prerequisite for recruitment[9,21]. Ultimately, 10 patients were included between 2014-2018. The median follow up was 23 mo, the median OS was 18 mo, and the median DFS was 4 mo. Disease recurrence was noted in 8/10 patients with isolated pulmonary metastasis seen in six patients. Overall, five patients were still alive on follow-up with two of them having no relapse at 23 mo and



26 mo after LT[21]. These outcomes established that LT could be considered in patients with unresectable liver-only CRLM only under strict selection criteria.

An international, multicenter, retrospective study of 12 patients was published by Toso *et al*[31] in 2017. Eleven of the patients had received chemotherapy prior to LT. The median follow-up time was 26 mo and the 1-, 3-, and 5-year OS was 83%, 62%, and 50%, respectively, while the 1-, 3-, and 5-year DFS was 56%, 38%, and 38%, respectively. Disease recurrence was noted in six patients with five of them presenting with pulmonary metastasis, while 5 out of the 11 patients were reported to be alive and without evidence of relapse at the end of the follow-up[31]. However, due to the nature of the study patients were not selected according to homogeneous criteria and they were not managed with the same interventions. Despite the limitations, this report demonstrated that LT for CRLM can provide a survival benefit in carefully selected patients, but additional refinement is necessary prior to the broader application of LT as an oncologic treatment for CRLM.

A recent worldwide systematic review and pooled analysis of 110 patients undergoing LT for CRLM reported that the 1-, 3-, and 5-year OS rates were 88.1%, 58.4%, and 50.5%, respectively[32].

Study characteristics and findings for the early experience and recent studies on LT for CRLM are shown in Table 1.

RISK STRATIFICATION CRITERIA

Similar to prior reports establishing specific selection criteria for other liver malignancies (Milan criteria for HCC and Mayo Clinic criteria for hilar cholangiocarcinoma)[33,34] the SECA studies introduced the Oslo score which was used a surrogate marker for favorable prognosis[9,21,30]. One point was assigned for each of the following characteristics: (1) Lesion larger than 5.5 cm; (2) Pre-LT plasma carcinoembryonic antigen (CEA) level above 80 μ g/L; (3) Time from primary tumor resection to LT less than 24 mo; and (4) Disease progression while on pre-LT chemotherapy. Each of these factors was significantly associated with poorer OS and the five patients who possessed all four factors comprised five of the six mortalities in the SECA-I trial[30]. Risk stratification was also done utilizing the Fong Clinical Risk Score (FCRS)[35], in which one point was given for the following: (1) Synchronous CRLM (less than 12 mo from diagnosis); (2) Primary tumor with positive lymph nodes; (3) More than one lesion; (4) Tumor larger than 5 cm; and (5) CEA level higher than $200 \ \mu g/L$. FCRS of 1-2 at the time of diagnosis was associated with significantly increased DFS compared to FCRS of 3-4[9].

The importance of stricter patient selection was also highlighted by the differences in OS and DFS between SECA-I and SECA-II studies, where 5-year OS was 60% and the 1-year DFS was 35% for SECA-I, while the 5-year OS was 83% and the 3-year DFS was 35% for SECA-II[9,29]. Smedman et al[21] attributed the poorer outcomes in terms of survival and disease recurrence of the D-arm of SECA-II in the significantly higher Oslo and FCRS scores of the patients compared to the patients of SECA-I and primary SECA-II trials[9,21,29,30]. Therefore, it is apparent that strict patient selection criteria and risk stratification are essential for the broader adoption of LT as a life prolonging oncologic treatment for liver-only CRLM.

IMMUNOSUPPRESSION

Immunosuppression is a controversial topic regarding LT for metastatic diseases in terms of achieving a balance between the risk of graft rejection and the risk of disease recurrence [19,32]. That is because attenuation of the native immune response from immunosuppression is essential to prevent graft rejection, however, it may contribute to unfavorable post-LT outcomes in patients with disseminated malignant disease, as it could facilitate disease recurrence. A study that assessed the impact of sirolimus post-LT for HCC, documented that immunosuppression improved the outcomes in the first few years post-LT and had no effect in DFS or OS beyond 5 years post-LT[36]. Notably, sirolimus was the immunosuppressive regimen used in the SECA trials[9,21, 29,30]. Data from a study that compared the growth of pulmonary metastasis in patients enrolled in the SECA trials vs patients with CRC and lung metastasis who did not receive immunosuppression, reported that there was significant difference between the two groups in terms of the time needed to double tumors' diameter and volume[37]. Moreover, the same study reported that there was no association between



Table 1 Study characterist	ics and findings						
A. Early experience							
Author, Yr	Study period	Number of patients	Clinical out	comes			
Mühlbacher et al[24], 1991	1982-1994	25	1-yr OS: 76%	, 3-yr OS: 32%,	5-yr OS: 12%		
Penn et al[27], 1991	N/A	10	70% recurren	ice rate			
Pichlmayr et al[28], 1997	1972-1995	4	2 post-operat	tive mortalities,	2 late mortaliti	es due to recur	rence
B. Recent era							
Author, Yr	Study period	Number of patients	Follow-up	1-yr OS, %	3-yr OS, %	5-yr OS, %	DFS
Hagness <i>et al</i> [30], 2013	2006-2011	21	27 mo	95	68	60	35% at $1^{\rm st} year$
Dueland <i>et al</i> [41], 2020	2012-2016	15	36 mo	100	83	83	13.7 mo
Smedman <i>et al</i> [21], 2020	2014-2018	10	23 mo	N/A	N/A	N/A	4 mo
Toso <i>et al</i> [31] , 2017	1995-2015	12	26 mo	83	62	50	56% at $1^{\rm st}{\rm year}$

All values reported for continuous variables are expressed in median. OS: Overall survival; DFS: Disease free survival, N/A: Not available.

sirolimus plasma levels and DFS or growth of pulmonary metastases[37]. However, the current level of evidence is relatively low, and future high-quality studies are required to draw solid conclusions for immunosuppressive therapies after LT for CRLM.

LIVER TRANSPLANTATION VERSUS CHEMOTHERAPY

In 2015, Dueland et al[38] published a study aiming to outline the differences in OS of patients with CRC and nonresectable CRLM treated by LT or chemotherapy. The investigators compared the SECA-I study with the NORDIC VII study, which was a multicenter randomized three-arm trial investigating the efficacy of cetuximab added to bolus fluorouracil/folinic acid and oxaliplatin[39]. The 21 patients from SECA-I study were compared with the 47 patients from the NORDIC VII study, in terms of DFS and OS. DFS was 8 to 10 mo in both groups[38]. However, the 5-year OS was 56% in patients undergoing LT compared to 9% in patients receiving chemotherapy[38]. The authors attributed this difference to the pattern of disease recurrence. While small and slow growing lung metastases were the most common recurrence pattern in the LT group, patients in the chemotherapy group presented with progression of the nonresectable CRLM, which has a less favorable prognosis[38].

LIVER TRANSPLANTATION VERSUS PORTAL VEIN EMBOLIZATION PLUS LIVER RESECTION

Emerging surgical advances have been proposed to increase the pool of patients with CRLM that can be subjected to liver resection. Implementation of portal vein embolization (PVE) could render initially nonresectable CRLM amenable to resection [40]. Dueland *et al*[41] compared 50 patients enrolled to SECA studies with a matched group of 53 intention-to-treat patients who have undergone PVE to expand the future liver remnant (FLR) and were planned to undergo liver resection (15 patients did not proceed to liver resection due to inadequate FLR or disease progression). Although the data for the whole LT cohort are not presented to clearly appreciate differences compared to the PVE cohort, the authors mentioned that the two groups had similar selection criteria. Additionally, patients were subclassified in two subgroups; the high tumor load (HTL) group was defined as patients having \geq 9 metastatic tumors or largest tumor diameter \geq 5.5 cm, while patients with CRLM below the aforementioned limits were included in the low tumor load (LTL) group[41]. The 5-year OS for patients with HTL was 33.4% in the LT arm (n = 29) compared to 6.7% in the PVE arm (n = 15) of the study without any between-group differences regarding tumor burden score. The 5-year OS for patients with LTL was 72.4% in the LT arm (n = 21) compared to 53.1% in the PVE arm (n = 30), while the tumor burden score was significantly higher



in the LT arm. Accounting that there are no randomized controlled trials comparing LT to PVE plus resection in patients with extensive liver-only CRLM, as well as the fact that these two modalities may not necessarily be applicable to the same pool of patients, this study provides some evidence that LT has promising future perspectives in the field of oncologic treatments for CRLM.

CURRENT AND FUTURE PERSPECTIVES

The International Liver Transplantation Society Transplant Oncology Consensus Conference recommendations, based on the findings of SECA trials, suggested that LT could be implemented in patients with unresectable CRLM with only liver involvement and with a maximum tumor diameter ≤ 5.5 cm, pre-LT CEA $\leq 80 \mu g/L$, response to pre-LT chemotherapy, and time interval from diagnosis to LT ≥ 1 years [42]. However, worldwide liver graft scarcity poses an ethical dilemma which is summarized as follows: How will the distribution of existing grafts to patients with CRLM impact patients with imperative need for a graft? In the United States, the Model for End-Stage Liver Disease score is used to prioritize patients for LT based on severity of liver derangements. However, patients with non-resectable CRLM have no portal hypertension or liver disease and thus are handicapped for access to deceased donors. Such patients could be good candidates for living donor liver transplantation (LDLT). Consequently, considering a long-term OS in the order of 60%, LDLT could offer a very good therapeutic alternative to this group of patients without jeopardizing the cadaveric donor pool.

Several trials attempting to elucidate the role of LT in CRLM are currently ongoing. In addition to the SECA-I (NCT00294827 - active, not recruiting: estimated study completion date May 2023) and SECA-II (NCT01479608 - active and recruiting: estimated study completion date December 2027), the Oslo group is also working on the SECA-III study, which aims to assess the efficacy of LT vs other therapies (chemotherapy and surgical resection) with a primary outcome of 2-year OS, and the RAPID (NCT02215889) trial, which aims to evaluate the outcomes of recipient left lateral segmentectomy and implantation of donor segments 2 and 3 followed by removal of the remaining recipient liver segments (second stage hepatectomy) at 4 wk post-LT. The LIVERT(W)OHEAL trial (NCT03488953) will evaluate the outcomes of LDLT in both the donors and the recipients. The largest ongoing trial, estimated to eventually enroll approximately 90 patients, is the TRANSMET (NCT02597348) phase III randomized controlled trial and will evaluate the 3-year OS and disease recurrence or progression in patients with CRLM undergoing LT plus chemotherapy vs chemotherapy only. Finally, a trial conducted by the Toronto group (NCT02864485), the COLT trial (NCT03803436), the TRIPLETE trial (NCT03231722), and the Swedish SOULMATE trial (NCT04161092) are also ongoing trials that investigate the utilization of LT as an oncologic treatment for CRLM.

Several other perspectives on the assessment for candidacy should also be incorporated into future studies. FDG is widely used to stage and monitor treatment response in metastatic CRC and the use of PET/CT scan to stage patients, as well as to assess response to therapy has been raised as a parameter of interest. Mutational profiling of CRC has been shown to have an impact on patient outcomes[43], and thus the role of selecting patients for LT based on mutational profiling will need to be addressed. Finally, the use of neoadjuvant radiotherapy to the native liver prior to LT to reduce intraoperative shedding of tumor cells during hepatectomy is also under consideration.

CONCLUSION

The SECA studies from Oslo have demonstrated promising results in prolonging survival with the use of LT as an oncologic treatment for carefully selected patients with unresectable liver-only CRLM. Further evidence is currently awaited from ongoing prospective trials in order to better define the role of LT for unresectable CRLM. The addition of unresectable CRLM as an indication for LT represents a paradigm shift and further confirms versatility of the emerging field of transplant oncology.

Zaisbideng® WJCO | https://www.wjgnet.com

REFERENCES

- Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. CA Cancer J Clin 2018; 68: 394-424 [PMID: 30207593 DOI: 10.3322/caac.21492]
- 2 Donadon M, Ribero D, Morris-Stiff G, Abdalla EK, Vauthey J-N. New paradigm in the management of liver-only metastases from colorectal cancer. Gastrointest Cancer Res 2007; 1: 20-27 [PMID: 19262699]
- Valderrama-Treviño AI, Barrera-Mera B, Ceballos-Villalva JC, Montalvo-Javé EE. Hepatic 3 Metastasis from Colorectal Cancer. Euroasian J Hepatogastroenterol 2017; 7: 166-175 [PMID: 29201802 DOI: 10.5005/jp-journals-10018-1241]
- Kow AWC. Hepatic metastasis from colorectal cancer. J Gastrointest Oncol 2019; 10: 1274-1298 4 [PMID: 31949948 DOI: 10.21037/jgo.2019.08.06]
- 5 Hugen N, van de Velde CJH, de Wilt JHW, Nagtegaal ID. Metastatic pattern in colorectal cancer is strongly influenced by histological subtype. Ann Oncol 2014; 25: 651-657 [PMID: 24504447 DOI: 10.1093/annonc/mdt5911
- Scheele J, Stangl R, Altendorf-Hofmann A. Hepatic metastases from colorectal carcinoma: impact of surgical resection on the natural history. Br J Surg 1990; 77: 1241-1246 [PMID: 2253003 DOI: 10.1002/bis.1800771115
- Hurwitz H, Fehrenbacher L, Novotny W, Cartwright T, Hainsworth J, Heim W, Berlin J, Baron A, 7 Griffing S, Holmgren E, Ferrara N, Fyfe G, Rogers B, Ross R, Kabbinavar F. Bevacizumab plus irinotecan, fluorouracil, and leucovorin for metastatic colorectal cancer. N Engl J Med 2004; 350: 2335-2342 [PMID: 15175435 DOI: 10.1056/NEJMoa032691]
- Kanas GP, Taylor A, Primrose JN, Langeberg WJ, Kelsh MA, Mowat FS, Alexander DD, Choti MA, 8 Poston G. Survival after liver resection in metastatic colorectal cancer: review and meta-analysis of prognostic factors. Clin Epidemiol 2012; 4: 283-301 [PMID: 23152705 DOI: 10.2147/CLEP.S34285]
- Dueland S, Syversveen T, Solheim JM, Solberg S, Grut H, Bjørnbeth BA, Hagness M, Line PD. 9 Survival Following Liver Transplantation for Patients With Nonresectable Liver-only Colorectal Metastases. Ann Surg 2020; 271: 212-218 [PMID: 31188200 DOI: 10.1097/SLA.00000000003404]
- 10 Garden OJ, Rees M, Poston GJ, Mirza D, Saunders M, Ledermann J, Primrose JN, Parks RW. Guidelines for resection of colorectal cancer liver metastases. Gut 2006; 55 Suppl 3: iii1-iii8 [PMID: 16835351 DOI: 10.1136/gut.2006.098053]
- 11 Fong Y. Surgical therapy of hepatic colorectal metastasis. CA Cancer J Clin 1999; 49: 231-255 [PMID: 11198884 DOI: 10.3322/canjclin.49.4.231]
- 12 Homayounfar K, Bleckmann A, Helms HJ, Lordick F, Rüschoff J, Conradi LC, Sprenger T, Ghadimi M, Liersch T. Discrepancies between medical oncologists and surgeons in assessment of resectability and indication for chemotherapy in patients with colorectal liver metastases. Br J Surg 2014; 101: 550-557 [PMID: 24756914 DOI: 10.1002/bjs.9436]
- 13 Folprecht G, Gruenberger T, Bechstein WO, Raab HR, Lordick F, Hartmann JT, Lang H, Frilling A, Stoehlmacher J, Weitz J, Konopke R, Stroszczynski C, Liersch T, Ockert D, Herrmann T, Goekkurt E, Parisi F, Köhne CH. Tumour response and secondary resectability of colorectal liver metastases following neoadjuvant chemotherapy with cetuximab: the CELIM randomised phase 2 trial. Lancet Oncol 2010; 11: 38-47 [PMID: 19942479 DOI: 10.1016/S1470-2045(09)70330-4]
- 14 Azoulay D, Lim C. Comment on "Survival Following Liver Transplantation for Patients With Nonresectable Liver-only Colorectal Metastases": Shaking the Coconut Tree. Ann Surg 2020; 271: e122-e124 [PMID: 32310616 DOI: 10.1097/SLA.00000000003605]
- 15 Passot G, Soubrane O, Giuliante F, Zimmitti G, Goéré D, Yamashita S, Vauthey JN. Recent Advances in Chemotherapy and Surgery for Colorectal Liver Metastases. Liver Cancer 2016; 6: 72-79 [PMID: 27995091 DOI: 10.1159/000449349]
- 16 Lv W, Zhang GQ, Jiao A, Zhao BC, Shi Y, Chen BM, Zhang JL. Chemotherapy Plus Cetuximab vs Chemotherapy Alone for Patients with KRAS Wild Type Unresectable Liver-Confined Metastases Colorectal Cancer: An Updated Meta-Analysis of RCTs. Gastroenterol Res Pract 2017; 2017: 8464905 [PMID: 28167959 DOI: 10.1155/2017/8464905]
- 17 Page AJ, Cosgrove DC, Herman JM, Pawlik TM. Advances in understanding of colorectal liver metastasis and implications for the clinic. Expert Rev Gastroenterol Hepatol 2015; 9: 245-259 [PMID: 25033964 DOI: 10.1586/17474124.2014.940897]
- Galjart B, van der Stok EP, Rothbarth J, Grünhagen DJ, Verhoef C. Posttreatment Surveillance in 18 Patients with Prolonged Disease-Free Survival After Resection of Colorectal Liver Metastasis. Ann Surg Oncol 2016; 23: 3999-4007 [PMID: 27393572 DOI: 10.1245/s10434-016-5388-8]
- Moris D, Tsilimigras DI, Chakedis J, Beal EW, Felekouras E, Vernadakis S, Schizas D, Fung JJ, 19 Pawlik TM. Liver transplantation for unresectable colorectal liver metastases: A systematic review. J Surg Oncol 2017; 116: 288-297 [PMID: 28513862 DOI: 10.1002/jso.24671]
- 20 Masi G, Vasile E, Loupakis F, Cupini S, Fornaro L, Baldi G, Salvatore L, Cremolini C, Stasi I, Brunetti I, Fabbri MA, Puglisi M, Trenta P, Granetto C, Chiara S, Fioretto L, Allegrini G, Crinò L, Andreuccetti M, Falcone A. Randomized trial of two induction chemotherapy regimens in metastatic colorectal cancer: an updated analysis. J Natl Cancer Inst 2011; 103: 21-30 [PMID: 21123833 DOI: 10.1093/jnci/djq456]
- Smedman TM, Line PD, Hagness M, Syversveen T, Grut H, Dueland S. Liver transplantation for 21



unresectable colorectal liver metastases in patients and donors with extended criteria (SECA-II arm D study). BJS Open 2020; 4: 467-477 [PMID: 32333527 DOI: 10.1002/bjs5.50278]

- 22 Starzl TE. The saga of liver replacement, with particular reference to the reciprocal influence of liver and kidney transplantation (1955-1967). J Am Coll Surg 2002; 195: 587-610 [DOI: 10.1016/s1072-7515(02)01498-9]
- Pavel M, Baudin E, Couvelard A, Krenning E, Öberg K, Steinmüller T, Anlauf M, Wiedenmann B, 23 Salazar R. ENETS consensus guidelines for the management of patients with liver and other distant metastases from neuroendocrine neoplasms of foregut, midgut, hindgut, and unknown primary. Neuroendocrinology 2012; 95: 157-176 [PMID: 22262022 DOI: 10.1159/000335597]
- 24 Mühlbacher F, Huk I, Steininger R, Gnant M, Götzinger P, Wamser P, Banhegyi C, Piza F. Is orthotopic liver transplantation a feasible treatment for secondary cancer of the liver? Transplant Proc 1991; 23: 1567-1568 [PMID: 1989293]
- Gorgen A, Muaddi H, Zhang W, McGilvray I, Gallinger S, Sapisochin G. The New Era of Transplant 25 Oncology: Liver Transplantation for Nonresectable Colorectal Cancer Liver Metastases. Can J Gastroenterol Hepatol 2018; 2018: 9531925 [PMID: 29623268 DOI: 10.1155/2018/9531925]
- Kappel S, Kandioler D, Steininger R, Längle F, Wrba F, Ploder M, Berlakovich G, Soliman T, Hetz 26 H, Rockenschaub S, Roth E, Mühlbacher F. Genetic detection of lymph node micrometastases: a selection criterion for liver transplantation in patients with liver metastases after colorectal cancer. Transplantation 2006; 81: 64-70 [PMID: 16421478 DOI: 10.1097/01.tp.0000189711.98971.9c]
- Penn I. Hepatic transplantation for primary and metastatic cancers of the liver. Surgery 1991; 110: 27 726-34; discussion 734 [PMID: 1656538]
- 28 Pichlmayr R, Weimann A, Tusch G, Schlitt HJ. Indications and Role of Liver Transplantation for Malignant Tumors. Oncologist 1997; 2: 164-170 [PMID: 10388047]
- 29 Hagness M, Foss A, Egge TS, Dueland S. Patterns of recurrence after liver transplantation for nonresectable liver metastases from colorectal cancer. Ann Surg Oncol 2014; 21: 1323-1329 [PMID: 24370906 DOI: 10.1245/s10434-013-3449-9]
- 30 Hagness M, Foss A, Line PD, Scholz T, Jørgensen PF, Fosby B, Boberg KM, Mathisen O, Gladhaug IP, Egge TS, Solberg S, Hausken J, Dueland S. Liver transplantation for nonresectable liver metastases from colorectal cancer. Ann Surg 2013; 257: 800-806 [PMID: 23360920 DOI: 10.1097/SLA.0b013e3182823957
- Toso C, Pinto Marques H, Andres A, Castro Sousa F, Adam R, Kalil A, Clavien PA, Furtado E, 31 Barroso E, Bismuth H; Compagnons Hépato-Biliaires Group. Liver transplantation for colorectal liver metastasis: Survival without recurrence can be achieved. Liver Transpl 2017; 23: 1073-1076 [PMID: 28544246 DOI: 10.1002/lt.24791]
- Giannis D, Sideris G, Kakos CD, Katsaros I, Ziogas IA. The role of liver transplantation for 32 colorectal liver metastases: A systematic review and pooled analysis. Transplant Rev (Orlando) 2020; 34: 100570 [PMID: 33002670 DOI: 10.1016/j.trre.2020.100570]
- 33 Mazzaferro V, Regalia E, Doci R, Andreola S, Pulvirenti A, Bozzetti F, Montalto F, Ammatuna M, Morabito A, Gennari L. Liver transplantation for the treatment of small hepatocellular carcinomas in patients with cirrhosis. N Engl J Med 1996; 334: 693-699 [PMID: 8594428 DOI: 10.1056/NEJM199603143341104]
- Heimbach JK, Gores GJ, Haddock MG, Alberts SR, Nyberg SL, Ishitani MB, Rosen CB. Liver 34 transplantation for unresectable perihilar cholangiocarcinoma. Semin Liver Dis 2004; 24: 201-207 [PMID: 15192792 DOI: 10.1055/s-2004-828896]
- 35 Fong Y, Fortner J, Sun RL, Brennan MF, Blumgart LH. Clinical score for predicting recurrence after hepatic resection for metastatic colorectal cancer: analysis of 1001 consecutive cases. Ann Surg 1999; 230: 309-18; discussion 318 [PMID: 10493478 DOI: 10.1097/00000658-199909000-00004]
- Geissler EK, Schnitzbauer AA, Zülke C, Lamby PE, Proneth A, Duvoux C, Burra P, Jauch KW, 36 Rentsch M, Ganten TM, Schmidt J, Settmacher U, Heise M, Rossi G, Cillo U, Kneteman N, Adam R, van Hoek B, Bachellier P, Wolf P, Rostaing L, Bechstein WO, Rizell M, Powell J, Hidalgo E, Gugenheim J, Wolters H, Brockmann J, Roy A, Mutzbauer I, Schlitt A, Beckebaum S, Graeb C, Nadalin S, Valente U, Turrión VS, Jamieson N, Scholz T, Colledan M, Fändrich F, Becker T, Söderdahl G, Chazouillères O, Mäkisalo H, Pageaux GP, Steininger R, Soliman T, de Jong KP, Pirenne J, Margreiter R, Pratschke J, Pinna AD, Hauss J, Schreiber S, Strasser S, Klempnauer J, Troisi RI, Bhoori S, Lerut J, Bilbao I, Klein CG, Königsrainer A, Mirza DF, Otto G, Mazzaferro V, Neuhaus P, Schlitt HJ. Sirolimus Use in Liver Transplant Recipients With Hepatocellular Carcinoma: A Randomized, Multicenter, Open-Label Phase 3 Trial. Transplantation 2016; 100: 116-125 [PMID: 26555945 DOI: 10.1097/TP.0000000000000965]
- Grut H, Solberg S, Seierstad T, Revheim ME, Egge TS, Larsen SG, Line PD, Dueland S. Growth rates of pulmonary metastases after liver transplantation for unresectable colorectal liver metastases. Br J Surg 2018; 105: 295-301 [PMID: 29168565 DOI: 10.1002/bjs.10651]
- 38 Dueland S, Guren TK, Hagness M, Glimelius B, Line PD, Pfeiffer P, Foss A, Tveit KM. Chemotherapy or liver transplantation for nonresectable liver metastases from colorectal cancer? Ann Surg 2015; 261: 956-960 [PMID: 24950280 DOI: 10.1097/SLA.0000000000000786]
- 39 Tveit KM, Guren T, Glimelius B, Pfeiffer P, Sorbye H, Pyrhonen S, Sigurdsson F, Kure E, Ikdahl T, Skovlund E, Fokstuen T, Hansen F, Hofsli E, Birkemeyer E, Johnsson A, Starkhammar H, Yilmaz MK, Keldsen N, Erdal AB, Dajani O, Dahl O, Christoffersen T. Phase III trial of cetuximab with continuous or intermittent fluorouracil, leucovorin, and oxaliplatin (Nordic FLOX) vs FLOX alone in first-line treatment of metastatic colorectal cancer: the NORDIC-VII study. J Clin Oncol 2012; 30:



1755-1762 [PMID: 22473155 DOI: 10.1200/JCO.2011.38.0915]

- Moris D, Ronnekleiv-Kelly S, Kostakis ID, Tsilimigras DI, Beal EW, Papalampros A, Dimitroulis D, 40 Felekouras E, Pawlik TM. Operative Results and Oncologic Outcomes of Associating Liver Partition and Portal Vein Ligation for Staged Hepatectomy (ALPPS) Versus Two-Stage Hepatectomy (TSH) in Patients with Unresectable Colorectal Liver Metastases: A Systematic Review and Meta-Analysis. World J Surg 2018; 42: 806-815 [PMID: 28798996 DOI: 10.1007/s00268-017-4181-6]
- Dueland S, Yaqub S, Syversveen T, Carling U, Hagness M, Brudvik KW, Line PD. Survival 41 Outcomes After Portal Vein Embolization and Liver Resection Compared With Liver Transplant for Patients With Extensive Colorectal Cancer Liver Metastases. JAMA Surg 2021; 156: 550-557 [PMID: 33787838 DOI: 10.1001/jamasurg.2021.0267]
- 42 Hibi T, Rela M, Eason JD, Line PD, Fung J, Sakamoto S, Selzner N, Man K, Ghobrial RM, Sapisochin G. Liver Transplantation for Colorectal and Neuroendocrine Liver Metastases and Hepatoblastoma. Working Group Report From the ILTS Transplant Oncology Consensus Conference. Transplantation 2020; 104: 1131-1135 [PMID: 32217939 DOI: 10.1097/TP.000000000003118]
- Sinicrope FA, Shi Q, Smyrk TC, Thibodeau SN, Dienstmann R, Guinney J, Bot BM, Tejpar S, 43 Delorenzi M, Goldberg RM, Mahoney M, Sargent DJ, Alberts SR. Molecular markers identify subtypes of stage III colon cancer associated with patient outcomes. Gastroenterology 2015; 148: 88-99 [PMID: 25305506 DOI: 10.1053/j.gastro.2014.09.041]



WJC0

World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1202-1214

DOI: 10.5306/wjco.v12.i12.1202

ISSN 2218-4333 (online)

ORIGINAL ARTICLE

Basic Study Increased tensin 4 expression is related to the histological type of gastric cancer

Marcin Nizioł, Justyna Zińczuk, Konrad Zaręba, Katarzyna Guzińska-Ustymowicz, Anna Pryczynicz

ORCID number: Marcin Nizioł 0000-0003-4981-7938; Justyna Zińczuk 0000-0003-2391-3307; Konrad Zareba 0000-0002-8093-5214; Katarzyna Guzinska-Ustymowicz 0000-0002-0479-3509; Anna Pryczynicz 0000-0002-2786-7179.

Author contributions: Nizioł M and Pryczynicz A contributed to the conceptualization and data curation; Nizioł M, Zińczuk J, Zarę ba K, Guzińska-Ustymowicz K and Pryczynicz A contributed to the methodology and investigation; Nizioł M contributed to the resources, writing-original draft preparation, visualization, project administration and funding acquisition; Pryczynicz A contributed to the writing-review and editing, supervision; all authors have read and agreed to the published version of the manuscript.

Institutional review board

statement: The study was reviewed and approved by the Institutional Review Board at the Medical University of Bialystok. The research was approved by the Bioethics Committee of the Medical University of Bialystok, permission no.: R-I-002/29/2019.

Conflict-of-interest statement: All authors have nothing to disclose.

Marcin Nizioł, Katarzyna Guzińska-Ustymowicz, Anna Pryczynicz, Department of General Pathomorphology, Medical University of Bialystok, Białystok 15-089, Poland

Justyna Zińczuk, Department of Clinical Laboratory Diagnostics, Medical University of Białystok, Bialystok 15-089, Poland

Konrad Zaręba, The Second Clinical Department of General and Gastroenterological Surgery, Medical University of Bialystok, Białystok 15-089, Poland

Corresponding author: Anna Pryczynicz, PhD, Associate Professor, Postdoc, Department of General Pathomorphology, Medical University of Bialystok, Kilińskiego 1, Białystok 15-089, Poland. anna.pryczynicz@umb.edu.pl

Abstract

BACKGROUND

Gastric cancer (GC) is one of the most common malignant tumors worldwide. Tensin 4 (TNS4) is an adhesive protein belonging to the tensin family. This protein is located in focal adhesion sites. The TNS4 gene is considered an oncogene in numerous cancers. This protein plays an important role in adhesion, migration and proliferation of cells.

AIM

To evaluate expression of TNS4 protein in GC tissues and analysis of the clinical and histopathological parameters as well as the overall survival rate of patients.

METHODS

The expression of TNS4 was assessed in 89 patients using immunohistochemistry.

RESULTS

Positive expression of TNS4 was observed in 49 of 89 patients (55.06%). Higher TNS4 expression was more common in GC tumors with a diameter \geq 5 cm (*P* = 0.040). We demonstrated that an increase in TNS4 expression was more frequent in tumors of the histological type without mucinous components than in tumors from mucosal cancers (P = 0.023). Furthermore, TNS4 expression was higher in moderately differentiated tumors than in poorly differentiated and non-differentiated tumors (P = 0.002). Increased TNS4 expression was also noted in the intestinal type of GC according to Lauren's classification (P = 0.020). No statistically significant correlation was found between the expression of TNS4 and the



Data sharing statement: No additional data are available.

Supported by the Medical University of Bialystok, No. SUB/1/DN/20/002/3314.

Country/Territory of origin: Poland

Specialty type: Gastroenterology and hepatology

Provenance and peer review: Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): 0 Grade B (Very good): 0 Grade C (Good): 0 Grade D (Fair): D Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution Noncommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work noncommercially, and license their derivative works on different terms, provided the original work is properly cited and the use is noncommercial. See: http ://creativecommons.org/Licenses /by-nc/4.0/

Received: April 15, 2021 Peer-review started: April 15, 2021 First decision: June 17, 2021 Revised: June 29, 2021 Accepted: November 18, 2021 Article in press: November 18, 2021 Published online: December 24, 2021

P-Reviewer: Huang Y S-Editor: Zhang H L-Editor: A P-Editor: Zhang H



overall survival rate of patients.

CONCLUSION

TNS4 expression was significantly higher in tumors with a diameter \geq 5 cm of the moderately differentiated intestinal type (according to Lauren's classification) of GC without a mucinous component. Therefore, increased TNS4 expression is related to the histological type of GC with a better prognosis.

Key Words: Gastric cancer; Tensin 4; Adhesion proteins; Immunohistochemistry

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: Tensin 4 (TNS4) is an adhesive protein belonging to the tensin family that plays an important role in cell adhesion, migration, and proliferation. These processes are important in cancer development and may limit cancer cell growth and improve patient survival. By applying immunohistochemistry, we investigated TNS4 expression in gastric cancer (GC) tissues and discovered that TNS4 expression was significantly higher in tumors with a diameter greater than 5 cm, in tumors of the moderately differentiated intestinal type (according to Lauren's classification) and in GC without a mucinous component. We concluded that enhanced TNS4 expression was associated with the histological type of GC with a better prognosis.

Citation: Nizioł M, Zińczuk J, Zaręba K, Guzińska-Ustymowicz K, Pryczynicz A. Increased tensin 4 expression is related to the histological type of gastric cancer. *World J Clin Oncol* 2021; 12(12): 1202-1214

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1202.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1202

INTRODUCTION

In 2018, more than one million people were diagnosed with gastric cancer (GC), accounting for 5.7% of all malignant cancers worldwide; this cancer is ranked 5th in terms of incidence. Moreover, 8.2% of patients with GC died, which reveals the aggressive nature of this disease. GC was diagnosed twice as often in men than in women[1]. The development of this cancer is a multistage process, and there are numerous environmental and genetic factors that contribute to its progression, including infection with Helicobacter pylori (H. pylori) bacteria and Epstein-Barr virus, consumption of excess salt and alcohol or smoking[2]. An alarming phenomenon is the incidence of this type of cancer in young people under 40 years of age, most often as a result of a genetic predisposition. A previous report showed that 30% of patients under 30 had GC in their family history^[3]. The diagnosis of this cancer at early stages of its progression is of key importance. Preoperative chemotherapy followed by surgical removal of the tumor mass in early cancer results in a 90% 5-year survival rate [4]. Therefore, it is crucial to search for molecular and biochemical changes typical of early stages of cancer development to inhibit tumor progression and increase the survival rate of patients with this type of cancer.

Tensin 4 (TNS4), also known as C-terminal tensin-like protein, belongs to the tensin family, which also includes tensins 1–3. The *TNS4* gene is located on the long arm of chromosome 17q12-21. The TNS4 protein has a molecular weight of 77 kDa. In terms of structure, TNS4 is composed of the Src homology 2 domain (SH2) and the phosphotyrosine-binding domain (PTB). The SH2 domain allows the TNS4 protein to participate in the transmission of intracellular signals, while the PTB domain binds to membrane integrator receptors. Unlike tensins 1–3, TNS4 does not have an actin-binding domain[5]. Intracellularly, TNS4 is located in focal adhesion sites, where it enables signaling between the extracellular matrix and the cell interior. From a physiological perspective, the expression of TNS4 has been detected in the prostate and placenta[6]. Pathologically increased TNS4 expression appears in cancerous tumors. *TNS4* was originally classified as a suppressor gene in prostate cancer[6], but as research on tensins progressed, its role as an oncogene was demonstrated in

colorectal, breast, colon, esophagus, and lung cancers and thymoma, where this protein was overexpressed[7-15]. TNS4 plays an important role in biological processes connected with carcinogenesis, such as proliferation, migration, cell adhesion and invasiveness[16].

The aim of our study was to evaluate the IHC expression of TNS4 protein in GC (stages I-IV) as well as to assess the relationship between protein expression and selected clinical and pathological parameters and the overall survival rate of patients.

MATERIALS AND METHODS

Study group

The study was conducted on a group of 89 patients diagnosed with GC who were treated surgically in the 2nd Clinical Department of General and Gastroenterological Surgery of the Medical University of Bialystok in the years 2005–2015. The tissue material was obtained from the archives of the Academic Center for Pathomorphological and Genetic-Molecular Diagnostics in Bialystok. The inclusion criterion was diagnosed adenocarcinoma at any stage of its progression; the exclusion criteria were diagnosed squamous cell carcinoma and other non-epithelial cancers, metastases of other cancers to the stomach and the lack of complete medical records. The normal mucous membrane was used as the control tissue. The research was approved by the Bioethics Committee of the Medical University of Bialystok, permission No. R-I-002/29/2019. The study was conducted in accordance with the World Medical Association Declaration of Helsinki for ethical principles for medical research involving human subjects. The characteristics of the study group are presented in Table 1.

Tissue preparations

Tissues collected during the operation were fixed in 4% buffered formalin and embedded in paraffin. Paraffin blocks were then cut on a microtome into approximately 4-µm-thick sections and stained with hematoxylin and eosin. A routine histopathological examination included the assessment of the histological type of cancer, malignancy grade (G), anatomoclinical stage (pT) and presence of lymph node metastases (pN), infiltration of blood and lymphatic vessels, perineural infiltration, peritumoral inflammation and degree of desmoplasia. Moreover, *H. pylori* infection was assessed by Giemsa staining. The following information was selected from the records with histopathological diagnosis: age and sex of patients, diameter and location of the tumor, presence of distant metastases and the type of cancer according to Lauren's classification.

Immunohistochemistry

Immunohistochemical (IHC) staining was performed on 89 GC tissues. Paraffin blocks were cut on a microtome into approximately 4-µm-thick sections and mounted onto silanized slides. The microscopic sections were incubated overnight at 60 °C and then deparaffinized in xylene solutions and hydrated in a series of alcohol solutions of decreasing concentration (2 × 99.9%, 96%, 70%). The next step was blocking end-ogenous peroxidase activity by using 3% hydrogen peroxide solution (10 min) as well as nonspecific antibody binding by means of horse serum (anti-mouse/rabbit serum produced in horse, Vector Laboratories, Germany) (20 min). In the following step, the sections were incubated with polyclonal anti-TNS4 antibody at a dilution of 1:75 (Biorbyt, orb186458 produced in rabbit) for 30 min at room temperature. The antibody binding sites were visualized with an ImmPress Universal Antibody Polymer Reagent kit (Vector Laboratories, Germany) and ImmPACT DAB chromogen (Vector Laboratories, Germany). Cell nuclei were stained with hematoxylin. The preparations were then dehydrated in a series of alcohol solutions of increasing concentration and overexposed in xylene solutions.

Validation of TNS4 expression detection

We particularly focused on obtaining reliable results of TNS4 IHC staining. To optimize the TNS4 staining procedure, we used positive and negative controls, and selected primary antibody dilutions (1:50, 1:75, 1:100, 1:200) and incubation times (30 min, 60 min, 120 min) were tested. For the control, antigen retrieval was also performed in buffers with pH = 6.0 and pH = 9.0.

Zaisbideng® WJCO | https://www.wjgnet.com

Table 1 The characteristics of study group	
Parameter	Number of cases
Age	
< 60	29 (32.58%)
≥ 60	60 (67.42%)
Gender	
Female	29 (32.58%)
Male	60 (67.42%)
Tumor diameter	
< 5 cm	18 (20.22%)
≥5 cm	71 (79.78%)
Tumor localization	
Upper 1/3	17 (19.10%)
Middle1/3	33 (37.08%)
Lower 1/3	16 (17.98%)
Whole stomach	23 (25.84%)
Histological type	
Adenocarcinoma	54 (60.67%)
Adenocarcinoma mucinosum	35 (39.33%)
Histological differentiation	
Moderately differentiated	25 (28.09%)
Poorly differentiated	35 (39.33%)
Non differentiated	29 (32.58%)
Depth of invasion	
T1	6 (6.74%)
T2	7 (7.87%)
Τ3	66 (74.16%)
T4	10 (11.23%)
Lymph node metastasis	
Absent	17 (19.10%)
Present	72 (80.90%)
Distant metastasis	
Absent	61 (68.54%)
Present	28 (31.46%)
Blood vessel infiltration	
Absent	48 (84.21%)
Present	9 (15.79%)
Lymphatic vessel infiltration	
Absent	21 (31.82%)
Present	45 (68.18%)
Perineural infiltration	
Absent	28 (34.15%)
Present	54 (65.85%)

Baisbideng® WJCO | https://www.wjgnet.com

Nizioł M et al. TNS4 expression in gastric cancer

Peritumoral inflammation	
Absent	42 (50.00%)
Present	42 (50.00%)
Desmoplasia	
Small	55 (66.27%)
Large	28 (33.73%)
Helicobacter pylori infection	
Absent	61 (73.49%)
Present	22 (26.51%)
Lauren's classification	
Intestinal	44 (55.00%)
Diffuse	36 (45.00%)

Microscopic evaluation

The preparations were examined under an Olympus BX41 light microscope by two independent pathomorphologists. TNS4 protein expression was evaluated under 400× magnification in 10 representative fields of view. In each field of view, 100 or more cancer cells were assessed. Protein expression was observed in both the cell membrane and the cytoplasm. The presence of TNS4 protein in $\geq 20\%$ of neoplastic cells was considered positive expression.

Statistical analysis

The comparison of TNS4 expression with the selected clinicopathological parameters was carried out by the Mann-Whitney U test for two groups and the Kruskal-Wallis test for 3 or more groups. Dunn's multiple comparison post hoc tests were conducted for the Kruskal-Wallis test. A value of P < 0.05 was considered statistically significant. The overall survival analysis was based on the Kaplan-Meier test. The program Statistica 13 (Statsoft, Krakow, Poland) was used for the analysis. Missing data were removed in pairs.

RESULTS

TNS4 expression in GC samples

TNS4 protein expression was tested immunohistochemically in 89 GC samples and 20 normal stomach tissues. Statistical analysis demonstrated that positive TNS4 expression in cancer cells occurred in 49 of 89 patients (55%). In neoplastic cells, the expression of TNS4 was observed in the cell membrane and cytoplasm (Figure 1)

Analysis of the correlation of TNS4 expression with clinical and pathological parameters of GC

The statistical analysis did not show any significant relationships between TNS4 protein expression and the age and sex of patients, tumor location, anatomoclinical stage-pT, presence of lymph node and distant metastases, blood and lymphatic vessel infiltration, perineural infiltration, peritumoral inflammation, desmoplasia or H. pylori infection. However, a significant correlation was found between TNS4 expression and tumor diameter. Tumors with a diameter of < 5 cm were present in 20.22% of the patients, while tumors with a diameter of \geq 5 cm were found in 79.78% of the patients. Positive expression of the TNS4 protein was more frequent in gastric tumors with a diameter of \geq 5 cm (60.56% of the patients with positive expression) than in tumors < 5 cm in diameter (33.33% of the patients) (P = 0.040) (Figure 2A). Furthermore, a significant correlation with the histological type of tumor was demonstrated. Cancers without mucinous components were diagnosed in 60.67% of the patients, and mucosal tumors were diagnosed in 39.33% of the patients. A positive reaction to the TNS4 protein was considerably more frequent in the histological type without a mucinous component (64.81% of patients) than in mucinous adenocarcinomas (40.00% of patients) (P = 0.023) (Figure 2B). Statistical analysis also showed a significant





Figure 1 Tensin 4 expression in normal gastric mucosa. A: 100× magnification and gastric cancer samples; B and C: Poorly differentiated gastric cancer (GC); D: Moderately differentiated GC (magnification 200×).

correlation with the malignancy grade. In the study group, there were 28.09% moderately differentiated, 39.32% poorly differentiated and 32.59% non-differentiated tumors. The TNS4 protein occurred much more frequently in moderately differentiated cancers (80.00% of the patients) than in poorly differentiated (57.14%) and nondifferentiated (31.03%) tumors (P = 0.002) (Figure 2C). It was also proven that TNS4 protein expression correlates with the type of tumor according to Lauren's classification. The intestinal type was present in 55% of the patients, while 45% of the patients suffered from the diffuse type of GC. The presence of TNS4 was more frequently observed in intestinal-type tumors (70.45%) than in diffuse-type tumors (44.44%) (*P* = 0.020) (Figure 2D).

Table 2 presents a summary of the correlation between clinicopathological parameters and TNS4 protein expression found in our study.

Analysis of the correlation of TNS4 protein expression with the overall survival rate

Statistical analysis did not reveal a significant correlation between TNS4 protein expression and overall patient survival (P = 0.25) (Figure 3).

DISCUSSION

In our study, we performed an IHC assessment of the expression of the TNS4 protein in GC patients diagnosed with stage I to IV disease. We observed that TNS4 expression was higher in tumor cells than in normal tissue, which is consistent with the results presented by Qi et al[17]. Sawazaki et al[18] evaluated the clinical significance of TNS4 mRNA expression in GC and demonstrated that its expression is significantly higher in tumor tissue, as confirmed by Sakashita *et al*[19]. Based on these observations, it can be concluded that changes in the expression of the TNS4 gene occur in tumor cells compared to normal gastric mucosa cells. This change is reflected in increased expression of TNS4 protein in GC.

In our research, we demonstrated that positive expression of TNS4 protein occurs twice as often in GC tumors with a diameter of ≥ 5 cm than in tumors with a diameter smaller than 5 cm. Similar observations were reported by other authors who noted that TNS4 protein expression increases with the size of breast tumors and hepatocellular



Table 2 Correlation between clinical pathological parameters and tensin 4 protein expression

Deservation	Expression of tensin 4 prot	tein	10
Parameter	Negative	Positive	* P value
Age			0.666
< 60	14 (48.28%)	15 (51.72%)	
≥60	26 (43.33%)	34 (56.67%)	
Gender			0.182
Female	16 (55.17%)	13 (44.83%)	
Male	24 (40.00%)	36 (60.00%)	
Tumor diameter			0.040
< 5 cm	12 (66.67%)	6 (33.33%)	
≥5 cm	28 (39.44%)	43 (60.56%)	
Tumor localization			0.081
Upper 1/3	10 (58.82%)	7 (41.18%)	
Middle1/3	9 (27.27%)	24 (72.73%)	
Lower 1/3	9 (56.25%)	7 (43.75%)	
Whole stomach	12 (52.17%)	11 (47.83%)	
Histological type			0.023
Adenocarcinoma	19 (35.19%)	35 (64.81%)	
Adenocarcinoma mucinosum	21 (60.00%)	14 (40.00%)	
Histological differentiation			0.002 (< 0.001)
Moderately differentiated	5 (20.00%)	20 (80.00%)	
Poorly differentiated	15 (42.86%)	20 (57.14%)	
Non differentiated	20 (68.97%)	9 (31.03%)	
Depth of invasion			0.208
T1	5 (83.33%)	1 (16.67%)	
T2	2 (28.57%)	5 (71.43%)	
T3	28 (42.42%)	38 (57.58%)	
T4	5 (50.00%)	5 (50.00%)	
Lymph node metastasis			0.071
Absent	11 (64.71%)	6 (35.29%)	
Present	29 (40.28%)	43 (59.72%)	
Distant metastasis			0.794
Absent	28 (45.90%)	33 (54.10%)	
Present	12 (42.86%)	16 (57.14%)	
Blood vessel infiltration			0.980
Absent	21 (43.75%)	27 (56.25%)	
Present	4 (44.44%)	5 (55.56%)	
Lymphatic vessel infiltration			0.358
Absent	10 (47.62%)	11 (52.38%)	
Present	16 (35.56%)	29 (64.44%)	
Perineural infiltration			0.628
Absent	13 (46.43%)	15 (53.57%)	



Nizioł M et al. TNS4 expression in gastric cancer

Present	22 (40.74%)	32 (59.26%)	
Peritumoral inflammation			0.453
Absent	19 (46.34%)	22 (53.66%)	
Present	16 (38.10%)	26 (61.90%)	
Desmoplasia			0.710
Small	24 (43.64%)	31 (56.36%)	
Large	11 (39.29%)	17 (60.71%)	
Helicobacter pylori infection			0.824
Absent	26 (42.62%)	35 (57.38%)	
Present	10 (45.45%)	12 (54.55%)	
Lauren's classification			0.020
Intestinal	13 (29.55%)	31 (70.45%)	
Diffuse	20 (55.56%)	16 (44.44%)	

^aIn bracket is *P* value before Dunn's multiple comparison post hoc test



Figure 2 Tensin 4 expression with clinical and pathological parameters of gastric cancer. A: Comparison of tensin 4 (TNS4) expression and (A) malignancy grade, G2 vs G4, statistically significant at P = 0.002; B: Tumor diameter; C: Histological type; D: Lauren's classification. The comparison of TNS4 expression with the selected clinicopathological parameters was conducted using the Mann-Whitney U test for two groups and the Kruskal-Wallis test for three groups. Dunn's multiple comparison post hoc tests were implemented for the Kruskal-Wallis test. A value of P < 0.05 was considered significant. ^aP < 0.05 G2 vs G4 TNS4 expression level in gastric cancer. G2: Moderately differentiated gastric cancer (GC); G3: Poorly differentiated GC; G4: non-differentiated GC.

carcinoma tumors[11,20]. This finding indicates a correlation between TNS4 expression and the progression of the tumor and may be associated with the activation of signaling pathways, including PI3/AKT and Ras/MAPK, the stimulation of which intensifies the processes of cell proliferation and migration^[13]. During the activation of these pathways, TNS4 expression is increased under the influence of epidermal growth factor at the levels of mRNA transcription and protein translation. Increased TNS4 expression leads to enhanced migration and proliferative potential of cancer cells.

We observed significant correlations between the expression of TNS4 and malignancy grade, histological type and type of tumor according to Lauren's classification. TNS4 protein was much more frequently present in moderately differentiated





Figure 3 Analysis of the correlation of tensin 4 protein expression with the overall survival rate.

tumors than in poorly differentiated and non-differentiated tumors, in adenocarcinomas without mucinous components and in the intestinal type of GC. The classification of GC based on the degree of glandular cell differentiation is a prognostic factor. The more differentiated the gastric adenocarcinoma, the better the prognosis is for a patient. Poorly and non-differentiated adenocarcinoma cells differ phenotypically from each other, forming clusters without visibly formed glands[21]. There are no reports in the literature suggesting a significant relationship between TNS4 protein expression and the grade of tumor malignancy, which indicates the uniqueness of our study. In contrast, Sakashita *et al*[19] evaluated the *TNS4* mRNA expression and observed that patients with moderately or poorly differentiated GC had higher levels of TNS4 expression. The differences in the results obtained by these researchers and us may be due to numerous mutations in the TNS4 gene in tumor cells, entailing changes in the structure of this protein. We demonstrated higher expression of this protein in moderately differentiated gastric adenocarcinoma, which suggests that TNS4 overexpression is associated with a better prognosis of the tumor.

Poorly differentiated GCs are characterized by weak or no developed glandular tubules. Under normal conditions, TNS4 is involved in glandular duct morphogenesis. This phenomenon was described by Wu et al[22] after their study on a 3D model of RWPE-1 cells of the prostate. These researchers demonstrated that a decrease in protein expression does not affect the phenotype of the newly formed glands, although their number decreased. The authors suggested that lowering TNS4 expression inhibits the proliferation of normal epithelial cells of the prostate by stopping the cell cycle. However, increased expression of TNS4 leads to disturbances in the formation of glandular ducts through β_1 -integrin/FAK pathway induction. Albasri et al[8] identified a link between TNS4 and the morphology of colorectal cancer cells. By means of confocal microscopy, these researchers demonstrated that induced expression of TNS4 affects the shape of cancer cells. Under the conditions of TNS4 overexpression, HCT116 cells assume a spindle shape compared to cells with physiologically low expression of TNS4.

In our research, we demonstrated significant differences in TNS4 expression between the two histological types of GC. We noted that positive expression of TNS4 protein was more frequent in GCs without mucinous components than in mucinous adenocarcinomas. Gastric mucosal adenocarcinomas have been linked to poorer prognosis, lower sensitivity to oncological treatment and increased risk of acquiring resistance to chemotherapy^[23]. Currently, there are no data available to indicate a link between TNS4 expression and gastric adenocarcinomas without a mucinous component. Phenotypically, GC with a mucinous component is characterized by a considerable amount of mucin in the extracellular space (> 50%)[24].

In our study, we correlated the expression of TNS4 protein with the cancer type according to Lauren's classification. Statistical analysis showed that positive TNS4 staining correlates with the intestinal type of tumor. The presence of this protein was more often observed in adenocarcinomas of the intestinal than diffuse type. The results presented by Chen *et al*^[25] indicate that the intestinal type correlates with better prognosis and better clinical and pathological parameters than the diffuse type. From a histological point of view, intestinal-type GC is phenotypically characterized by cells forming glandular ducts in the lumen of which the cancer grows. Usually, these cells do not secrete mucus, but if it is produced, it remains in the lumen of the glands^[26]. In the literature, we did not find any studies assessing TNS4 expression in relation to the histological type of GC according to Lauren's classification. The positive



correlation between the expression of TNS4 and the intestinal type of cancer suggests further directions of research as well as the need to investigate the role of this protein in the histogenesis of GC.

Our study did not show any correlation between the level of TNS4 protein expression and tumor progression (pT, pN, pM status), although the literature shows that TNS4 is located in focal adhesion sites and participates in cell migration, thus increasing the metastatic potential of tumor cells. Qi et al[17] observed a significant correlation between TNS4 expression and the tumor stage. Their results indicated a correlation between TNS4 expression and deeper infiltrating tumors as well as lymph node and distant metastases. We did not observe similar significant relationships in our study. Other researchers have shown that stimulated TNS4 expression results in an increased ability of cells to migrate, infiltrate into adjacent tissues and metastasize. TNS4 was shown to play an important role in epithelial-mesenchymal transition by stimulating several signaling pathways, including E-cadherin, Akt, Src, TGF-β/Smad and Snail[8,17,27-29]. Other signaling pathways that are stimulated under TNS4 overexpression are PI3K/Akt and Ras/MAPK. Their activation may intensify cell proliferation and migration^[13]. Furthermore, *in vitro* studies revealed that TNS4 may promote colony formation, as demonstrated in a study on pancreatic cancer cells[30]. Another important signaling protein in cancer development is MET, which is stabilized by TNS4 and acts as a mediator in carcinogenesis[31]. The exact mechanisms regulating TNS4, e.g., a mutation that stabilizes this protein or amplifies the TNS4 gene, have not been discovered thus far. Thorpe et al[32] reported that TNS4 can be regulated by the EGFR/KRAS and STAT3 pathways. Other studies have indicated that the expression of this protein may be modulated by growth factors and cytokines, e.g., FGF2, PDGF, IGF-1, TGF-β, IL-6 and IL-13[33]. Other literature reports suggest that TNS4 may be engaged in tubulogenesis[34] as one of the stages of angiogenesis. The formation of blood vessel networks by tumor cells is the way in which the tumor microenvironment acquires nutrients and oxygen needed for tumor growth. Various publications have presented evidence for the oncogenic role of TNS4. In our research, we did not confirm the procancer role of the TNS4 protein. Our results prove that TNS4 may be related to the formation of a certain type of cancer, as observed in prostate cancer[35].

Moreover, we did not demonstrate any significant correlations between TNS4 expression level and the patients' overall survival rate. According to the works of Qi et *al*[17] and Sakashita *et al*[19], TNS4 overexpression correlates with shorter survival time and worse prognosis for patients with GC. In our study, the lack of this correlation may result from a small number of evaluated patients and the lack of data on the survival of many of these patients, which is a limitation of our research and does not allow us to draw broader conclusions about the participation of TNS4 in GC carcinogenesis. Additionally, it is possible that differences between the mRNA level in the study by Sakashita et al[19] and the protein expression level in our research are because the regulation of TNS4 transcription depends on the type of cells or tissues as well as on potential mutations in the TNS4 gene in GC cells. Thus, the expression level of this protein in different tissues is different. The control of this gene expression may also result from epigenetic modifications[36].

CONCLUSION

The expression of TNS4 was significantly higher in tumors with a diameter of ≥ 5 cm, of a moderately differentiated type of GC without a mucinous component, and of the intestinal type according to Lauren's classification. Increased levels of TNS4 expression are linked to the histological type of GC with a better prognosis.

ARTICLE HIGHLIGHTS

Research background

Gastric cancer (GC) is still one of the most common malignant neoplasms worldwide in terms of incidence and cancer motility. The development of GC is a multistage process. Tensin 4 (TNS4) belongs to the tensin family. This protein can participate in the transmission of intracellular signals. TNS4 plays an important role in biological processes connected with carcinogenesis, such as proliferation, migration, cell adhesion and invasiveness. It is very important to search for new biomarkers that



could help to diagnose GC at the early stages of its development.

Research motivation

We wanted to evaluate the role of the TNS4 protein in the development of GC. This protein may be a promising biomarker in the diagnosis of GC at the early stages of its development.

Research objectives

The study objective was to show the expression of TNS4 in GC tissues and assess the relationship between protein expression and clinical and pathological parameters and the overall survival rate of patients.

Research methods

In our study, we used immunohistochemistry to evaluate the expression of the TNS4 protein. The research was conducted on a group of 89 patients.

Research results

We observed that higher TNS4 expression was more often observed in GCs with a larger diameter (P = 0.040). Our results also showed that an increase in TNS4 expression was more frequently observed in tumors without mucinous components than in tumors from mucosal cancers (P = 0.023). Furthermore, higher TNS4 expression was demonstrated in moderately differentiated tumors (P = 0.002). Increased TNS4 expression was also noted in the intestinal type of GC according to Lauren's classification (P = 0.020). No significant correlation was found between the expression of TNS4 and the overall survival rate of patients.

Research conclusions

The expression of TNS4 was significantly higher in tumors with a diameter of ≥ 5 cm, of a moderately differentiated type of GC without a mucinous component, and of the intestinal type according to Lauren's classification. Increased levels of TNS4 expression are linked to the histological type of GC with a better prognosis.

Research perspectives

It is possible to develop this study with cell line methods. It will be possible to investigate the potential role of TNS4 in GC cell proliferation.

REFERENCES

- Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. CA Cancer J Clin 2018; 68: 394-424 [PMID: 30207593 DOI: 10.3322/caac.21492]
- Guggenheim DE, Shah MA. Gastric cancer epidemiology and risk factors. J Surg Oncol 2013; 107: 2 230-236 [PMID: 23129495 DOI: 10.1002/jso.23262]
- Rona KA, Schwameis K, Zehetner J, Samakar K, Green K, Samaan J, Sandhu K, Bildzukewicz N, 3 Katkhouda N, Lipham JC. Gastric cancer in the young: An advanced disease with poor prognostic features. J Surg Oncol 2017; 115: 371-375 [PMID: 28008624 DOI: 10.1002/jso.24533]
- Song Z, Wu Y, Yang J, Yang D, Fang X. Progress in the treatment of advanced gastric cancer. Tumour Biol 2017; 39: 1010428317714626 [PMID: 28671042 DOI: 10.1177/1010428317714626]
- Lo SH. Tensin. Int J Biochem Cell Biol 2004; 36: 31-34 [PMID: 14592531 DOI: 5 10.1016/s1357-2725(03)00171-7
- 6 Lo SH, Lo TB. Cten, a COOH-terminal tensin-like protein with prostate restricted expression, is down-regulated in prostate cancer. Cancer Res 2002; 62: 4217-4221 [PMID: 12154022]
- 7 Liao YC, Chen NT, Shih YP, Dong Y, Lo SH. Up-regulation of C-terminal tensin-like molecule promotes the tumorigenicity of colon cancer through beta-catenin. Cancer Res 2009; 69: 4563-4566 [PMID: 19487278 DOI: 10.1158/0008-5472.CAN-09-0117]
- 8 Albasri A, Seth R, Jackson D, Benhasouna A, Crook S, Nateri AS, Chapman R, Ilyas M. C-terminal Tensin-like (CTEN) is an oncogene which alters cell motility possibly through repression of Ecadherin in colorectal cancer. J Pathol 2009; 218: 57-65 [PMID: 19214987 DOI: 10.1002/path.2508]
- 9 Liao YC, Lo SH. Tensins - emerging insights into their domain functions, biological roles and disease relevance. J Cell Sci 2021; 134: jcs254029 [PMID: 33597154 DOI: 10.1242/jcs.254029]
- 10 Aratani K, Komatsu S, Ichikawa D, Ohashi T, Miyamae M, Okajima W, Imamura T, Kiuchi J, Nishibeppu K, Kosuga T, Konishi H, Shiozaki A, Fujiwara H, Okamoto K, Tsuda H, Otsuji E. Correction: Overexpression of CTEN relates to tumor malignant potential and poor outcomes of adenocarcinoma of the esophagogastric junction. Oncotarget 2019; 10: 5726 [PMID: 31620248 DOI:



10.18632/oncotarget.27229]

- 11 Albasri A, Aleskandarany M, Benhasouna A, Powe DG, Ellis IO, Ilyas M, Green AR. CTEN (Cterminal tensin-like), a novel oncogene overexpressed in invasive breast carcinoma of poor prognosis. Breast Cancer Res Treat 2011; 126: 47-54 [PMID: 20390342 DOI: 10.1007/s10549-010-0890-3]
- 12 Albasri A, Al-Ghamdi S, Fadhil W, Aleskandarany M, Liao YC, Jackson D, Lobo DN, Lo SH, Kumari R, Durrant L, Watson S, Kindle KB, Ilyas M. Cten signals through integrin-linked kinase (ILK) and may promote metastasis in colorectal cancer. Oncogene 2011; 30: 2997-3002 [PMID: 21339732 DOI: 10.1038/onc.2011.26]
- 13 Chan LK, Chiu YT, Sze KM, Ng IO. Tensin4 is up-regulated by EGF-induced ERK1/2 activity and promotes cell proliferation and migration in hepatocellular carcinoma. Oncotarget 2015; 6: 20964-20976 [PMID: 26035355 DOI: 10.18632/oncotarget.4122]
- 14 Sasaki H, Moriyama S, Mizuno K, Yukiue H, Konishi A, Yano M, Kaji M, Fukai I, Kiriyama M, Yamakawa Y, Fujii Y. Cten mRNA expression was correlated with tumor progression in lung cancers. Lung Cancer 2003; 40: 151-155 [PMID: 12711115 DOI: 10.1016/s0169-5002(03)00037-0]
- Sasaki H, Yukiue H, Kobayashi Y, Fukai I, Fujii Y. Cten mRNA expression is correlated with tumor 15 progression in thymoma. Tumour Biol 2003; 24: 271-274 [PMID: 15001839 DOI: 10.1159/0000761411
- Albasri A, Fadhil W, Scholefield JH, Durrant LG, Ilyas M. Nuclear expression of phosphorylated 16 focal adhesion kinase is associated with poor prognosis in human colorectal cancer. Anticancer Res 2014; 34: 3969-3974 [PMID: 25075018]
- Qi X, Sun L, Wan J, Xu R, He S, Zhu X. Tensin4 promotes invasion and migration of gastric cancer 17 cells via regulating AKT/GSK-3β/snail signaling pathway. Pathol Res Pract 2020; 216: 153001 [PMID: 32534709 DOI: 10.1016/j.prp.2020.153001]
- Sawazaki S, Oshima T, Sakamaki K, Aoyama T, Sato T, Shiozawa M, Yoshikawa T, Rino Y, Imada 18 T, Masuda M. Clinical Significance of Tensin 4 Gene Expression in Patients with Gastric Cancer. In Vivo 2017; 31: 1065-1071 [PMID: 29102927 DOI: 10.21873/invivo.11171]
- 19 Sakashita K, Mimori K, Tanaka F, Kamohara Y, Inoue H, Sawada T, Hirakawa K, Mori M. Prognostic relevance of Tensin4 expression in human gastric cancer. Ann Surg Oncol 2008; 15: 2606-2613 [PMID: 18612693 DOI: 10.1245/s10434-008-9989-8]
- Chen J, Zhang Y, Deng G, Ma J, Wu X, Qu Y, Zeng S. [Correlation between the expression of C-20 terminal tensin-like protein and the prognosis of hepatocellular carcinoma]. Zhong Nan Da Xue Xue Bao Yi Xue Ban 2014; 39: 1233-1239 [PMID: 25544169 DOI: 10.11817/j.issn.1672-7347.2014.12.003]
- Adachi Y, Yasuda K, Inomata M, Sato K, Shiraishi N, Kitano S. Pathology and prognosis of gastric 21 carcinoma: well versus poorly differentiated type. Cancer 2000; 89: 1418-1424 [PMID: 11013353]
- Wu WM, Liao YC. Downregulation of C-Terminal Tensin-Like Protein (CTEN) Suppresses Prostate 22 Cell Proliferation and Contributes to Acinar Morphogenesis. Int J Mol Sci 2018; 19 [PMID: 30332774 DOI: 10.3390/ijms19103190]
- 23 Machlowska J, Pucułek M, Sitarz M, Terlecki P, Maciejewski R, Sitarz R. State of the art for gastric signet ring cell carcinoma: from classification, prognosis, and genomic characteristics to specified treatments. Cancer Manag Res 2019; 11: 2151-2161 [PMID: 30936747 DOI: 10.2147/CMAR.S188622
- Jass JR, Sobin LH, Watanabe H. The World Health Organization's histologic classification of 24 gastrointestinal tumors. A commentary on the second edition. Cancer 1990; 66: 2162-2167 [PMID: 2171747 DOI: 10.1002/1097-0142(19901115)66:10<2162::aid-cncr2820661020>3.0.co;2-n]
- Chen YC, Fang WL, Wang RF, Liu CA, Yang MH, Lo SS, Wu CW, Li AF, Shyr YM, Huang KH. 25 Clinicopathological Variation of Lauren Classification in Gastric Cancer. Pathol Oncol Res 2016; 22: 197-202 [PMID: 26502923 DOI: 10.1007/s12253-015-9996-6]
- 26 Lauren P. The two histological main types of gastric carcinoma: diffuse and so-called intestinal-type carcinoma. An attempt at a histo-clinical classification. Acta Pathol Microbiol Scand 1965; 64: 31-49 [PMID: 14320675 DOI: 10.1111/apm.1965.64.1.31]
- 27 Asiri A, Toss MS, Raposo TP, Akhlaq M, Thorpe H, Alfahed A, Asiri A, Ilyas M. Cten promotes Epithelial-Mesenchymal Transition (EMT) in colorectal cancer through stabilisation of Src. Pathol Int 2019; 69: 381-391 [PMID: 31290243 DOI: 10.1111/pin.12811]
- Lu X, Gao J, Zhang Y, Zhao T, Cai H, Zhang T. CTEN induces epithelial-mesenchymal transition 28 (EMT) and metastasis in non small cell lung cancer cells. PLoS One 2018; 13: e0198823 [PMID: 29985912 DOI: 10.1371/journal.pone.0198823]
- 29 Thorpe H, Asiri A, Akhlaq M, Ilyas M. Cten promotes epithelial-mesenchymal transition through the post-transcriptional stabilization of Snail. Mol Carcinog 2017; 56: 2601-2609 [PMID: 28691764 DOI: 10.1002/mc.22704]
- 30 Al-Ghamdi S, Cachat J, Albasri A, Ahmed M, Jackson D, Zaitoun A, Guppy N, Otto WR, Alison MR, Kindle KB, Ilyas M. C-terminal tensin-like gene functions as an oncogene and promotes cell motility in pancreatic cancer. Pancreas 2013; 42: 135-140 [PMID: 22750970 DOI: 10.1097/MPA.0b013e3182557ceb]
- Muharram G, Sahgal P, Korpela T, De Franceschi N, Kaukonen R, Clark K, Tulasne D, Carpén O, 31 Ivaska J. Tensin-4-Dependent MET Stabilization Is Essential for Survival and Proliferation in Carcinoma Cells. Dev Cell 2014; 29: 629-630 [PMID: 28898622 DOI: 10.1016/j.devcel.2014.05.018]
- Thorpe H, Akhlaq M, Jackson D, Al Ghamdi S, Storr S, Martin S, Ilyas M. Multiple pathways 32 regulate Cten in colorectal cancer without a Tensin switch. Int J Exp Pathol 2015; 96: 362-369



[PMID: 26852686 DOI: 10.1111/iep.12154]

- Hung SY, Shih YP, Chen M, Lo SH. Up-regulated cten by FGF2 contributes to FGF2-mediated cell 33 migration. Mol Carcinog 2014; 53: 787-792 [PMID: 23625726 DOI: 10.1002/mc.22034]
- Kwon SH, Nedvetsky PI, Mostov KE. Transcriptional profiling identifies TNS4 function in epithelial 34 tubulogenesis. Curr Biol 2011; 21: 161-166 [PMID: 21236678 DOI: 10.1016/j.cub.2010.12.037]
- 35 Li Y, Mizokami A, Izumi K, Narimoto K, Shima T, Zhang J, Dai J, Keller ET, Namiki M. CTEN/tensin 4 expression induces sensitivity to paclitaxel in prostate cancer. Prostate 2010; 70: 48-60 [PMID: 19725034 DOI: 10.1002/pros.21037]
- Yang K, Wu WM, Chen YC, Lo SH, Liao YC. $\Delta Np63\alpha$ Transcriptionally Regulates the Expression 36 of CTEN That Is Associated with Prostate Cell Adhesion. PLoS One 2016; 11: e0147542 [PMID: 26784942 DOI: 10.1371/journal.pone.0147542]



WJC0

World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1215-1226

DOI: 10.5306/wjco.v12.i12.1215

ISSN 2218-4333 (online)

ORIGINAL ARTICLE

Basic Study Tumor irradiation may facilitate the detection of tumor-specific mutations in plasma

Ekaterina Kuligina, Fedor Moiseyenko, Sergey Belukhin, Ekaterina Stepanova, Maria Zakharova, Vera Chernobrivtseva, Ikram Aliev, Tatiana Sharabura, Vladimir Moiseyenko, Svetlana Aleksakhina, Tatiana Laidus, Aleksandr Martianov, Maksim Kholmatov, Aldon Whitehead, Grigoriy Yanus, Evgeny Imyanitov

ORCID number: Ekaterina Kuligina 0000-0002-2396-6540; Fedor Moiseyenko 0000-0003-2544-9042; Sergey Belukhin 0000-0002-3106-175X: Ekaterina Stepanova - 0000-0002-4013-181X; Maria Zakharova 0000-0003-1885-7574; Vera Chernobrivtseva 0000-0001-7037-177X: Ikram Aliev 0000-0001-6562-9422: Tatiana Sharabura 0000-0002-7111-7631; Vladimir Moiseyenko 0000-0003-4807-7915; Svetlana Aleksakhina 0000-0002-2149-7728: Tatiana Laidus 0000-0003-4988-1796; Aleksandr Martianov 0000-0002-7690-8328; Maksim Kholmatov 0000-0002-4201-8124; Aldon Whitehead 0000-0001-5921-3738; Grigoriy Yanus 0000-0002-9844-4536; Evgeny Imyanitov 0000-0003-4529-7891.

Author contributions: Kuligina ES, Moiseyenko FV, Moiseyenko VM, Yanus GA and Imyanitov EN designed and coordinated the study; Belukhin SA, Stepanova EO, Zakharova MS, Chernobrivtseva VV, Aliev II, Tatiana M Sharabura TM, Laidus TA, Martianov AS and Kholmatov MM performed the experiments, acquired the data and analyzed the results; Kuligina ES, Moiseenko FV, Aleksakhina SN, Laidus TA and Whitehead AJ interpreted the data; Kuligina ES, Moiseenko FV, Moiseyenko VM,

Ekaterina Kuligina, Svetlana Aleksakhina, Tatiana Laidus, Aleksandr Martianov, Maksim Kholmatov, Grigoriy Yanus, Evgeny Imyanitov, Department of Tumor Biology, N.N. Petrov Institute of Oncology, St.-Petersburg 197758, Russia

Ekaterina Kuligina, Svetlana Aleksakhina, Aleksandr Martianov, Grigoriy Yanus, Evgeny Imyanitov, Department of Medical Genetics, St.-Petersburg Pediatric Medical University, St.-Petersburg 194100, Russia

Fedor Moiseyenko, Vladimir Moiseyenko, Department of Therapy, City Cancer Center, St.-Petersburg 197758, Russia

Sergey Belukhin, Ikram Aliev, Department of Surgery, City Cancer Center, St.-Petersburg 197758, Russia

Ekaterina Stepanova, Maria Zakharova, Vera Chernobrivtseva, Tatiana Sharabura, Department of Radiology, City Cancer Center, St.-Petersburg 197758, Russia

Aldon Whitehead, Internal Medicine Residency Program, The University of Illinois College of Medicine, Chicago, IL 60612, United States

Evgeny Imyanitov, Department of Oncology, I.I. Mechnikov Northwestern Medical University, St.-Petersburg 191015, Russia

Corresponding author: Evgeny Imyanitov, DSc, Professor, Department of Tumor Biology, N.N. Petrov Institute of Oncology, Pesochny, St.-Petersburg 19775, Russia. evgeny@imyanitov.spb.ru

Abstract

BACKGROUND

The mutation-based analysis of circulating tumor DNA (ctDNA) is a promising diagnostic tool for clinical oncology. However, it has low success rate because many cancer patients do not have detectable ctDNA in the bloodstream.

AIM

To evaluate whether preoperative tumor irradiation results in a transient increase of plasma ctDNA concentration due to the induction of apoptosis in radiation-



Whitehead AJ and Imyanitov EN prepared the manuscript; all authors approved the final version of the article.

Institutional review board statement: The study was approved by the local Ethics Committee.

Conflict-of-interest statement: All authors have nothing to disclose.

Data sharing statement: All data generated or analyzed during this study are included in this

Supported by the Russian Science Foundation, No. 20-75-10163.

Country/Territory of origin: Russia

Specialty type: Oncology

published article.

Provenance and peer review:

Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): A Grade B (Very good): 0 Grade C (Good): C Grade D (Fair): D Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt p://creativecommons.org/License s/by-nc/4.0/

Received: April 28, 2021 Peer-review started: April 28, 2021 First decision: June 13, 2021 Revised: June 26, 2021 Accepted: November 28, 2021 Article in press: November 28, 2021 exposed cells.

METHODS

This study focused on patients with locally advanced rectal cancer, because preoperative tumor irradiation is a part of their standard treatment plan. Nine subjects, whose tumors contained KRAS, NRAS or BRAF mutations, donated serial blood samples 1 h prior to the first fraction of irradiation (at baseline), immediately after the first fraction (time 0), and 1, 3, 6, 12, 24, 36, 48, 72 and 96 h after the first fraction. The amount of mutated gene copies was measured by droplet digital PCR.

RESULTS

Five out of nine patients were mutation-negative by ctDNA test at baseline; two of these subjects demonstrated an emergence of the mutated DNA copies in the bloodstream within the follow-up period. There were 4 patients, who had detectable ctDNA in the plasma at the start of the experiment; three of them showed an evident treatment-induced increase of the content of mutated RAS/ RAF alleles.

CONCLUSION

Local tumor irradiation may facilitate the detection of tumor-specific DNA in the bloodstream. These data justify further assessment of the clinical feasibility of irradiation-assisted liquid biopsy.

Key Words: Liquid biopsy; Rectal cancer; KRAS; BRAF; Mutations; Tumor response; Radiotherapy

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: The detection of circulating tumor DNA (ctDNA) in cancer patients is compromised by the low sensitivity of this assay. We hypothesized that tumor irradiation may lead to the transient increase of ctDNA content due to induction of cell death. Nine patients with locally advanced RAS/RAF-mutated rectal cancer provided serial blood samples at baseline and during the first 96 h after the first dose of tumor irradiation. Treatment-induced elevation of the concentration of mutated RAS/RAF alleles in the blood was revealed in five of these subjects. In conclusion, local tumor irradiation may facilitate the detection of plasma ctDNA and thus improve the efficacy of liquid biopsy.

Citation: Kuligina E, Moiseyenko F, Belukhin S, Stepanova E, Zakharova M, Chernobrivtseva V, Aliev I, Sharabura T, Moiseyenko V, Aleksakhina S, Laidus T, Martianov A, Kholmatov M, Whitehead A, Yanus G, Imyanitov E. Tumor irradiation may facilitate the detection of tumorspecific mutations in plasma. World J Clin Oncol 2021; 12(12): 1215-1226

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1215.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1215

INTRODUCTION

"Liquid biopsy" is a popular diagnostic tool, which is based on the identification of tumor-specific markers in plasma or other body fluids. The analysis of several proteins, e.g., prostate-specific antigen, carcinoembryonic antigen, CA-125 etc. has been utilized for years, however, these assays have significant limitations with regard to specificity and sensitivity[1,2]. The examination of tumor-derived mutations in circulating DNA (ctDNA) is considered to be more promising[3]. Indeed, some methods of genetic testing, for instance, droplet digital PCR (ddPCR) or nextgeneration sequencing allow the detection of a single mutated allele within a huge excess of wild-type nucleic acids[4,5]. In addition, while the protein-based liquid biopsy is not truly cancer-specific but rather tissue-specific, oncogenic mutations are strongly associated with the malignant phenotype. For the time being, clinical use of



Published online: December 24, 2021

P-Reviewer: Chen SY, Suzuki R S-Editor: Lin M L-Editor: A P-Editor: Liu M



ctDNA tests is largely limited to the analysis of secondary mutations emerging during targeted therapy, as these assays may help to identify mechanisms of acquired drug resistance and therefore guide the subsequent treatment choice. It is anticipated that in the near future liquid biopsy will support other components of cancer care, such as screening, early diagnosis, analysis of treatment outcome and monitoring of relapse of tumor disease[6-11].

ctDNA-based liquid biopsy may have unacceptably low sensitivity. It appears that many categories of neoplasms (medulloblastomas; gliomas; cancers of kidney, thyroid, breast, etc.) are composed of relatively well-preserved cells, which do not shed DNA in the bloodstream, at least when the tumor is small[12,13]. Consequently, the analysis of ctDNA cannot substitute the detection of mutations in tumors tissue, *i.e.*, tissue biopsy cannot be easily replaced by liquid biopsy. The development of tools, which allow non-invasive examination of tumor characteristics, is of great value. One of the options may involve utilization of various agents, which induce tumor cell death and thus facilitate DNA shedding. In particular, tumor irradiation may increase tumor-specific ctDNA level due to the involvement of the above mechanism[14-16].

While considering the appropriate model for the validation of this assumption, we found it reasonable to focus on rectal cancer. Approximately 40%-50% of rectal carcinomas contain missense mutations in KRAS, NRAS or BRAF oncogenes, which can be used for ctDNA assays[17-19]. Furthermore, many rectal cancer patients undergo preoperative radiotherapy (RT) as a part of the treatment plan. Here we present the results of the study, which involved consecutive patients with mutationpositive rectal cancer. We demonstrate, that tumor irradiation indeed results in a transient increase of concentration of tumor-derived DNA and thus can be considered as a liquid biopsy supporting tool.

MATERIALS AND METHODS

Study workflow

The study considered treatment-naive patients with histologically verified locally advanced rectal cancer (T1-2/N1-2/M0, T3-4/N0-2/M0), who were referred to the St.-Petersburg City Cancer Center between February 2019 and April 2020 and who planned to undergo preoperative RT. The study was approved by the local Ethics Committee. Thirty patients provided informed consent and underwent RAS/RAF mutation testing (Figure 1). Thirteen analyzed tumors carried nucleotide substitutions in the mentioned genes. Four subjects failed to participate in the study due to various reasons (two tumors contained "rare" RAS mutations (KRAS A59G and NRAS G12C), which could not be detected by available ddPCR assays; 1 patient experienced rapid disease progression and was not subjected to RT; 1 patient developed acute paraproctitis and went to another hospital for treatment). Finally, nine patients were included in the ctDNA study and provided serial blood samples. Clinical characteristics of invited patients are summarized in the Supplementary Table 1.

RT and response evaluation

RT was performed according to routine procedures either with 45-50 Gy in 25-28 fractions or short-course radiation therapy (25 Gy in 5 fractions) with or without concurrent fluoropyrimidine-based chemotherapy. Chemotherapy was delivered according to standard regimens (capecitabine 825 mg/m² twice daily given within 5 d per week for 6-8 weeks). After RT patients were restaged with magnetic resonance imaging and the response was evaluated according to the TRG (tumor regression grade) system^[20]. When the patients were surgically treated, the pathologic response was evaluated according to Mandard criteria^[21]. The treatment results summary is presented in Supplementary Table 2.

Sample collection and processing

Patients provided blood at 11 different time points: 1 h before the first fraction of radiation (at baseline), immediately after the first fraction (time 0), and 1, 3, 6, 12, 24, 36, 48, 72 and 96 h after the first fraction (Figure 2). Ten milliliters of blood were collected into PAXgene Blood ccfDNA Tubes (Qiagen) or cf-DNA/cf-RNA Preservative Tubes (Norgen). Plasma samples were separated from the cellular fraction within 2-8 h after the blood-draw by two-step centrifugation (400 g for 10 min at room temperature followed by 14400 g for 10 min at 4 °C). The supernatants were aliquoted into 2 mL tubes and stored at -70 °C until further use. Cell-free DNA was extracted with the QIAmp Circulating Nucleic Acid Kit (Qiagen) as recommended by



Kuligina E et al. Tumor irradiation increases blood ctDNA concentration



Figure 1 The flowchart of patients screening.



Figure 2 Schedule for serial blood-takes and irradiation fractions.

Diefenbach et al[22]. Isolated DNA was subsequently diluted in sterile distilled water and frozen at -24 °C until further analysis.

ctDNA analysis

The fractions of KRAS/NRAS mutations in codons 12, 13, 61 or BRAF V600E allele were measured by ddPCR using the QX100 Bio-Rad System[23]. ddPCR reactions were performed in triplicate. These reactions contained 2X ddPCR Supermix for Probes (no UTP, Bio-Rad), mutation-specific oligonucleotides (see Supplementary Tables 3 and 4) and 2-3 µL of the template DNA in a total reaction volume of 22-23 µL. Data analyses were performed with the QuantaSoft Software version 1.7.4 as recommended by the manufacturer. All ddPCR reactions, which yielded 10 or more droplets with the target DNA molecule, were considered informative. The absolute number of tumor-derived "mutated" DNA copies in 1 mL of plasma (C_{mut}) was calculated according to the formula:



N mut copies/1 mL plasma = -

V plasma

Concentration $\left(\frac{\text{copies}}{\mu L} \text{ cfDNA}\right) \times \text{V}$ templat

Where: Concentration - number of «mutated» droplets per 1 µL of ddPCR reaction; $V_{temolate}$ - volume of ctDNA aliquot taken into ddPCR, μ L; $V_{dilution}$ - total volume of diluted ctDNA sample collected from the plasma, µL; V _{plasma} - volume of processed plasma, mL.

Statistics

The change of ctDNA content after tumor irradiation was evaluated according to the following formula:

Percentage change (%) = $100\% \times \frac{\text{final Cmut} - \text{initial Cmut}}{100\% \times 100\%}$

Quantitative data were present as a median values/range or means ± 95% confidence interval (1.960ox). The non-parametric Wilcoxon Signed Rank Test and Mann–Whitney *U* test were utilized to compare the medians. *P* value of < 0.05 was considered statistically significant. All calculations were performed using IBM SPSS v.23 software package.

RESULTS

Nine rectal patients were included in the study of ctDNA fluctuations occurring within the first hours after RT. Individual characteristics of the patients are given in Table 1. Four out of nine (44%) analyzed subjects had detectable RAS/RAF mutations in plasma DNA at baseline (at least 5 mutation-specific signals per reaction). The probability of detecting ctDNA in plasma did not correlate with any clinical characteristics, e.g. age, gender, mutation type, T/N stages, tumor grade, tumor location within the rectum, extramural venous invasion, circumferential resection margin, tumor response to treatment or PFS (statistical data not shown).

Three of 5 patients, who were negative for plasma RAS/RAF-mutated DNA at baseline, did not show the presence of mutated copies in subsequent serial samples obtained after tumor irradiation. The remaining two patients (#ArAS and #GaZM) demonstrated an appearance of the mutated DNA copies within the follow-up period (Table 2 and Figure 3).

Four patients were recognized as "plasma-positive" at baseline. The mean concentration of mutated copies in plasma samples was 82 copies per 1 mL (22, 23, 39, and 244, respectively). The variant allele frequency (VAF) of circulating mutations ranged from 0.5% to 27.2%. The analysis of changes in mutated ctDNA concentration occurring within the first 96 h of treatment revealed a pronounced increase in the number of circulating RAS/RAF mutant copies in patients #DaKS, #ArTP, #MaNK (with the maximum percent changes equal to 509%, 174% and 71%, respectively). Patient #MaLI showed less consistent variations in ctDNA content, with a maximal concentration at the start (23 mutant copies per 1 mL plasma, VAF 2.3%) and a number of subsequent spikes and drops (Figure 3).

There was no correlation between the content of mutated ctDNA and the total irradiation dose accumulated during the blood collection time (r = -0.400; P (2-tailed) = 0.253, Spearman's Rho).

DISCUSSION

Patients with locally advanced rectal cancer provide a good opportunity for the analysis of RT-induced changes in the ctDNA level, as these malignancies frequently contain RAS/RAF mutations and the tumor irradiation is a part of routine clinical management of this disease[21]. The data obtained within this study are consistent with prior investigations, which were performed on lung cancer patients and demonstrated that radiotherapeutic or chemoradiotherapeutic intervention may result in a transient increase of the level of ctDNA in some cases[14-16]. As compared to published reports[14-16], our study considered multiple evenly distributed time points within the first day after tumor irradiation. We anticipated, that this additional effort may help us to identify a time interval characterized by maximal RT-induced ctDNA release. However, there was a significant interpatient variability with regard to the



Table 1	Patients, 1	umor	s and	l trea	itment c	naracte	eristics													
Patient	Gender	Age	сT	cN	Stage	CRM	EMV	Tumor	RAS/RAFstatus	Total RT	Chemosensibiliza-	RECIST	Surgery	MRI	Mandard	урТ	уN	Progression ²	Follow-	ctDNA positive at
								location		uose, oy	tion			ING	ino			Yes/no	up, mo	baseline
ArAS	М	44	3	1	3	No	No	L	KRAS G12S	50	Yes	SD	Yes	III	NA	3	0	No	6.90	No
GaZM	F	66	3	2	3	Yes	No	U-M	KRAS G13D	46	Yes	PR	Yes	IV	4	4	1	No	4.03	No
DaKS	М	73	4	1	3	Yes	No	L	KRAS G12A	25 (short course)	No	PD	No	IV	NA	NA	NA	Yes	4.73	Yes
ArTP	F	81	4	1	3	Yes	Yes	L	KRAS Q61L	50	No	SD	Yes	III	3	3	0	Nd	5.80	Yes
MaLI	F	78	3	1	3	No	No	L	NRAS G12D	25(short course)	No	PD	No	IV	NA	NA	NA	Yes	3.97	Yes
MaNK	F	48	3	2	3	No	Yes	М	KRAS G12D	50	Yes	SD	Yes	III	2	3	2	No	6.57	Yes
ZuNM	F	63	3	2	3	No	No	L-M	BRAF V600E	44	Yes	PR	Yes	II	2	3	0	No	5.13	No
MiMF	F	74	3	1	3	No	Yes	М	NRAS G12D	25(short course)	No	PR	No	III	NA	NA	NA	No	3.13	No
SaVV	М	65	3	1	3	Yes	No	L-M	NRAS Q61R	50	Yes	SD	No	III	NA	NA	NA	Nd	1.13	No

¹Capecitabine 825 mg/m² twice daily was delivered on the days of RT (D1-5, D8-12, D 15-19, D22-26, D29-33).

²Disease status at March 1, 2021; Nd – lost from follow-up.

Tumor localization: U: Upper rectum, M: Middle rectum, L: Lower rectum; EMV: Extramural venous invasion; CRM: Circumferential resection margin; TRG: Tumor regression grade; CR: Complete response; PR: Partial response; SD: Stable disease; PD: Progressive disease; NA: Not evaluated; ctDNA: Circulating tumor DNA.

timing of ctDNA concentration peaks (Table 2 and Figure 3).

There are several limitations of this investigation. Human studies involving multiple serial blood takes are logistically complicated and need to be well balanced with ethical issues, therefore it is understandable that our study and similar reports [14-16] are of limited size. Furthermore, the range of "natural" variations of ctDNA measurements occurring due to imperfect reproducibility of laboratory protocols or physiological fluctuations of ctDNA content is largely unknown. Therefore, although our study demonstrated a trend towards the RT-induced increase of ctDNA concentration in some rectal cancer patients, it is not clear how these observations need to be adjusted for the described above confounding factors. This limitation is also applicable to other published data sets[14-16].

The analysis of tumor-specific mutations at the initial diagnostic work-up is usually not complicated, given that the management of cancer patients always requires morphological visualization of transformed cells and thus implies the availability of malignant tissue. However, the detection of actionable mutations acquired during the

	anges in circui	aling tumor DNA content dum	ig radiotherapy for	locally	auvai	iceu ie	cial co	IIICEI							
Patient ID	Mutation	Baseline mutated ctDNA	ctDNA analysis	-1 h	0	1 h	3 h	6 h	12 h	24 h	36 h	48 h	72 h	96 h	Radiation dose per day / number of fractions / total dose accumulated for the blood collection time
ArAS	KRAS G12S	Neg	C (mut) ¹	0	0	0	3	0	0	0	0	14	12	nd	$2.0 \text{ Gy}/4 \text{ fr}/8 \text{ Gy} + \text{capecitabine}^3$
			$C (mut + wt)^2$	642	1006	1337	1963	633	906	829	762	4298	3088	nd	
			VAF, %	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.3	0.4	nd	
GaZM	KRAS G13D	Neg	C (mut)	0	0	0	26	37	59	40	50	46	5	80	$2.0 \text{ Gy}/5 \text{ fr}/10 \text{ Gy} + \text{capecitabine}^3$
			C (mut + wt)	6	7	8	866	808	1450	604	441	506	112	1219	
			VAF, %	0.0	0.0	0.0	3.0	4.6	4.1	6.6	11.3	9.1	4.5	6.6	
DaKS	KRAS G12A	Pos	C(mut)	22	22	52	134	42	33	32	102	27	65	36	5.0 Gy/5fr/25 Gy
			C (mut + wt)	295	118	655	1802	588	292	170	549	359	391	331	
			VAF, %	7.5	18.6	7.9	7.4	7.1	11.3	18.8	18.6	7.5	16.6	10.9	
ArTP	KRAS Q61L	Pos	C (mut)	39	0	0	26	10	0	39	nd	42	107	nd	2.0 Gy/4 fr/8 Gy
			C (mut + wt)	7606	9430	4036	5028	3187	3033	12574	nd	10622	14443	nd	
			VAF, %	0.5	0.0	0.0	0.5	0.3	0.0	0.3	nd	0.4	0.7	nd	
MaLI	NRAS G12D	Pos	C (mut)	23	15	0	0	0	5	13	15	10	21	15	5.0 Gy/5 fr/25 Gy
			C (mut + wt)	981	1011	1073	1303	1309	1160	1010	724	815	673	1088	
			VAF, %	2.3	1.5	0.0	0.0	0.0	0.0	1.3	2.1	1.2	3.1	1.4	
MaNK	KRAS G12D	Pos	C (mut)	244	257	nd	nd	nd	Nd	254	335	415	418	387	$2.0 \text{ Gy}/5 \text{ fr}/10 \text{ Gy} + \text{capecitabine}^3$
			C (mut + wt)	897	1091	nd	nd	nd	Nd	1424	1832	2241	1719	1955	
			VAF, %	27.2	23.6	nd	nd	nd	Nd	17.8	18.0	18.5	24.3	19.8	
ZuNM	BRAF V600E	Neg	C (mut)	0	0	0	0	0	0	0	0	0	0	0	$2.0 \text{ Gy}/5 \text{ fr}/10 \text{ Gy} + \text{capecitabine}^3$
		-	C (mut + wt)	1652	372	522	522	522	522	671	837	1004	791	2805	
			VAF, %	0	0	0	0	0	0	0	0	0	0	0	
MiMF	NRAS G12D	Neg	C (mut)	0	0	0	0	0	0	0	0	0	0	0	5.0 Gy/5 fr/25 Gy
		0	C (mut + wt)	594	500	740	2294	804	681	681	386	644	2891	1536	
			VAF, %	0	0	0	0	0	0	0	0	0	0	0	
SaVV	NRAS Q61R	Neg	C (mut)	0	0	0	0	0	0	0	0	0	0	0	$2.0 \text{ Gy}/5 \text{ fr}/10 \text{ Gy} + \text{capecitabine}^3$
SaVV	NRAS Q61R	Neg	C (mut)	0	0	0	0	0	0	0	0	0	0	0	$2.0 \text{ Gy}/5 \text{ fr}/10 \text{ Gy} + \text{capecitabine}^3$

C (mut + wt)	196	237	541	840	621	748	353	372	272	261	249
VAF, %	0	0	0	0	0	0	0	0	0	0	0

¹C (mut): Number of mutated copies per 1 mL of plasma measured by droplet digital PCR.

²C (mut + wt): Total number of target fragment (both *wt* and *mut*) copies per 1 mL of plasma measured by droplet digital PCR.

 3 Capecitabine 825 mg/m² twice daily was delivered on the 1-5 d of RT.

ctDNA: Circulating tumor DNA; VAF: Variant allele frequency.

course of therapy presents a challenge. For example, the management of lung cancer patients, whose tumors progressed during gefitinib, erlotinib or afatinib treatment, involves the analysis of EGFR T790M mutation. The presence of this mutation justifies the administration of osimertinib, while the absence of this substitution calls for other treatment options. Re-biopsy of multiple visceral tumor lumps is often not feasible; therefore, the analysis of EGFR T790M mutation usually relies on liquid biopsy[24,25]. Clinical studies demonstrate that the detection of EGFR T790M mutation in plasma is seriously compromised by the low sensitivity of the test, especially in patients with limited tumor burden[26,27].

This study utilized patients with localized rectal cancer, who had a moderate volume of tumor masses. It is therefore explainable that only 4 out of 9 patients had detectable ctDNA at baseline. These data are comparable with the results obtained in other studies[12,28]. We deliberately focused on rectal cancer disease, as these patients often receive irradiation during the standard preoperative treatment, so no additional interventions were involved within this investigation. We demonstrated that two out of five subjects, who were initially ctDNA-negative, showed the presence of mutated DNA copies in the plasma after the start of the therapy. In addition, 3 out of 4 initially ctDNA-positive subjects experienced a RT-related increase of ctDNA content. The obtained data look promising, so further studies, which involve tumor irradiation not as a part of regular treatment plan, but as an additional intervention aimed to support ctDNA analysis, appear to be justified.

The clinical utility of this approach deserves to be evaluated in lung cancer patients, who demonstrate the disease progression during the therapy by first- or second-generation EGFR inhibitors and therefore require the diagnostic detection of EGFR T790M substitution. It is feasible to organize a prospective study, where the tumor lumps observed in these patients will be subjected to irradiation in order to provoke the release of tumor DNA in the bloodstream. It is essential to minimize the risks of this procedure by considering the anatomic location of targeted tumor foci (particularly, the vicinity of large blood vessels), ensuring a highly precise topical delivery of the irradiation dose and accounting for potentially significant comorbidities. If this intervention was to increase the rate of EGFR T790M allele detection in the plasma while being sufficiently safe, the proposed approach would have significant potential for clinical use.



Figure 3 Changes in ctDNA content occurring within first 96 h after the start of radiotherapy.

CONCLUSION

Local tumor irradiation may facilitate the detection of plasma ctDNA. This study calls for a comprehensive evaluation of the clinical feasibility of irradiation-assisted liquid biopsy.

ARTICLE HIGHLIGHTS

Research background The detection of circulating tumor DNA (ctDNA) is a valuable diagnostic tool,



Raishideng® WJCO | https://www.wjgnet.com

however many cancer patients do not have detectable amount of ctDNA in their plasma.

Research motivation

We evaluated whether tumor irradiation may provoke the release of tumor DNA in the bloodstream and thus improve the efficiency of liquid biopsy.

Research objectives

We have chosen for the study patients with locally advanced rectal cancer as they usually receive preoperative tumor irradiation as a part of standard treatment plan.

Research methods

The study included 9 patients with RAF/RAF mutations. Multiple serial blood draws were taken within first 96 h after the first fraction of radiotherapy. The amount of mutated RAF/RAF copies in the plasma was quantified by the droplet digital PCR.

Research results

Five out of nine patients demonstrated increased ctDNA content at least at some plasma samples obtained after the beginning of radiotherapy.

Research conclusions

Radiotherapy is a promising tool for the improvement of performance of liquid biopsy.

Research perspectives

It is feasible to extend this study to lung cancer patients, who receive tyrosine kinase inhibitors and may experience acquired tumor resistance due to the gain of secondary mutation.

REFERENCES

- 1 Locker GY, Hamilton S, Harris J, Jessup JM, Kemeny N, Macdonald JS, Somerfield MR, Hayes DF, Bast RC Jr; ASCO. ASCO 2006 update of recommendations for the use of tumor markers in gastrointestinal cancer. J Clin Oncol 2006; 24: 5313-5327 [PMID: 17060676 DOI: 10.1200/JCO.2006.08.2644]
- 2 Lakemeyer L, Sander S, Wittau M, Henne-Bruns D, Kornmann M, Lemke J. Diagnostic and Prognostic Value of CEA and CA19-9 in Colorectal Cancer. Diseases 2021; 9: 21 [PMID: 33802962 DOI: 10.3390/diseases90100211
- Cheng ML, Pectasides E, Hanna GJ, Parsons HA, Choudhury AD, Oxnard GR. Circulating tumor 3 DNA in advanced solid tumors: Clinical relevance and future directions. CA Cancer J Clin 2021; 71: 176-190 [PMID: 33165928 DOI: 10.3322/caac.21650]
- 4 Zhang H, Liu R, Yan C, Liu L, Tong Z, Jiang W, Yao M, Fang W, Chen Z. Advantage of Next-Generation Sequencing in Dynamic Monitoring of Circulating Tumor DNA over Droplet Digital PCR in Cetuximab Treated Colorectal Cancer Patients. Transl Oncol 2019; 12: 426-431 [PMID: 30562681 DOI: 10.1016/j.tranon.2018.11.015]
- 5 Suzuki T, Suzuki T, Yoshimura Y, Yahata M, Yew PY, Nakamura T, Nakamura Y, Park JH, Matsuo R. Detection of circulating tumor DNA in patients of operative colorectal and gastric cancers. Oncotarget 2020; 11: 3198-3207 [PMID: 32922660 DOI: 10.18632/oncotarget.27682]
- 6 Liu L, Toung JM, Jassowicz AF, Vijayaraghavan R, Kang H, Zhang R, Kruglyak KM, Huang HJ, Hinoue T, Shen H, Salathia NS, Hong DS, Naing A, Subbiah V, Piha-Paul SA, Bibikova M, Granger G, Barnes B, Shen R, Gutekunst K, Fu S, Tsimberidou AM, Lu C, Eng C, Moulder SL, Kopetz ES, Amaria RN, Meric-Bernstam F, Laird PW, Fan JB, Janku F. Targeted methylation sequencing of plasma cell-free DNA for cancer detection and classification. Ann Oncol 2018; 29: 1445-1453 [PMID: 29635542 DOI: 10.1093/annonc/mdy119]
- 7 Reece M, Saluja H, Hollington P, Karapetis CS, Vatandoust S, Young GP, Symonds EL. The Use of Circulating Tumor DNA to Monitor and Predict Response to Treatment in Colorectal Cancer. Front Genet 2019; 10: 1118 [PMID: 31824558 DOI: 10.3389/fgene.2019.01118]
- Tie J, Cohen JD, Wang Y, Christie M, Simons K, Lee M, Wong R, Kosmider S, Ananda S, 8 McKendrick J, Lee B, Cho JH, Faragher I, Jones IT, Ptak J, Schaeffer MJ, Silliman N, Dobbyn L, Li L, Tomasetti C, Papadopoulos N, Kinzler KW, Vogelstein B, Gibbs P. Circulating Tumor DNA Analyses as Markers of Recurrence Risk and Benefit of Adjuvant Therapy for Stage III Colon Cancer. JAMA Oncol 2019; 5: 1710-1717 [PMID: 31621801 DOI: 10.1001/jamaoncol.2019.3616]
- Formica V, Lucchetti J, Doldo E, Riondino S, Morelli C, Argirò R, Renzi N, Nitti D, Nardecchia A, 9 Dell'Aquila E, Ferroni P, Guadagni F, Palmieri G, Orlandi A, Roselli M. Clinical Utility of Plasma KRAS, NRAS and BRAF Mutational Analysis with Real Time PCR in Metastatic Colorectal Cancer



Patients-The Importance of Tissue/Plasma Discordant Cases. J Clin Med 2020; 10: 87 [PMID: 33383664 DOI: 10.3390/jcm10010087]

- 10 Naidoo M, Gibbs P, Tie J. ctDNA and Adjuvant Therapy for Colorectal Cancer: Time to Re-Invent Our Treatment Paradigm. Cancers (Basel) 2021; 13: 346 [PMID: 33477814 DOI: 10.3390/cancers13020346]
- 11 Rodriguez-Casanova A, Costa-Fraga N, Bao-Caamano A, López-López R, Muinelo-Romay L, Diaz-Lagares A. Epigenetic Landscape of Liquid Biopsy in Colorectal Cancer. Front Cell Dev Biol 2021; 9: 622459 [PMID: 33614651 DOI: 10.3389/fcell.2021.622459]
- Bettegowda C, Sausen M, Leary RJ, Kinde I, Wang Y, Agrawal N, Bartlett BR, Wang H, Luber B, 12 Alani RM, Antonarakis ES, Azad NS, Bardelli A, Brem H, Cameron JL, Lee CC, Fecher LA, Gallia GL, Gibbs P, Le D, Giuntoli RL, Goggins M, Hogarty MD, Holdhoff M, Hong SM, Jiao Y, Juhl HH, Kim JJ, Siravegna G, Laheru DA, Lauricella C, Lim M, Lipson EJ, Marie SK, Netto GJ, Oliner KS, Olivi A, Olsson L, Riggins GJ, Sartore-Bianchi A, Schmidt K, Shih IM, Oba-Shinjo SM, Siena S, Theodorescu D, Tie J, Harkins TT, Veronese S, Wang TL, Weingart JD, Wolfgang CL, Wood LD, Xing D, Hruban RH, Wu J, Allen PJ, Schmidt CM, Choti MA, Velculescu VE, Kinzler KW, Vogelstein B, Papadopoulos N, Diaz LA Jr. Detection of circulating tumor DNA in early- and latestage human malignancies. Sci Transl Med 2014; 6: 224ra24 [PMID: 24553385 DOI: 10.1126/scitranslmed.3007094]
- Zhang Y, Yao Y, Xu Y, Li L, Gong Y, Zhang K, Zhang M, Guan Y, Chang L, Xia X, Jia S, Zeng Q. 13 Pan-cancer circulating tumor DNA detection in over 10,000 Chinese patients. Nat Commun 2021; 12: 11 [PMID: 33397889 DOI: 10.1038/s41467-020-20162-8]
- Kageyama SI, Nihei K, Karasawa K, Sawada T, Koizumi F, Yamaguchi S, Kato S, Hojo H, Motegi 14 A, Tsuchihara K, Akimoto T. Radiotherapy increases plasma levels of tumoral cell-free DNA in nonsmall cell lung cancer patients. Oncotarget 2018; 9: 19368-19378 [PMID: 29721209 DOI: 10.18632/oncotarget.25053]
- Nygård L, Ahlborn LB, Persson GF, Chandrananda D, Langer JW, Fischer BM, Langer SW, 15 Gabrielaite M, Kjær A, Rosenfeld N, Mouliere F, Østrup O, Vogelius IR, Bentzen SM. Circulating cell free DNA during definitive chemo-radiotherapy in non-small cell lung cancer patients - initial observations. PLoS One 2020; 15: e0231884 [PMID: 32343749]
- 16 Walls GM, McConnell L, McAleese J, Murray P, Lynch TB, Savage K, Hanna GG, de Castro DG. Early circulating tumour DNA kinetics measured by ultra-deep next-generation sequencing during radical radiotherapy for non-small cell lung cancer: a feasibility study. Radiat Oncol 2020; 15: 132 [PMID: 32471446 DOI: 10.1186/s13014-020-01583-7]
- 17 Vaughn CP, Zobell SD, Furtado LV, Baker CL, Samowitz WS. Frequency of KRAS, BRAF, and NRAS mutations in colorectal cancer. Genes Chromosomes Cancer 2011; 50: 307-312 [PMID: 21305640 DOI: 10.1002/gcc.20854]
- Cancer Genome Atlas Network. Comprehensive molecular characterization of human colon and 18 rectal cancer. Nature 2012; 487: 330-337 [PMID: 22810696 DOI: 10.1038/nature11252]
- 19 Yanus GA, Belyaeva AV, Ivantsov AO, Kuligina ESh, Suspitsin EN, Mitiushkina NV, Aleksakhina SN, Iyevleva AG, Zaitseva OA, Yatsuk OS, Gorodnova TV, Strelkova TN, Efremova SA, Lepenchuk AY, Ochir-Garyaev AN, Paneyah MB, Matsko DE, Togo AV, Imyanitov EN. Pattern of clinically relevant mutations in consecutive series of Russian colorectal cancer patients. Med Oncol 2013; 30: 686 [PMID: 23943423 DOI: 10.1007/s12032-013-0686-5]
- 20 Patel UB, Brown G, Rutten H, West N, Sebag-Montefiore D, Glynne-Jones R, Rullier E, Peeters M, Van Cutsem E, Ricci S, Van de Velde C, Kjell P, Quirke P. Comparison of magnetic resonance imaging and histopathological response to chemoradiotherapy in locally advanced rectal cancer. Ann Surg Oncol 2012; 19: 2842-2852 [PMID: 22526897 DOI: 10.1245/s10434-012-2309-3]
- Mandard AM, Dalibard F, Mandard JC, Marnay J, Henry-Amar M, Petiot JF, Roussel A, Jacob JH, 21 Segol P, Samama G. Pathologic assessment of tumor regression after preoperative chemoradiotherapy of esophageal carcinoma. Clinicopathologic correlations. Cancer 1994; 73: 2680-2686 [PMID: 8194005 DOI: 10.1002/1097-0142(19940601)73:11<2680::aid-cncr2820731105>3.0.co;2-c]
- Diefenbach RJ, Lee JH, Kefford RF, Rizos H. Evaluation of commercial kits for purification of 22 circulating free DNA. Cancer Genet 2018; 228-229: 21-27 [PMID: 30553469 DOI: 10.1016/j.cancergen.2018.08.005]
- Sacher AG, Paweletz C, Dahlberg SE, Alden RS, O'Connell A, Feeney N, Mach SL, Jänne PA, 23 Oxnard GR. Prospective Validation of Rapid Plasma Genotyping for the Detection of EGFR and KRAS Mutations in Advanced Lung Cancer. JAMA Oncol 2016; 2: 1014-1022 [PMID: 27055085 DOI: 10.1001/jamaoncol.2016.0173]
- Jenkins S, Yang JC, Ramalingam SS, Yu K, Patel S, Weston S, Hodge R, Cantarini M, Jänne PA, 24 Mitsudomi T, Goss GD. Plasma ctDNA Analysis for Detection of the EGFR T790M Mutation in Patients with Advanced Non-Small Cell Lung Cancer. J Thorac Oncol 2017; 12: 1061-1070 [PMID: 28428148 DOI: 10.1016/j.jtho.2017.04.003]
- Akhoundova D, Mosquera Martinez J, Musmann LE, Britschgi C, Rütsche C, Rechsteiner M, Nadal 25 E, Garcia Campelo MR, Curioni-Fontecedro A. The Role of the Liquid Biopsy in Decision-Making for Patients with Non-Small Cell Lung Cancer. J Clin Med 2020; 9: 3674 [PMID: 33207619 DOI: 10.3390/jcm9113674]
- 26 Passiglia F, Rizzo S, Di Maio M, Galvano A, Badalamenti G, Listì A, Gulotta L, Castiglia M, Fulfaro F, Bazan V, Russo A. The diagnostic accuracy of circulating tumor DNA for the detection of EGFR-T790M mutation in NSCLC: a systematic review and meta-analysis. Sci Rep 2018; 8: 13379 [PMID:



30190486 DOI: 10.1038/s41598-018-30780-4]

- 27 Cortiula F, Pasello G, Follador A, Nardo G, Polo V, Scquizzato E, Del Conte A, Miorin M, Giovanis P, D'Urso A, Girlando S, Settanni G, Picece V, Veccia A, Corvaja C, Indraccolo S, De Maglio G. A Multi-Center, Real-Life Experience on Liquid Biopsy Practice for EGFR Testing in Non-Small Cell Lung Cancer (NSCLC) Patients. Diagnostics (Basel) 2020; 10: 765 [PMID: 32998450 DOI: 10.3390/diagnostics10100765]
- Sclafani F, Chau I, Cunningham D, Hahne JC, Vlachogiannis G, Eltahir Z, Lampis A, Braconi C, 28 Kalaitzaki E, De Castro DG, Wotherspoon A, Capdevila J, Glimelius B, Tarazona N, Begum R, Lote H, Hulkki Wilson S, Mentrasti G, Brown G, Tait D, Oates J, Valeri N. KRAS and BRAF mutations in circulating tumour DNA from locally advanced rectal cancer. Sci Rep 2018; 8: 1445 [PMID: 29362371 DOI: 10.1038/s41598-018-19212-5]



WJC0

World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1227-1243

DOI: 10.5306/wjco.v12.i12.1227

ISSN 2218-4333 (online)

SYSTEMATIC REVIEWS

Mixed odontogenic tumors: A review of the clinicopathological and molecular features and changes in the WHO classification

Celeste Sánchez-Romero, Oslei Paes de Almeida, Ronell Bologna-Molina

ORCID number: Celeste Sánchez-Romero 0000-0001-5365-2692; Oslei Paes de almeida 0000-0003-2002-8003; Ronell Bologna-Molina 0000-0001-9755-4779.

Author contributions: Sánchez-Romero C, Bologna-Molina R and Paes de Almeida O participated in the conceptualization,

bibliographic search, selection of information, interpretation of data, writing of the manuscript as well as in the subsequent revision of the manuscript.

Conflict-of-interest statement: All

the authors have indicated that they have no potential conflicts of interest and no financial relationships relevant to this article to disclose

PRISMA 2009 Checklist statement:

The authors have read the PRISMA 2009 Checklist, and the manuscript was prepared and revised according to the PRISMA 2009 Checklist.

Country/Territory of origin: Uruguay

Specialty type: Dentistry, oral surgery and medicine

Provenance and peer review: Invited manuscript; Externally peer reviewed.

Celeste Sánchez-Romero, Ronell Bologna-Molina, Department of Oral Pathology, Faculty of Dentistry, University Juarez of the Durango State, Durango 33106, Mexico

Oslei Paes de Almeida, Department of Diagnosis, Faculty of Piracicaba, State University of Campinas, São Paulo Brazil, Piracicaba 13414-903, Sao Paulo, Brazil

Ronell Bologna-Molina, Department of Oral and Maxillofacial Pathology, Universidad de la República UDELAR (URUGUAY), Montevideo 16400, Uruguay

Corresponding author: Ronell Bologna-Molina, DDS, MSc, PhD, Professor, Department of Oral and Maxillofacial Pathology, Universidad de la República UDELAR (URUGUAY), Montevideo 16400, Uruguay. ronellbologna@hotmail.com

Abstract

BACKGROUND

Ameloblastic fibromas and ameloblastic fibrosarcomas are rare odontogenic tumors, and controversy exists in the classification of cases presenting hard-tissue production: Ameloblastic fibrodentinoma (AFD) and ameloblastic fibro-odontoma (AFO). These cases are currently considered "developing odontomas" (hamartomatous lesions).

AIM

To analyze the clinicopathologic features of these lesions and discuss the changes in the 2017 World Health Organization classification.

METHODS

An electronic literature search was performed in the PubMed/MEDLINE database. An electronic search of the English language literature was performed and last updated in September 2020 in the PubMed/MEDLINE database using the following terms: "ameloblastic fibroma", "ameloblastic fibrodentinoma", "ameloblastic fibro-odontoma", "ameloblastic sarcoma", "ameloblastic fibrosarcoma", "ameloblastic fibrodentinosarcoma", "ameloblastic fibroodontosarcoma" and "odontogenic carcinosarcoma". The inclusion criteria were odontogenic tumor series, case reports and systematic reviews that provided sufficient clinical, radiological and microscopic documentation to confirm the diagnosis.

RESULTS

The database search strategy resulted in 947 papers. Articles focusing on other



Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): 0 Grade B (Very good): B, B Grade C (Good): 0 Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt ps://creativecommons.org/Licens es/by-nc/4.0/

Received: January 19, 2021 Peer-review started: January 19, 2021 First decision: May 14, 2021 Revised: May 25, 2021 Accepted: November 25, 2021 Article in press: November 25, 2021 Published online: December 24, 2021

P-Reviewer: Mesquita RA, Rattan V S-Editor: Ma YI

L-Editor: A P-Editor: Ma YJ



topics, articles that were not in English, duplicate articles, and articles without fulfilling the inclusion criteria were excluded. Finally, 96 publications were included in this review to describe and discuss the main features of the searched entities. Several aspects of AFO and AFD, such as biological behavior, age of occurrence, amount of hard tissue, and potential for malignant transformation into odontogenic sarcomas, support the neoplastic nature in most of the reported cases. Considering the clinical, radiographic, histopathological and molecular characteristics of odontogenic lesions with hard tissue production, we suggest that these types of lesions should continue to be recognized as odontogenic tumors by maintaining the classically used terms.

CONCLUSION

This recommendation will be relevant for future clinical, microscopic, and molecular studies to better understand the biology of these interesting odontogenic tumors.

Key Words: Ameloblastic fibroma; Ameloblastic fibrosarcoma; Odontogenic carcinosarcoma; Odontogenic tumors

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.

Core Tip: We consider that the recent 2017 World Health Organization classification does not clarify the subject when considering ameloblastic fibrodentinoma (AFD) and ameloblastic fibro-odontoma (AFO) as "developing odontomas". According to the clinical, radiographical, histopathological and molecular features of the cases reviewed, we suggest that AFD and AFO should continue to be considered benign neoplasms. Thus, the nomenclature of these mixed benign odontogenic tumors would be congruent with the classification of ameloblastic/odontogenic sarcomas. Additionally, further studies are warranted to compare these interesting odontogenic tumors and finally better clarify and understand their similarities and differences.

Citation: Sánchez-Romero C, Paes de Almeida O, Bologna-Molina R. Mixed odontogenic tumors: A review of the clinicopathological and molecular features and changes in the WHO classification. World J Clin Oncol 2021; 12(12): 1227-1243

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1227.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1227

INTRODUCTION

Ameloblastic fibroma (AF) is a rare, benign odontogenic tumor formed by odontogenic ectomesenchyme that resembles the dental papilla, has embedded epithelial strands and nests, and is similar to dental lamina and enamel organs but without the presence of hard tissues. If the lesion has dentinoid material, it must be denominated as ameloblastic fibrodentinoma (AFD); when it produces enamel/enamel matrix, it is known as ameloblastic fibro-odontoma (AFO), independent of the amount of hard tissue present[1]. In 2005, WHO classified AF and AFD together, making no distinctions regarding epidemiological and clinical features. Microscopically, the only difference between AF and AFD is the presence of dentinoid in the latter. AFO affects younger patients and has shown a better prognosis than AF/AFD[1]. However, the WHO classification of 2017 states that AF rarely produces dental hard tissues and cases formerly considered AFD/AFO rather represent developing odontomas^[2].

This group of mixed odontogenic tumors (AF and related-lesions) histologically resembles different tooth formation stages, particularly when dentin and enamel are produced, sharing similar morphologic features with the so-called "developing odontoma", which is considered a tumor-like malformation or hamartoma by WHO. Nevertheless, unlike odontomas, these mixed tumors present characteristics that support the concept of a true neoplasm, such as biological behavior, age of occurrence, and well-documented cases of malignant transformation into odontogenic sarcomas, namely, ameloblastic fibrosarcoma (AFS), ameloblastic fibrodentinosarcoma (AFDS)



and ameloblastic fibro-odontosarcoma (AFOS). Moreover, the recent publication of a few reports of odontogenic carcinosarcomas led to its inclusion as a specific tumor by WHO in 2017[2-4].

This review is based on the WHO classification of 2005, because most of the literature is based on this nomenclature, and it was performed to analyze the clinicopathologic features of these lesions to discuss the changes in the 2017 WHO classification.

MATERIALS AND METHODS

An electronic search of the English language literature was performed and last updated in September 2020 in the PubMed/MEDLINE database using the following terms: "ameloblastic fibroma", "ameloblastic fibrodentinoma", "ameloblastic fibroodontoma", "ameloblastic sarcoma", "ameloblastic fibrosarcoma", "ameloblastic fibrodentinosarcoma", "ameloblastic fibroodontosarcoma" and "odontogenic carcinosarcoma".

Previous cases that did not use the current terminology for these tumors, recently identified as AF, AFD, AFO, AFS, AFDS or AFOS, were also found and evaluated for possible inclusion.

The inclusion criteria were odontogenic tumor series, case reports and systematic reviews including AF, AFD, AFO, AFS, AFDS or AFOS, which provided sufficient clinical, radiological and microscopic documentation to confirm the diagnosis. Reports without this information were excluded.

RESULTS

The database search strategy resulted in 947 papers. Articles focusing on other topics, articles that were not in English, duplicate articles, and articles without fulfilling the inclusion criteria were excluded. Finally, 96 publications were included in this review to describe and discuss the main features of the searched entities.

AF

Clinical characteristics: This uncommon benign mixed odontogenic tumor occurs preferentially in children and young adults, with a mean age of 14.9 years, ranging from 7 wk to 57 years. Only 20% of the cases are diagnosed in patients older than 20 years. Considering all odontogenic tumors, AF represents only 0.6% to 3.1% of these neoplasms. Most of the cases affect the mandible, with a slight predilection for male patients, with a male/female ratio of 1.4:1. The size of AF when diagnosed varies from 0.7 to 16 cm (mean of 4.05 cm)[5].

Most cases present as painless jaw swelling or are discovered during routine radiographical examination due to delayed tooth eruption, eventually causing cortical expansion and facial asymmetry. Approximately 80% of AF involves the posterior region of the mandible but has also been found on the posterior maxilla and rarely in the anterior region of the jaws[5,6].

Radiographic features: Radiographically, AF presents as a well-defined, unilocular (56%) or multilocular (44%) radiolucent lesion, with regular and well-defined margins, typically sclerotic (94%). Tumors measuring less than 5 cm usually tend to be unilocular. Approximately 80% of cases are associated with a single or several unerupted teeth, usually of permanent dentition. Root resorption and cortical perforation are uncommon and described in 8.1% and 5.2% of cases, respectively [5,6].

Histopathology: Microscopically, AF is a mixed tumor with variable amounts of epithelial and ectomesenchymal components in different areas of the same lesion. The ectomesenchyme resembles the embryonic dental papilla, comprising a myxoid cellrich stroma involving odontogenic epithelial elements that may present different patterns: epithelial strands, comprising a double layer of cuboidal cells (Figure 1A); cords with tooth bud-like projections of cuboidal cells (Figure 1B); epithelial follicles comprising a layer of peripheral tall columnar ameloblast-like cells and a central area, displaying more loosely arranged stellate/spindle-shaped cells, similar to the stellate reticulum of the enamel organ (Figure 1C); clefts of mesenchymal tissue surrounding follicular epithelial proliferation can be present (Figure 1D); and smaller epithelial rosette-like islands that resemble remnants of dental lamina may be observed




Figure 1 Diverse aspects of the odontogenic epithelium in ameloblastic fibromas within the cell-rich myxoid stroma. A: Epithelial strands, comprising a double layer of cuboidal cells (HE, 20×); B: Epithelial proliferation with primitive appearance that resembles tooth bud-like structures (HE, 20×); C: Epithelial component with a follicular pattern comprising columnar cells at the periphery of the nests with central stellate reticulum-like cells (HE, 10×); D: Clefts of mesenchymal tissue surrounding follicular epithelial proliferations (HE, 20×); E: Mild hyalinization surrounding the basal layer of the epithelial nest (left). Smaller epithelial rosette-like islands resemble remnants of dental lamina (right) (HE, 40×); F: A very low rate of proliferation in both mesenchymal and epithelial components, showing the benign behavior of ameloblastic fibromas (IHC for Ki-67, 20×).

> (Figure 1E). Mitotic figures in either the epithelium or mesenchyme are uncommon, a finding consistent with the benign nature of the tumor. According to the 2005 WHO classification, no hard tissues, such as enamel or dentin, are present[1]. In the 2017 classification, AF rarely presented dental hard tissue formation that eventually reached an exceptional size[2]. According to the histopathological criteria of 2005, 280 cases of AF were identified in the literature. The proliferation rate is low, with a Ki-67 Labeling index generally lower than 3% (Figure 1F)[5-7].

> Treatment and prognosis: Most reported AFs were treated conservatively by enucleation and curettage. Radical surgery is used in more extensive tumors or recurrent lesions. Recurrence was reported in 16.3% of cases, and malignant transformation into AFS was cited in 6.4%. Recurrence seems to be more common in younger patients and malignant transformation more common in older patients^[5].

AFD

Clinical characteristics: AFD is a rare benign odontogenic tumor with histopathological features of AF and the formation of dysplastic dentin. The WHO classification



Zaishidena® WJCO | https://www.wjgnet.com

of 2005 describes AF and AFD together, without further considerations of the latter, beyond the presence of dentin/dentinoid. No strong evidence of differences in the biological behavior of AF and AFD is available[1,8]. However, the 2017 WHO classification of tumors cited that lesions referred to as AFDs are more likely "developing odontomas", and the editors suggest that they are no longer being considered mixed odontogenic tumors, as in the previous classification^[2].

AFD is rare, corresponding to less than 1% of all odontogenic tumors in most reported series. It usually presents as asymptomatic swelling, more frequently at the posterior mandible (mandible/maxilla ratio of 2.4:1), often associated with a permanent unerupted tooth. When a deciduous tooth is involved, the lesions are generally located in the incisor area. From the 45 cases reviewed, we found a slight male predilection, corresponding to 59.5% of cases, usually in the first and second decades of life; however, 17 of 45 (37.7%) cases occurred in the third decade and beyond. The mean age was 17.8 years, with an age range of 1 to 63 years (Table 1).

Radiographic features: Radiographically, AFD presents as a well-defined radiolucency with varying degrees of radiopacity, depending on the amount of calcified dentinoid. In 2012, Giraddi and Garg[9] reported a large and aggressive AFD with irregular borders, with considerable expansion and perforation of the cortical bone; however, the possibility of eventual foci of malignant transformation to AFDS should be considered in this case.

Histopathology: Microscopically, AFD is formed by odontogenic epithelium and ectomesenchyme arranged in an indistinguishable pattern from AF, in addition to the presence of dentinoid (Figure 2A, B). The epithelial cords and islands resemble the dental lamina and enamel organ, lying in myxoid cell-rich ectomesenchymal tissue with stellate-shaped fibroblasts resembling dental papilla. The amount of dentinoid material is variable, but minimal evidence is sufficient for the diagnosis to be accepted [1]. We found only 45 cases of AFD in the English literature according to these characteristics. Similar to AF, the Ki-67 index in AFD is low in both epithelial and mesenchymal components[8].

Treatment and prognosis: The treatment of choice is surgical, with enucleation of the lesion and unerupted tooth involvement. Recurrence is uncommon (9%) and likely a consequence of incomplete surgical removal. Radical surgery has been used in aggressive, atypical or recurrent lesions[9]. AFD rarely progresses into ameloblastic fibrodentinosarcoma, in which only the mesenchymal component shows malignant transformation. Only 4 cases of AFDS with a preexisting benign lesion have been described in the English literature (Table 2).

AFO

Clinical characteristics: AFO is a slow-growing, expansive, benign mixed odontogenic tumor that is histologically similar to AFD but also contains enameloid material in variable amounts^[1]. Similar to AFD, the term AFO was excluded from the latest WHO classification, in which lesions with these characteristics are considered developing odontomas^[2].

According to the literature, AFO occurs mainly in children, with a mean age of 9.6 years. It has a male predilection, with a ratio of 1.85:1 and an average size of 3.3 cm, ranging from 0.8 to 14 cm. More than 80% of cases affect the posterior portion of the mandible, eventually causing facial asymmetry[10,11]. To our knowledge, 222 cases have been reported in the English literature, among which 211 were reviewed by Chrcanovic *et al*^[5] and 11 additional cases were published later^[12-22], including one peripheral case^[23]. One case was associated with paresthesia of the chin and lower lip in a 12-year-old girl[22].

Radiographic features: AFO usually appears as a well-defined unilocular mixed radiolucent-radiopaque lesion, frequently in close association with the crown of an unerupted tooth. It commonly causes painless cortical expansion but no perforation [10,11].

We reviewed the literature and found 82 cases with optimal radiographic documentation, among which 22 (26.8%) presented radiographically as a single large opaque mass similar to odontoma and 11 (13.4%) presented several foci of opacities; however, most cases were poor in hard tissues, with 43 cases presenting few opacities (52.4%) and 6 cases appearing as radiolucent lesions (7.3%).

Histopathology: Similar to AF, AFO comprises odontogenic epithelium and ectomesenchyme, but it also contains hard dental tissues in variable amounts and degrees of



WJCO | https://www.wjgnet.com

Table 1 Reported 45 cases of ameloblastic fibrodentinoma found in the English language literature

Case	Ref.	Year	Sex/age	Location
1	Straith[42]	1936	F/30	Posterior mandible
2	Field and Ackerman ^[43]	1942	NA/9	Posterior mandible
3	Stafne[44]	1943	F/25	Posterior mandible
4	Stafne[44]	1943	NA/23	Posterior mandible
5	Stafne[45]	1946	NA	NA
6	Thoma and Goldman[46]	1946	M/6	Maxillary sinus area
7	Ingham[47]	1952	F/19	Posterior mandible
8	Sirsat[48]	1952	M/36	Maxillary sinus area
9	Husted and Pindborg[49]	1953	M/4	Anterior maxilla
10	Husted and Pindborg[49]	1953	F/63	Posterior mandible
11	Hitchin and White[50]	1955	M/4	Anterior mandible
12	Pindborg[51]	1955	M/20	Posterior mandible
13	Gorlin <i>et al</i> [52]	1961	F/4	Anterior maxilla
14	Gorlin <i>et al</i> [52]	1961	F/13	Posterior mandible
15	Gorlin <i>et al</i> [52]	1961	M/8	Posterior mandible
16	Azaz et al[53]	1967	M/4.5	Anterior mandible
17	Manning and Browne ^[54]	1970	F/55	Posterior mandible
18	Hoggins and Browne ^[55]	1976	M/24	Posterior mandible
19	Gulmen et al[56]	1976	M/30	Anterior mandible
20	Godjesk et al[57]	1980	M/3.5	Anterior mandible
21	Rennie and Critchlow[58]	1981	M/7	Posterior maxilla
22	van Wyk and van der Vyver <mark>[59]</mark>	1983	M/8	Posterior mandible
23	Villafañe <i>et al</i> [60]	1986	F/22	Posterior maxilla
24	Lukinmaa et al <mark>[61</mark>]	1897	M/11	Posterior mandible
25	Anker and Radden[62]	1989	F/24	Posterior mandible
26	Ulmansky <i>et al</i> [63]	1994	M/60	Posterior maxilla
27	Ulmansky et al[63]	1994	M/8	Posterior maxilla
28	Cassidy <i>et al</i> [64]	1987	M/12	Posterior maxilla
29	Akal et al[65]	1997	M/9	Posterior mandible
30	Akal et al[65]	1997	M/22	Mandible
31	Takeda <i>et al</i> [66]	2000	M/21	Mandibular retromolar area
32	Karasu <i>et al</i> [67]	2004	F/21	NA
33	Bhargava <i>et al</i> [68]	2011	M/51	Anterior maxilla
34	Giraddi and Garg[9]	2012	F/17	Mandible
35	Bologna-Molina et al[8]	2013	F/1.5	Mandible
36	Sankireddy <i>et al</i> [69]	2013	M/14	Anterior maxilla
37	Salehinejad <i>et al</i> [70]	2013	F/13	Anterior mandible
38	Ikeda et al[71]	2014	F/8	Posterior mandible
39	Lee et al[72] ¹	2014	F/4	Anterior mandible
40	Unsal et al[73]	2014	M/11	Anterior mandible
41	Joseph et al[74]	2015	M/12	Anterior Maxilla



42	Costa et al[75]	2015	F/12	Posterior mandible
43	Bhargava et al[76]	2016	M/1	Anterior mandible
44	Bavle <i>et al</i> [77]	2017	F/14	Posterior mandible
45	Sabu et al[78]	2018	M/20	Mandible (left body to right parasymphysis)

¹Associated with calcifying cystic odontogenic tumor, only dentinoid production. F: Female; M: Male; NA: Not available.

Table 2 Main data of 21 cases reported of ameloblastic fibrodentinosarcoma/ameloblastic fibro-odontosarcoma in the literature

Case	Ref.	Sex/age	Location	Mineralized tissues	Preexisting tumor	Progression
1	Villa[79]	F/20 yr	Posterior mandible	Enamel	Yes (NA)	Recurrence
2	Forman and Garret[80]	M/17 yr	Posterior mandible	Dentin and enamel	No	No recurrence
3	Altini and Smith[81]	M/27 yr	Mandible	Dentin	No	NA
4	Takeda et al[32]	M/19 yr	Maxilla	Dentin	AF	Recurrence and death
5	Howell and Burkes[31]	F/18 yr	Posterior mandible	Dentin and enamel	AFO	Recurrence, metastasis and death
6	Howell and Burkes[31]	M/36 yr	Posterior mandible	Dentin	AFO	Recurrence
7	Altini <i>et al</i> [<mark>41</mark>]	M/25 yr	Mandible	Dentin	No	No recurrence
8	Takeda et al[33]	M/23 yr	Mandible	Dentin and enamel	No	Recurrence and death
9	Corominas-Villafañe <i>et al</i> [82]	M/12 yr	Mandible	NA	AF	No recurrence
10	Herzog <i>et al</i> [83] ¹	F/14 yr	Mandible	NA	AFO	NA
11	Bregni et al[25]	M/32 yr	Mandible	Dentin	No	NA
12	Muller <i>et a</i> [84]	M/83 yr	Mandible	Dentin and enamel	AFO	Recurrence
13	Zabolinejad et al[35]	M/4 mo	Maxillary sinus	Dentin	No	No recurrence
14	Mainenti et al[34]	F/12 yr	Mandible	Dentin and enamel	AFO	No recurrence
15	Wang et al[30]	F/45 yr	Posterior mandible	Dentin and enamel	No	No recurrence
16	Reiser et al[85]	F/6 yr	Mandible	Dentin and enamel	No	No recurrence
17	Khan et al[<mark>86</mark>]	F/17 yr	Mandible	NA	No	NA
18	Gatz et al et al[87]	F/14 yr	Maxilla	Dentin	AFO	Recurrence
19	Chen et al[88]	M/4 yr	Mandible	Dentin and enamel	No	No recurrence
20	Niu et al[<mark>89</mark>]	F/31 yr	Mandible	Dentin and enamel	No	No recurrence at 3 months, lost follow-up
21	Atarbashi-Moghadam <i>et al</i> [90]	F/32 yr	Mandible	Dentin	No	Recurrence and metastasis

¹Article in German, abstract in English.

F: Female; M: Male; AF: Ameloblastic fibroma; AFO: Ameloblastic fibro-odontoma; AFDS: Ameloblastic fibrodentinosarcoma; AFOS: Ameloblastic fibroodontosarcoma; NA: Not available.

> maturation, such as enameloid and dentinoid (Figure 2C-F). Frequently, Ki-67 is lower than 1% in epithelial and mesenchymal cells[1,24].

> Treatment and prognosis: The treatment of choice is conservative surgery through enucleation and curettage, with removal of the unerupted tooth. Recurrence is uncommon, and malignant transformation is very rare, with only 6 cases reported to



Raishideng® WJCO https://www.wjgnet.com



Figure 2 Mineralized tissue formation in ameloblastic fibrodentinoma and ameloblastic fibro-odontoma. A, B: Dentinoid induction by epithelial cells in ameloblastic fibrodentinoma; note the presence of tubules in (A) (HE, 20x); C: Prominent proliferation of soft tissue similar to ameloblastic fibromas and focal areas of dentinoid and enamel matrix production in close relationship with the epithelial component in ameloblastic fibro-odontoma (HE, 2.5×; inset 20×); D: Structures similar to tubules are observed in the dentinoid (arrowhead), which can be associated with odontogenic epithelium or ectomesenchymal tissue, while enamel matrix (arrow) associated with columnar odontogenic appears more basophilic, with different patterns of deposition that can resemble prisms or globules (HE, 20x); E: Calcificated material, compatible with enameloid, in direct relationship with epithelial cells of the stellate reticulum-like area (HE, 20x); F: Details of the columnar ameloblast-like cells with reverse polarization producing enamel matrix in which the "fish scale" pattern is visible. Flattened cells between the columnar cells and stellate reticulum-like area resemble the stratum intermedium of the tooth germ, which is believed to assist the ameloblast in producing enamel during odontogenesis (HE, 40×).

date[5,10].

AFS

Clinical characteristics: AFS is a very rare malignant odontogenic tumor, considered the malignant counterpart of AF, in which the ectomesenchymal component shows features of sarcoma, while the odontogenic epithelium remains bland[1]. To date, up to 100 cases of AFS have been reported in the English language literature. The mean age of the affected patients was 28 years (range: 3 to 89 years), with a slight predilection for male patients (male-to-female ratio: 1.6:1). AFS is more frequent in the mandible, and up to one-third of the cases have been derived from previously documented AF. Patients with AFS originating from AF have a mean age of 33 years, and those with de novo AFS (previous AF not demonstrated) are one decade younger. AFS usually appears as painful swelling, and paraesthesia may be present[5,25,26].



Zaishidena® WJCO | https://www.wjgnet.com

Radiographic features: As with most malignant intraosseous tumors, AFS presents as expansive ill-defined unilocular or multilocular radiolucency with bone destruction areas, perforation of cortical areas, irregular margins and occasional root resorption. It can be associated with unerupted teeth and eventually cause diffuse expansion and thinning of the cortex[2,27].

Histopathology: Histologically, AFS is similar to AF; however, the ectomesenchyme is hypercellular and malignant, while the epithelial component tends to decrease and virtually disappears in recurrent tumors (Figure 3A-C)[1,28]. Epithelial nests and cords remain inactive, presenting an immunohistochemical profile similar to AF and positivity for AE1/AE3[15-17] (Figure 3E). Proliferation-related markers such as Ki-67 and p53 can help distinguish AF and AFS, because they are virtually negative in the mesenchymal component and epithelium of AF and positive in a variable percentage in the malignant cells of AFS (Figure 3F).

Treatment and prognosis: Because AFS is locally very aggressive, with a high tendency to relapse, treatment includes wide surgical removal and long-term followup. Adjuvant radiotherapy has been used in some cases with favorable results [29], while the usefulness of chemotherapy has not been confirmed^[26]. Recurrence occurs in approximately 20% and metastasis in only 4.5% of cases, but the mortality after 5 years of treatment is relatively high, estimated in 25.4% of cases [1,25,27].

Ameloblastic fibrodentinosarcoma/ameloblastic fibro-odontosarcoma

In the 2005 and 2017 WHO classifications of tumors, AFDS/AFOS are described together as tumors with histological features of AFS presenting dentinoid (fibrodentinosarcoma) or dentinoid and enameloid (fibro-odontosarcoma) (Figure 3D) [1,2].

Clinical and histopathological features: Clinically, AFDS/AFOS present as painful swelling of the jaws, with only 21 cases reported in the literature, as summarized in Table 2[9,30]. From these cases, 10 described enameloid formation, corresponding to AFOS. The age range of the reported cases was from 4 mo to 83 years, with a peak in the third decade. Approximately 40% of the cases recurred, two developed metastasis, and three patients died because of aggressive local invasion[31-33].

Immunohistochemically, AFDS/AFOS are similar to AFS, with odontogenic epithelium positive for AE1/AE3, facilitating the localization of epithelial nests in cases of mesenchymal predominance, excluding the diagnosis of other sarcomas. As discussed previously, proliferation markers such as Ki-67 and p53 confirm the aggressiveness of the lesion, helping to differentiate it from its respective benign counterpart [30,34,35].

Radiographic features: Radiographically, they appear as a uni- or multilocular radiolucency with variable dense opacities associated with impacted teeth. Irregular borders, expansion and perforation of the cortex are common, indicating a malignant tumor^[34].

Treatment and prognosis: Treatment is based on wide local surgical excision, and long-term follow-up is advised[30]. We found 21 cases of AFDS/AFOS in the English language literature, 8 of which (38%) recurred, 2 developed distant metastasis, and 3 cases (15%) caused death.

Molecular characterization of mixed odontogenic tumors

To date, few studies have investigated the genetic/molecular profiling of mixed odontogenic tumors. Molecular testing (polymerase chain reaction followed by direct sequencing and next-generation sequencing) has revealed that 33% to 100% of benign mixed odontogenic tumors (AF, AFD, and AFO) and 71% of AFS harbor BRAF p. V600E mutation in their mesenchymal component (and rare cases in both components), unlike odontomas, which are BRAF wild-type. This finding suggests that a subset of AF, AFD and AFO differs molecularly from odontomas, likely supporting the distinct nature of these entities (neoplastic vs hamartomatous). The BRAF p.V600E mutation is involved in the pathogenesis of several tumors, including ameloblastoma, playing a role as a downstream activator of the MAPK signaling pathway, which regulates several cell processes, such as proliferation, survival and apoptosis[36-38].Confirming these findings, immunohistochemical reactions against BRAFV600E exhibited specific staining only in the stromal component, supporting the role of this mutation as a driver of the malignant stromal component[38]. Although the BRAF p.V600E mutation seems to be present in most AFSs in the study of Agaimy et al





Figure 3 Histopathological aspects of ameloblastic fibromas and ameloblastic fibro-odontosarcoma. Marked pleomorphism and atypia in mesenchymal cells (HE, 20×) (A-C). A: Several mitotic figures, nuclear hyperchromatism and multinucleated and aberrant cells are seen in a highly pleomorphic sarcomatous component of ameloblastic fibromas, while epithelial islands remain benign; B: A follicular benign epithelial island is surrounded by hypercellularized sarcomatous proliferation; C: Malignant mesenchymal tissue resembling a storiform pattern and haphazard disposition of sarcomatous cells around an epithelial branching cord; D: Production of enamel matrix (arrowhead) and dentinoid (arrow) as well as malignant mesenchymal tissue (left side) are components of this ameloblastic fibro-odontosarcoma (A-D: HE, 20×); E: Cytokeratins can help to localize odontogenic epithelial cells within dominant sarcomatous proliferation (IHC for AE1/AE3, 20×); F: Most mesenchymal malignant cells show nuclear positivity for p53 antigen (IHC for p53, 20×).

> [38], the NRAS p.Gln61Lys mutation was also detected in one AFS case, and another case was wild type.

DISCUSSION

AF, AFD and AFO present similar clinical, radiographic and microscopic features and were accepted as different entities until the 2017 WHO classification, which considers these entities as representing diverse stages of maturation of a developing odontoma, as suggested by Cahn and Blum in the past[39]. In summary, we suggest that the lesions referred to as AFD and AFO are more likely "developing odontomas" and are no longer considered mixed odontogenic tumors, as in the previous classifications of odontogenic tumors.

We consider that the WHO classification of these tumors in 2017 is unclear because it states that it is not possible to differentiate histologically between AF (true neoplasms) and early-stage odontomas before they differentiate and mature; therefore,



Zaisbideng® WJCO | https://www.wjgnet.com

the existence of both lesions is accepted. However, no evidence has shown that AF matures to odontoma. If maturation of AF occurred, it would be expected that AF (immature lesion) would be diagnosed at an earlier age than AFO (mature lesion). However, an opposite trend has occurred: as the mean age of diagnosis of AFO is 9.6 years and that of AF is 14.9 years. However, because several AFOs have been reported in children in areas of odontogenesis, some cases might represent odontomas. Nevertheless, some cases of AFD/AFO arise in age groups that are not consistent with a hamartoma: from the 45 reviewed cases of AFD, 17 cases (37.7%) were aged \geq 20 years, and 7 cases (15.5%) were aged \geq 30 years, 4 of which were in the sixth and seventh decades of life.

The terms AFD and AFO were practically discarded from the latest WHO tumor classification, considering that once hard tissues are produced, these tumors are more likely to form odontomas[2,40]. Nevertheless, in this WHO classification, AFD and AFO might conceptually be neoplastic when reaching an exceptionally large size but without establishing a measure for this statement^[2].

To avoid concepts that may be confusing and that are not appropriately supported by scientific evidence, we suggest not using the term developing odontomas and simply continuing to use odontomas if the clinical, radiographical and microscopic characteristics support this well-established diagnosis. Cases of a typical odontoma associated with AF could be termed AF associated with odontoma because odontomas can eventually be associated with other odontogenic tumors.

We accept that some cases are difficult to classify as AFO or odontoma because of the large amounts of hard dental tissues and because some cases of odontoma have been diagnosed as AFO. However, a cut off could be considered based on the proportion of hard and epithelial-ectomesenchymal tissues, as well as clinical (size, location, age, and clinical behavior) and radiographic features.

No evidence exists that all AF/AFD/AFO are "developing odontomas" because each of these tumors has its own clinicopathological features. AF is a well-recognized entity, and it should also be emphasized that no evidence is available that AF matures and forms small or large amounts of hard dental tissues, even in cases of recurrence.

AFD has no potential to produce enamel/enameloid; therefore, it cannot mature to an odontoma. However, some AFOs can produce large amounts of hard dental tissues and may mimic radiographically and microscopically odontomas; nevertheless, most AFOs present relatively few calcified areas. We reviewed 82 cases in the English literature with adequate radiographic documentation, most of which had small amounts of hard tissues: 59.8% presented few opacities or radiolucent images, 13.4% showed a higher number of scattered opacities, and only 26.8% presented a single opaque mass similar to odontoma. Even considering these cases rich and poor in calcified dental tissues diagnosed as AFO, evidence exists that cases poor in dental calcified structures evolve to those that mimic odontomas.

Recent molecular studies have shown genetic differences (principally, BRAFV600E mutation) between odontoma (BRAF wild type) and a subset of AF, AFD, AFO and most AFS, supporting that these lesions may represent distinct entities with a neoplastic nature[36-38].

In summary, we propose to continue to use the classical terms AFD and AFO because it is part of the 2017 WHO classification for malignant counterparts. This recommendation can be relevant for future clinical, microscopic and molecular studies to better clarify the subject and better understand the biology of these interesting odontogenic tumors.

Several aspects support the neoplastic nature of AF, AFD and AFO, such as their biological behavior, significant frequency of BRAF mutation, age of occurrence, amount of hard tissue and potential for malignant transformation into odontogenic sarcomas with or without the production of dental hard tissues. Among the 18 cases of AFDS/AFOS reported in the literature, 6 were related to a preexisting AFO, and this malignant transformation would not be expected in a hamartomatous lesion as a developing odontoma. The 2017 WHO classification accepts AFS, AFDS and AFOS as entities, and they can be de novo or derived from AF. This inconsistency in the nomenclature between benign and malignant corresponding tumors probably occurred because the topics of "odontogenic sarcomas" and "ameloblastic fibroma" were written by different authors in the 2017 WHO classification.

Odontogenic carcinosarcoma was added to the 2017 WHO classification based on 6 case reports, considering that it may arise de novo or can be derived from previous AF or AFS. However, we also found in the literature that, in two cases, ameloblastoma and malignant ameloblastoma were reported as the preceding tumors (Table 3). In contrast to AFS, in which metastasis is rare, 33% (3 cases) of odontogenic/ameloblastic carcinosarcomas presented biphasic metastasis (epithelial and sarcomatous



Table	Table 3 Main data of 9 cases reported of ameloblastic/odontogenic carcinosarcoma in the literature													
Case	Ref.	Sex/age	Location	Preexisting tumor	Progression									
1	Tanaka et al[91]	M/63	Maxilla	Malignant ameloblastoma	Recurrence, metastasis and death									
2	Slama <i>et al</i> [92] ¹	F/26	Mandible	AF	Metastasis and death									
3	Kunkel <i>et al</i> [3]	M/52	Mandible	No	Recurrence, metastasis and death									
4	DeLair et al[93]	F/19	Mandible	AF	No recurrence									
5	Chikosi <i>et al</i> [94]	F/9	Mandible	Ameloblastoma	Recurrence and death									
6	Kim <i>et al</i> [4]	M/61	Mandible	No	No recurrence									
7	Dos Santos et al[95]	M/42	Maxilla	No	Unknown									
8	Soares et al[96]	M/22	Mandible	No	No recurrence									
9	Soares et al[96]	F/19	Mandible	Rhabdomyosarcoma (parotid region) ²	Post-surgical systemic infection and death									

¹Article in French, abstract in English.

²Thirteen years before, treated with surgical resection followed by radiotherapy.

M: Male: F: Female: AF: Ameloblastic fibroma

components), and 5 of 9 cases resulted in death[3]. Thus, this entity was recently recognized at the present WHO classification. Immunohistochemically, positivity for p53 and a Ki-67 index > 45% in both carcinomatous and sarcomatous components can be useful to confirm the diagnosis^[2].

It is reasonable to consider that basic benign and malignant neoplasms are AF and AFS and that the presence of small amounts of dental hard tissues does not significantly alter the biological characteristics and clinical behaviors of these entities [1,13]. Although not clearly established, the presence and higher amount of hard tissues may indicate less aggressiveness and possibly lower potential of malignant transformation. In this context, AFO should have a better prognosis than AF/AFD, with a lesser tendency for malignant transformation. AFDS/AFOS seem to have a similar rate of recurrence as AF; however, the metastasis and mortality indexes seem to be higher in AFSs. Additionally, the number of cases of AFD/AFO and AFDS/ AFOS reported is very small, making comparisons of these tumors with AF/AFS difficult.

Reports of AFSs have been present for several years, possibly as AF/AFO/AFD that have suddenly followed an aggressive course before being treated, indicating a possible malignant transformation[25,41].

CONCLUSION

In summary, we reviewed the principal clinical, histopathological and molecular characteristics of AF, AFD and AFO and their malignant counterparts. Odontogenic/ameloblastic carcinosarcoma was cited because, according to reports, it can arise from preexisting AF. We consider that the recent 2017 WHO classification does not clarify the subject when considering AFD and AFO as developing odontomas. According to the clinical, radiographical, histopathological and molecular features of the cases reviewed, we suggest that AFD and AFO should continue to be considered benign neoplasms. Thus, the nomenclature of these mixed benign odontogenic tumors would be congruent with the classification of ameloblastic/odontogenic sarcomas. Additionally, further studies are warranted to compare these interesting odontogenic tumors and finally better clarify and understand their similarities and differences.

ARTICLE HIGHLIGHTS

Research background

Ameloblastic fibromas and ameloblastic fibrosarcomas are rare odontogenic tumors, and controversy exists in the classification of cases presenting hard-tissue production: Ameloblastic fibrodentinoma (AFD) and ameloblastic fibro-odontoma (AFO). These



cases are currently considered "developing odontomas" (hamartomatous lesions). There is still controversy as to whether they are true hamartomas or neoplasms.

Research motivation

The authors consider that the recent 2017 WHO classification does not clarify the subject when considering AFD and AFO as "developing odontomas". According to the clinical, radiographical, histopathological and molecular features of the cases reviewed, we suggest that AFD and AFO should continue to be considered benign neoplasms.

Research objectives

The objective was to analyze the clinicopathologic features of these lesions and discuss the changes in the 2017 WHO classification.

Research methods

For this systematic review an electronic literature search was performed in the PubMed/MEDLINE database. An exhaustive search was made of all the existing information on these mixed odontogenic tumors.

Research results

Several aspects of AFO and AFD, such as biological behavior, age of occurrence, amount of hard tissue, and potential for malignant transformation into odontogenic sarcomas, support the neoplastic nature in most of the reported cases.

Research conclusions

Considering the clinical, radiographic, histopathological, and molecular characteristics of odontogenic lesions with hard tissue production, we suggest that these types of lesions should continue to be recognized as odontogenic tumors by maintaining the classically used terms. This recommendation will be relevant for future clinical, microscopic, and molecular studies to better understand the biology of these interesting odontogenic tumors. This new information will be relevant for the clinical conduct to be followed in these tumors.

Research perspectives

Future research should be focused on the comparative molecular study between these odontogenic neoplasms and odontomas; trying to clarify molecular differences between neoplasia and hamartoma.

REFERENCES

- 1 Barnes L, Eveson JW, Reichart P, Sidransky D. World Health Organization Classification of Tumours: Pathology and Genetics of Head and Neck Tumours. Lyon: IARC Press; 2005
- El-Naggar A, Chan JK, Grandis JR, Takata T, Slootweg PJ. WHO Classification of Head and Neck 2 Tumours. 4th ed. Lvon: IARC Press: 2017
- Kunkel M, Ghalibafian M, Radner H, Reichert TE, Fischer B, Wagner W. Ameloblastic fibrosarcoma 3 or odontogenic carcinosarcoma: a matter of classification? Oral Oncol 2004; 40: 444-449 [PMID: 14969825 DOI: 10.1016/j.oraloncology.2003.09.010]
- 4 Kim IK, Pae SP, Cho HY, Cho HW, Seo JH, Lee DH, Park IS. Odontogenic carcinosarcoma of the mandible: a case report and review. J Korean Assoc Oral Maxillofac Surg 2015; 41: 139-144 [PMID: 26131431 DOI: 10.5125/jkaoms.2015.41.3.139]
- Chrcanovic BR, Brennan PA, Rahimi S, Gomez RS. Ameloblastic fibroma and ameloblastic 5 fibrosarcoma: A systematic review. J Oral Pathol Med 2018; 47: 315-325 [PMID: 28776760 DOI: 10.1111/jop.12622]
- 6 Chen Y, Li TJ, Gao Y, Yu SF. Ameloblastic fibroma and related lesions: a clinicopathologic study with reference to their nature and interrelationship. J Oral Pathol Med 2005; 34: 588-595 [PMID: 16202078 DOI: 10.1111/j.1600-0714.2005.00361.x]
- Sano K, Yoshida S, Ninomiya H, Ikeda H, Ueno K, Sekine J, Iwamoto H, Uehara M, Inokuchi T. Assessment of growth potential by MIB-1 immunohistochemistry in ameloblastic fibroma and related lesions of the jaws compared with ameloblastic fibrosarcoma. J Oral Pathol Med 1998; 27: 59-63 [PMID: 9526730 DOI: 10.1111/j.1600-0714.1998.tb02094.x]
- 8 Bologna-Molina R, Salazar-Rodríguez S, Bedoya-Borella AM, Carreón-Burciaga RG, Tapia-Repetto G, Molina-Frechero N. A histopathological and immunohistochemical analysis of ameloblastic fibrodentinoma. Case Rep Pathol 2013; 2013: 604560 [PMID: 23476862 DOI: 10.1155/2013/604560]
- Giraddi GB, Garg V. Aggressive atypical ameloblastic fibrodentinoma: Report of a case. Contemp



Clin Dent 2012; 3: 97-102 [PMID: 22557908 DOI: 10.4103/0976-237X.94557]

- Boxberger NR, Brannon RB, Fowler CB. Ameloblastic fibro-odontoma: a clinicopathologic study of 10 12 cases. J Clin Pediatr Dent 2011; 35: 397-403 [PMID: 22046699 DOI: 10.17796/jcpd.35.4.t3387t25865758w3]
- 11 Buchner A, Kaffe I, Vered M. Clinical and radiological profile of ameloblastic fibro-odontoma: an update on an uncommon odontogenic tumor based on a critical analysis of 114 cases. Head Neck Pathol 2013; 7: 54-63 [PMID: 23001451 DOI: 10.1007/s12105-012-0397-9]
- 12 Melo Filho MR, Pêgo SPB, Cardoso CM, Rocha BA, Martelli-Júnior H, Flores IL, Dos Santos LAN, Paranaiba LMR. Metachronous ameloblastic fibro-odontoma and dentigerous cyst in the posterior mandible. Gen Dent 2017; 65: 69-72 [PMID: 29099370]
- 13 Aly N, Amer H, Khatib OE. Ameloblastic fibro-odontoma with chondroid tissue formation. Contemp Oncol (Pozn) 2018; 22: 50-53 [PMID: 29692665 DOI: 10.5114/wo.2018.74395]
- 14 Kufta K, Kang S, Alawi F, Moran A, Panchal N. Ameloblastic Fibro-Odontoma of the Maxilla in a Pierre-Robin Sequence Patient. Fetal Pediatr Pathol 2017; 36: 416-422 [PMID: 28557592 DOI: 10.1080/15513815.2017.1324547
- Mashhadiabba F, Shamloo N, Jafari M, Vafadar S. Ameloblastic Fibro-Odontoma in a 7-Month-Old 15 Infant: A Case Report. J Dent (Shiraz) 2017; 18: 234-236 [PMID: 29034280]
- Peters SM, Bergen MS, Philipone EM, Yoon AJ. Ameloblastic Fibro-Odontoma in an Adolescent: A 16 Case Report and Review of Literature. J Clin Pediatr Dent 2018; 42: 458-460 [PMID: 30085878 DOI: 10.17796/1053-4625-42.6.10]
- 17 Kalra A, Pajpani M, Webb R. Ameloblastic Fibro-Odontoma. J Dent Child (Chic) 2018; 85: 143-146 [PMID: 30869592]
- Kale SG, Shetty A, Balakrishnan J, Purvey P. Ameloblastic Fibro-odontoma with a Predominant 18 Radiopaque Component. Ann Maxillofac Surg 2017; 7: 304-307 [PMID: 29264304 DOI: 10.4103/ams.ams 84 17]
- Thulasirman SK, Thuasidoss G, Prabhu NK, Krishnakumar Raja VB. A Rare Case of Ameloblastic 19 Fibro-Odontoma of Mandible with Literature Review. Ann Maxillofac Surg 2018; 8: 324-326 [PMID: 30693255 DOI: 10.4103/ams.ams_127_18]
- Abdulla AM, Sivadas G, Surej Kumar LK, Sheejith Hari Peeceeyen CS, Vedam V. Ameloblastic 20 Fibroodontoma: Uncommon Case Presentation in a 6-Year-Old Child with Review of the Literature. Case Rep Med 2017; 2017: 9483738 [PMID: 28883834 DOI: 10.1155/2017/9483738]
- Rao AJP, Reddy M, Mahanthi VL, Chalapathi KV. Ameloblastic fibro-odontoma in a 14 year old 21 girl: A case report. J Cancer Res Ther 2019; 15: 715-718 [PMID: 31169249 DOI: 10.4103/jcrt.JCRT 215 17]
- Saeed DM, Setty S, Markiewicz MR, Cabay RJ. Ameloblastic fibro-odontoma associated with 22 paresthesia of the chin and lower lip in a 12-year-old girl. SAGE Open Med Case Rep 2019; 7: 2050313X19870642 [PMID: 31452891 DOI: 10.1177/2050313X19870642]
- Chrcanovic BR, Gomez RS. Ameloblastic Fibrodentinoma and Ameloblastic Fibro-Odontoma: An 23 Updated Systematic Review of Cases Reported in the Literature. J Oral Maxillofac Surg 2017; 75: 1425-1437 [PMID: 28153756 DOI: 10.1016/j.joms.2016.12.038]
- Martínez Martínez M, Romero CS, Piña AR, Palma Guzmán JM, de Almeida OP. Pigmented 24 ameloblastic fibro-odontoma: clinical, histological, and immunohistochemical profile. Int J Surg Pathol 2015; 23: 52-60 [PMID: 25339415 DOI: 10.1177/1066896914553663]
- 25 Bregni RC, Taylor AM, García AM. Ameloblastic fibrosarcoma of the mandible: report of two cases and review of the literature. J Oral Pathol Med 2001; 30: 316-320 [PMID: 11334469 DOI: 10.1034/j.1600-0714.2001.300510.x]
- Pourdanesh F, Mohamadi M, Moshref M, Soltaninia O. Ameloblastic Fibrosarcoma of the Mandible 26 With Distant Metastases. J Oral Maxillofac Surg 2015; 73: 2067.e1-2067.e7 [PMID: 26207695 DOI: 10.1016/j.joms.2015.07.003]
- 27 Lai J, Blanas N, Higgins K, Klieb H. Ameloblastic fibrosarcoma: report of a case, study of immunophenotype, and comprehensive review of the literature. J Oral Maxillofac Surg 2012; 70: 2007-2012 [PMID: 22177804 DOI: 10.1016/j.joms.2011.09.012]
- Gilani SM, Raza A, Al-Khafaji BM. Ameloblastic fibrosarcoma: a rare malignant odontogenic tumor. 28 Eur Ann Otorhinolaryngol Head Neck Dis 2014; 131: 53-56 [PMID: 23845294 DOI: 10.1016/j.anorl.2013.03.001
- Hu YY, Deng MH, Yuan LL, Niu YM. Ameloblastic fibrosarcoma of the mandible: A case report and 29 mini review. Exp Ther Med 2014; 8: 1463-1466 [PMID: 25289041 DOI: 10.3892/etm.2014.1940]
- 30 Wang S, Shi H, Wang P, Yu Q. Ameloblastic fibro-odontosarcoma of the mandible: imaging findings. Dentomaxillofac Radiol 2011; 40: 324-327 [PMID: 21697160 DOI: 10.1259/dmfr/80061108
- Howell RM, Burkes EJ Jr. Malignant transformation of ameloblastic fibro-odontoma to ameloblastic 31 fibrosarcoma. Oral Surg Oral Med Oral Pathol 1977; 43: 391-401 [PMID: 265043 DOI: 10.1016/0030-4220(77)90326-7
- Takeda Y, Kaneko R, Suzuki A. Ameloblastic fibrosarcoma in the maxilla, malignant transformation 32 of ameloblastic fibroma. Virchows Arch A Pathol Anat Histopathol 1984; 404: 253-263 [PMID: 6437063 DOI: 10.1007/BF00694891]
- 33 Takeda Y, Kuroda M, Suzuki A. Ameloblastic odontosarcoma (ameloblastic fibro-odontosarcoma) in the mandible. Acta Pathol Jpn 1990; 40: 832-837 [PMID: 2077816 DOI: 10.1111/j.1440-1827.1990.tb02497.x



- Mainenti P, Oliveira GS, Valério JB, Daroda LS, Daroda RF, Brandão G, Rosa LE. Ameloblastic 34 fibro-odontosarcoma: a case report. Int J Oral Maxillofac Surg 2009; 38: 289-292 [PMID: 19150219 DOI: 10.1016/j.ijom.2008.11.025]
- 35 Zabolinejad N, Hiradfar M, Anvari K, Razavi AS. Ameloblastic fibrosarcoma of the maxillary sinus in an infant: a case report with long-term follow-up. J Pediatr Surg 2008; 43: e5-e8 [PMID: 18280269 DOI: 10.1016/j.jpedsurg.2007.09.077]
- Coura BP, Bernardes VF, de Sousa SF, Diniz MG, Moreira RG, de Andrade BAB, Romañach MJ, 36 Pontes HAR, Gomez RS, Odell EW, Gomes CC. Targeted Next-Generation Sequencing and Allele-Specific Ouantitative PCR of Laser Capture Microdissected Samples Uncover Molecular Differences in Mixed Odontogenic Tumors. J Mol Diagn 2020; 22: 1393-1399 [PMID: 32966885 DOI: 10.1016/j.jmoldx.2020.08.005
- You Z, Xu LL, Li XF, Zhang JY, DU J, Sun LS. [BRAF gene mutations in ameloblastic fibromas]. 37 Beijing Da Xue Xue Bao Yi Xue Ban 2019; 51: 4-8 [PMID: 30773536 DOI: 10.19723/j.issn.1671-167X.2019.01.002]
- Agaimy A, Skalova A, Franchi A, Alshagroud R, Gill AJ, Stoehr R, Baumhoer D, Bauer S. 38 Ameloblastic fibrosarcoma: clinicopathological and molecular analysis of seven cases highlighting frequent BRAF and occasional NRAS mutations. Histopathology 2020; 76: 814-821 [PMID: 31899815 DOI: 10.1111/his.140531
- 39 Cahn LR, Blum T. Ameloblastic odontoma: case report critically analyzed. J Oral Surg (Chic) 1952; 10: 169-170
- Wright JM, Vered M. Update from the 4th Edition of the World Health Organization Classification 40 of Head and Neck Tumours: Odontogenic and Maxillofacial Bone Tumors. Head Neck Pathol 2017; 11: 68-77 [PMID: 28247226 DOI: 10.1007/s12105-017-0794-1]
- 41 Altini M, Thompson SH, Lownie JF, Berezowski BB. Ameloblastic sarcoma of the mandible. J Oral Maxillofac Surg 1985; 43: 789-794 [PMID: 3862778 DOI: 10.1016/0278-2391(85)90336-2]
- 42 Straith FE. Odontoma. A rare type. Report of a case. Dent Dig 1936; 42: 196
- 43 Field H, Ackerman A. Calcifying fibro-adamantoblastoma. Am J Orthod 1942; 28: 543-545
- Stafne E. Dentinoma; report of 2 cases. Am J Orthod 1943; 29: 156-159 44
- Stafne EC. Dentinoma; report of a case. J Oral Surg (Chic) 1946; 4: 145-147 [PMID: 20983332] 45
- Thoma KH, Goldman HM. Odontogenic tumors; a survey of seventy-five cases. Am J Orthod 1946; 46 32: 763-791 [PMID: 20276219 DOI: 10.1016/0096-6347(46)90040-3]
- Ingham GG. Dentinoma. Oral Surg Oral Med Oral Pathol 1952; 5: 353-358 [PMID: 14920026 DOI: 47 10.1016/0030-4220(52)90290-9
- 48 Sirsat MV. Odontogenic tumors. Indian J Med Res 1952; 40: 555-567 [PMID: 13061064]
- 49 Husted E, Pindborg JJ. Odontogenic tumours; clinical and roentgenological aspects, treatment and pathology. Odontol Tidskr 1953; 61: 275-292 [PMID: 13145176]
- 50 Hitchin A, White J. A dentinoma related to the deciduous dentition. Br Dent J 1955; 98: 163-165
- 51 Pindborg JJ. On dentinomas; with report of a case. Acta Pathol Microbiol Scand Suppl 1955; 105: 135-144 [PMID: 14398285]
- Gorlin RJ, Chaudhry AP, Pindborg JJ. Odontogenic tumors. Classification, histopathology, and 52 clinical behavior in man and domesticated animals. Cancer 1961; 14: 73-101 [PMID: 13707265 DOI: 10.1002/1097-0142(196101/02)14:1<73::aid-cncr2820140111>3.0.co;2-t]
- 53 Azaz B, Ulmansky M, Lewin-Epstein J. Dentinoma. Report of a case. Oral Surg Oral Med Oral Pathol 1967; 24: 659-663 [PMID: 5234284 DOI: 10.1016/0030-4220(67)90212-5]
- Manning GL, Browne RM. Dentinoma. Br Dent J 1970; 128: 178-181 [PMID: 5270266 DOI: 54 10.1038/sj.bdj.4802444]
- 55 Hoggins GS, Browne RM. Dentinoma: a case report. Br J Oral Surg 1976; 14: 179-184 [PMID: 1070346 DOI: 10.1016/0007-117x(76)90038-x]
- Gulmen S, Adams RJ, Boggiano JJ. "Dentinoma" of the mandible. J Oral Surg 1976; 34: 921-926 56 [PMID: 1067390]
- 57 Godjesk JE, Dolinsky HB, Schneider LC, Doyle JL. Ameloblastic fibro-dentinoma in the gingiva: report of a case. J Oral Med 1980; 35: 59-61 [PMID: 6931877]
- Rennie JS, Critchlow HA. Dentinoma of the maxilla. Br J Oral Surg 1981; 19: 138-141 [PMID: 58 6942880 DOI: 10.1016/0007-117x(81)90040-8]
- 59 van Wyk CW, van der Vyver PC. Ameloblastic fibroma with dentinoid formation/immature dentinoma. A microscopic and ultrastructural study of the epithelial-connective tissue interface. J Oral Pathol 1983; 12: 37-46 [PMID: 6403684 DOI: 10.1111/j.1600-0714.1983.tb00314.x]
- 60 Villafañe OC, Fonseca MM, Gendelman H, Grotti MA. Dentinoma of the maxilla: report of a case. Acta Stomatol Belg 1986; 83: 203-209 [PMID: 3471068]
- 61 Lukinmaa PL, Hietanen J, Laitinen JM, Malmström M. Mandibular dentinoma. J Oral Maxillofac Surg 1987; 45: 60-64 [PMID: 3467039 DOI: 10.1016/0278-2391(87)90088-7]
- Anker AH, Radden BG. Dentinoma of the mandible. Oral Surg Oral Med Oral Pathol 1989; 67: 731-62 733 [PMID: 2740094 DOI: 10.1016/0030-4220(89)90016-9]
- Ulmansky M, Bodner L, Praetorius F, Lustmann J. Ameloblastic fibrodentinoma: report on two new 63 cases. J Oral Maxillofac Surg 1994; 52: 980-984 [PMID: 8064465 DOI: 10.1016/s0278-2391(10)80085-0]
- 64 Cassidy JP, Crocker DJ, Grau WH. Ameloblastic fibrodentinoma. J Oral Maxillofac Surg 1987; 45: 734-736 [PMID: 3475448 DOI: 10.1016/0278-2391(87)90323-5]



- Akal UK, Günhan O, Güler M. Ameloblastic fibrodentinoma. Report of two cases. Int J Oral 65 Maxillofac Surg 1997; 26: 455-457 [PMID: 9418150 DOI: 10.1016/s0901-5027(97)80013-6]
- 66 Takeda Y, Sato H, Satoh M, Nakamura S, Yamamoto H, Pigmented ameloblastic fibrodentinoma: a novel melanin-pigmented intraosseous odontogenic lesion. Virchows Arch 2000; 437: 454-458 [PMID: 11097374 DOI: 10.1007/s004280000249]
- Karasu HA, Akman H, Uyanik LO, Sayan NB. Ameloblastic fibrodentinoma. A case report. NY 67 State Dent J 2004; 70: 22-23 [PMID: 15683218]
- 68 Bhargava D, Dave A, Sharma B, Nanda KD. Ameloblastic fibrodentinoma. Indian J Dent Res 2011; 22: 345-347 [PMID: 21891911 DOI: 10.4103/0970-9290.84287]
- Sankireddy S, Kaushik A, Krishna BA, Reddy AL, Vinod VC, Sridevi V. Ameloblastic 69 fibrodentinoma involving anterior maxilla: a rare case report. J Indian Soc Pedod Prev Dent 2013; 31: 275-278 [PMID: 24262404 DOI: 10.4103/0970-4388.121832]
- 70 Salehinejad J, Langaroodi AJ, Shahakbari R, Yazdani N. Ameloblastic fibrodentinoma: report of a rare case. J Contemp Dent Pract 2013; 14: 548-551 [PMID: 24172005 DOI: 10.5005/jp-journals-10024-1360]
- 71 Ikeda H, Minamizato T, Fujita S, Asahina I. Ameloblastic fibrodentinoma with a congenitally missing second premolar tooth: a case report. Oral Surg Oral Med Oral Pathol Oral Radiol 2014; 117: e88-e91 [PMID: 23830803 DOI: 10.1016/j.0000.2013.05.009]
- 72 Lee J, Song YG, Moon SY, Choi B, Kim BC, Yoon JH. Calcifying cystic odontogenic tumor associated with ameloblastic fibro-odontoma of the anterior mandible. J Craniofac Surg 2014; 25: e259-e260 [PMID: 24785751 DOI: 10.1097/SCS.000000000000563]
- Unsal H, Eren H, Inceoglu B, Oncul AM. Mature form of ameloblastic fibrodentinoma: a case report. 73 J Craniofac Surg 2014; 25: e299-e301 [PMID: 24777020 DOI: 10.1097/SCS.0000000000000728]
- 74 Joseph S, Priya L, Gopal D, Devachen M, Narayan A, Afnan M. Ameloblastic fibrodentinoma presenting as a false gingival enlargement in the maxillary anterior region. Case Rep Dent 2015; 2015: 812087 [PMID: 25709845 DOI: 10.1155/2015/812087]
- Costa FW, Silva PG, Soares EC, Sousa FB, Alves AP, Scarparo HC. Surgical approach in a large 75 ameloblastic fibrodentinoma. J Craniofac Surg 2015; 26: 950-951 [PMID: 25978781 DOI: 10.1097/SCS.00000000001398]
- 76 Bhargava M, Sood S, Rathore P. Ameloblastic Fibrodentinoma: Report of a Case in an Infant. J Clin Diagn Res 2016; 10: ZD06-ZD07 [PMID: 26894185 DOI: 10.7860/JCDR/2016/16518.7070]
- 77 Bavle RM, Muniswammappa S, Venugopal R, R AS. Ameloblastic Fibrodentinoma: A Case with Varied Patterns of Dysplastic Dentin. Cureus 2017; 9: e1349 [PMID: 28721317 DOI: 10.7759/cureus.1349]
- Sabu AM, Gandhi S, Singh I, Solanki M, Sakharia AR. Ameloblastic Fibrodentinoma: A Rarity in 78 Odontogenic Tumors. J Maxillofac Oral Surg 2018; 17: 444-448 [PMID: 30344385 DOI: 10.1007/s12663-017-1062-31
- Villa VG. Ameioblastic sarcoma in the mandible; report of a case. Oral Surg Oral Med Oral Pathol 79 1955; 8: 123-129 [PMID: 13236294 DOI: 10.1016/0030-4220(55)90180-8]
- 80 Forman G, Garrett J. Ameloblastic sarcoma: report of case. J Oral Surg 1972; 30: 50-54 [PMID: 4500330
- Altini M, Smith I. Ameloblastic dentinosarcoma- a case report. Int J Oral Surg 1976; 5: 142-147 81 [PMID: 820663 DOI: 10.1016/s0300-9785(76)80063-4]
- 82 Corominas-Villafañe O, Cuestas-Carnero R, Corominas O Jr, Gendelman H. Ameloblastic odontosarcoma: report of a case. Acta Stomatol Belg 1993; 90: 149-156 [PMID: 8122585]
- Herzog U, Putzke HP, Bienengräber V, Radke C. [The ameloblastic fibro-odontoma--an odontogenic 83 mixed tumor progressing into an odontogenic sarcoma]. Dtsch Z Mund Kiefer Gesichtschir 1991; 15: 90-93 [PMID: 1816940]
- Muller S, Parker DC, Kapadia SB, Budnick SD, Barnes EL. Ameloblastic fibrosarcoma of the jaws. 84 A clinicopathologic and DNA analysis of five cases and review of the literature with discussion of its relationship to ameloblastic fibroma. Oral Surg Oral Med Oral Pathol Oral Radiol Endod 1995; 79: 469-477 [PMID: 7614208 DOI: 10.1016/s1079-2104(05)80130-1]
- 85 Reiser V, Alterman M, Shuster A, Kaplan I. Pediatric ameloblastic fibro-odontosarcoma of the mandible: a challenge of diagnosis and treatment. J Oral Maxillofac Surg 2013; 71: e45-e57 [PMID: 23245775 DOI: 10.1016/j.joms.2012.08.027]
- Khan M, Ramachandra VK, Kampasi N. Clinical, radiographic, and histologic presentation of 86 fibrodentinosarcoma. Quintessence Int 2014; 45: 169-172 [PMID: 24389571 DOI: 10.3290/j.qi.a30991]
- Gatz SA, Thway K, Mandeville H, Kerawala C, MacVicar D, Chisholm J. Chemotherapy responsiveness in a patient with multiply relapsed ameloblastic fibro-odontosarcoma of the maxilla. Pediatr Blood Cancer 2015; 62: 2029-2032 [PMID: 26178860 DOI: 10.1002/pbc.25627]
- 88 Chen SJ, Zheng XW, Lin X, Liu H. Ameloblastic fibro-odontosarcoma of the mandible in a pediatric patient. Eur Ann Otorhinolaryngol Head Neck Dis 2016; 133: 419-421 [PMID: 27130809 DOI: 10.1016/j.anorl.2015.11.010
- 89 Niu H, Liu J, Chen Y, Geng N. Ameloblastic fibro-odontosarcoma of the mandible with active epithelial proliferation: A rare case report. Mol Clin Oncol 2017; 7: 971-975 [PMID: 29285358 DOI: 10.3892/mco.2017.1448]
- Atarbashi-Moghadam S, Lotfi A, Mokhtari S. A mixed odontogenic sarcoma: A challenging 90 histopathologic case and brief review of the literature. J Oral Maxillofac Pathol 2018; 22: S29-S34



[PMID: 29491601 DOI: 10.4103/jomfp.JOMFP_74_17]

- Tanaka T, Ohkubo T, Fujitsuka H, Tatematsu N, Oka N, Kojima T, Morishita Y, Yoshimi N, Mori 91 H. Malignant mixed tumor (malignant ameloblastoma and fibrosarcoma) of the maxilla. Arch Pathol Lab Med 1991; 115: 84-87 [PMID: 1987921]
- Slama A, Yacoubi T, Khochtali H, Bakir A. [Mandibular odontogenic carcinosarcoma: a case report]. 92 Rev Stomatol Chir Maxillofac 2002; 103: 124-127 [PMID: 11997741]
- 93 DeLair D, Bejarano PA, Peleg M, El-Mofty SK. Ameloblastic carcinosarcoma of the mandible arising in ameloblastic fibroma: a case report and review of the literature. Oral Surg Oral Med Oral Pathol Oral Radiol Endod 2007; 103: 516-520 [PMID: 17395065 DOI: 10.1016/j.tripleo.2006.02.025]
- 94 Chikosi R, Segall N, Augusto P, Freedman P. Odontogenic carcinosarcoma: case report and literature review. J Oral Maxillofac Surg 2011; 69: 1501-1507 [PMID: 21195529 DOI: 10.1016/j.joms.2010.05.071]
- Dos Santos JN, Servato JPS, Cardoso SV, de Faria PR, Pires BC, Loyola AM. Odontogenic 95 carcinosarcoma: morphologic and immunohistochemical description of a case. Oral Surg Oral Med Oral Pathol Oral Radiol 2018; 126: e264-e270 [PMID: 30554629 DOI: 10.1016/j.0000.2018.05.013]
- 96 Soares CD, Delgado-Azañero W, Morais TML, de Almeida OP, Ghersi Miranda H. Odontogenic Carcinosarcoma: Clinicopathologic Features of 2 Cases. Int J Surg Pathol 2020; 28: 421-426 [PMID: 31786969 DOI: 10.1177/1066896919888578]



World Journal of Clinical Oncology

Submit a Manuscript: https://www.f6publishing.com

World J Clin Oncol 2021 December 24; 12(12): 1244-1263

DOI: 10.5306/wjco.v12.i12.1244

ISSN 2218-4333 (online)

SYSTEMATIC REVIEWS

Carcinosarcoma of gallbladder: A world review

Thomas Zheng Jie Teng, Branden Qi Yu Chua, Vishal G Shelat

ORCID number: Thomas Zheng Jie Teng 0000-0001-7355-9591; Branden Qi Yu Chua 0000-0003-4096-531X; Vishal G Shelat 0000-0003-3988-8142.

Author contributions: Teng TZJ, Chua BQY and Shelat VG contributed to the conception of the idea and writing of the paper.

Conflict-of-interest statement:

There is no conflict of interest to declare

PRISMA 2009 Checklist statement:

The authors have read the PRISMA 2009 Checklist, and the manuscript was prepared and revised according to the PRISMA 2009 Checklist.

Country/Territory of origin: Singapore

Specialty type: Surgery

Provenance and peer review:

Invited article; Externally peer reviewed.

Peer-review model: Single blind

Peer-review report's scientific quality classification

Grade A (Excellent): 0 Grade B (Very good): 0 Grade C (Good): C Grade D (Fair): 0 Grade E (Poor): 0

Open-Access: This article is an open-access article that was

Thomas Zheng Jie Teng, Branden Qi Yu Chua, Vishal G Shelat, Department of General Surgery, Tan Tock Seng Hospital, Singapore 308433, Singapore

Thomas Zheng Jie Teng, Undergraduate Medicine, Lee Kong Chian School of Medicine, Singapore 308232, Singapore

Corresponding author: Vishal G Shelat, FRCS (Gen Surg), Adjunct Associate Professor, Department of General Surgery, Tan Tock Seng Hospital, 11 Jln Tan Tock Seng, Singapore 308433, Singapore. vgshelat@rediffmail.com

Abstract

BACKGROUND

Gallbladder carcinosarcoma is a rare hepatobiliary tumor comprising of both carcinomatous and sarcomatous components. Due to its rarity, the literature with regards to the topic is scarce and currently lacking, spanning less than 100 cases.

AIM

To summarize the current literature on gallbladder carcinosarcoma.

METHODS

A literature review was performed on the PubMed database using the keywords "Gallbladder" AND "Carcinosarcoma" from 1970 to 2021. Additionally, similar searches were performed on MEDLINE and Web of Science.

RESULTS

Risk factors noted include female gender, gallstones and chronic cholecystitis. In the absence of any diagnostic biochemical testing or tumor markers, imaging modality serves as the key initial impression tool, which can be histologically confirmed only post-resection. While surgery is the only curative option, the use of adjunctive chemotherapy has been considered on top of excision in recent years, with some success.

CONCLUSION

While this study has taken steps to bridge the gap in the literature, more cases should be reported to further ascertain the current associations and management potential for gallbladder carcinosarcoma.

Key Words: Carcinosarcoma; Gallbladder; Gallstone; Malignancy; Carcinoma; Sarcoma

©The Author(s) 2021. Published by Baishideng Publishing Group Inc. All rights reserved.



selected by an in-house editor and fully peer-reviewed by external reviewers. It is distributed in accordance with the Creative Commons Attribution NonCommercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: htt p://creativecommons.org/License s/by-nc/4.0/

Received: March 14, 2021 Peer-review started: March 14, 2021 First decision: May 4, 2021 Revised: May 14, 2021 Accepted: November 24, 2021 Article in press: November 24, 2021 Published online: December 24, 2021

P-Reviewer: Ruess DA S-Editor: Gong ZM L-Editor: Filipodia P-Editor: Gong ZM



Core Tip: Gallbladder carcinosarcoma (GBCS) while rare, is an important histological subtype of gallbladder malignancy as it is associated with poor prognosis. Most GBCS patients tend to present late. As of now, the primary method of diagnosis is that of a pathological analysis with the main stay of treatment being surgical excision. Furthermore, the clinical diagnosis of GBCS remains extremely challenging given its seemingly nonspecific clinical features. We aim to provide an in-depth world review of the known cases of GBCS in order to identify unifying features of the disease and to assess effective management strategies that have been employed by clinicians.

Citation: Teng TZJ, Chua BQY, Shelat VG. Carcinosarcoma of gallbladder: A world review. World J Clin Oncol 2021; 12(12): 1244-1263

URL: https://www.wjgnet.com/2218-4333/full/v12/i12/1244.htm DOI: https://dx.doi.org/10.5306/wjco.v12.i12.1244

INTRODUCTION

Gallbladder carcinosarcoma (GBCS) is defined by the presence of both carcinomatous and sarcomatous components in the tumor, making it a rarity even amongst the uncommon gallbladder cancer family[1]. While its history is deep-rooted, with the first case being reported by Karl[2] in 1907, less than 100 cases have been reported since. In 2008, Zhang et al[3] sought to collectively analyze the 70 cases in the literature at that time. However, Zhang *et al*[3] noted the need for a larger scale case series to provide more information on the neoplasm for better accuracy and reliability. Since then, there has been a gap in the literature for such an analysis (Figure 1). This study aims to fill this gap by providing a comprehensive overview of GBCS.

MATERIALS AND METHODS

A literature review was performed on the PubMed database using the keywords "Gallbladder" AND "Carcinosarcoma" from 1970 to 2021. Additionally, similar searches were performed on Medline and Web of Science. The last search was performed on January 31, 2021. After removing duplicate results from similar databases, the search yielded 105 articles: 16 non-English and non-Japanese studies and 12 unrelated topics (animal studies, gallbladder carcinoma and non-gallbladder pathology) were excluded. Out of the remaining 77 articles, seven were not case reports or case series on GBCS and thus excluded. The remaining 70 articles including 76 patients were included in the final analysis (Table 1)[1,4-72]. Article filtering and exclusion was done according to PRISMA guidelines (Figure 2). Data extracted included study year, age and gender of the patient, clinical presentation, risk factors, laboratory investigations, tumor markers, the ultrasound imaging findings, location of the lesion within the gallbladder, size of the lesion, initial diagnosis, method of confirming the diagnosis, immunohistochemical results (vimentin, cytokeratin, Ki-67), management and prognosis of the patient. Kaplan-Meier survival curves were compared between lesions larger than 5 cm and those smaller than 5 cm as data by Zhang et al[3] suggested that tumors smaller than 5 cm had better survival. For all statistical tests, a P value of 0.05 was used to determine statistical significance.

RESULTS

Seventy-eight patients with a mean age of 66.4 years (range: 40-91 years) were reported during the study period. The patients were predominantly female (n = 55, 72.4%) with a gender ratio of 2.62. Nine patients (11.8%) had chronic cholecystitis, and 1 patient each had hepatitis C and abnormal pancreaticobiliary maljunction (APBJ). Of those who reported the presence of gallstones, a majority noted the presence of gallstones (n = 35/42, 83.3%). The majority of patients complained of abdominal pain (n = 58, 76.3%), most of which was localized to the right upper quadrant. Twenty-two patients (28.9%) presented with constitutional symptoms (either unexplained loss of



Teng TZJ et al. Carcinosarcoma of gallbladder

Table 1 Summary of 76 reported cases of gallbladder carcinosarcoma from 1970 to 2021 Tumor markers Liver Risk factors for GB Clinical Position Size **Confirmatory diagnosis** Stage Survival No. Year Ref. Age/sex function (CEA, Initial diagnosis CA (stones) (UICC) (mo) presentation of CA (mm) (mode) AFP, CA tests 19-9) 2020 Khurram et al 64/F AST, GGT 132 × NA NIL 1 No stones RUQ pain, Fundus Normal Hepatic abscess Cholecystectomy 97 × [4] intermittent elevated mentioned fever, 110 abdominal distension 2 2020 Ayoub *et al*[5] 66/M NA RUQ pain Normal Body Normal 150 × Gallbladder Cholecystectomy and IVA 12+ 80 × malignancy lymphadenectomy 60 3 2020 Kaneko et al[6] 70/F NA Obstructive NA NA 110 × Gallbladder Cholecystectomy NA 44+ Normal 70 × malignancy jaundice 34 4 2020 Siddiqui et al[7] 57/M NA Abdominal ALP, total Fundus NA 620 Gallbladder ERCP (unsuccessful), PTC NA NA bilirubin malignancy with internal-external biliary pain, nausea, LOW, LOA drainage catheter elevated 5 2020 Mochizuki et al 88/F Gallstones Chills, NA Body NA 60 × Acute cholecystitis Cholecystectomy NA 10 +25 [8] tremors, vomiting radical cholecystectomy with NA 2019 Varshney et al 50/M Gallstones RUQ pain, AST, ALT, Fundus Normal 65 × Gallbladder 6+ 6 obstructive bilirubin 55 malignancy standard lymphadenectomy [9]

					jaundice	elevated									
7	2019	Aldossary <i>et al</i> [10]	40/M	Gallstones	RUQ pain	Normal	Entire gallbladder	Normal	115 × 92 × 50	Gallbladder malignancy	Open lap, radical cholecystectomy, extended R hemi w IC anastomosis, liver resection	IVB	6	Vimentin	Adjuvant chemotherapy
8	2019	Aldossary <i>et al</i> [10]	52/F	No stones	RUQ pain	ALT, AST elevated	Fundus	CA19-9 level of 154.3 IU/mL, with normal levels of AFP and CEA	136 × 120 × 95	Gallbladder Malignancy	Open lap, radical CCY, transverse chole, Roux en Y + distal gastrectomy	IVB	3	Vimentin, CK	NA
9	2019	Aldossary <i>et al</i> [10]	62/F	Gallstones	RUQ pain, nausea, anorexia	Normal	Body	Normal	27 × 9	Gallbladder malignancy	Lap CCY	Π	86+	Vimentin, CK	Adjuvant chemotherapy

IHC

CK

CK

Ki-67

CK

Further

positives management

NA

Vimentin, NA

CK, Ki-67 NA

Vimentin NA

NA

chemotherapy

Vimentin, Adjuvant

10	2019	Alratroot <i>et al</i> [11]	52/F	Xanthogranulomatous cholecystitis	RUQ pain	GGT elevated	Fundus	CA 19-9 154.33 IU/mL	110 × 60	Gallbladder malignancy	Laparotomy with radical cholecystectomy, transverse colectomy, distal gastrectomy, omentectomy and liver bed resection	III	1.5+	Vimentin, CK	Adjuvant chemotherapy
11	2019	Matsubayashi et al <mark>[12]</mark>	72/F	Pancreaticobiliary maljunction	RUQ pain	ALP, GGT elevated	Entire gallbladder	Normal	90 × 85	Gallbladder malignancy	Laparotomy and extended cholecystectomy	IIIA	73+	Vimentin, CK	NA
12	2018	Doniparthi <i>et al</i> [<mark>13</mark>]	49/M	NA	Epigastric pain	AST, ALT, lipase elevated	NA	Normal	32	Acute cholecystitis	Lap cholecystectomy, followed up by robotic liver resection and lymphadenectomy	NA	NA	NA	NA
13	2018	Koustav <i>et al</i> [<mark>14</mark>]	40/F	NA	RUQ pain		NA	CA19-9 elevated	43 × 51	Gallbladder malignancy	Staging laparoscopy + extended cholecystectomy	NA	NA	NA	NA
14	2018	Trautman <i>et al</i> [<mark>15</mark>]	73/F	Chronic cholecystitis	Abdominal distension, constipation, vomiting, LOW	AST, ALT, ALP elevated	NA	Beta- HCG elevated		Gallbladder malignancy	Diagnostic laparoscopy	NA	0.5	Vimentin	Palliative (NM)
15	2017	Furuya <i>et al</i> [<mark>16</mark>]	61/F	NA	RUQ pain	Normal	NA	Normal	15 × 15	Chronic cholecystitis with stone	Cholecystectomy	NA	NA	NA	NA
16	2016	Hu et al[17]	68/F	Cholelithiasis	RUQ pain, fever	Normal	Body	CA19-9 elevated	16 × 15 × 13	Gallbladder malignancy	Cholecystectomy	NA	1	NA	NA
17	2016	Cruz et al[<mark>18</mark>]	52/F	Gallstones	RUQ pain	ALT AST elevated	Entire gallbladder	Normal	170 × 125	Gallbladder malignancy	Cholecystectomy	NA	1	Vimentin, CK	Palliative (NM)
18	2016	Dong et al[19]	61/M	NA	Abdominal distension	NA	NA	Normal	180	Gallbladder malignancy	Resection (not specified)	NA	NIL mentioned	Ki-67	NA
19	2016	Gupta <i>et al</i> [20]	46/F	NA	RUQ pain	NA	Fundus	All normal	350 × 250 × 200	Gallbladder malignancy	Radical cholecystectomy with hepato-duodenal ligament lymph node clearance and segment 4b/5 liver resection	NA	15 (still alive)	Vimentin, CK	Adjuvant chemotherapy
20	2016	Wong et al[21]	52/F	NA	Abdominal pain	NA	Entire gallbladder	CA19-9 elevated	75	NA	Autopsy	III	6	Vimentin, CK	Adjuvant chemotherapy
21	2016	Ansari <i>et al</i> [22]	50/F	NA	RUQ pain	Normal	Entire gallbladder	Normal	50 × 40	NA	Radical cholecystectomy	II	13 mo (still alive)	Vimentin, CK, Ki-67	Adjuvant chemotherapy
22	2015	Gao et al[23]	62/M	Chronic cholecystitis	RUQ pain	Normal	Entire gallbladder	Normal	50 × 40	Gallbladder malignancy	Simple cholecystectomy	II	0	Vimentin, CK	NA
23	2015	Tonouchi <i>et al</i> [<mark>24</mark>]	87/M	No stones	Abdominal pain	NA	NA	NA	60 × 55	Diffuse peritonitis	Cholecystectomy with partial transverse colectomy around the fistula	NA	Lost to follow-up	Vimentin, CK	NA

Teng TZJ et al. Carcinosarcoma of gallbladder

24	2015	Faujdar et al [<mark>25</mark>]	60/F	NA	RUQ pain, fever	Normal	Entire gallbladder		120 ×70 × 60	Gallbladder malignancy	Cholecystectomy	NA	60+	Vimentin, CK	NA
25	2014	Wada <i>et al</i> [<mark>26</mark>]	68/M	NA	Right flank pain	GGT elevated	NA	Normal	85 × 70	Gallbladder malignancy	Extended right hepatectomy with portal thrombectomy with hepatoduodenal ligament lymphadenectomy	NA	51+	Vimentin, CK, Ki-67	Adjuvant chemotherapy
26	2014	Kishino et al [27]	70s/F	NA	Referred for suspected GB cancer (presenting complaint not mentioned)	NA	Fundus	NA	68	Gallbladder malignancy	Cholecystectomy	NA	1.5+	Vimentin, CK	NA
27	2013	Wang et al[28]	68/F	Chronic cholecystitis, cholecystolithiasis	RUQ pain, jaundice, fever	ALT, ALP elevated	NA	CEA, CA19-9, AFP elevated	100 × 70 × 50	Gallbladder malignancy	Cholecystectomy with liver segmentectomy (S4a+S5) and a lymph node dissection, followed by resection of the extrahepatic bile duct and a Roux-en-Y type hepatic cholangiojejunostomy	NA	6+	Vimentin, CK	NA
28	2013	Khanna[<mark>29</mark>]	45/F	NA	RUQ pain	Normal	Body		60 × 40	Gallbladder malignancy	Laparotomy and simple cholecystectomy with wedge resection	NA	3	Vimentin, CK, Ki-67	NA
29	2013	Li et al <mark>[30]</mark>	64/M	Chronic cholecystitis	RUQ pain	NA	NA	CEA, CA19-9 elevated	40 × 30 × 30	NA	Cholecystectomy, R hemicolectomy, resection of multiple hepatic metastases	NA	3+	Vimentin, CK, Ki-67	NA
30	2012	Kim <i>et a</i> l[<mark>31</mark>]	72/F	Gallstones	Abdominal pain	Normal	Fundus	Normal	65 × 45 × 45	Gallbladder malignancy	Radical cholecystectomy with wedge resection of liver combined with hepatoduodenal ligament lymphadenectomy	NA	4	NA	Adjuvant chemotherapy
31	2012	Kim et al <mark>[31</mark>]	81/M	NA	Epigastric pain	Normal	Fundus	Normal		Gallbladder malignancy	Cholecystectomy with liver segmentectomy (S4a,5) and lymph node dissection	NA	13	Vimentin, CK	NA
32	2012	Sadamori <i>et al</i> [<mark>32</mark>]	80/M	NA	RUQ pain, fever		Entire gallbladder		76 × 27	Gallbladder malignancy	Cholecystectomy with liver segmentectomy (S4a and S5) and lymph node dissection	NA	2+		Adjuvant chemotherapy
33	2012	Kataria <i>et al</i> [<mark>33</mark>]	55/F	NA	RUQ pain	Normal	Fundus	Normal	70 × 50 × 30	NA	Cholecystectomy, wedge resection of liver with resection of transverse colon and paraduodenal lymph node	NA	6	Vimentin, CK	NA
34	2012	Parreira <i>et al</i> [34]	59/F	NA	RUQ pain	Normal	NA	NA	NA	NA	Conventional cholecystectomy	NA	2	NA	NA

35	2012	Park et al[35]	77/F	NA	RUQ pain	AST, ALT elevated	NA	CA19-9, CA-125 elevated	78 × 55 × 12	Gallbladder malignancy	Laparotomy, followed by cholecystectomy and lymph node dissection	IIIB	10+	NA	NA
36	2012	Ishida <i>et a</i> l[<mark>36</mark>]	62/F	NA	Incidental finding on radiograph for left calcaneal fracture	Normal	Entire gallbladder	Normal	52 × 38	Gallbladder malignancy	Open cholecystectomy	NA	8	NA	NA
37	2011	Lee et al[37]	77/F	No stones	RUQ pain	Not mentioned	Body	CA19-9, CA-125 elevated	80 × 70 × 30	Gallbladder malignancy	Cholecystectomy	NA	1.5+	Vimentin, CK	NA
38	2011	Pu et al[<mark>38</mark>]	59/F	Cholecystolithiasis	RUQ pain, fever	Normal	Body	CA19-9 elevated	120 × 25 × 60	Gallbladder malignancy	Exploratory laparotomy, followed by radical LN resection and hepatocholangojejunostomy Roux-En-Y	Π	0	СК	Adjuvant chemotherapy
39	2011	Krishnamurthy et al[<mark>39</mark>]	83/M	No stones	Abdominal pain	NA	NA	NA	NA	NA	Laparoscopic cholecystectomy	NA	48+	Vimentin, CK	NA
40	2009	Kohtani <i>et al</i> [40]	84/M	Chronic cholecystitis	RUQ pain	Serum glutamic oxaloacetic transaminase, GGT elevated	Neck	NA		Gallbladder malignancy	Open cholecystectomy	Π	3+	Vimentin, CK, Ki-67	Adjuvant chemotherapy
41	2009	Agarwal <i>et al</i> [<mark>41</mark>]	60/F	NA	RUQ pain, fever	Normal	Neck	NA	70 × 50 × 40	Gallbladder malignancy	Staging laparoscopy, laparotomy, simplex cholecystectomy	NA	24+	Vimentin, CK	NA
42	2009	Magata <i>et al</i> [<mark>42</mark>]	78/F	NA	RUQ pain	NA	Body	CEA elevated	115 x 40 x 35	Gallbladder malignancy	Whole-layer cholecystectomy with regional lymph node dissection	NA	6+	Vimentin, CK	NA
43	2009	Shimada et al [43]	69/M	Choledocholithiasis	Fever	Normal	Entire gallbladder	AFP elevated	90 × 50	Gallbladder malignancy	Laparotomy, cholecystectomy, lymph node dissection	NA	54+	Vimentin, CK, Ki-67	NA
44	2009	Uzun <i>et al</i> [44]	70/M	NA	RUQ pain	Normal	Fundus	Normal	100 × 60 × 30	Gallbladder malignancy	Radical cholecystectomy, wedge resection of liver- gallbladder bed with hepatoduodenal ligament lymphadenectomy	NA	8	CK, Ki-67	NA
45	2006	Kubota <i>et al</i> [<mark>45</mark>]	72/M	NA	RUQ pain, fever	AST, ALT, ALP elevated	NA	Normal	70 × 55 × 40	Gallbladder malignancy	En bloc resection of the gallbladder and segments 4a and 5 of the liver, partial colectomy, and lymph node dissection	NA	6	NA	NA
46	2005	Akatsu <i>et al</i> [46]	76/F	Gallstones	Incidental	Normal	NA	Normal		Gallbladder	Extended cholecystectomy,	NA	2	Vimentin,	NA

Teng TZJ et al. Carcinosarcoma of gallbladder

					finding on follow-up for cholelithiasis					malignancy	liver 4b and 5 resection			CK	
47	2005	Huguet <i>et al</i> [47]	64/F	Cholecystitis	RUQ pain, fever	Normal	Entire gallbladder	Normal	120 × 100 × 70	Gallbladder malignancy	A cholecystectomy with wedge resection of the gallbladder fossa (involving liver segments 4 and 5), extrahepatic bile duct excision, non-pylorus- preserving pancreaticoduodenectomy with excision of 15 cm of proximal jejunum, and right hemicolectomy	NA	60	Vimentin, CK	NA
48	2005	Sodergren <i>et al</i> [48]	64/F	NA	Malaise and LOA	ALP Elevated	NA	NA	20 × 12 × 12	NA	Extrahepatic radical bile duct resection with hepatic and coeliac lymph node clearance followed by right hepaticodochojejunostomy to a jejunal Roux loop	NA	5	Vimentin, CK	NA
49	2005	Sodergren <i>et al</i> [48]	60/F	NA	Painless jaundice	NA	NA	NA	90	Gallbladder malignancy	Cholecystectomy and extrahepatic bile duct resection with lymph node clearance	NA	2	Vimentin, CK	Palliative (NM)
50	2004	Takahashi <i>et al</i> [49]	84/F	NA	RUQ pain	NA	Body	CEA, CA19-9 elevated	84 × 40 × 30	Gallbladder malignancy	Cholecystectomy and transverse colon partial colectomy	NA	2	Vimentin, CK	
51	2003	Kim <i>et al</i> [50]	61/F	No stones	RUQ pain	Normal	Neck	Normal	45 × 40 × 40	Gallbladder malignancy	Cholecystectomy with common bile duct resection	NA	2	Vimentin	Palliative (NM)
52	2002	Al-Sheneber <i>et</i> al[<mark>1</mark>]	68/F	Acute cholecystitis, gallstones	RUQ pain	Normal	NA	CEA elevated	148 × 80	Gallbladder malignancy	CT guided needle biopsy of the upper abdominal mass	NA	7	Vimentin, CK	NA
53	2002	Hotta <i>et al</i> [<mark>51</mark>]	53/M	Chronic cholecystitis, gallstones	RUQ Pain	Normal	NA	Normal	1100	Gallbladder malignancy	Cholecystectomy with resection of subsegmentectomy of liver S5 and a resection of transverse colon at the second operation	Π	2	NA	Adjuvant chemotherapy
54	2002	Ajiki et al <mark>[52</mark>]	69/F	Gallstones, left renal tumor	Epigastric pain	Normal	NA	CA19-9 elevated	NA	Double cancers of the left kidney and gallbladder	Left renal excision, cholecystectomy with liver segmentectomy (S4a, S5), and lymph node dissection	NA	NA	NA	Adjuvant chemotherapy
55	2000	Yavuz et al[53]	50/F	NA	RUQ pain	NA	Body	NA	80 × 60 × 60		Exploratory laparotomy -> cholecystectomy, liver wedge biopsy	NA	NA	NA	NA

56	1999	Eriguchi <i>et al</i> [<mark>54</mark>]	65/F	Gallstones	RUQ pain	Normal	Entire gallbladder	NA		Gallbladder malignancy	Cholecystectomy	Ι	16+	NA	NA
57	1997	Ryś et al[55]	67/F	Gallstones	Abdominal pain, LOW	NA	Fundus	NA	10 × 15	Gallbladder malignancy	Hemicolectomy and cholecystectomy	NA	2	Vimentin	NA
58	1996	Nakagawa et al [<mark>56</mark>]	60/F	NA	Abdominal pain, fever	Normal	Body	NA	70 × 40	Gallbladder malignancy	Mass reduction surgery	NA	NA	NA	NA
59	1994	Fagot et al[57]	83/F	Gallstones	Vomiting, fever, right RHC pain	Total bilirubin elevated	Fundus	NA	45	NA	Surgery (not defined)	NA	12+	NA	NA
60	1992	Nakazawa et al [<mark>58</mark>]	63/F	NA	Nausea	Normal	Body	Normal	30 × 30	Gallbladder malignancy	Pancreaticoduodenectomy	NA	NA	NA	NA
61	1990	Ishihara <i>et al</i> [<mark>59</mark>]	58/F	NA	Abdominal pain	NA	Fundus	NA	50 × 80	Gallbladder malignancy	Cholecystectomy	NA	11+	Vimentin	NA
62	1988	Lumsden <i>et al</i> [<mark>60</mark>]	81/F	Gallstones	RUQ pain, LOW, LOA	Total bilirubin, ALP, GOT elevated	NA	NA	50 × 20 × 20	Biliary neoplasm	Cholecystectomy	NA	12+	NA	NA
63	1987	Hasegawa et al [<mark>61</mark>]	61/M	NA	RUQ pain	Normal	Entire gallbladder	NA	150	Gallbladder malignancy	Resection (not specified)	NA	6	NA	NA
64	1987	Herrera- Goepfert <i>et al</i> [<mark>62</mark>]	60/F	Gallstones	Abdominal pain, jaundice, LOW		Entire gallbladder	NA	70 × 40	Gallbladder malignancy	Autopsy	NA	NA	NA	NA
65	1986	Inoshita <i>et al</i> [<mark>63</mark>]	53/M	Gallstones	RUQ pain, jaundice	Total bilirubin, ALP, GOT, GPT elevated	Neck	NA	NA	Choledocholithiasis	Open laparotomy	NA	17	NA	NA
66	1985	Lopez et al[<mark>64</mark>]	78/F	No stones	Anorexia, LOW	Normal	NA	NA	NA	Gallbladder empyema	Open laparotomy	NA	NA	NA	NA
67	1984	Born <i>et al</i> [65]	90/F	Gallstones	Anorexia, nausea, vomiting	Amylase elevated	NA	NA	150 × 150 × 10	NA	Exploratory laparotomy	NA	3	NA	NA
68	1982	von Kuster <i>et al</i> [<mark>66</mark>]	91/F	Gallstones	RUQ pain, fever	GOT, ALP elevated	NA	NA	20	Gallbladder empyema	Exploratory laparotomy	NA	0	NA	NA
69			77/F	NA	Bleeding in GI tract (lower)	Normal	NA	NA	30	NA	Exploratory laparotomy	III	31+	NA	NA
70	1982	Aldovini <i>et al</i> [67]	75/F	Gallstones	Abdominal pain	ALP, SGT elevated	NA	NA	90	NA	Cholecystectomy	NA	8+	NA	NA
71	1982	Yamagiwa <i>et al</i> [68]	78/F	NA	RUQ pain	NA	NA	NA	NA	NA	NA	NA		NA	NA

72	1980	Mansori <i>et al</i> [<mark>69</mark>]	81/M	Gallstones	Abdominal pain	GOT, ALP elevated	NA	NA	NA	NA	Exploratory laparotomy	NA	0.5	NA	NA
73	1973	Higgs et al[70]	77/M	Gallstones	Jaundice	ALP, GOT elevated	NA	NA	NA	NA	Cholecystectomy and CBDE	NA	1	NA	NA
74	1971	Mehrotra <i>et al</i> [<mark>71</mark>]	45/F	Gallstones	RUQ pain	Normal	Neck	NA	50 × 40 × 30	NA	Open laparotomy	NA	4	NA	NA
75	1970	Appelman <i>et al</i> [72]	91/M	Gallstones, chronic cholecystitis	Obstructive jaundice	AST, ALT, ALP elevated	Fundus	NA	NA	Pancreatic cancer	Autopsy	NA	0.5	NA	NA
76	1970	Appelman <i>et al</i> [72]	75/F	Gallstones, chronic cholecystitis	RUQ pain	ALP elevated	Fundus	NA	50 × 50 × 20	NA	Cholecystectomy	NA	1	NA	NA

GBCA: Gallbladder cancer; CEA: Carcinoembryonic antigen; AFP: Alpha-fetoprotein; CA 19-9: Carbohydrate antigen 19-9; UICC: Union for International Cancer Control; IHC: Immunohistochemistry; RUQ: Right upper quadrant; AST: Aspartate aminotransferase; ALT: Alanine aminotransferase; ALP: Alkaline phosphatase; GGT: Gamma-glutamyl transferase; ERCP: Endoscopic retrograde cholangiopancreatography; PTC: Percutaneous transhepatic cholangiogram; CCY: Cholecystectomy; NA: Not available; NM: Not mentioned; NIL: None; CR: Complete response; LOA: Loss of appetite; LOW: Loss of weight; R hemi w C: Right hemihepatectomy with cholecystectomy; M: Male; F: Female; Laparoscopic; Chole: Cholecystectomy; LN: Lymph node; CT: Computerized tomography; RH: Right hepatectomy; GI: Gastrointestinal; CBDE: Common bile duct exploration; HCG: Human chorionic gonadotropin; SGOT: Serum glutamic-oxaloacetic transaminase; SGPT: Serum glutamic-pyruvic transaminase.

weight, anorexia or lethargy). Nineteen patients (25.0%) had nausea and vomiting, and 13 patients (17.1%) were febrile. Two patients (2.6%) were asymptomatic when diagnosed.

Liver function test was the common serum biochemical test reported (n = 57). Deranged liver function tests were reported in 25 (43.9%) patients. Tumor markers were variably reported. The following tumor markers were elevated: carbohydrate antigen 19-9 (CA19-9) (n = 9/27, 33.3%), carcinoembryonic antigen (n = 5/27, 18.5%) and alpha-fetoprotein (n = 2/12, 16.6%) in some patients. Also, CA-125 was elevated in 2 patients.

Forty-three patients had the location of the gallbladder tumor reported. Fundus was the most common location (n = 15, 34.9%), followed by body (n = 10, 23.3%) and neck (n = 5, 11.6%). In 14 patients (32.5%), the tumor filled the entire gallbladder lumen, and thus exact position could not be determined. Fifty-nine patients had initial diagnosis reported. Out of these 59 patients, gallbladder malignancy was the primary diagnosis in the majority of patients (n = 49, 83.1%). Ten patients (16.9%) were initially diagnosed with other pathologies: cholelithiasis (n = 1), cholecystitis (n = 3), gallbladder empyema (n = 2), diffuse peritonitis (n = 1), pancreatic cancer (n = 1), biliary neoplasm (n = 1) and pyogenic liver abscess (n = 1).

Confirmation of diagnosis was reported in all but 1 patient (n = 75). It was mostly done *via* surgical resection, either diagnostic cholecystectomy or laparotomy (n = 70, 93.3%). In the remaining 5 patients, diagnosis was made by fluid analysis from percutaneous cholecystostomy (n = 1, 1.3%), computerized tomography (CT) scan guided needle biopsy (n = 1, 1.3%) and autopsy (n = 3, 4.0%). Staging of the cancer was



Figure 1 Paucity of gallbladder carcinosarcoma reports and trends by decade.





reported infrequently, with TNM system being the most common (n = 15, 19.7%). The majority of patients had stage II (n = 6, 40.0%) and stage III disease (n = 5, 33.3%). Three patients had stage IV disease (20.0%), and 1 patient had stage I disease (6.67%). Immunohistochemical stains (vimentin for mesenchymal components and cytokeratin for epithelial components) were reported in 50 patients (68.5%). Vimentin (n = 42, 84.0%), cytokeratin (n = 39, 78.0%) and Ki-67 staining (n = 7, 14.0%) were variably positive.

Fourteen patients (18.4%) received adjuvant chemotherapy. Various chemotherapy combinations included: gemcitabine and cisplatin, leucovorin and 5-fluorouracil (5-FU), cisplatin and doxorubicin, cisplatin and 5-FU, tegafur-uracil and gemcitabine and oxaliplatin and 5-FU. Palliative treatment was chosen in 4 patients (5.26%). Amongst all those reported, 32 patients contained both survival and tumor size data. Kaplan-Meier survival analysis was performed (Figure 3), and there was no significant difference in survival times (P = 0.301) for patients with tumors less than 5 cm in diameter compared to those with larger tumors.

WJCO | https://www.wjgnet.com



Figure 3 Kaplan-Meier survival curve measuring the difference in survival between patients with gallbladder carcinosarcoma of less than 5 cm diameter and more than 5 cm diameter (P = 0.301).

DISCUSSION

Gallbladder cancer is a rare neoplasm, accounting for about 0.5% of all gastrointestinal malignancies^[73]. Most common gallbladder cancer is adenocarcinoma. GBCS is a rare form of gallbladder cancer, with only 78 cases reported. GBCS is characterized by carcinomatous and sarcomatous components and is made up of both epithelial and mesenchymal components. Commonly, the epithelial component consists of adenocarcinoma followed by the less common squamous cell carcinoma^[74]. While there are multiple theories to justify the mixture of the epithelial and mesenchymal components, there is no consensus on the pathophysiology of the neoplasm. GBCS is considered the most aggressive biliary tract malignancy, usually discovered at late stages, and has poor prognosis[3].

Incidence

In terms of patient demographics, our results are consistent with the report of Zhang et al[3]. In a report including 68 GBCS patients, those authors reported a median age of 68 years (range: 45 to 91 years) with female predominance (female:male = 2.7:1), consistent with our results with a gender ratio of 2.32 and a mean age of 66.0 years (range: 40-91 years). Female preponderance is likely due to increased prevalence of gallstones in females. Zhang et al[3] noted gallstones in 66.7% of their patients. In our study, the incidence of gallstones was high (83%). However, gallstone presence was not specific nor sensitive in the diagnosis of GBCS, as not only are they a common finding in cancers of the gallbladder, only 1%-5% of patients with gallstones develop gallbladder malignancies. In our analysis of the literature, gallstone presence was only noted in 83.3% of patients where the presence of gallstones was assessed.

APBJ is also another risk factor of gallbladder malignancy[12]. Matsubayashi et al [12] reported a 72-year-old female patient with symptoms of abdominal pain. Laboratory investigations revealed raised alkaline phosphatase and gamma-glutamyl transpeptidase. CT scan confirmed a polypoid gallbladder mass. Magnetic resonance cholangiopancreatography scan showed ABPJ, and this was confirmed at subsequent endoscopic retrograde cholangiopancreatography. While APBJ is a well-known risk factor for gallbladder cancers^[75], this was the first case of APBJ in GBCS noted in the literature.

Other risk factors mentioned include chronic cholecystitis, which could be both a risk factor and the manifestation of gallbladder malignancy. Unique to the gallbladder is a cycle of gallbladder epithelium damage and repair, enabling a chronic inflammatory environment from chronic cholecystitis^[76]. This cycle of inflammation, injury, repair and regeneration increases cell turnover and oxidative stress. Yildiz et al [77] stated biliary tract to be the "consummate example of inflammation-associated carcinoma". Chronic inflammation from gallstone disease can lead to protein damage,



WJCO | https://www.wjgnet.com

genetic mutations, inhibition of apoptosis, promotion of angiogenesis, modulation of cell adhesion and motility as well as immunosuppression. Chronic cholecystitis leads to gallbladder wall thickening, and CT or magnetic resonance imaging (MRI) scans are sensitive to detect wall thickness. However, it is not possible to distinguish if thickening of the gallbladder wall is due to inflammation or malignancy [78]. Thus, multidisciplinary discussion involving experienced radiologists and hepatobiliary surgical team is essential to make management plans for patients with suspicious gallbladder lesions.

Signs and symptoms

Clinical manifestations of GBCS are nonspecific, with symptoms such as abdominal pain localized to the right upper quadrant, constitutional symptoms, nausea, vomiting and fever. The mechanism resulting in constitutional symptoms in patients with cancer is multifactorial and not yet fully understood. It is thought that multiple pathways involving pro-cachectic and pro-inflammatory signals from tumor cells along with systemic inflammation of the host combine with widespread metabolic changes contribute to the manifestations of symptoms like anorexia and cachexia^[79]. In particular, cholecystokinin is an integral peptide involved in satiety and regulating diet intake^[80]. Given its role in gallbladder contraction, dysregulation of cholecystokinin could be involved in the manifestation of constitutional symptoms of anorexia in patients with GBCS.

Similarly, the pathophysiology of febrile response in malignancies is complex. Released pyrogenic cytokines from tumor cells and tissue macrophages induces a chain of events that result in reset of hypothalamic thermostat due to prostaglandin E2 and related pathways[81]. On physical examination, the presence of a right hypochondria tenderness or mass is not specific, and it does not rule out malignancy. Thus, if a patient is managed for suspected acute or chronic cholecystitis, a follow-up physical examination and imaging needs to be arranged to document resolution of inflammatory process. In this review, 2 asymptomatic patients were diagnosed with GBCS. From our analysis, Ishida et al[36] and Akatsu et al[46] reported incidental findings of GBCS on imaging findings for unrelated issues. Ishida et al[36] reported a 62-year-old female with unexpected calcification in the right upper abdomen in a CT meant for follow-up of percutaneous pinning of a left calcaneal fracture. Akatsu et al [46] reported a 76-year-old female who was on regular follow-up for cholelithiasis. Abdominal ultrasound revealed a heterogeneously hypoechoic mass around the gallbladder bed. In both patients, a preoperative diagnosis of possible gallbladder malignancy was made, and surgical exploration with subsequent cholecystectomy was performed.

Biochemical investigations

Biochemical abnormalities in GBCS are also mostly nonspecific. The most common derangements were transaminitis, hyperbilirubinemia and anemia. This was consistent with Ayoub et al[5] who reported that hepatic and inflammatory markers were often normal.

Presurgical diagnosis of gallbladder malignancies is difficult due to its varying presentations. Differentials to consider for such lesions when calcification is present include calcified gallstones, porcelain gallbladder and GBCS[78]. Our analysis noted cases where GBCS was initially diagnosed with cholelithiasis, acute cholecystitis, gallbladder empyema, diffuse peritonitis, pancreatic cancer and pyogenic liver abscess.

Imaging

As there are no radiological signs identified in the current literature that distinguishes GBCS from other gallbladder malignancies[50,51], the diagnosis is difficult even with imaging. For instance, Appelman et al[72] described a 91-year-old male presenting with yellow sclera, pale stools, dark urine and pruritus. His liver function tests were deranged with obstructive pattern, and a diagnosis of pancreatic cancer with biliary tract obstruction was made. The patient refused surgical intervention and died within 2 wk. Autopsy confirmed the diagnosis of metastatic disease with GBCS primary.

Khurram et al[4] reported a 64-year-old lady presented with right upper quadrant mass, intermittent fever and abdominal distension following a recent travel history to Ghana. CT scan revealed a hepatic lesion with coexisting gallbladder distension consistent with pyogenic liver abscess. Due to failure to respond to intravenous antibiotics, MRI scan was done. MRI scan showed a gallbladder fundus soft tissue lesion with local invasion into the liver[22]. Histopathological diagnosis of GBCS was



made after surgical excision. Hence, in the absence of a confirmatory preoperative diagnosis, all suspicious gallbladder lesions must be reviewed at multidisciplinary meetings.

Porcelain gallbladder, gallbladder tuberculosis and xanthogranulomatous cholecystitis are common benign conditions that can be confused with malignancy. Porcelain gallbladder is described as a hyperechoic focus with posterior acoustic shadowing on an ultrasound scan[82]. Ultrasound scan is not sensitive for regional and distant spread of malignancy. CT and MRI scans are more sensitive to detect contiguous spread to liver, regional lymph node involvement and distant metastases. Diffuse nodular thickening without layering, early enhancement, low apparent diffusion coefficient and high lesion to spinal cord ratio are MRI features suggestive of gallbladder cancer^[83]. In addition, CT and MRI scans provide details that assist in surgical planning. 18-fluorodeoxyglucose-positron emission tomography-CT can aid in distinguishing between benign and malignant gallbladder lesions. Malignant lesions have high standardized uptake value. In a study reporting 30 patients with a mean age of 48.22 ± 31.33 years and gallbladder wall thickening (focal > 4 mm and diffuse > 7 mm), Gupta et al[84] reported that 18-fluorodeoxyglucose-positron emission tomography had high overall sensitivity (91%), specificity (79%), positive predictive value (77%), negative predictive value (92%) and diagnostic accuracy (84%).

Histological diagnosis

Diagnosis of GBCS is usually made after pathological analysis of a surgical specimen. In patients with unresectable neoplasms, tissue diagnosis can be achieved by percutaneous biopsy. This is essential to plan definitive chemotherapy[85]. In clearly resectable lesions, the role of percutaneous biopsy is debated due to risk of needle-tract seeding[86]. Furthermore, as the gallbladder is a hollow organ, bile spill and peritonitis remain a risk too[87]. As GBCS are rare tumors with poor prognostic outcomes, treatment options are not well defined, with little evidence supporting or refuting any postoperative adjuvant therapy. Okabayashi *et al*[88] and Mochizuki *et al*[8] both corroborate that surgical treatment remains the only cure for GBCS. While the histopathological features between GBCS and adenocarcinoma of the gallbladder are different, management is similar.

Surgical management

Currently, the consensus for treatment involves surgical excision of the gallbladder and extrahepatic bile duct, regional lymphadenectomy and even a pancreaticoduodenectomy depending on the extent of the growth[88]. Completion liver resection with or without lymphadenectomy and/or bile duct resection is an accepted standard for post simple cholecystectomy discovered gallbladder cancer with T_{1b} and higher stage. This approach not only involves two surgeries but also increases the risk of cutting through the tumor with potential for tumor seeding and dissemination. Yip *et al*[89] in a series of 40 patients with incidental gallbladder cancer reported that the majority of patients were not amenable for further curative resection. A report from Memorial Sloan-Kettering Cancer Centre involving 116 patients showed that survival of patients with residual disease was not different than survival of patients with stage IV disease, and neither group of patients benefit from reoperation[90]. Thus, single surgery may be better.

Radical cholecystectomy has higher morbidity as compared to simple cholecystectomy. Thus, the concept of something intermediate, *i.e.*, extended cholecystectomy, is attractive. Fujisaki *et al*[91] reported a case describing the concept of laparoscopic extended cholecystectomy with 1 cm liver margin; however, they proposed open conversion when intraoperative histology showed gallbladder cancer invading the subserosal layer. With current advancements, laparoscopic extended cholecystectomy was noted to have lesser intraoperative and postoperative complications than open extended cholecystectomy[92].

The key differences between a 'radical' and 'extended' cholecystectomy are restricting the liver parenchyma transection to the 2 cm wedge of liver tissue and performing regional lymphadenectomy and choledochectomy only in selected patients. Radical cholecystectomy can be done by open, laparoscopic or robot assisted approach, with comparable short-term outcomes[93]. Overall, more data is required to determine the safety and feasibility of minimal access techniques in gallbladder malignancies. Due to the absence of histological diagnosis, management of suspicious gallbladder lesions must be determined by local resources, surgeon experience and access to technology.

Zaishideng® WJCO | https://www.wjgnet.com

In a recent systematic review, Frountzas et al[94] reported that many patients with xanthogranulomatous cholecystitis were managed with complex procedures like wedge hepatic resection and bile duct excision with high open conversion rate (35.0%) at planned cholecystectomy. Intraoperative frozen section analysis is a useful adjunct in surgical planning. While intraoperative frozen tissue diagnosis is relatively reliable to determine whether lesions are benign or malignant, it does not reliably detail the depth of invasion of gallbladder malignancies [95]. Furthermore, the accuracy of intraoperative frozen tissue diagnosis for GBCS has yet to be determined due to paucity of scientific data.

Adjuvant treatment

The adjuvant treatment reduces recurrence risk and improves survival outcomes by eliminating or controlling the micrometastatic disease. A meta-analysis of retrospective studies including 6712 gallbladder cancer patients reported that lymph node positive patients enjoyed the survival benefit[96]. Few reported patients consider the use of UFT: tegafur/uracil, gemcitabine or a combination of tegafur/gimeracil/ oteracil. The median survival of GBCS is 7.8 mo[10], and the addition of such regimes has not shown to improve survival[38]. There is a report by Pu et al[38] of using a combination of 5-FU (commonly used in gallbladder cancer) and oxaliplatin (commonly used in sarcomas). They reported a 59-year-old female coming in with right upper quadrant pain, fever and a raised CA19-9 level of 12000 U/mL, which was confirmed to be GBCS. The patient received oxaliplatin 150 mg and 5-FU 500 mg intravenously every 30 d for 6 cycles. At 6-mo follow-up, she did not reveal any signs of recurrence.

Adjuvant radiotherapy is shown to be of value in reducing local recurrence in selected patients with gallbladder cancer. In a study including 4180 patients with resected gallbladder cancer diagnosed from 1988 to 2003 from the Surveillance, Epidemiology, and End results database, Wang et al[97] reported that adjuvant radiotherapy provides survival benefit in node positive or T2 and higher stage disease. A single arm phase II study conducted by South West Oncology Group reported that gemcitabine plus capecitabine, followed by radiation (45 Gy to regional lymphatics, 54-59.4 Gy to tumor bed) and capecitabine resulted in 56% 2-year survival rate for patients with gallbladder cancer. Based on these results, the American Society of Clinical Oncology guidelines recommend chemotherapy plus radiation in gallbladder cancer patients with R1 resection[98]. There is no data to support neoadjuvant chemotherapy. Due to aggressive biological behavior, rapid progression or recurrence is common, and this is associated with a myriad of constitutional symptoms. For holistic care, management of the patients' subjective symptoms of anorexia and lethargy needs to be considered. Testosterone replacement therapy helps alleviate such symptoms in male patients with advanced cancer[99].

Prognosis

Generally, the prognosis of GBCS is poor. The majority of patients presenting to the hospital are locally advanced, with liver metastasis and peritoneal dissemination. Other metastasis sites reported include adrenal glands, pancreas, diaphragm and the lower thoracic vertebrae. Zhang et al[3] reported a mean survival time of 17.5 mo, with 1-year and 5-year survival rates at $(19 \pm 5)\%$ and $(16 \pm 5)\%$, respectively. While it was previously noted the longest survival time to be reported as 54 mo by Uzun et al[44], our review noted 86 mo to be the longest survival time [10]. Aldossary *et al* [10] reported a 62-year-old female patient who complained of severe intermittent right upper quadrant pain of 2 mo duration. Laboratory investigations were normal, and ultrasound suggested a gallbladder with large stones and a non-mobile echogenic mass. A stage II (pT2, pN0, M0) moderately differentiated GBCS was noted on histology after laparoscopic cholecystectomy. The patient underwent 14 cycles of adjuvant chemotherapy. She had local recurrence at 2 years. Wide local excision of the mass with wedge resection of the liver, lymphadenectomy and partial gastrectomy was done. The patient remained disease free for 86 mo. Zhang et al[3] also claim that tumors smaller than 5 cm had a more prolonged survival, however we did not observe this. More data is required to confirm this, as only 28 patients detailing both the survival data and size of tumor have been reported.

Role of tumor markers

GBCS is not noted to have association with any tumor markers. Consistent with the current literature, most of the patients did not note any raised tumor markers^[5]. However, it is still common practice for physicians to perform tumor marker levels



WJCO | https://www.wjgnet.com

such as CA19-9, carcinoembryonic antigen and alpha-fetoprotein when considering possible differentials for masses in the gallbladder as well as for prognostication. For instance, Hayashi et al[100] propose that alpha-fetoprotein-producing carcinomas of the gallbladder are more likely to metastasize to the liver and have poor prognosis. CA19-9 is typically associated with pancreatobiliary malignancies but has a limited role in clinical practice[101]. Thus, prognostication is relied typically on histological features, pathologic stage as well as immunohistochemistry. Immunohistochemistry for the mesenchymal and epithelial components yield positive staining for vimentin and cytokeratin [45]. Our review shows that the majority of the patients had positive staining for vimentin (81.2%) and cytokeratin (79.2%). Additionally, Ki-67 was suggested by Kubota et al[45] to have prognostic value, whereby its presence signifies a possibly higher malignant proliferative potential for GBCS. However, this claim needs to be further investigated as Kubota et al^[45] examined this immunohistochemical marker in only 1 patient with CSGB.

Comparison to gallbladder adenocarcinoma

There is substantial overlap of risk factors, diagnosis and treatment of GBCS with gallbladder adenocarcinoma. Thus, the majority of authors extrapolate the clinical characteristics of gallbladder adenocarcinoma to determine the best approach to diagnosis and management of GBCS. From this review, we can determine three key differences between GBCS and gallbladder adenocarcinoma. First, tumor markers have limited utility in patients with GBCS. In a study of 55 cases by Shukla *et al*[102], it is noted that the combination of CA-125 and CA19-9 helped detect gallbladder malignancy in patients with gallstones (80.7%). Second, the prognosis of GBCS may be marginally better compared to carcinoma of the gallbladder. In the meta-analysis by Zhang *et al*[3], it was noted that the survival rate was slightly better ($16\% \pm 5\%$ 5-year survival) compared to carcinoma of the gallbladder (0-10% 5-year survival). Thus, the identification of GBCS will be useful to determine the prognosis for patients albeit with only a small variation between the two. Third, immunohistochemistry markers like vimentin and cytokeratin are associated with diagnosis of GBCS.

CONCLUSION

In conclusion, GBCS is more common in females. Gallstones and chronic cholecystitis are risk factors for GBCS. Serum biochemistry and tumor markers have a limited role in diagnosis. Typical imaging modalities can assist to establish a diagnosis in patients with suspicious gallbladder lesions. Multiple imaging modalities are complementary. Multidisciplinary oncology board discussions are essential to guide management plans. Surgery is currently the only curative option for GBCS, and size of the tumor does not impact prognosis. While most features of GBCS parallel that of carcinomas of the gallbladder clinically, identification of GBCS specifically allows clinicians to determine overall prognosis. Due to paucity of reported cases, more evidence is required before meaningful and valid evidence-based patient-centric recommendations can be made. This review serves to educate and raise awareness among the clinicians dealing with gallbladder malignancies. It is likely that there are more clinical differences between GBCS and common forms of gallbladder cancer; active reporting of cases will help enhance understanding of this rare cancer.

ARTICLE HIGHLIGHTS

Research background

Literature on gallbladder carcinosarcoma (GBCS) is currently scarce, with less than 100 cases reported since the first case by Karl Lansteiner.

Research motivation

While there has been efforts by Zhang et al in 2008 to consolidate the literature, there has not been a review of the current literature since.

Research objectives

This study aims to fill this gap by providing a comprehensive overview of GBCS.



Research methods

A literature review was performed on the PubMed database using the keywords "Gallbladder" AND "Carcinosarcoma" from 1970 to 2021, where relevant articles were included. Animal studies, gallbladder carcinoma and non-gallbladder pathology as well as articles that were not in English or Japanese were excluded.

Research results

GBCS is more common in females. Gallstones and chronic cholecystitis are risk factors for GBCS. Serum biochemistry and tumor markers a have limited role in diagnosis. Typical imaging modalities can assist to establish a diagnosis in patients with suspicious gallbladder lesions. Multiple imaging modalities are complementary. Multidisciplinary oncology board discussions are essential to guide management plans. Surgery is currently the only curative option for GBCS, and size of the tumor does not impact prognosis.

Research conclusions

While most features of GBCS parallel that of carcinomas of the gallbladder clinically, identification of GBCS specifically allows clinicians to determine overall prognosis. Due to paucity of reported cases, more evidence is required before meaningful and valid evidence-based patient-centric recommendations can be made.

Research perspectives

Due to the paucity of the number of reported cases, more active reporting of such should be encouraged to further understand this malignancy.

REFERENCES

- 1 Al-Sheneber IF, Jaber T, Huttner I, Arseneau J, Loutfi A. Carcinosarcoma of the gallbladder: A case report and review of literature. Saudi J Gastroenterol 2002; 8: 22-24 [PMID: 19861787]
- 2 Karl L. Plattenepithelkarzinom und sarkom der gallenblase in einem falle von cholelithiasis. Z Klin Med 1907; 62: 427-433
- 3 Zhang L, Chen Z, Fukuma M, Lee LY, Wu M. Prognostic significance of race and tumor size in carcinosarcoma of gallbladder: a meta-analysis of 68 cases. Int J Clin Exp Pathol 2008; 1: 75-83 [PMID: 18784825]
- Khurram R, Khamar R, Lunat R, Novelli D, Ozretić L, Chaudhary K. Gallbladder carcinosarcoma masquerading as a hepatic abscess. Radiol Case Rep 2021; 16: 152-156 [PMID: 33240458 DOI: 10.1016/j.radcr.2020.10.058
- 5 Ayoub M, Jabi R, Achraf M, Benani A, Soumia EA, Imane K, Mohamed B. Surgical management of gallbladder carcinosarcoma: A case report and review of the Literature. Int J Surg Case Rep 2020; 75: 460-463 [PMID: 33076195 DOI: 10.1016/j.ijscr.2020.09.114]
- 6 Kaneko R, Kimura Y, Sakata H, Ikehara T, Mitomi H, Uekusa T, Ohbu M, Kubo S. A case of primary hepatic mixed neuroendocrine-non-neuroendocrine tumor (MiNEN) associated with gallbladder carcinosarcoma. Clin J Gastroenterol 2020; 13: 1280-1288 [PMID: 32779146 DOI: 10.1007/s12328-020-01202-8]
- Siddiqui M, Hegde S, Nguyen T, DePaul S. Sarcomatoid carcinoma of the gallbladder: A rare form 7 of gallbladder cancer. SAGE Open Med Case Rep 2020; 8: 2050313X20906739 [PMID: 32095246 DOI: 10.1177/2050313X20906739]
- Mochizuki K, Hata H, Naitou K, Motosugi U, Kondo T. Carcinosarcoma (adenocarcinoma, neuroendocrine carcinoma, undifferentiated carcinoma and chondrosarcoma) of the gallbladder. Clin J Gastroenterol 2020; 13: 110-115 [PMID: 31264080 DOI: 10.1007/s12328-019-01012-7]
- 9 Varshney VK, Bharti JN, Sureka B, Soni SC. Gallbladder Carcinosarcoma with Mirizzi Syndrome: a Rare Presentation. J Gastrointest Cancer 2019; 50: 997-1000 [PMID: 30178399 DOI: 10.1007/s12029-018-0167-3]
- 10 Aldossary MY, AlQattan AS, Alghamdi YM, Alayed AA, Alquraish F, AlAnzi OA, Alabdulrahim N, Alateeq A, Alqahtani MS. Surgical outcomes of primary carcinosarcoma of the gallbladder after curative resection: A rare case series. Int J Surg Case Rep 2019; 65: 32-39 [PMID: 31678697 DOI: 10.1016/i.ijscr.2019.10.056
- 11 Alratroot JA, Joudeh AA, Amr SS. Carcinosarcoma of the Gallbladder with Chondrosarcomatous Differentiation and Intracytoplasmic Eosinophilic Hyaline Globules (Thanatosomes): A Report of a Case and Review of the Literature. Case Rep Pathol 2019; 2019: 9697235 [PMID: 30881717 DOI: 10.1155/2019/9697235
- Matsubayashi H, Matsui T, Sugiura T, Makuuchi R, Kaneko J, Satoh J, Satoh T, Fujie S, Ishiwatari 12 H, Sasaki K, Ono H. A Large Carcinosarcoma of the Gallbladder Accompanied by Pancreaticobiliary Maljunction: A Case with a Six-year Survival. Intern Med 2019; 58: 2809-2817 [PMID: 31243200 DOI: 10.2169/internalmedicine.2783-19]



- 13 Doniparthi M, Omar H, Haider A, O'Riordan K. P0042 - Primary Gallbladder Carcinosarcoma: A Rare Biliary Malignancy. ACG 2018; 113: S713-S714
- 14 Koustav Jana AKD, Suman Saha, Ranajoy Ghosh, Sukanta Ray. Carcinosarcoma Gallbladder: A Rare Variety of Gallbladder Malignancy. J Clin Diagn Res 2018; 12: PD01-PD02 [DOI: 10.7860/jcdr/2018/31830.11387]
- 15 Trautman J, Wood BE, Craig SJ. A rare case report of gallbladder carcinosarcoma. J Surg Case Rep 2018; 2018: rjy167 [PMID: 30046436 DOI: 10.1093/jscr/rjy167]
- Furuya Y, Hiroshima K, Wakahara T, Akimoto H, Yanagie H, Harigaya K, Yasuhara H. 16 Undifferentiated carcinoma of the gallbladder with endothelial differentiation: A case report and literature review. Mol Clin Oncol 2016; 5: 773-776 [PMID: 28101355 DOI: 10.3892/mco.2016.1046]
- Hu W, Zhou G, Gao S, Wu L, Bao H, Yu H, Zhang Q. Carcinosarcoma of the gallbladder 17 accompanied with cholecystolithiasis: a case report. Int J Clin Exp Pathol 2016; 9: 2583-2586
- Cruz J, Matos AP, Neta JO, Ramalho M. Carcinosarcoma of the gallbladder-an exceedingly rare 18 tumour. BJR Case Rep 2016; 2: 20160019 [PMID: 30460034 DOI: 10.1259/bjrcr.20160019]
- 19 Dong A, Dong H, Jing W, Zuo C. FDG PET/CT in Sarcomatoid Carcinoma of the Gallbladder With Chondroid Differentiation. Clin Nucl Med 2016; 41: 638-640 [PMID: 26859217 DOI: 10.1097/rlu.000000000001160
- Gupta S, Kori C, Kumar V. Primary Carcinosarcoma of the Gall Bladder: A Rare Entity. Indian J 20 Surg Oncol 2016; 7: 101-105 [PMID: 27065693 DOI: 10.1007/s13193-015-0420-1]
- 21 Wong MW, Chen MJ, Chen CJ, Hung CY, Wang HY, Chang CW. Gallbladder sarcomatoid carcinoma: A case report and review of current literature. Adv Digest Med 2016; 3: 61-64 [DOI: 10.1016/j.aidm.2014.10.003]
- 22 Ansari FA, Bhatnagar M, Katiyar DC, Gairola M, Gupta V, Princee A. Carcinosarcoma of Gall Bladder. Trop Gastroenterol 2016; 37: 63-65 [PMID: 29668183 DOI: 10.7869/tg.325]
- Gao S, Huang L, Dai S, Chen D, Dai R, Shan Y. Carcinosarcoma of the gallbladder: a case report 23 and review of the literature. Int J Clin Exp Pathol 2015; 8: 7464-7469 [PMID: 26261654]
- Tonouchi A, Yokoyama N, Hashidate H, Matsuzawa N, Katayanagi N, Otani T. Education and 24 Imaging, Gastroenterology: Carcinosarcoma of the gallbladder presenting as a cholecysto-colic fistula. J Gastroenterol Hepatol 2015; 30: 1112 [PMID: 26094660 DOI: 10.1111/jgh.12906]
- 25 Faujdar M, Gupta S, Jain R, Wadhwani D. Carcinosarcoma of the gallbladder with heterologous differentiation: a case report. J Gastrointest Cancer 2015; 46: 175-177 [PMID: 25698405 DOI: 10.1007/s12029-015-9692-5
- 26 Wada Y, Takami Y, Tateishi M, Ryu T, Mikagi K, Momosaki S, Saitsu H. Carcinosarcoma of the gallbladder: report of a case. Clin J Gastroenterol 2014; 7: 455-459 [PMID: 26184028 DOI: 10.1007/s12328-014-0522-2
- Kishino T, Mori T, Kawai S, Mori H, Nishikawa K, Hirano K, Matsushima S, Ohtsuka K, Ohnishi 27 H, Watanabe T. Carcinosarcoma, an atypical subset of gallbladder malignancies. J Med Ultrason (2001) 2014; 41: 487-490 [PMID: 27278030 DOI: 10.1007/s10396-014-0534-z]
- 28 Wang Y, Gu X, Li Z, Xiang J, Chen Z. Gallbladder carcinosarcoma accompanied with bile duct tumor thrombi: A case report. Oncol Lett 2013; 5: 1809-1812 [PMID: 23833646 DOI: 10.3892/ol.2013.1289]
- Khanna M, Khanna A, Manjari M. Carcinosarcoma of the gallbladder: a case report and review of 29 the literature. J Clin Diagn Res 2013; 7: 560-562 [PMID: 23634423 DOI: 10.7860/JCDR/2013/4924.2825
- 30 Li BF, Li PW, Shi R, Liu ZL, Liu YF. Clinicopathologic analysis of gallbladder carcinosarcoma. Am Surg 2013; 79: E37-E39 [PMID: 23317603]
- 31 Kim HH, Hur YH, Jeong EH, Park EK, Koh YS, Kim JC, Kim HJ, Kim JW, Cho CK. Carcinosarcoma of the gallbladder: report of two cases. Surg Today 2012; 42: 670-675 [PMID: 22391981 DOI: 10.1007/s00595-012-0160-6]
- 32 Sadamori H, Fujiwara H, Tanaka T, Yanai H, Satoh D, Yagi T, Fujiwara T. Carcinosarcoma of the gallbladder manifesting as cholangitis due to hemobilia. J Gastrointest Surg 2012; 16: 1278-1281 [PMID: 22350725 DOI: 10.1007/s11605-012-1836-3]
- Kataria K, Yadav R, Seenu V. Sarcomatoid carcinoma of the gall bladder. J Surg Case Rep 2012; 33 2012: 5 [PMID: 24960781 DOI: 10.1093/jscr/2012.2.5]
- 34 Parreira JM, Siqueira DE, Menacho AM, Pelizzari LL, Santos LC. Carcinosarcoma of the gallbladder: case report. Arq Bras Cir Dig 2012; 25: 65-66 [PMID: 22569983 DOI: 10.1590/s0102-67202012000100015]
- Park SB, Kim YH, Rho HL, Chae GB, Hong SK. Primary carcinosarcoma of the gallbladder. J Korean Surg Soc 2012; 82: 54-58 [PMID: 22324048 DOI: 10.4174/jkss.2012.82.1.54]
- 36 Ishida J, Ajiki T, Hara S, Ku Y. Gallbladder calcification leads to discovery of carcinosarcoma of the gallbladder. Surgery 2012; 152: 934-935 [PMID: 21875736 DOI: 10.1016/j.surg.2011.07.033]
- 37 Lee S, Kim SY, Hong SK. Sarcomatoid Carcinoma of the Gallbladder with Pure Squamous Cell Carcinoma: A Brief Case Report. J Pathol Transl Med 2011; 45(2): 209-211 [DOI: 10.4132/KoreanJPathol.2011.45.2.209]
- Pu JJ, Wu W. Gallbladder carcinosarcoma. BMJ Case Rep 2011; 2011 [PMID: 22700610 DOI: 38 10.1136/bcr.05.2010.3009
- Krishnamurthy V, Shivalingiah SD, Ravishankar S, Manjunath GV. Morphologic heterogeneity in 39 carcinosarcoma of the gallbladder: report of a rare cases. Case Rep Pathol 2011; 2011: 929654



[PMID: 22937397 DOI: 10.1155/2011/929654]

- Kohtani T, Masuda J, Hisaki T, Shimase K, Mizuguchi K. Long-Term Survival of an Elderly 40 Patient with Carcinosarcoma of the Gallbladder after Cholecystectomy. Case Rep Gastroenterol 2009; **3**: 235-239 [PMID: 21103281 DOI: 10.1159/000229592]
- Agarwal T, Jain M, Goel A, Visayaragavan P, Gupta RK. Carcinosarcoma of the gallbladder. Indian J Pathol Microbiol 2009; 52: 244-245 [PMID: 19332928 DOI: 10.4103/0377-4929.48933]
- 42 Magata S, Kai K, Mori M, Edakuni G, Tokunaga O, Miyazaki K. Fibroadenoma of the gallbladder with borderline malignancy: a poorly recognized gallbladder tumor. J Hepatobiliary Pancreat Surg 2009; 16: 874-877 [PMID: 19350196 DOI: 10.1007/s00534-009-0080-z]
- Shimada K, Iwase K, Aono T, Nakai S, Takeda S, Fujii M, Koma M, Nishikawa K, Matsuda C, 43 Hirota M, Fushimi H, Tanaka Y. Carcinosarcoma of the gallbladder producing alpha-fetoprotein and manifesting as leukocytosis with elevated serum granulocyte colony-stimulating factor: report of a case. Surg Today 2009; 39: 241-246 [PMID: 19280285 DOI: 10.1007/s00595-008-3833-4]
- 44 Uzun MA, Koksal N, Gunerhan Y, Celik A, Gunes P. Carcinosarcoma of the gallbladder: report of a case. Surg Today 2009; 39: 168-171 [PMID: 19198999 DOI: 10.1007/s00595-008-3805-8]
- 45 Kubota K, Kakuta Y, Kawamura S, Abe Y, Inamori M, Kawamura H, Kirikoshi H, Kobayashi N, Saito S. Nakajima A. Undifferentiated spindle-cell carcinoma of the gallbladder: an immunohistochemical study. J Hepatobiliary Pancreat Surg 2006; 13: 468-471 [PMID: 17013725 DOI: 10.1007/s00534-006-1100-x]
- 46 Akatsu T, Ueda M, Shimazu M, Wakabayashi G, Aiura K, Tanabe M, Kawachi S, Kameyama K, Kitajima M. Primary undifferentiated spindle-cell carcinoma of the gallbladder presenting as a liver tumor. J Gastroenterol 2005; 40: 993-998 [PMID: 16261437 DOI: 10.1007/s00535-005-1684-y]
- Huguet KL, Hughes CB, Hewitt WR. Gallbladder carcinosarcoma: a case report and literature 47 review. J Gastrointest Surg 2005; 9: 818-821 [PMID: 15985237 DOI: 10.1016/j.gassur.2004.12.009]
- Sodergren MH, Silva MA, Read-Jones SL, Hubscher SG, Mirza DF. Carcinosarcoma of the biliary tract: two case reports and a review of the literature. Eur J Gastroenterol Hepatol 2005; 17: 683-685 [PMID: 15879734 DOI: 10.1097/00042737-200506000-00016]
- 49 Takahashi Y, Fukushima J, Fukusato T, Shiga J. Sarcomatoid carcinoma with components of small cell carcinoma and undifferentiated carcinoma of the gallbladder. Pathol Int 2004; 54: 866-871 [PMID: 15533231 DOI: 10.1111/j.1440-1827.2004.01771.x]
- 50 Kim MJ, Yu E, Ro JY. Sarcomatoid carcinoma of the gallbladder with a rhabdoid tumor component. Arch Pathol Lab Med 2003; 127: e406-e408 [PMID: 14521442 DOI: 10.1043/1543-2165(2003)1272.0.CO;2
- Hotta T, Tanimura H, Yokoyama S, Ura K, Yamaue H. So-called carcinosarcoma of the 51 gallbladder; spindle cell carcinoma of the gallbladder: report of a case. Surg Today 2002; 32: 462-467 [PMID: 12061703 DOI: 10.1007/s005950200077]
- 52 Ajiki T, Nakamura T, Fujino Y, Suzuki Y, Takeyama Y, Ku Y, Kuroda Y, Ohbayashi C. Carcinosarcoma of the gallbladder with chondroid differentiation. J Gastroenterol 2002; 37: 966-971 [PMID: 12483254 DOI: 10.1007/s005350200162]
- 53 Yavuz E, Bilgiç B, Cevikbaç U, Demiryont M. Test and teach. Number Ninety Nine. Carcinosarcoma of the gallbladder. Pathology 2000; 32: 41, 63-64 [PMID: 10740804 DOI: 10.1080/003130200104556]
- 54 Eriguchi N, Aoyagi S, Hara M, Hashino K, Imamura M, Sato S, Imamura I, Kutami R, Jimi A. A so-called carcinosarcoma of the gallbladder in a patient with multiple anomalies--a case report. Kurume Med J 1999; 46: 175-179 [PMID: 10659595 DOI: 10.2739/kurumemedj.46.175]
- 55 Ryś J, Kruczak A, Iliszko M, Babińska M, Wasilewska A, Limon J, Niezabitowski A. Sarcomatoid carcinoma (carcinosarcoma) of the gallbladder. Gen Diagn Pathol 1998; 143: 321-325 [PMID: 96539151
- 56 Nakagawa T, Yamakado K, Takeda K, Nakagawa T. An ossifying carcinosarcoma of the gallbladder: radiologic findings. AJR Am J Roentgenol 1996; 166: 1233-1234 [PMID: 8615287 DOI: 10.2214/ajr.166.5.8615287]
- Fagot H, Fabre JM, Ramos J, Laffay V, Guillon F, Domergue J, Baumel H. Carcinosarcoma of the 57 gallbladder. A case report and review of the literature. J Clin Gastroenterol 1994; 18: 314-316 [PMID: 8071517 DOI: 10.1097/00004836-199406000-00011]
- 58 Nakazawa T, Sotoyama K, Tsuchiya N, Naito Y, Honma J, Noto M, Nakazawa H, Takada I, Kaneyama M, Nasu Y. [A case of common bile duct neoplasm with carcinosarcoma of the gallbladder]. Nihon Naika Gakkai Zasshi 1992; 81: 908-910 [PMID: 1402245]
- Ishihara T, Kawano H, Takahashi M, Yokota T, Uchino F, Matsumoto N, Fukuyama N. 59 Carcinosarcoma of the gallbladder. A case report with immunohistochemical and ultrastructural studies. Cancer 1990; 66: 992-997 [PMID: 2386928 DOI:

10.1002/1097-0142(19900901)66:5<992::aid-cncr2820660532>3.0.co;2-2]

- 60 Lumsden AB, Mitchell WE, Vohman MD. Carcinosarcoma of the gallbladder: a case report and review of the literature. Am Surg 1988; 54: 492-494 [PMID: 3395026]
- 61 Hasegawa H, Takada T, Yasuda H, Uchiyama K, Tsuchiya S, Misu Y, Saitoh Y, Shikata J, Suzuki H. [A case of huge carcinosarcoma of the gallbladder]. Nihon Shokakibyo Gakkai Zasshi 1987; 84: 1690-1694 [PMID: 3320418]
- 62 Herrera-Goepfert R, Manrique Ortega JJ, Rodríguez-Martínez HA. Carcinosarcoma of the gallbladder. Histol Histopathol 1987; 2: 273-275 [PMID: 2980729]
- Inoshita S, Iwashita A, Enjoji M. Carcinosarcoma of the gallbladder. Report of a case and review of 63



the literature. Acta Pathol Jpn 1986; 36: 913-920 [PMID: 3766137 DOI: 10.1111/j.1440-1827.1986.tb03124.x]

- Lopez GE, Strimel W, Herrera-Ornelas L. Carcinosarcoma of the gallbladder: report of a case. J 64 Surg Oncol 1985; 29: 224-226 [PMID: 3908822 DOI: 10.1002/jso.2930290405]
- 65 Born MW, Ramey WG, Ryan SF, Gordon PE. Carcinosarcoma and carcinoma of the gallbladder. Cancer 1984; 53: 2171-2177 [PMID: 6704904 DOI: 10.1002/1097-0142(19840515)53:10<2171::aid-cncr2820531027>3.0.co;2-m]
- Von Kuster LC, Cohen C. Malignant mixed tumor of the gallbladder: report of two cases and a 66 review of the literature. Cancer 1982; 50: 1166-1170 [PMID: 7104957 DOI: 10.1002/1097-0142(19820915)50:6<1166::aid-cncr2820500622>3.0.co;2-1
- Aldovini D, Piscioli F, Togni R. Primary malignant mixed mesodermal tumor of the gallbladder. 67 Report of a case and critical review of diagnostic criteria. Virchows Arch A Pathol Anat Histol 1982; **396**: 225-230 [PMID: 6289513 DOI: 10.1007/BF00431243]
- 68 Yamagiwa H, Yoshimura H, Tomiyama H, Kawahara S, Ito F. [So-called carcinosarcoma]. Rinsho Byori 1982; 30: 1096-1102 [PMID: 7182581]
- 69 Mansori KS, Cho SY. Malignant mixed tumor of the gallbladder. Am J Clin Pathol 1980; 73: 709-711 [PMID: 6246794 DOI: 10.1093/ajcp/73.5.709]
- 70 Higgs WR, Mocega EE, Jordan PH Jr. Malignant mixed tumor of the gallbladder. Cancer 1973; 32: 471-475 [PMID: 4353018 DOI:

10.1002/1097-0142(197308)32:2<471::aid-cncr2820320227>3.0.co;2-o] 71 Mehrotra TN, Gupta SC, Naithani YP. Carcino-sarcoma of the gall bladder. J Pathol 1971; 104: 145-148 [PMID: 5286390 DOI: 10.1002/path.1711040209]

- 72 Appelman HD, Coopersmith N. Pleomorphic spindle-cell carcinoma of the gallbladder. Relation to sarcoma of the gallbladder. Cancer 1970; 25: 535-541 [PMID: 5416826 DOI: 10.1002/1097-0142(197003)25:3<535::aid-cncr2820250307>3.0.co;2-1
- Pandey M. Risk factors for gallbladder cancer: a reappraisal. Eur J Cancer Prev 2003; 12: 15-24 73 [PMID: 12548106 DOI: 10.1097/00008469-200302000-00004]
- 74 Kuroki T, Tajima Y, Matsuo K, Kanematsu T. Genetic alterations in gallbladder carcinoma. Surg Today 2005; 35: 101-105 [PMID: 15674488 DOI: 10.1007/s00595-004-2906-2]
- 75 Kimura K, Ohto M, Saisho H, Unozawa T, Tsuchiya Y, Morita M, Ebara M, Matsutani S, Okuda K. Association of gallbladder carcinoma and anomalous pancreaticobiliary ductal union. Gastroenterology 1985; 89: 1258-1265 [PMID: 4054518 DOI: 10.1016/0016-5085(85)90641-9]
- Espinoza JA, Bizama C, García P, Ferreccio C, Javle M, Miquel JF, Koshiol J, Roa JC. The 76 inflammatory inception of gallbladder cancer. Biochim Biophys Acta 2016; 1865: 245-254 [PMID: 26980625 DOI: 10.1016/j.bbcan.2016.03.004]
- 77 Yıldız İ, Koca YS, Barut İ. Overlap of Acute Cholecystitis with Gallstones and Squamous Cell Carcinoma of the Gallbladder in an Elderly Patient. Case Reports in Surgery 2015; 2015: 767196 [DOI: 10.1155/2015/767196]
- Mohan R, Shelat VG, Junnarkar SP. Large gallstone initially diagnosed as porcelain gallbladder 78 with malignant change. Postgrad Med J 2021 [PMID: 33541934 DOI: 10.1136/postgradmedj-2020-139303]
- Donohoe CL, Ryan AM, Reynolds JV. Cancer cachexia: mechanisms and clinical implications. 79 Gastroenterol Res Pract 2011; 2011: 601434 [PMID: 21760776 DOI: 10.1155/2011/601434]
- Dhanapal R, Saraswathi T, Govind RN. Cancer cachexia. J Oral Maxillofac Pathol 2011; 15: 257-80 260 [PMID: 22144825 DOI: 10.4103/0973-029X.86670]
- Foggo V, Cavenagh J. Malignant causes of fever of unknown origin. Clin Med (Lond) 2015; 15: 292-294 [PMID: 26031983 DOI: 10.7861/clinmedicine.15-3-292]
- 82 Machado NO. Porcelain Gallbladder: Decoding the malignant truth. Sultan Qaboos Univ Med J 2016; 16: e416-e421 [PMID: 28003886 DOI: 10.18295/squmj.2016.16.04.003]
- Gupta P, Marodia Y, Bansal A, Kalra N, Kumar-M P, Sharma V, Dutta U, Sandhu MS. Imaging-83 based algorithmic approach to gallbladder wall thickening. World J Gastroenterol 2020; 26: 6163-6181 [PMID: 33177791 DOI: 10.3748/wjg.v26.i40.6163]
- 84 Gupta V, Vishnu KS, Yadav TD, Sakaray YR, Irrinki S, Mittal BR, Kalra N, Vaiphei K. Radiopathological Correlation of 18F-FDG PET in Characterizing Gallbladder Wall Thickening. J Gastrointest Cancer 2019; 50: 901-906 [PMID: 30397856 DOI: 10.1007/s12029-018-0176-2]
- Akosa AB, Barker F, Desa L, Benjamin I, Krausz T. Cytologic diagnosis in the management of 85 gallbladder carcinoma. Acta Cytol 1995; 39: 494-498 [PMID: 7762339]
- 86 Berger-Richardson D, Chesney TR, Englesakis M, Govindarajan A, Cleary SP, Swallow CJ. Trends in port-site metastasis after laparoscopic resection of incidental gallbladder cancer: A systematic review. Surgery 2017; 161: 618-627 [PMID: 27743715 DOI: 10.1016/j.surg.2016.08.007]
- Rana C, Krishnani N, Kumari N. Ultrasound-guided fine needle aspiration cytology of gallbladder 87 lesions: a study of 596 cases. Cytopathology 2016; 27: 398-406 [PMID: 26990137 DOI: 10.1111/cyt.12296]
- 88 Okabayashi T, Sun ZL, Montgomey RA, Hanazaki K. Surgical outcome of carcinosarcoma of the gall bladder: a review. World J Gastroenterol 2009; 15: 4877-4882 [PMID: 19842216 DOI: 10.3748/wjg.15.4877]
- Yip VS, Gomez D, Brown S, Byrne C, White D, Fenwick SW, Poston GJ, Malik HZ. Management 89 of incidental and suspicious gallbladder cancer: focus on early referral to a tertiary centre. HPB



(Oxford) 2014; 16: 641-647 [PMID: 24279377 DOI: 10.1111/hpb.12189]

- Butte JM, Kingham TP, Gönen M, D'Angelica MI, Allen PJ, Fong Y, DeMatteo RP, Jarnagin WR. 90 Residual disease predicts outcomes after definitive resection for incidental gallbladder cancer. J Am Coll Surg 2014; 219: 416-429 [PMID: 25087941 DOI: 10.1016/j.jamcollsurg.2014.01.069]
- 91 Fujisaki S, Saitoh Y, Tomita R, Fukuzawa M. Laparoscopic extended cholecystectomy. J Laparoendosc Adv Surg Tech A 2001; 11: 219-222 [PMID: 11569511 DOI: 10.1089/109264201750539736]
- Zhao X, Li XY, Ji W. Laparoscopic versus open treatment of gallbladder cancer: A systematic 92 review and meta-analysis. J Minim Access Surg 2018; 14: 185-191 [PMID: 28782743 DOI: 10.4103/jmas.JMAS_223_16]
- 93 Goel M, Khobragade K, Patkar S, Kanetkar A, Kurunkar S. Robotic surgery for gallbladder cancer: Operative technique and early outcomes. J Surg Oncol 2019; 119: 958-963 [PMID: 30802316 DOI: 10.1002/jso.25422]
- 94 Frountzas M, Schizas D, Liatsou E, Economopoulos KP, Nikolaou C, Apostolou KG, Toutouzas KG, Felekouras E. Presentation and surgical management of xanthogranulomatous cholecystitis. Hepatobiliary Pancreat Dis Int 2021; 20: 117-127 [PMID: 33536138 DOI: 10.1016/j.hbpd.2021.01.002]
- 95 Yamaguchi K, Chijiiwa K, Saiki S, Shimizu S, Tsuneyoshi M, Tanaka M. Reliability of frozen section diagnosis of gallbladder tumor for detecting carcinoma and depth of its invasion. J Surg Oncol 1997; 65: 132-136 [PMID: 9209526 DOI: 10.1002/(sici)1096-9098(199706)65:2<132::aid-jso11>3.0.co;2-7]
- 96 Horgan AM, Amir E, Walter T, Knox JJ. Adjuvant therapy in the treatment of biliary tract cancer: a systematic review and meta-analysis. J Clin Oncol 2012; 30: 1934-1940 [PMID: 22529261 DOI: 10.1200/JCO.2011.40.5381]
- Wang SJ, Fuller CD, Kim JS, Sittig DF, Thomas CR Jr, Ravdin PM. Prediction model for 97 estimating the survival benefit of adjuvant radiotherapy for gallbladder cancer. J Clin Oncol 2008; 26: 2112-2117 [PMID: 18378567 DOI: 10.1200/JCO.2007.14.7934]
- Ben-Josef E, Guthrie KA, El-Khoueiry AB, Corless CL, Zalupski MM, Lowy AM, Thomas CR Jr, 98 Alberts SR, Dawson LA, Micetich KC, Thomas MB, Siegel AB, Blanke CD. SWOG S0809: A Phase II Intergroup Trial of Adjuvant Capecitabine and Gemcitabine Followed by Radiotherapy and Concurrent Capecitabine in Extrahepatic Cholangiocarcinoma and Gallbladder Carcinoma. J Clin Oncol 2015; 33: 2617-2622 [PMID: 25964250 DOI: 10.1200/JCO.2014.60.2219]
- 99 Teng TZJ, Shelat VG. Testosterone gel improves appetite and reduces tiredness in males with advanced cancer. BMJ Support Palliat Care 2021; 11: 145 [PMID: 33468510 DOI: 10.1136/bmjspcare-2020-002662]
- 100 Hayashi K, Hiraki M, Yamashita Y, Kurohiji T, Kimitsuki H, Watanabe J, Kakegawa T. A case of cancer of the gallbladder with a high level of Alphafetoprotein (AFIP). J Jap Pract Surg Soc 1994; 55: 3161-3165 [DOI: 10.3919/ringe1963.55.3161]
- 101 Lee T, Teng TZJ, Shelat VG. Carbohydrate antigen 19-9 - tumor marker: Past, present, and future. World J Gastrointest Surg 2020; 12: 468-490 [PMID: 33437400 DOI: 10.4240/wjgs.v12.i12.468]
- 102 Shukla VK, Gurubachan, Sharma D, Dixit VK, Usha. Diagnostic value of serum CA242, CA 19-9, CA 15-3 and CA 125 in patients with carcinoma of the gallbladder. Trop Gastroenterol 2006; 27: 160-165 [PMID: 17542293]



WJCO | https://www.wjgnet.com



Published by Baishideng Publishing Group Inc 7041 Koll Center Parkway, Suite 160, Pleasanton, CA 94566, USA Telephone: +1-925-3991568 E-mail: bpgoffice@wjgnet.com Help Desk: https://www.f6publishing.com/helpdesk https://www.wjgnet.com

